# FINAL SITE CHARACTERIZATION REPORT REGION 8 DRY CLEANING SITES CRYSTAL CLEANERS SITE SITE NO. 8-51-022

WORK ASSIGNMENT NO. D004434-19

**Prepared for:** 

New York State Department of Environmental Conservation Albany, New York

**Prepared by:** 

MACTEC Engineering and Consulting, PC Portland, Maine

# MACTEC NO. 3612072075

#### **MARCH 2007**

This document was prepared for the sole use of New York State Department of Environmental Conservation, the only intended beneficiary of our work. No other party shall rely on the information contained herein without prior written consent of MACTEC Engineering and Consulting, PC.

# FINAL SITE CHARACTERIZATION REPORT REGION 8 DRY CLEANING SITES CRYSTAL CLEANERS SITE SITE NO. 8-51-022

# WORK ASSIGNMENT NO. D004434-19

Prepared for:

### New York State Department of Environmental Conservation Albany, New York

Prepared by:

MACTEC Engineering and Consulting, PC Portland, Maine

#### MACTEC NO. 3612072075

### **MARCH 2007**

This document was prepared for the sole use of New York State Department of Environmental Conservation, the only intended beneficiary of our work. No other party shall rely on the information contained herein without prior written consent of MACTEC Engineering and Consulting, PC.

Submitted by:

Approved by:

John W. Peterson Project Manager

William I. Weber, P.E. Program Manager

# **TABLE OF CONTENTS**

| LIST C | OF TABLES   | iii |
|--------|---|-----|
| LIST C | OF FIGURES  | iv  |
| LIST C | DF ACRONYMS   | v   |
| 1.0    | INTRODUCTION  | 1-1 |
| 2.0    | SITE BACKGROUND AND PHYSICAL SETTING                              | 2-1 |
| 2.1    | SITE LOCATION   | 2-1 |
| 2.2    | SITE HISTORY  | 2-1 |
| 2.3    | PREVIOUS INVESTIGATIONS   | 2-2 |
| 2.4    | PHYSICAL SETTING  | 2-3 |
| 2.5    | FILE REVIEW   | 2-5 |
| 2.6    | SUMMARY OF DATA RECORDS SEARCH AND ASSESSMENT FINDINGS            | 2-6 |
| 3.0    | SCOPE OF WORK   | 3-1 |
| 3.1    | GENERAL FIELD ACTIVITIES  | 3-1 |
|        | 3.1.1 Mobilization  | 3-2 |
|        | 3.1.2 Health and Safety   | 3-2 |
|        | 3.1.3 Decontamination   | 3-2 |
|        | 3.1.4 Investigation Derived Wastes                                | 3-2 |
| 3.2    | BASEMENT OPEN WATER SAMPLING                                      | 3-3 |
| 3.3    | GEOPROBE® BORINGS AND SAMPLING                                    | 3-3 |
| 3.4    | SUB-SLAB SOIL VAPOR SAMPLING                                      | 3-6 |
| 3.5    | ADDITIONAL GEOPROBE <sup>®</sup> BORINGS AND GROUNDWATER SAMPLING | 3-7 |
|        | 3.5.1 Geoprobe Sampling and Microwell Installations               | 3-7 |
|        | 3.5.2 Existing Microwell Sampling                                 | 3-8 |
| 3.6    | WATER LEVEL SURVEY  | 3-9 |
| 3.7    | SITE SURVEY   | 3-9 |
| 4.0    | DATA ASSESSMENT   | 4-1 |
| 4.1    | ANALYTICAL RESULTS  | 4-1 |
|        | 4.1.1 Soil Sample Results   | 4-3 |
|        | 4.1.2 Groundwater Sample Results                                  | 4-3 |
|        | 4.1.3 Surfacewater Sample Results                                 | 4-5 |
|        | 4.1.4 Soil Vapor Sample Results                                   | 4-5 |
| 4.2    | POTENTIOMETRIC SURFACE MAP  | 4-6 |
| 5.0    | INVESTIGATION FINDINGS  | 5-1 |
| 6.0    | REFERENCES  | 6-1 |
| TABLE  | <u>ES</u>   |     |

**FIGURES** 

# TABLE OF CONTENTS (CONTINUED)

#### **APPENDICES:**

| APPENDIX A | FACILITY INFORMATION REPORT AND PREVIOUS INVESTIGATIONS |
|------------|---|
| APPENDIX B | SITE PHOTOGRAPHS  |
| APPENDIX C | FIELD DATA RECORDS                                      |
| APPENDIX D | SITE SURVEY   |
| APPENDIX E | DATA USABILITY SUMMARY REPORT                           |

# LIST OF TABLES

# 3.1 Microwell Data

- 4.1 Soil Sample VOC Results
- 4.2 Groundwater VOC Results
- 4.3 Oct/Nov Groundwater Results
- 4.4 Surfacewater VOC Results
- 4.5 Soil Vapor VOC Results

# LIST OF FIGURES

- 1.1 Site Location
- 2.1 Historic Sample Locations
- 3.1 Sample Locations
- 4.1 PCE Results in Groundwater
- 4.2 Groundwater Elevations and Flow Direction

# LIST OF ACRONYMS

| ASP               | Analytical Services Protocol                            |
|-------------------|---|
| ASTM              | American Society for Testing and Materials              |
| bgs               | below ground surface                                    |
| Chemtech          | Chemtech Consulting Group, Inc.                         |
| cis-1,2-DCE       | cis-1,2-Dichloroethene                                  |
| EDR               | Environmental Data Resources, Inc.                      |
| °F                | degrees Fahrenheit                                      |
| MACTEC<br>msl     | MACTEC Engineering and Consulting, P.C. mean sea level  |
| NYCRR             | New York Codes, Rules, and Regulations                  |
| NYS               | New York State  |
| NYSDEC            | New York State Department of Environmental Conservation |
| NYSDOH            | New York State Department of Health                     |
| PCE               | tetrachloroethene                                       |
| PID               | photoionization detector                                |
| ppm               | parts per million                                       |
| PVC               | polyvinyl chloride                                      |
| Report            | Site Characterization Report                            |
| ROW               | right of way  |
| SC                | Site Characterization                                   |
| Site              | Crystal Cleaners Site                                   |
| SVOC              | Semi-Volatile Organic Compounds                         |
| 1,1,1-TCA         | 1,1,1-trichloroethane                                   |
| 1,1,2-TCA         | 1,1,2-trichloroethane                                   |
| TCE               | trichloroethene   |
| TCL               | Target Compound List                                    |
| TICS              | tentatively identified compounds                        |
| trans-1,2-DCE     | trans-1,2-Dichloroethene                                |
| μg/Kg             | micrograms per Kilogram                                 |
| μg/L              | micrograms per Liter                                    |
| μg/m <sup>3</sup> | micrograms per cubic meter                              |
| USEPA             | United States Environmental Protection Agency           |
| VOC               | volatile organic compound                               |
| WA                | Work Assignment   |

# **1.0 INTRODUCTION**

MACTEC Engineering and Consulting, P.C. (MACTEC), is submitting this Site Characterization Report (Report) to the New York State Department of Environmental Conservation (NYSDEC). This Report addresses the Site Characterization (SC) at the Crystal Cleaners site (Site) in Corning, New York (Figure 1.1). This Report was prepared in response to Work Assignment (WA) No. D004434-19 (NYSDEC, 2005), and in accordance with the requirements of the July 1997 Superfund Standby Contract No. D004434 between the NYSDEC and MACTEC.

This Report is one of five site-specific Reports for the Region 8 Dry Cleaning Sites multiple site Site Characterizations WA. The other four Reports address the sites listed below:

- Loohns Corning (Site No. 8-51-028 replaces Former Your Way Cleaners)
- Former American Dry Cleaners (Site No. 8-08-036)
- Castle Cleaners (Site No. 8-08-034)
- Former Helwigs Dry Cleaners (Site No. 8-51-023)

The Crystal Cleaners site, Site No. 8-51-022, is currently listed as a potential hazardous waste site, or "P" site, by the NYSDEC, because insufficient information exists to determine whether wastes were disposed of at the site and whether, if present, those wastes pose a potential significant threat to public health or the environment (New York State (NYS), 2006).

The purpose of the SC is to provide information to be used by the NYSDEC to reclassify the site to one of the following categories:

| Class 1    | Hazardous waste constitutes a significant threat to public health or the environment, as described in Title 6 of the New York Codes, Rules, and Regulations (NYCRR) Part 375 (NYS, 2006); and the significant threat to public health or the environment is causing, or presents an imminent danger of causing, either irreversible or irreparable damage to the environment. |
|------------|---|
| Class 2    | Hazardous waste constitutes a significant threat to public health or the environment as defined in 6 NYCRR 375 (NYS, 2006)  |
| Class 3    | Hazardous waste does not presently constitute a significant threat to public health or the environment as defined in 6 NYCRR 375 (NYS, 2006).   |
| Not Listed | Sites where hazardous waste disposal is not documented.   |

To complete its reclassification, the NYSDEC requires information to establish the following:

• The existence of documented hazardous waste disposal, as defined in Title 6 of the NYCRR Part 371 (NYS, 1999a).

- The site's significance with respect to the threat it poses to public health and the environment as defined in 6 NYCRR Part 375 (NYS, 2006).
- Identification of contaminant source.

To implement the SC, this WA was authorized and was divided into the following three tasks:

Task 1: Work Plan Development, Task 2: Subsurface Investigations, and Task 3: Reporting.

This Report presents reclassification documentation collected by MACTEC during Task 1 and Task 2 so the NYSDEC can recommend follow up action for the site (i.e., reclassify, delist, or perform additional investigations). Resources used to prepare this Report include: (1) information provided in the Work Assignment, (2) appropriate guidelines in the NYSDEC Draft DER-10 Guidance (NYSDEC, 2002), (3) results of previous investigations, if applicable, and (4) results of the SC investigation.

This Report is divided into five sections. Section 1 is the introduction. Section 2 presents information collected during Task 1, which included a search of state and county site records, and a site inspection. Because Task 1 activities did not develop adequate data on which to base a delist or reclassification recommendation Task 2, Subsurface Investigation, was conducted. Section 3 of this Report presents the work conducted during Task 2: Subsurface Investigations. Section 4 presents results of the field investigation. Section 5 presents an investigation summary.

#### 2.0 SITE BACKGROUND AND PHYSICAL SETTING

On September 9 and 10, 2005, MACTEC personnel reviewed available records from the NYSDEC office in Albany, New York, and visited the City of Corning, New York town offices. Information was also collected from the Site owner by the NYSDEC. As part of the review, MACTEC ordered a copy of an Environmental Data Resources, Inc. (EDR) report which provides a listing of federal and state governmental information pertaining to potential and documented environmental impacts, both at the Site and within the American Society for Testing and Materials (ASTM) recommended search radii. Complete lists of all recommended ASTM record searches for standard due diligence requirements are included in the EDR report provided under separate cover. The information was reviewed to support a Site classification, and to help prepare the scope of work for the SC field investigations. The information collected from these sources is summarized below.

# 2.1 SITE LOCATION

Crystal Cleaners is located at 343 West Pulteney Street, in the City of Corning, Steuben County, New York (Figure 1.1). The Crystal Cleaners property consists of approximately 0.58 acres including a retail building and a large parking lot and service station. The current Site building was constructed in 1970 and includes a mini-mart and service station, a dry cleaning business, and a vacant store. It is a one story building with a basement located only under the dry cleaner.

Residential property is located north, south and east of the Site and commercial property is located west of the Site.

# 2.2 SITE HISTORY

The property lot was purchased from Corning, Inc. in December 1969. The current Site building was constructed in 1970. It is a one story building with a basement located only under the dry cleaner. The water table appears to be located at approximately the level of the basement slab (there is an open 1 ft. deep pool of water in the basement - water level is only two inches below grade of concrete slab). There is a sump pump located in the basement and the basement apparently floods during most rain events (Davis, 2005).

The property has contained a gas station since at least 1974, when four 4000 gallon gasoline tanks were installed at the site. An additional 1000 gallon kerosene tank was installed in 1984. The gasoline tanks were removed in 1992 and replaced with two 8000 gallon gasoline tanks that are still in service. The 1000 gallon kerosene tank was abandoned in place in 2000 and a new 1000 gallon kerosene tank was installed. The tank information is listed in the Facility Information Report, included in Appendix A.

The date of first dry cleaner is not known, but Corning One Hour Martinizing appeared in the 1981 Corning City Guide at 343 Pulteney West. The 1989 Corning City Guide lists the property as One Hour Techni Clean. The current manager of the dry cleaner took over lease of the property in 1994 and changed the name to Crystal Cleaner. He stated that the original operation was a wet to dry system. This was converted to a dry to dry system in the mid 1980's. He updated the equipment and added a spill protection in the mid-1990's (Davis, 2005).

According to the City of Corning Department of Public Works, the water main along Pultney Road was installed in 1907 and the sewer line was installed around 1908 (Panton, 2005). It is therefore assumed that Crystal Cleaners has always been serviced by public water and sewer.

## 2.3 **PREVIOUS INVESTIGATIONS**

Chlorinated solvents were first detected in the City of Corning's water supply wells number 1 and 2 in the early 1980's. These wells are located approximately 950 feet and 1300 feet from Crystal Cleaners, respectively, along the banks of the Cohocton River (Figure 1.1). These two wells are both screened from approximately 50 to 70 feet below ground surface (bgs). Pumping tests indicate that the wells can produce up to one million gallons a day, although they are currently run on an alternating 10 day schedules, with one well producing approximately one million gallons over the ten day period (total running time of approximately 24 hours), and then rotating to the next well for the subsequent ten day period (Panton, 2005).

Tetrachloroethene (PCE) has been detected at low concentrations in both wells. Concentrations typically range from non-detect to 14 micrograms per liter ( $\mu$ g/L), with slightly higher concentrations detected in Well 2 then Well 1. PCE was detected in the samples collected in the 2004 round at concentrations of 1.1  $\mu$ g/L in Well 1 and 11  $\mu$ g/L in Well 2. The NYS Class GA standard for PCE is 5  $\mu$ g/L.

In preparation for selling the Site property, the owner of the plaza that includes Crystal Cleaner hired Teeter Environmental Services, Inc. to conduct a Phase II Site assessment in 2005, primarily for the purpose of determining the condition of the underground fuel tanks for the Site's gas station (Teeter, 2005). Previous investigation boring locations are shown of Figure 2.1. The assessment included the completion of six soil borings (BS-1 to BS-6) to approximately 16 feet bgs; including the collection of groundwater grab samples from four locations. Groundwater samples collected from B-1 and B-3 were analyzed for volatile petroleum hydrocarbons (United States Environmental Protection Agency (USEPA) Method 8260-STARS list) and fuel related semi-volatile organic compounds (SVOCs) (USEPA Method 8270C-STARS list). Groundwater samples collected from B-4 and B-5 were analyzed for volatile organic compounds (VOCs) via the full USEPA Method 8260 list. Because no visual evidence of contamination was noted, and no readings above background were detected on the photoionization detector (PID) when screening soil samples collected from the borings, no soil samples were submitted for laboratory analyses.

SVOCs were not detected in the groundwater samples analyzed. Naphthalene was detected in borings B-1, B-3, and B-4 at concentrations of 29.7  $\mu$ g/L, 9.9  $\mu$ g/L, and 5.8  $\mu$ g/L, respectively, in relation to the NYS guidance value of 10  $\mu$ g/L. Toluene and xylene were detected in B-1 at concentrations of 6.2  $\mu$ g/L and 5.2  $\mu$ g/L respectively, in comparison to the NYS groundwater standard of 5  $\mu$ g/L. PCE was detected in the two samples analyzed for the full list of VOCs, at concentrations of 7  $\mu$ g/L in B-5 and 43  $\mu$ g/L in B-4, in comparison with the NYS groundwater standard of 5  $\mu$ g/L. The highest concentration was detected at the southern edge of the site property. Although PCE was determined to be present in Site groundwater at concentrations above the NYS groundwater standards of 5  $\mu$ g/L, no groundwater contours were plotted and no source of the chlorinated solvents in site soils was identified. The Teeter Phase II Report is included in Appendix A.

### 2.4 PHYSICAL SETTING

#### Topography

The Site is located in the Cohocton/Chemung River Valley, which runs east-west. The Site property is located at 940 feet above mean sea level (msl), sloping slightly to the south. The surrounding area slopes slightly down to the south, before reaching the dike at the edge of the Chemung River, located 900 feet from the Site. The Chemung River is located at an elevation of

approximately 930 feet above msl, just south of the dike. The topography to the northeast of the site is relatively flat for approximately 0.7 miles, and then rises to a ridge at 1600 feet above msl approximately 1.5 miles from the Site.

#### Climate

The climate of the area is characterized by moderately warm summers and cold winters. Mean monthly temperatures range from 23 degrees Fahrenheit (°F) in January to 68°F in July. Average annual precipitation is 32 inches. Average annual snowfall is 37 inches (National Climatic Data Center, 2004).

#### **Surface Water Hydrology**

The site is not located in an area mapped as either a 100 year or 500 year flood zone (EDR, 2006). Surface drainage from the site generally follows the topography, flowing toward the municipal storm drains located on Pulteney Street. These storm drains flow to a treatment plant located approximately 2.4 miles east of the site (Panton, 2005). The treatment plant discharges to the Chemung River downstream of the site.

#### **Groundwater Hydrology**

The Chemung River is a local groundwater discharge area. Groundwater at the Site was encountered at approximately 10 to 12 feet bgs, and is interpreted to flow south towards the River. Groundwater contours for the greater Corning area indicate that groundwater at the site flows in a south-easterly direction, towards the river (USGS, 1982).

#### Geology

Overburden soils at the Site consisted primarily of fluvial silts, sands and gravels. Surficial geology is mapped as oxidized, non calcareous, fine sand to gravel (Muller, 1986). Teeter described site soils as varying horizontally and vertically and generally consisting of brown and reddish brown gravelly silt with varying amounts of sand, sandy gravel with little silt and clayey silt with some sand and gravel. Based on regional geologic mapping (Rickard and Fisher, 1970) bedrock is expected to consist of shale and siltstones associated with the Upper Devonian West

Falls Group; specifically, the Gardeau Formation, consisting of shale and siltstone; and/or Roricks Glen shale (Rickard and Fisher, 1970).

### Site Walkover

On September 9, 2005 MACTEC and the NYSDEC personnel conducted a walkover of the Site area.

| NAME            | TITLE  | AFFILIATION/TELEPHONE   |
|-----------------|--|---|
| Charles Staples | Site Lead  | MACTEC Engineering and<br>Consulting<br>207-775-5401                    |
| Mathew Dunham   | Environmental Engineer<br>NYSDEC Project Manager | NYSDEC Division of<br>Environmental Remediation, Albany<br>518-402-9812 |
| Richard Davis   | Site Manager                                     | Owner/Manager of Crystal Cleaner  |
| Donald Styker   | Site Owner                                       | Owner of Site Property  |

#### SITE WALKOVER ATTENDEES

The site walkover consisted of viewing the Crystal Cleaners property, including the inside of the facility and the basement, and the surrounding neighborhood to assess possible contamination sources and the logistical concerns for the field program. MACTEC personnel documented the walkover with photographs (Included in Appendix B).

Obvious sources of contamination were not observed (i.e., leaking drums); however, detailed inspections of potential sources, including site soils were not conducted during the site walkover. Additional information for the purpose of identifying potential sources was obtained during Task 2.

# 2.5 FILE REVIEW

MACTEC reviewed files from various state and local agency offices to develop information to support a reclassification or delisting, and to help prepare the scope of work for the SC field investigations. The EDR report was also reviewed for relevant site information.

#### 2.6 SUMMARY OF DATA RECORDS SEARCH AND ASSESSMENT FINDINGS

Under federal and state regulations a solid waste may be regulated as a hazardous waste if it is a material included in one of USEPA's or the NYSDEC's lists of hazardous wastes. If a material is regulated because of its inclusion on a federal or state list, it is commonly referred to as a "listed hazardous waste." A waste may also be regulated under the Resource Conservation and Recovery Act as a "characteristic hazardous waste" if it exhibits one of the characteristics of toxicity, corrosivity, reactivity, or flammability.

Results of sampling and analysis of the Corning supply wells No. 1 and 2 and of the groundwater at the Site property indicated the presence of chlorinated solvents (PCE) in groundwater. Spent chlorinated solvents not originating from a household sources, including PCE are included on both the USEPA's and the NYSDEC's lists of hazardous wastes. Under 6 NYCRR Part 371.4(a) (1), these spent solvents constitute hazardous waste from non-specified sources. Disposal of these chlorinated solvents has been confirmed by available analytical results, but the source area has not been identified.

As defined by 6 NYCRR Part 375, significant threat can be established by documenting a contravention of environmental standards. Surface water and groundwater are the only media for which NYS has promulgated standards. Under NYS Water Quality Regulations (6 NYCRR Parts 700-705) the state has set numeric standards that are the maximum concentration of compounds in groundwater and surface water that protect public health and/or the environment (NYS, 1999b).

Groundwater analytical data collected at the Site indicates that PCE contamination in Site groundwater exceeds the NYS groundwater standards. It was not known if this PCE contamination originated from the Site, or if it was migrating off-site and contaminating the public supply wells. It was therefore not known if the Site posed a significant threat. As a result, the SC field investigations were conducted to:

- collect the data necessary to verify the likelihood of uncontrolled waste disposal,
- determine if potential contamination is present on Site, and is migrating offsite, and
- provide sufficient information to allow the NYSDEC to re-classify the site.

# 3.0 SCOPE OF WORK

To reclassify the Site, the NYSDEC requires data documenting hazardous waste disposal as set forth in 6 NYCRR Part 371, and the potential significant threat to human health and the environment as defined by 6 NYCRR Part 375. Because data necessary to determine if the contaminants present in Site groundwater (specifically PCE) originated from the Site and are migrating off-site, or if those contaminants pose a potential significant threat to human health and the environment were not available in federal and state files reviewed during Task 1, additional field investigations were performed as described below. Task 2 activities included the Field Investigation. The objective of Task 2 activities was to determine, if possible, whether the VOCs detected in the Site groundwater originated from the Site, and whether potential onsite VOCs contamination is migrating offsite. An additional objective was to collect data to determine if the Site is the source of PCE contamination in the two village supply wells. Task 3 is the preparation and distribution of this Report.

#### **TASK 2 - FIELD INVESTIGATIONS**

The following subsections describe the activities conducted during the field investigation portion of the Site SC. The work followed the scope of work as outlined in the SC Work Plan (MACTEC, 2005). The field investigation was conducted in accordance with the specifications presented in the Quality Assurance Program Plan (ABB Environmental Services, 1995) and the site specific Quality Assurance Project Plan. Off-site laboratory analyses was performed by Chemtech Consulting Group, Inc. (Chemtech), a New York State Department of Health (NYSDOH) approved laboratory. Off-site laboratory analysis complied with the NYSDEC Analytical Services Protocols (ASP) (NYSDEC, 2000).

#### **3.1 GENERAL FIELD ACTIVITIES**

General field activities, including mobilization, health and safety, and decontamination, are described in the following subsections.

#### 3.1.1 Mobilization

After receiving the NYSDEC authorization to begin fieldwork, MACTEC and its subcontractors conducted utility clearance, mobilized to the Site and began the field exploration program.

A field team orientation meeting was held on-site with MACTEC personnel to familiarize field workers with site history, health and safety requirements, equipment calibration procedures, and other field procedures.

### 3.1.2 Health and Safety

Field investigation activities were conducted at Level D personal protection. Based on PID readings, no upgrades on personal protection were warranted.

### 3.1.3 Decontamination

Sampling methods and equipment for this field program were chosen to minimize investigation derived waste and minimize possibility of cross contamination. Disposable sampling equipment was used as much as practical to minimize decontamination time and water disposal.

Non disposable sampling equipment was decontaminated by 1) scrubbing the sample collection equipment with potable water and Liquinox, rinsing with potable water, rinsing with deionized water, and then allowing the equipment to air dry, or 2) steam cleaning the equipment and then allowing the equipment to air dry. Decontamination fluids did not exhibit visual or olfactory evidence of contamination and were released to the ground surface in the area of the exploration, so as to allow the liquids to infiltrate into the soil.

#### 3.1.4 Investigation Derived Wastes

The field investigation did not result in the generation of wastes that were considered hazardous (i.e., no visual or olfactory signs of contamination, and no PID readings above 5 parts per million (ppm) were detected). Therefore drill cuttings and purge water resulting from the investigation were placed on the ground surface in the area of exploration and personal protective equipment and disposable sampling equipment were double bagged and disposed of as non-hazardous refuse.

#### **3.2 BASEMENT OPEN WATER SAMPLING**

Two samples were obtained from surface water present in the basement below the dry cleaner on January 23, 2006. One sample (SW-2) was collected from the approximate 4 foot by 4 foot by one foot deep open "pool" located in the center of the basement, and one sample (SW-1) was collected from the sump, located in the southeast corner of the basement. Approximate Sample locations are shown on Figure 3.1. Samples were collected directly into the sample vials.

The samples were submitted to the analytical laboratory and analyzed for target compound list (TCL) VOCs using USEPA OLM04.2 methods as described in the NYSDEC ASP of June 2000.

# 3.3 ROUND ONE GEOPROBE® BORINGS AND SAMPLING

Field investigation activities included the completion of Geoprobe<sup>®</sup> borings, the collection and analysis of groundwater, soil, and soil vapor samples, and the installation of microwells. Geoprobe sampling was conducted from January 23 to 26, 2006. The purpose of the activities was to provide groundwater data for comparison to NYS Class GA Groundwater Quality Standards set forth under 6 NYCRR Parts 700-705 (NYS, 1999b), and to assist the NYSDEC in evaluating significant threat to public health and the environment as defined by 6 NYCRR Part 375 (NYS, 2006). Soil sample results were used to assess whether hazardous waste constituents are present in site soils, and, if possible, confirm a source of chlorinated solvents. Soil vapor sample results were used to evaluate whether VOCs present in soil and/or groundwater are migrating towards occupied buildings via vapor migration.

MACTEC used a Geoprobe<sup>®</sup> sampling device to collect groundwater and soil samples to identify potential chlorinated solvents. The Geoprobe<sup>®</sup> pushes and/or hammers rods and probe tips into the subsurface for sample collection. A total of 17 borings and three soil vapor collection points were completed, including the installation of four microwells. A total of 35 Groundwater, four soil, and three soil vapor samples were collected at this Site.

In addition, 10 groundwater samples were collected from six Geoprobe<sup>®</sup> borings on February 6 and 7, 2006 for an investigation conducted by MACTEC at the Former Helwigs Cleaners Site (MACTEC, 2006). These boring locations are also shown on Figure 3.1 (HGW-1 to HGW-6).

MACTEC worked closely with the NYSDEC, the Crystal Cleaners owner, the neighboring property owners, and utility companies to obtain access to the exploration locations. Boring locations are shown on Figure 3.1. Locations were chosen to determine groundwater conditions upgradient and downgradient of, as well as adjacent to, the site building. Additional downgradient samples were collected to determine if potential contaminants are migrating in groundwater towards the City supply wells.

**Soil Sampling.** Soil samples were collected using a four-foot long 2-inch diameter core sampler with an acrylic liner for the collection of discrete subsurface soil samples. Soil samples were collected continuously from the ground surface to the top of the groundwater table. PID headspace readings were used to screen soil samples for the presence of VOCs as each soil sample was removed from the sample collection tube. Samples were described using the Unified Soil Classification System. The sample description and classification, VOC headspace reading, and boring observations were recorded on the Field Data Record, included in Appendix C. Based on the PID readings and physical evidence such as color or odor, four unsaturated soil samples were submitted to the laboratory for analysis. Samples exhibiting the highest PID readings and physical evidence of contamination were selected for analysis. Soil samples were shipped to Chemtech for analyses of TCL VOCs using USEPA OLM04.2 methods as described in the NYSDEC ASP of June 2000. Off-site laboratory analysis included Category B deliverables.

**Groundwater Sampling.** Groundwater samples were collected using a small diameter stainless steel wire wound screen that was exposed to the aquifer, after being pushed to the desired depth interval. A peristaltic pump was used for the collection of discrete groundwater samples. One tubing volume of water was purged and one set of parameters including temperature, conductivity, pH, and turbidity were collected before sampling. VOC samples were collected at a low purge rate (approximately 100 milliliters per minute) to minimize potential volatilization.

To assess vertical extent of contamination, groundwater samples were collected from two locations in each boring, the water table and 8-10 feet into the water table (8 feet below the first sample). For boring GW-1, only one water sample could be collected due to cobbles, and three water samples were collected from boring GW-4. Each boring was completed to at least 10 feet into the water table, present at approximately 10 feet bgs. Groundwater samples were shipped to Chemtech for analyses TCL VOCs using USEPA OLM04.2 methods. Off-site laboratory analysis included Category B deliverables.

**Microwell Installation.** To determine groundwater flow direction at the Site, four Geoprobe<sup>®</sup> borings (GW-1, GW-5, GW-6, and GW-10) were completed as microwells. Microwell locations are shown on Figure 3.1. Groundwater was encountered at approximately 10 feet bgs. The one-inch diameter microwells were installed after soil and groundwater samples were collected from each boring. The microwells were installed as piezometers and used for water level measurements only. Microwells were constructed with schedule 40 polyvinyl chloride (PVC), with 10 foot lengths of 0.01-inch machine slotted well screens. The well screens for GW-5, GW6, and GW-11 were set with approximately 2 feet of screen above the water table to determine water table elevations and create a potentiometric map. GW-1 was set from 15 to 25 feet bgs, approximately five feet below the water table. The wells were constructed with a # 0 sand pack or native soil backfill and sealed at the ground surface with bentonite. The wells were completed with a locking cap and a six inch flush mount cover. The wells were developed for twenty minutes with a peristaltic pump to clean the screen and determine if the wells were conductive with groundwater.

**Soil Vapor Sampling.** Based on proximity to nearby residences and/or businesses, and discussions with the NYSDEC Project Manager, three soil vapor samples (GV-1 to GV-3) were collected to evaluate the potential vapor migration of contaminants from the groundwater. Soil vapor samples were collected using a Geoprobe<sup>®</sup> sampling device.

The Geoprobe<sup>®</sup> rods were pushed to between 6 and 8 feet bgs (expected to be below the rain infiltration line, but above the water table fringe zone). Soil vapor collected just above the water table gives an indication of the possible vapor migration from potentially contaminated groundwater.

Soil vapor samples were collected from the Geoprobe<sup>®</sup> PRT (Post Run Tubing) points. Upon reaching 6 feet bgs, the Geoprobe<sup>®</sup> rods were pulled back 0.5 feet, leaving a disposable stainless steel tip in the ground and exposing the bottom of the open rods to the soil. A <sup>1</sup>/<sub>4</sub>-inch outside diameter low density polyethylene tubing was then attached to the bottom of the Geoprobe rods using a stainless steel adapter and o-rings, allowing air to be pulled from below the bottom of the rods. The tubing was then purged at the ground surface and the soil vapor sample was collected from the tubing into the sample container. In addition, the outside of the rods were sealed at the ground surface with pre-hydrated bentonite. Approximately 2 liters of soil vapor, plus the volume of the tubing, was purged at a rate of 400 ml/min using 580B OVM PID pump before collecting samples. During the soil vapor purge, vapors were screened using a PID. In addition, helium leak

tests were conducted on a subset of the Region 8 Dry Cleaners Sites soil vapor samples to ensure samples were representative of sub-surface conditions and not outdoor ambient air. Helium tests were set up by encapsulating the sample point with a bucket sealed to the ground surface with bentonite. The soil vapor samples were collected with one-liter SUMMA<sup>®</sup>-type canisters with flow valves (set to approximately 30 minutes per sample). Flow into the canisters was less than 0.1 liters per minute, as requested by the NYSDOH. Samples were sent to Chemtech for VOC analysis by USEPA Method TO-15.

#### 3.4 SUB-SLAB SOIL VAPOR SAMPLING

Based on site observations during the field program, and discussions with the NYSDEC and the property owner, one sub-slab soil vapor sample was collected on January 26, 2006 from below the Site building concrete slab, within the vacant retail space in the east end of the building (SV-1; approximate location shown on Figure 3.1). To complete the sampling, a one-inch diameter hole was drilled through the concrete slab with a hammer drill. The hole was then swept to remove drill cuttings/dust from the area. A ¼-inch piece of Teflon tubing was inserted through a 1" diameter rubber stopper, and placed into the hole, so that the bottom of the tubing was below the slab floor and the stopper rested inside the one-inch hole, forming a seal. The stopper was then covered with bees wax to provide an impenetrable seal for the migration of indoor air into the sub-slab. One 60 cubic centimeter (cc) volume of air was purged from the tubing with a polyethylene syringe. A 6-liter SUMMA<sup>®</sup>-type canister with a 24-hour flow valve was connected to the tubing. The time of sample collection, canister vacuum (in inches Mercury), weather conditions, and barometric pressure were recorded in the field log book.

Approximately 24 hours after sample collection, the flow valves were shut off. The time, remaining vacuum in the canister, and barometric pressure were noted in the field log book. The samples were shipped to Chemtech laboratories for analyses of VOCs via USEPA Method TO-15. Laboratory analysis included Category B deliverables.

Upon completion of the sampling, the tubing and stopper were removed from the building floor and the hole was filled completely with a fast drying hydraulic concrete (i.e. Quickcrete).

# 3.5 ROUND TWO GEOPROBE<sup>®</sup> BORINGS AND GROUNDWATER SAMPLING

Based on analytical results from the January 2006 sampling event, it was determined that additional groundwater samples and water elevation points were necessary to better characterize the Site and groundwater flow direction. The additional field investigation activities, conducted from October 31 to November 2, 2006, included the completion of eight Geoprobe<sup>®</sup> borings (GS-18 to GS-25) and the collection and analysis of 16 groundwater samples and the installation of three microwells.

### 3.5.1 Round Two Geoprobe Sampling and Microwell Installations

During Round Two, MACTEC used a Geoprobe<sup>®</sup> 66 DT rubber-mounted track rig sampling device to collect groundwater and soil samples to identify potential chlorinated solvents.

MACTEC worked closely with the NYSDEC, the Crystal Cleaners owner, the neighboring property owners, and utility companies to obtain access to the exploration locations. The additional locations for the borings are shown on Figure 3.1. Groundwater boring locations were chosen to determine if groundwater concentrations upgradient and crossgradient of the Site exceeded the NYS Class GA criteria.

**Soil Sampling.** Soil samples were collected using a 4-foot long 2-inch diameter core sampler with an acrylic liner for the collection of discrete subsurface soil samples. Soil samples were collected continuously from the ground surface to 16 feet bgs at all boring locations for visual characterization. Samples were described using the Unified Soil Classification System. PID headspace readings were used to screen soil samples for the presence of VOCs. The sample description and classification, VOC headspace reading, and boring observations were recorded on the Field Data Record, included in Appendix C. No samples were collected for laboratory analyses.

**Groundwater Sampling.** Geoprobe groundwater samples were collected as described in Section 3.3 above.

To assess vertical extent of contamination, MACTEC collected groundwater samples from two depth intervals in each boring, the water table and 10 feet into the water table (10 feet below the first sample). The water table was encountered at approximately 15 feet bgs. Groundwater

sampling records are provided in Appendix C. Groundwater samples were shipped to Chemtech for analyses of TCL VOCs using USEPA OLM04.2 Methods as described in the NYSDEC ASP of June 2000. Off-site laboratory analysis included Category B deliverables.

**Microwell Installation.** To better characterize groundwater flow direction at the Site, three additional Geoprobe<sup>®</sup> borings (GW-18, GW-19, and GW-23) were completed as microwells. Microwell GW-18 was installed up gradient of the Site in the City right of way (ROW) of Taylor Street. Microwell, GW-19, was also installed up gradient of the Site in the ROW of Townsend Avenue. Microwell, GW-23, was installed cross-gradient of the Site in the ROW of Remington Boulevard. Microwell locations are shown on Figure 3.1.

Groundwater was encountered at approximately 15 feet bgs. The one-inch diameter microwells were installed after groundwater samples were collected from each boring. The microwells were installed as piezometers and used primarily for water level measurements. Microwells were constructed with schedule 40 PVC, with 10 foot lengths of 0.01-inch machine slotted well screens. The well screens for GW-18, and GW-19 were set from 11 to 21 feet bgs, with approximately 4 feet of screen above the water table. GW-23 was set from 13.5 to 23.5 feet bgs, with approximately 2 feet of screen above the water table. From these piezometers, water table elevations were determined and a potentiometric map of the Site was created. The wells were constructed with a # 0 sand pack or the native soil backfill and sealed at the ground surface with cement. The wells were completed with a locking cap and a six inch flush mount cover. The wells were developed for twenty minutes with a peristaltic pump to clean the screen and determine if the wells were conductive with groundwater. Well construction diagrams are included in Appendix C.

#### 3.5.2 Existing Microwell Sampling

Four existing 1-inch diameter microwells (GW-1, GW-5, GW-6, and GW-10) were sampled on November 1, 2006. All four microwells were previously installed by MACTEC in January 2006. The existing microwells were sampled in accordance with the USEPA "low flow" guidance. Groundwater parameters including water levels, turbidity, temperature, dissolved oxygen, specific conductance, pH and redox potential were recorded in a field log and on a field data record. The low flow sampling requirements were met while sampling these four existing wells. Groundwater Field Data Records are available in Appendix C. These samples were submitted to Chemtech and analyzed for TCL VOCs using USEPA OLM04.2 methods as described in the NYSDEC ASP of June 2000.

# **3.6 WATER LEVEL SURVEY**

Water levels were measurements were conducted on February 9, 2006 and February 1, 2007. The February 9, 2006 round included water level measurements at the initial microwells (GW-1, GW-5, GW-6, and GW-10). The February 1, 2007 round consisted of measuring water levels at six of the seven microwells (GW-10 was buried under snow and ice and not accessible). Well caps were opened and the wells were allowed to equilibrate to atmospheric pressure. The depths of the wells as well as the depth to water were measured using a conductivity probe from the top of well risers. Groundwater table elevations were calculated from the well riser elevations (subsection 3.6). Well information and groundwater measurements are presented in Table 3.1.

# 3.7 SITE SURVEY

Upon completion of field investigation activities, MACTEC's survey subcontractor, Lu Engineers, complete a survey of the Site and surrounding area and created a base map. Horizontal locations were tied to the NYS Plane Coordinate System using North American Datum (NAD) of 1983. The site plan provides horizontal locations of relevant Site features, including surrounding homes and businesses at a scale of 1 inch to 50 feet. Relevant features include, but are not limited to all structures, buildings, roads, fences, new monitoring wells, underground utilities, fire plugs, and power poles.

Vertical elevations of the three new micro wells were tied to msl, North Atlantic Vertical Datum (NAVD) of 1988, and measured to an accuracy of 0.01 feet. Horizontal well measurements were to an accuracy of 0.1 feet.

The base map was used to accurately locate all Geoprobe<sup>®</sup> sample points, microwells, and any other media sampling locations. Temporary sample points were located using a Trimble global positioning system. Sample points are included on Figure 3.1, and the Lu Engineers survey map is included in Appendix D.

### 4.0 DATA ASSESSMENT

This section presents results of the laboratory analyses for soil, groundwater, air samples and additional groundwater samples collected during Task 2, as well as results of the water level survey.

#### 4.1 ANALYTICAL RESULTS

Soil, groundwater, and soil vapor analytical results were compared to appropriate standards or guidelines. Reported concentrations of individual analytes indicating contravention of standards or guidelines are summarized in the following sections, and noted on Tables 4.1, 4.2, 4.3, and 4.5.

A Data Usability Summary Report (DUSR) was completed in accordance with the NYSDEC's Guidance for the Development of Data Usability Summary Reports (NYSDEC, 1997). This report and complete analytical results including tentatively identified compounds (TICS) are presented in Appendix E.

Based on laboratory or data usability review, some of the data was qualified with a J, B, D and/or an R. Compounds were qualified J if the concentration listed was an estimated value, which was less than the specified minimum reporting limit but greater than the instrument detection limit. Compounds qualified J were analyzed for and determined to be present in the sample and the mass spectrum of the compound met the identification criteria of the method. The reporting limits for most target VOCs using the OLM04.2 Methods, including the target chlorinated solvents compounds were 10  $\mu$ g/L. This is above most of the NYS Class GA groundwater standards; however, the actual instrument detection limit was below the NYS Class GA groundwater standards. A list of Chemtech's instrument reporting limits for the OLM04.2 Method is included in Appendix E.

Compounds qualified B indicate that the compound was found in the trip blank, or laboratory blank, and in the sample. It indicates possible sample contamination and warns the data user to use caution when applying the results of this analyte.

Compounds qualified D indicate that the compound was reported from an analytical run that required a dilution due to concentrations greater than the highest calibration standard.

Compounds qualified R indicate that the compound was rejected during data validation. Results are rejected when validation criteria are not met by the laboratory.

Analytical results were compared to the standards, criteria, or guidelines (SCGs) described below.

**Soil Samples.** Analytical results were compared to NYS Soil SCGs. Soil SCGs are based on the NYSDEC's Cleanup Objectives ("Technical and Administrative Guidance Memorandum 4046; Determination of Soil Cleanup Objectives and Cleanup Levels" and 6 NYCRR Subpart 375-6 - Remedial Program Soil Cleanup Objectives for unrestricted use).

**Groundwater Samples.** Analytical results were compared to: (1) the NYS Class GA Groundwater Quality Standards from 6 NYCRR Parts 700-706 (NYS, 1999b) or, where applicable, (2) the NYS Class GA Groundwater Quality Guidance Values from the Division of Water Technical and Operational Guidance Series 1.1.1 "Ambient Water Quality Standards and Guidance Values" (NYSDEC, 1998).

**Surfacewater Samples.** Analytical results were compared to: (1) the NYS Class GA Groundwater Quality Standards from 6 NYCRR Parts 700-706 (NYS, 1999b) or, where applicable, (2) the NYS Class GA Groundwater Quality Guidance Values from the Division of Water Technical and Operational Guidance Series 1.1.1 "Ambient Water Quality Standards and Guidance Values" (NYSDEC, 1998).

**Geoprobe**<sup>®</sup> Soil Vapor Samples. There are currently no SCGs for concentrations of compounds in soil vapor. Soil vapor samples were collected to determine whether this environmental medium is contaminated, characterize the nature and extent of contamination, and identify possible sources of the contamination.

**Sub-Slab Soil Vapor Sample.** The sub-slab soil vapor sample was collected to identify potential exposure associated with soil vapor intrusion and to characterize the nature and extent of subsurface vapor contamination. The sub-slab sample results were compared to Matrix 1 and Matrix 2 from the New York State Department of Health, Guidance for Evaluating Soil Vapor Intrusion in the State of New York (NYSDOH, 2006). Those SCGs are typically used in conjunction with indoor air samples, but no indoor air samples were collected.

# 4.1.1 Soil Sample Results

A summary of target VOCs detected in soil samples is presented in Table 4.1. Boring locations are shown on Figure 3.1 (GS locations are same as GW locations).

VOCs were not detected in soil samples at concentrations above the NYSDEC Soil Cleanup Objectives, and all detected compounds were at concentrations less than 2 micrograms per kilogram ( $\mu$ g/kg) (NYSDEC, 1994).

Trace concentrations of methylene chloride were detected at sample locations GS-2 and GS-3. Benzene and toluene were found at trace concentrations at sample locations GS-3 and GS-4 and trace concentrations of 1,2,4-trichlorobenzene, 1,4-dichlorobenezene, benzene, PCE, toluene, trichlorofluoromethane, and m/p-xylene were detected at location GS-1.

#### 4.1.2 Groundwater Sample Results

A summary of target VOCs detected in groundwater samples is presented in Tables 4.2 (January 2006) and 4.3 (October/November 2006) and a summary of PCE concentrations is presented on Figure 4.1. Formal results from the SC at the Former Helwigs Dry Cleaners are not presented in this document, although due to its proximity to the site, concentrations of PCE detected in the Helwig site Geoprobe borings (HGW-1 to HGW-6) are presented on Figure 4.1.

Over the two sampling events, fifty-three samples were collected at twenty five sample locations. PCE was detected in 18 of the 25 sample locations with detections ranging from 0.74 J  $\mu$ g/L (GW-3) to 820 D  $\mu$ g/L (GW-13). PCE detections at Sixteen of the 18 locations exceeded the NYS Class GA groundwater standard for PCE of 5  $\mu$ g/L (see Table 4.2 and 4.3). The highest concentrations were detected at locations GW-4 (610 D  $\mu$ g/L) and GW-13 (820 D  $\mu$ g/L), located approximately 5 feet and 270 feet southeast of the property building, respectively, in the presumed direction of groundwater flow. PCE was not detected in the four presumed upgradient sample locations GW-18, GW-19, GW-20, and GW-21. Low concentrations of PCE were detected slightly upgradient of

the Site building at locations GW-1 (11  $\mu$ g/L) and GW-11 (9.7 J  $\mu$ g/L). Groundwater sample locations and PCE detections are presented on Figure 4.3.

Cis-1,2-dichloroethene (cis-1,2-DCE) was detected at ten of the 25 sample locations at detections ranging from 0.53  $\mu$ g/L (GW-7) to 580  $\mu$ g/L (GW-13). Five of the 25 locations had detections that exceeded the NYS Class GA groundwater standard for Cis-1,2-DCE of 5  $\mu$ g/L.

Trichloroethene (TCE) was detected in ten of the 25 sample locations at concentrations ranging from 0.57  $\mu$ g/L (GW-5) to 180  $\mu$ g/L (GW-13). Three of the seventeen locations had detections that exceeded the NYS Class GA groundwater standard for TCE of 5  $\mu$ g/L.

1,1,2-trichloroethane (1,1,2-TCA) was detected at sample location GW-4 at a concentration of 3.9  $\mu$ g/L which exceeds the NYS Class GA groundwater standard for 1,1,2-TCA of 1  $\mu$ g/L.

Trans-1,2-dichloroethene (trans-1,2-DCE) was detected at one of the 25 sample locations (GW-13) at both sample depths at concentrations ranging from 6.3 J  $\mu$ g/L (17 ft bgs) to 6.5 J  $\mu$ g/L (25 ft bgs). Both sample depths had detected concentrations which exceed the NYS Class GA groundwater standard for Trans-1,2-DCE of 5  $\mu$ g/L.

Vinyl chloride was detected at one of the 25 sample locations (GW-13) at both sample depths at concentrations ranging from 14 J  $\mu$ g/L (25 ft bgs) to 56 J  $\mu$ g/L (17 ft bgs). Both sample depths had detections which exceed the NYS Class GA groundwater standard for vinyl chloride of 2  $\mu$ g/L.

Trace concentrations of 1,1-dichloroethene were detected at sample location GW-13 (1.6 J µg/L).

Additional non-chlorinated compounds were also detected in groundwater samples. Benzene was detected at location GW-21 (1.1  $\mu$ g/L at a concentration slightly above the NYS standard of 1  $\mu$ g/L). Trace (below groundwater standard) concentrations of one or more of the following compounds: toluene benzene, ethylbenzene, isopropyl benzene, cyclohexane, methyl cyclohexane, xylene, 2-butanone, and acetone; were detected at locations GW-6, GW-10 GW-18, GW-20 and GW-21.

In addition to these analytical results, groundwater analytical results from the investigation at the Former Helwigs Cleaners site were reviewed (MACTEC, 2006). VOCs were not detected in the

six borings (HGW-1 to HGW-6) at concentration above the NYS Class GA groundwater standards. PCE and TCE were not detected in the six borings. The only chlorinated solvent detected was a trace concentration (1.4 J  $\mu$ g/L) of cis-1,2-DCE detected at sample location HGW-3. Former Helwigs Cleaners boring locations are shown on Figure 4.1.

### 4.1.3 Surfacewater Sample Results

A summary of target VOCs detected in surface samples is presented in Table 4.3. Since surface samples were collected from a basement sump, and is more representative of groundwater conditions than ecologically viable surface water, analytical results are compared to groundwater standards.

PCE was detected at both of the surface water sample locations at concentrations ranging from 8.1 J  $\mu$ g/L (SW-2) to 17  $\mu$ g/L (SW-1). Both detections exceeded the NYS Class GA Groundwater standard for PCE of 5  $\mu$ g/L.

Cis-1,2-DCE was detected in samples SW-1 (88  $\mu$ g/L) and SW-2 (1.2 J  $\mu$ g/L). The detection of cis-1,2-DCE in sample SW-1 exceeded the NYS Class GA groundwater standard of 5  $\mu$ g/L. Vinyl chloride (3.2 J  $\mu$ g/L) was also detected in sample SW-1. The detection of vinyl chloride in sample SW-1 exceeds the NYS Class GA Groundwater Standard of 2  $\mu$ g/L.

In addition to these compounds trace concentrations of trans-1,2-DCE (0.94 J  $\mu$ g/L) and TCE (1.7 J  $\mu$ g/L) were detected in sample SW-1.

# 4.1.4 Soil Vapor Sample Results

A summary of target VOCs detected in soil vapor samples is presented in Table 4.4.

The only VOCs for which the NYSDOH has promulgated guidance values for soil vapor are TCE, PCE, and 1,1,1-trichloroethane (1,1,1-TCA). These guidance values are only applicable when evaluating sub-slab soil vapor samples, and not exterior soil vapor samples.

**Sub-Slab Soil Vapor Sample Results.** PCE was detected in the sub-slab soil vapor sample at a concentration of 747 D micrograms per cubic meter ( $\mu g/m^3$ ) and TCE was detected in the sub-slab

sample at a concentration of  $30.2 \ \mu g/m^3$ . These concentrations are below the  $1000 \ \mu g/m^3$  and  $250 \ \mu g/m^3$  concentrations corresponding, respectively; however, in the absence of indoor air data, remedial actions and/or monitoring recommendations cannot be made at this time. 1,1,1-TCA was not detected in the sub-slab soil vapor sample.

**Geoprobe Soil Vapor Sample Results.** There are no current guidance or standards for exterior soil vapor sample results. PCE was detected in the three soil vapor samples at concentrations less than  $10 \ \mu g/m^3$ . TCE and 1,1,1-TCA were not detected in soil vapor samples.

#### 4.2 POTENTIOMETRIC SURFACE MAP

The microwell survey and depth to water measurements were used to evaluate groundwater flow. Microwell survey and water elevation data are presented in Table 3.1. Depth to water across the survey area varied from approximately 9.5 feet bgs to 15 feet bgs. Groundwater elevations varied from 922.76 feet above msl, to 923.17 feet above msl in the initial February 2006 round and from 921.12 feet above msl to 923.18 feet above msl during the February 2007 round. The groundwater table gradient appears to be relatively flat, varying by only 0.28 feet in elevation over 120 feet of distance (GW-6 to GW-5) in January 2006, or 2.03 feet in elevation over approximately 400 feet of distance (GW-19 to GW-23) in February 2007. The water elevation in microwell GW-1, located north of the Site building and presumably upgradient from the building, was slightly lower than the other microwell water elevations. Water elevations in GW-5 also appear to be relatively high in comparison to the other measurements. It is possible that utility lines located along Cutler and West Pulteney are affecting localized water elevations (i.e. potential groundwater mounding from water infiltrating along sewer lines). Based on measured groundwater elevations, interpreted groundwater flow is to the south/southeast, towards the Cohocton River, which is consistent with the USGSs groundwater study (USGS, 1982). Note that pumping on the Corning wells may induce the solvent plume to migrate towards the wells. The February 2007 groundwater data is presented on Figure 4.2.

# 5.0 INVESTIGATION FINDINGS

A review of physical and chemical data collected during the SC resulted in the following findings:

- 1) The site is located in a mixed residential/commercial neighborhood that is serviced by public water. Low concentrations of PCE have been detected in the City of Corning's public supply wells #1 and #2, located approximately 950 feet and 1300 feet from Crystal Cleaners, respectively, along the banks of the Cohocton River. Groundwater contours for the greater Corning area indicate that groundwater at the site flows in a southeasterly direction towards the river (USGS, 1982).
- 2) No chlorinated solvents were detected in the five soil samples collected. Four of these samples were collected from around the Site building.
- 3) PCE, a common dry cleaning solvent and a NYS listed hazardous waste, was detected in groundwater samples collected from 18 of the 25 sample locations; concentrations detected exceeded NYS groundwater standard of 5  $\mu$ g/L for PCE at 16 of the 25 locations. PCE was not detected in the four upgradient locations GW-18, GW-19, GW-20 (potentially cross-gradient) and GW-21. PCE was detected in groundwater samples collected from presumably slightly upgradient sample locations GW-1, GW-2, and GW-11 at concentrations ranging from 2.1  $\mu$ g/L to 11  $\mu$ g/L. PCE was also collected in presumably cross gradient sample locations GW-10 (110  $\mu$ g/L) and GW-9 (280  $\mu$ g/L) at concentrations in exceedance of groundwater standards. PCE was detected at the highest concentrations in groundwater samples collected from locations GW-4 (610 D  $\mu$ g/L) and GW-13 (820 D  $\mu$ g/L), located immediately downgradient of the Site building and approximately 270 feet downgradient of the Site building, respectively.
- 4) Chlorinated solvents were detected in both the open sump and the 4x4 hole in the basement below the dry cleaner. The highest concentrations were detected in the sump, which is located below the southeast corner of the building. Based on the depth of groundwater in microwell GW-4, the water in the sumps is presumed to represent groundwater. Concentrations of PCE (16  $\mu$ g/L), cis-1,2-DCE (88  $\mu$ g/L), and vinyl chloride (3.2 J) exceeded the NYS Class GA groundwater standards of 5  $\mu$ g/L, 5  $\mu$ g/L and 2  $\mu$ g/L, respectively.
- 5) The detection of PCE breakdown products in groundwater samples collected down gradient of the Site, including TCE (maximum concentration of 180  $\mu$ g/L), cis-1,2-DCE (maximum concentration of 580 D  $\mu$ g/L), trans-1,2-DCE (maximum concentration of 6.3 J  $\mu$ g/L), 1,1- DCE (maximum concentration of 1.6 J  $\mu$ g/L) and vinyl chloride (maximum concentration of 56 J  $\mu$ g/L), as well as a decrease in PCE concentrations as one moves further downgradient from GW-13, indicate that successive dechlorinated solvents are also likely diminishing downgradient of the Site due to dilution and diffusion within the groundwater column. Each of these breakdown products, with the exception of 1,1-DCE, exceeded NYS groundwater standards at a minimum of one down gradient location.
- 6) PCE was detected in the sub-slab soil vapor sample from SV-1 at a concentration of 747 D  $\mu g/m^3$ . This concentration is above the NYSDOH sub-slab soil vapor concentration recommended for monitoring (100  $\mu g/m^3$ ) but below the concentration recommended for mitigation (1000  $\mu g/m^3$ ). However, in the absence of indoor air data remedial actions

and/or monitoring recommendations cannot be made at this time. TCE was also detected in the sub-slab soil vapor sample at a concentration of  $30 \,\mu g/m^3$ .

- 7) Trace concentrations (less than  $10 \ \mu g/m^3$ ) of PCE were detected in the three Geoprobe soil vapor samples. TCE was not detected in the Geoprobe soil vapor samples.
- 8) Although the Site contains a fueling station and associated petroleum underground storage tanks, only trace concentrations (less than  $2 \mu g/kg$  soil and less than  $5 \mu g/L$  groundwater) of fuel related VOCs were detected. These concentrations are below applicable standards or guidance values.

Data Gaps. Based on the SC, the following data gaps still exist:

- 1) No soil samples were collected from below the Site building and potential source area soil concentrations below the dry cleaning facility are not known.
- 2) The extent of the chlorinated solvent groundwater plume above the NYS standards has not been fully defined.

#### 6.0 **REFERENCES**

- ABB Environmental Services, 1995. Program Quality Assurance Program Plan. Prepared for the New York State Department of Environmental Conservation, Albany, New York. June 1995.
- Davis, 2005. Conversation between MACTEC and Richard Davis, owner/manager of Crystal Cleaners. September 10, 2005.
- Environmental Data Resources, Inc. (EDR), 2006. EDR Radius Map with Goecheck, for 343 West Pulteney Street, Corning, New York. February 27, 2006.
- MACTEC Engineering and Consulting, Inc. P.C. (MACTEC), 2006. Draft Site Characterization Report, Region 8 Dry Cleaners Sites, Former Helwigs Dry Cleaners Site. Prepared for New York State Department of Environmental Conservation, Albany, New York. July, 2006.
- MACTEC Engineering and Consulting, Inc. P.C. (MACTEC), 2005. *Program Health and Safety Plan*. Prepared for New York State Department of Environmental Conservation, Albany, New York. 2005.
- Muller, et al., 1986. Surficial Geologic Map of New York, Finger Lakes Sheet, New York State Museum Geologic Survey, Map and Chart Series #40. 1986.
- National Climatic Data Center (NCDC), 2004. Comparative Climactic Data for the United States-1971-2000. February 2004.
- New York State (NYS), 1999a. New York Codes, Rules, and Regulations, Title 6, Part 371 Identification and Listing of Hazardous Wastes. Amended November 1999.
- New York State (NYS), 1999b. New York Codes, Rules, and Regulations, Title 6, Part 700-705 Water Quality Regulations Surface Water and Groundwater Classifications and Standards. Amended August 1999.
- New York State (NYS), 2006. New York Codes, Rules, and Regulations, Title 6, Part 375 Environmental Remediation Programs. Effective December 14, 2006.
- New York State Department of Environmental Conservation (NYSDEC), 2005. Work Assignment #D003826-20 Region 8 Dry Cleaning Sites- letter dated August 17, 2005.

- New York State Department of Environmental Conservation (NYSDEC), 2002. Draft DER-10, Technical Guidance for Site Investigation and Remediation. December 2002.
- New York State Department of Environmental Conservation (NYSDEC), 2000. "Analytical Services Protocols"; 6/00 Edition; June 2000.
- New York State Department of Environmental Conservation (NYSDEC), 1998; Division of Water Technical and Operational Guidance Series 1.1.1 "Ambient Water Quality Standards and Guidance Values"; Reissued June 1998.

New York State Department of Environmental Conservation (NYSDEC), 1997. "Guidance for the Development of Data Usability Reports"; Division of Environmental Remediation; September 1997.

- New York State Department of Environmental Conservation (NYSDEC), 1994. Revised Technical and Administrative Guidance Memorandum HWR 94-4046: Determination of Soil Cleanup Objectives and Cleanup Levels. January 1994.
- New York State Department of Health (NYSDOH), 2006. "Guidance for Evaluating Soil Vapor Intrusion in the State of New York", Final, October 2006.
- Panton, 2005. Telephone discussions between Steven Panton, Corning Public Works Water Superintendent and MACTEC. September 9, 2005.
- Rickard and Fisher, 1970. Geologic Map of New York, Finger Lakes Sheet, New York State Map and Chart Series 15. by L.V. Richard and D.W. Fisher. March, 1970.
- Teeter, 2005. "Phase II Environmental Site Assessment, Sugar Creek and Crystal Cleaners", prepared by Teeter Environmental Services, Inc., May 25, 2005.
- United States Geological Survey (USGS), 1982. Geohydrology of the Valley-Fill Aquifer in the Corning Area, Steuben County, New York. 1982.

TABLES

|--|

|          |           |           | Screen    |           |           |          | Water     |          | Water     |
|----------|-----------|-----------|-----------|-----------|-----------|----------|-----------|----------|-----------|
|          |           |           | Depth     | Casing    | Riser     | DTW      | Elevation | DTW      | Elevation |
| Location | Northing  | Easting   | (ft bgs)  | Elevation | Elevation | 2/9/2006 | 2/9/06    | 2/1/2007 | 2/1/07    |
| GW-01    | 784833.22 | 686874.93 | 16.6-25.6 | 938.05    | 937.96    | 15.20    | 922.76    | 16.08    | 921.88    |
| GW-05    | 784642.44 | 686939.03 | 6-16      | 932.42    | 932.26    | 9.37     | 922.89    | 9.53     | 922.73    |
| GW-06    | 784750.55 | 686899.83 | 6.8-16.8  | 934.48    | 934.25    | 11.08    | 923.17    | 12.02    | 922.23    |
| GW-10    | 784795.95 | 686785.46 | 8-18      | 934.89    | 934.70    | 11.62    | 923.08    | NA       | NA        |
| GW-18    | 785118.56 | 686578.18 | 14.1-24.1 | 938.02    | 937.76    | NA       | NA        | 14.93    | 922.83    |
| GW-19    | 784041.48 | 686769.82 | 10.8-20.8 | 939.12    | 938.75    | NA       | NA        | 15.57    | 923.18    |
| GW-23    | 784689.28 | 686586.14 | 12-22     | 938.35    | 938.11    | NA       | NA        | 16.99    | 921.12    |

Notes:

ft bgs = feet below ground surface.

DTW = Depth to Water as measured from top of PVC riser by MACTEC Engineering.

Wells surveyed by Lu Engineers.

Created by: CRS 2/6/07 Checked by: BAS 2/8/07
## Table 4.1: Soil Sample VOC Results

| Loca                   | tion Name    | GS-1 (    | GW-1)     | GS-2 (  | GW-2)     | GS-3 (GW-3) |           | <b>GS-4</b> ( | GW-4)     | GS-4 (  | GW-4)     | GS-5 ( | (GW-5)    |
|------------------------|--------------|-----------|-----------|---------|-----------|-------------|-----------|---------------|-----------|---------|-----------|--------|-----------|
| Field                  | Sample Id    | CRGS001   | 00601XX   | CRGS002 | 00701XX   | CRGS003     | 300701XX  | CRGS004       | 00701XD   | CRGS004 | 00701XX   | CRGS00 | 500801XX  |
| Sample Dep             | oth (ft bgs) | 6-        | 8         | 7-      | 9         | 7           | -9        | 7             | .9        | 7-      | 9         | 8-     | 10        |
| Field Sample Dat       |              | 1/24/2006 |           | 1/25/   | 2006      | 1/25/       | /2006     | 1/26/         | 2006      | 1/26/   | 2006      | 1/25   | /2006     |
|                        | QC Code      | FS        |           | FS      |           | FS          |           | FD            |           | FS      |           | FS     |           |
| Paramater              | Criteria     | Result    | Qualifier | Result  | Qualifier | Result      | Qualifier | Result        | Qualifier | Result  | Qualifier | Result | Qualifier |
| 1,2,4-Trichlorobenzene | 670*         | 0.5       | 8 J       | 1       | 1 U       | 1           | 2 U       | 1             | 2 U       | 1       | 1 U       | 12     | 2 UJ      |
| 1,4-Dichlorobenzene    | 1800         | 0.5       | 9 J       | 1       | 1 U       | 1           | 2 U       | 1             | 2 U       | 1       | 1 U       | 12     | 2 UJ      |
| Benzene                | 60           | 0.7       | 3 J       | 1       | 1 U       | 1           | 2 U       | 0.6           | 5 J       | 1       | 1 U       | 12     | 2 UJ      |
| Methylene chloride     | 50           | 1         | 0 U       | 1.      | 5 J       | 1.          | 6 J       | 1             | 2 U       | 1       | 1 U       | 12     | 2 UJ      |
| Tetrachloroethene      | 1300         | 0.6       | 4 J       | 1       | 1 U       | 1           | 2 U       | 1             | 2 U       | 1       | 1 U       | 12     | 2 UJ      |
| Toluene                | 700          | 0.6       | 6 J       | 1       | 1 U       | 1           | 2 U       | 0.            | 8 J       | 1       | 1 U       | 12     | 2 UJ      |
| Trichlorofluoromethane | NA           | 1.        | 6 J       | 1       | 1 UJ      | 1           | 2 UJ      | 1             | 2 U       | 1       | 1 U       | 12     | 2 UJ      |
| Xylene, m/p            | 260          | 1.        | 3 J       | 1       | 1 U       | 1           | 2 U       | 1             | 2 U       | 1       | 1 U       | 12     | 2 UJ      |

## Notes:

Only Detected Compounds shown. Samples analyzed for VOCs by USEPA Method OLM04.2.

Results in microgram per kilogram ( $\mu g/kg$ ). Detections are shown in bold.

ft bgs = feet below ground surface

QC Code:

FS = Field Sample

FD = Field Duplicate

#### Qualifiers:

U = Not detected at a concentration greater than the reporting limit

J = Estimated value

Criteria = Values from 6 NYCRR 375- Table 375-6.8(a):Unrestricted Use Soil Cleanup Objectives.

\*= Value from NYSDEC TAGM 4046.

NA = Not Available

Table Created by:ASZ 6/1/06Table Checked by:CRS 6/9/06

| Locat                    | tion Name                   | GV     | W-1       | GV        | V-2       | GV      | V-2       | G      | W-3       | GV     | N-3       | GV     | N-4       |
|--------------------------|-----------------------------|--------|-----------|-----------|-----------|---------|-----------|--------|-----------|--------|-----------|--------|-----------|
| Field                    | Sample Id                   | CRGW00 | 102601XX  | CRGW00    | 201801XA  | CRGW002 | 202601XX  | CRGW00 | 301801XA  | CRGW00 | 302601XX  | CRGW00 | 401201XA  |
| Sample Dep               | th (ft bgs)                 | 2      | 26        | 1         | .8        | 2       | 6         | 1      | 18        | 2      | 26        | 1      | 2         |
| Field Sa                 | Field Sample Date 1/24/2006 |        | /2006     | 1/25/2006 |           | 1/25/   | 2006      | 1/25   | /2006     | 1/25/  | /2006     | 1/26/  | /2006     |
|                          | QC Code                     | I      | <b>FS</b> | F         | ſS        | F       | S         | 1      | TS .      | F      | <b>TS</b> | F      | TS .      |
| Paramater                | Criteria                    | Result | Qualifier | Result    | Qualifier | Result  | Qualifier | Result | Qualifier | Result | Qualifier | Result | Qualifier |
| 1,1,2-Trichloroethane    | 1                           | 10     | ) U       | 10        | ) U       | 10      | 0 U       | 1(     | ) U       | 10     | ) U       | 10     | 0 U       |
| 1,1-Dichloroethene       | 5                           | 10     | ) U       | 10        | ) U       | 10      | 0 U       | 1(     | ) U       | 10     | ) U       | 10     | ) U       |
| Cis-1,2-Dichloroethene   | 5                           | 10     | ) U       | 10        | ) U       | 10      | 0 U       | 1(     | ) U       | 10     | ) U       | 10     | ) U       |
| Ethyl benzene            | 5                           | 10     | ) U       | 10        | ) U       | 10      | 0 U       | 1(     | ) U       | 10     | ) U       | 10     | ) U       |
| Isopropylbenzene         | 5                           | 10     | ) U       | 10        | ) U       | 10      | 0 U       | 1(     | ) U       | 10     | ) U       | 10     | 0 U       |
| Methyl cyclohexane       | NA                          | 10     | ) U       | 10        | ) U       | 10      | 0 U       | 10     | ) U       | 10     | ) U       | 10     | ) U       |
| Tetrachloroethene        | 5                           | 11     | l         | 2.1       | J         | 0.9     | 3 J       | 0.88   | 8 J       | 0.74   | 4 J       | 15     | 5         |
| Toluene                  | 5                           | 10     | ) U       | 10        | ) U       | 10      | 0 U       | 10     | ) U       | 10     | ) U       | 10     | ) U       |
| trans-1,2-Dichloroethene | 5                           | 10     | ) U       | 10        | ) U       | 10      | 0 U       | 10     | ) U       | 10     | ) U       | 10     | 0 U       |
| Trichloroethene          | 5                           | 10     | ) U       | 10        | ) U       | 10      | 0 U       | 10     | ) U       | 10     | ) U       | 10     | 0 U       |
| Vinyl chloride           | 2                           | 10     | ) U       | 10        | ) U       | 10      | 0 U       | 1(     | ) U       | 10     | ) U       | 10     | ) UJ      |
| Xylene, m/p              | 5                           | 10     | ) U       | 10        | ) U       | 10      | 0 U       | 1(     | ) U       | 10     | ) U       | 10     | ) U       |

## Notes:

Only Detected Compounds shown. Samples analyzed for VOCs by USEPA Method OLM04.2.

Results in microgram per liter ( $\mu g/L$ ). Detections are shown in bold.

ft bgs = feet below ground surface

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

 $\mathbf{U}=\mathbf{N}\mathbf{o}\mathbf{t}$  detected at a concentration greater than the reporting limit

J = Estimated value

D = Result reported from a diluted analytical run

Criteria = Values from Technical and Operational Guidance Series (TOGS) 1.1.1, "Ambient Water Quality Standards and Guidance Values and Groundwater Effluent Limitations" (NYSDEC, 1998).

All Criteria listed are New York State Groundwater Standards.

|                          | Loc Name     | GV     | N-4       | GV     | V-4        | GV     | V-5       | GV     | V-5       | GV     | V-5       | GV     | W-6       |
|--------------------------|--------------|--------|-----------|--------|------------|--------|-----------|--------|-----------|--------|-----------|--------|-----------|
| Field                    | Sample Id    | CRGW00 | 402001XB  | CRGW00 | 403001XX   | CRGW00 | 501601XA  | CRGW00 | 501601XD  | CRGW00 | 502601XX  | CRGW00 | 601401XA  |
| Sample Dep               | pth (ft bgs) | 2      | 20        | 3      | 0          | 1      | 6         | 1      | .6        | 2      | 6         | 1      | 4         |
| Field Sa                 | mple Date    | 1/26   | /2006     | 1/26/  | 2006       | 1/25/  | 2006      | 1/25/  | /2006     | 1/25/  | /2006     | 1/25   | /2006     |
|                          | QC Code      | F      | ſS        | F      | S          | F      | S         | F      | D         | F      | S         | F      | <b>TS</b> |
| Paramater                | Criteria     | Result | Qualifier | Result | Qualifier  | Result | Qualifier | Result | Qualifier | Result | Qualifier | Result | Qualifier |
| 1,1,2-Trichloroethane    | 1*           | 10     | ) U       | 3.9    | ) <b>1</b> | 1      | 0 U       | 1(     | ) UJ      | 10     | ) U       | 1(     | ) U       |
| 1,1-Dichloroethene       | 5*           | 10     | ) U       | 10     | U          | 1      | 0 U       | 10     | ) UJ      | 10     | ) U       | 10     | ) U       |
| Cis-1,2-Dichloroethene   | 5*           | 10     | ) U       | 10     | U          | 1.     | 8 J       | 2.1    | IJ        | 10     | ) U       | 10     | ) U       |
| Ethyl benzene            | 5*           | 10     | ) U       | 10     | U          | 1      | 0 U       | 10     | ) UJ      | 10     | ) U       | 10     | ) U       |
| Isopropylbenzene         | 5*           | 10     | ) U       | 10     | U          | 1      | 0 U       | 10     | ) UJ      | 10     | ) U       | 10     | ) U       |
| Methyl cyclohexane       | NA           | 10     | ) U       | 10     | U          | 1      | 0 U       | 10     | ) UJ      | 10     | ) U       | 10     | ) U       |
| Tetrachloroethene        | 5*           | 15     | 5         | 610    | D          | 5.     | 5 J       | 5.4    | 4 J       | 3      | 8 J       | 7.9    | ) J       |
| Toluene                  | 5*           | 1(     | ) U       | 10     | U          | 1      | 0 U       | 1(     | ) UJ      | 10     | ) U       | 1(     | ) U       |
| trans-1,2-Dichloroethene | 5*           | 10     | ) U       | 10     | U          | 1      | 0 U       | 10     | ) UJ      | 10     | ) U       | 10     | ) U       |
| Trichloroethene          | 5*           | 10     | ) U       | 10     | U          | 0.5    | 7 J       | 10     | ) UJ      | 10     | ) U       | 10     | ) U       |
| Vinyl chloride           | 2*           | 10     | ) UJ      | 10     | U          | 1      | 0 U       | 10     | ) UJ      | 10     | ) U       | 10     | ) U       |
| Xylene, m/p              | 5*           | 10     | ) U       | 10     | U          | 1      | 0 U       | 10     | ) UJ      | 10     | ) U       | 1(     | ) U       |

## Notes:

Only Detected Compounds shown. Samples analyzed for VOCs by USEPA Method OLM04.2.

Results in microgram per liter (µg/L). Detections are shown in bold.

ft bgs = feet below ground surface

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

U = Not detected at a concentration greater than the reporting limit

J = Estimated value

D = Result reported from a diluted analytical run

Criteria = Values from Technical and Operational Guidance Series (TOGS) 1.1.1, "Ambient Water Quality Standards and Guidance Values and Groundwater Effluent Limitations" (NYSDEC, 1998).

\* = New York State Standard

| L                        | oc Name                     | GV          | V-6       | GV     | W-7       | GV      | V-7       | G      | W-7       | GV     | W-8       | G      | W-8       |
|--------------------------|-----------------------------|-------------|-----------|--------|-----------|---------|-----------|--------|-----------|--------|-----------|--------|-----------|
| Field S                  | ample Id                    | CRGW00      | 602201XX  | CRGW00 | 702201XA  | CRGW007 | 703401XD  | CRGW00 | 703401XX  | CRGW00 | 802501XA  | CRGW0  | 0803701XX |
| Sample Dept              | h (ft bgs)                  | 2           | 2         | 2      | 22        | 3       | 4         |        | 34        | 2      | 25        |        | 37        |
| Field Sam                | Field Sample Date 1/25/2006 |             | /2006     | 1/24   | /2006     | 1/24/   | 2006      | 1/24   | /2006     | 1/23/  | /2006     | 1/23   | 8/2006    |
|                          | QC Code                     | F           | S         | I      | FS        | F       | D         | I      | <b>TS</b> | F      | TS .      | ]      | FS        |
| Paramater                | Criteria                    | Result      | Qualifier | Result | Qualifier | Result  | Qualifier | Result | Qualifier | Result | Qualifier | Result | Qualifier |
| 1,1,2-Trichloroethane    | 1*                          | 10          | ) U       | 10     | 0 U       | 10      | ) U       | 10     | U         | 10     | ) U       | 10     | U         |
| 1,1-Dichloroethene       | 5*                          | 10          | ) U       | 10     | 0 U       | 10      | ) U       | 10     | U         | 10     | ) U       | 10     | U         |
| Cis-1,2-Dichloroethene   | 5*                          | 10          | ) U       | 1.5    | 5 J       | 10      | ) U       | 0.53   | J         | 3.6    | όJ        | 0.85   | J         |
| Ethyl benzene            | 5*                          | 10          | ) U       | 10     | 0 U       | 10      | ) U       | 10     | U         | 10     | ) U       | 10     | U         |
| Isopropylbenzene         | 5*                          | 10          | ) U       | 10     | 0 U       | 10      | ) U       | 10     | U         | 10     | ) U       | 10     | U         |
| Methyl cyclohexane       | NA                          | 10          | ) U       | 10     | 0 U       | 10      | ) U       | 10     | U         | 10     | ) U       | 10     | U         |
| Tetrachloroethene        | 5*                          | <b>4.</b> ] | l J       | 110    | 0         | 200     | ) D       | 230    | D         | 24     | ļ         | 130    | 1         |
| Toluene                  | 5*                          | 0.55        | 5 J       | 10     | 0 U       | 10      | ) U       | 10     | U         | 10     | ) U       | 10     | U         |
| trans-1,2-Dichloroethene | 5*                          | 10          | ) U       | 10     | 0 U       | 10      | ) U       | 10     | U         | 10     | ) U       | 10     | U         |
| Trichloroethene          | 5*                          | 10          | ) U       | 1.4    | 4 J       | 1.3     | 3 J       | 1.5    | J         | 0.93   | 3 J       | 10     | U         |
| Vinyl chloride           | 2*                          | 10          | ) U       | 10     | 0 UJ      | 10      | ) U       | 10     | U         | 10     | ) UJ      | 10     | UJ        |
| Xylene, m/p              | 5*                          | 10          | ) U       | 1(     | 0 U       | 10      | ) U       | 10     | U         | 10     | ) U       | 10     | U         |

## Notes:

Only Detected Compounds shown. Samples analyzed for VOCs by USEPA Method OLM04.2.

Results in microgram per liter ( $\mu$ g/L). Detections are shown in bold.

ft bgs = feet below ground surface

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

U = Not detected at a concentration greater than the reporting limit

J = Estimated value

D = Result reported from a diluted analytical run

Criteria = Values from Technical and Operational Guidance Series (TOGS) 1.1.1, "Ambient Water Quality Standards and Guidance Values and Groundwater Effluent Limitations" (NYSDEC, 1998).

\* = New York State Standard

| L                        | Loc Name GW-9<br>Field Sample Id CRGW00901401 |        | W-9       | G      | W-9       | GW          | /-10      | GV     | V-11      | GW     | -11       | GV     | W-12      |
|--------------------------|---|--------|-----------|--------|-----------|-------------|-----------|--------|-----------|--------|-----------|--------|-----------|
| Field S                  | Field Sample Id<br>Sample Denth (ft bgs) 14   |        | 901401XA  | CRGW00 | 902201XX  | CRGW01      | 002601XX  | CRGW01 | 102201XA  | CRGW01 | 103001XX  | CRGW01 | 1202201XA |
| Sample Dept              | h (ft bgs)                                    | 1      | 4         |        | 22        | 2           | 6         | 2      | 2         | 3      | 0         | -      | 22        |
| Field San                | nple Date                                     | 1/25   | /2006     | 1/25   | /2006     | 1/24/       | 2006      | 1/24   | /2006     | 1/24/  | 2006      | 1/23   | 3/2006    |
|                          | QC Code                                       | F      | S         | l      | FS        | F           | S         | I      | ſS        | F      | S         | J      | FS        |
| Paramater                | Criteria                                      | Result | Qualifier | Result | Qualifier | Result      | Qualifier | Result | Qualifier | Result | Qualifier | Result | Qualifier |
| 1,1,2-Trichloroethane    | 1*  | 10     | ) U       | 10     | U         | 10          | ) U       | 10     | ) U       | 10     | ) U       | 10     | UJ        |
| 1,1-Dichloroethene       | 5*  | 10     | ) U       | 10     | U         | 10          | ) U       | 10     | ) U       | 10     | ) U       | 10     | UJ        |
| Cis-1,2-Dichloroethene   | 5*  | 10     | ) U       | 2.8    | J         | 7.2         | 2 J       | 10     | ) U       | 10     | ) U       | 5.5    | J         |
| Ethyl benzene            | 5*  | 10     | ) U       | 10     | U         | 2.8         | 8 J       | 10     | ) U       | 10     | ) U       | 10     | UJ        |
| Isopropylbenzene         | 5*  | 10     | ) U       | 10     | U         | 1.8         | 8 J       | 10     | ) U       | 10     | ) U       | 10     | UJ        |
| Methyl cyclohexane       | NA  | 1(     | ) U       | 10     | U         | 4.4         | l J       | 10     | ) U       | 10     | ) U       | 10     | UJ        |
| Tetrachloroethene        | 5*  | 74     | 4 J       | 280    | DJ        | 27          | 7         | 9.1    | / J       | 0.84   | J         | 42     | J         |
| Toluene                  | 5*  | 1(     | ) U       | 10     | U         | 10          | ) U       | 10     | ) U       | 10     | ) U       | 10     | UJ        |
| trans-1,2-Dichloroethene | 5*  | 10     | ) U       | 10     | U         | 10          | ) U       | 10     | ) U       | 10     | ) U       | 10     | UJ        |
| Trichloroethene          | 5*  | 1.9    | J         | 11     |           | <b>4.</b> 4 | l J       | 10     | ) U       | 10     | ) U       | 5      | J         |
| Vinyl chloride           | 2*  | 1(     | ) U       | 10     | U         | 10          | ) U       | 10     | ) UJ      | 10     | ) U       | 10     | UJ        |
| Xylene, m/p              | 5*  | 1(     | ) U       | 10     | U         | 2.4         | l J       | 10     | ) U       | 10     | ) U       | 10     | UJ        |

## Notes:

Only Detected Compounds shown. Samples analyzed for VOCs by USEPA Method OLM04.2.

Results in microgram per liter ( $\mu g/L$ ). Detections are shown in bold.

ft bgs = feet below ground surface

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

U = Not detected at a concentration greater than the reporting limit

J = Estimated value

D = Result reported from a diluted analytical run

Criteria = Values from Technical and Operational Guidance Series (TOGS) 1.1.1, "Ambient Water Quality Standards and Guidance Values and Groundwater Effluent Limitations" (NYSDEC, 1998).

\* = New York State Standard

| L                        | .oc Name   | GV     | V-12      | GV     | V-13      | GV     | V-13      | GV     | V-14      | GW     | /-14      | GV     | V-15      |
|--------------------------|------------|--------|-----------|--------|-----------|--------|-----------|--------|-----------|--------|-----------|--------|-----------|
| Field S                  | ample Id   | CRGW01 | 203401XX  | CRGW01 | 1301701XA | CRGW01 | 302501XX  | CRGW01 | 402201XA  | CRGW01 | 403001XX  | CRGW01 | 502101XA  |
| Sample Dept              | h (ft bgs) | 3      | 4         | 1      | 17        | 2      | 25        | 2      | 2         | 3      | 0         | 2      | 21        |
| Field San                | nple Date  | 1/23   | /2006     | 1/24   | /2006     | 1/24   | /2006     | 1/23   | /2006     | 1/23/  | 2006      | 1/23   | /2006     |
|                          | QC Code    | F      | ſS        | I      | FS        | I      | TS .      | I      | ſS        | F      | S         | ŀ      | FS .      |
| Paramater                | Criteria   | Result | Qualifier |
| 1,1,2-Trichloroethane    | 1*         | 10     | ) U       | 1(     | ) U       | 10     | ) U       | 1      | ) UJ      | 10     | ) UJ      | 10     | ) UJ      |
| 1,1-Dichloroethene       | 5*         | 10     | ) U       | 1.0    | 6 J       | 1      | J         | 1      | ) UJ      | 10     | ) UJ      | 10     | ) UJ      |
| Cis-1,2-Dichloroethene   | 5*         | 25     | 5         | 580    | ) D       | 44(    | ) D       | 2.2    | 2 J       | 5.2    | 2 J       | 7      | 7 J       |
| Ethyl benzene            | 5*         | 10     | ) U       | 1(     | ) U       | 1(     | ) U       | 1      | ) UJ      | 10     | ) UJ      | 10     | ) UJ      |
| Isopropylbenzene         | 5*         | 10     | ) U       | 10     | ) U       | 10     | ) U       | 1      | ) UJ      | 10     | ) UJ      | 10     | ) UJ      |
| Methyl cyclohexane       | NA         | 10     | ) U       | 10     | ) U       | 1(     | ) U       | 1      | ) UJ      | 10     | ) UJ      | 10     | ) UJ      |
| Tetrachloroethene        | 5*         | 21(    | DJ        | 820    | ) D       | 390    | ) D       | 3.     | 4 J       | 13     | 3 J       | 65     | 5 J       |
| Toluene                  | 5*         | 10     | ) U       | 1(     | ) U       | 1(     | ) U       | 1      | ) UJ      | 10     | ) UJ      | 10     | ) UJ      |
| trans-1,2-Dichloroethene | 5*         | 10     | ) U       | 6.3    | 3 J       | 6.5    | 5 J       | 1      | ) UJ      | 10     | ) UJ      | 10     | ) UJ      |
| Trichloroethene          | 5*         | 6.9    | J         | 180    | 0         | 120    | )         | 1      | ) UJ      | 1.7    | / J       | 2.3    | 3 J       |
| Vinyl chloride           | 2*         | 10     | ) UJ      | 50     | 6 J       | 14     | ł J       | 1      | ) UJ      | 10     | ) UJ      | 10     | ) UJ      |
| Xylene, m/p              | 5*         | 10     | U         | 1(     | ) U       | 1(     | ) U       | 1      | ) UJ      | 10     | ) UJ      | 10     | ) UJ      |

## Notes:

Only Detected Compounds shown. Samples analyzed for VOCs by USEPA Method OLM04.2.

Results in microgram per liter (µg/L). Detections are shown in bold.

ft bgs = feet below ground surface

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

 $\mathbf{U}=\mathbf{N}\mathbf{o}\mathbf{t}$  detected at a concentration greater than the reporting limit

J = Estimated value

D = Result reported from a diluted analytical run

Criteria = Values from Technical and Operational Guidance Series (TOGS) 1.1.1, "Ambient Water Quality Standards and Guidance Values and Groundwater Effluent Limitations" (NYSDEC, 1998).

\* = New York State Standard

|                          | Loc Name    | GV     | V-15      | GV     | V-16      | GV     | V-16      | GV     | V-17      | GW     | /-17      |
|--------------------------|-------------|--------|-----------|--------|-----------|--------|-----------|--------|-----------|--------|-----------|
| Field                    | Sample Id   | CRGW01 | 502501XX  | CRGW01 | 1601701XA | CRGW01 | 602101XX  | CRGW01 | 701601XA  | CRGW01 | 702001XX  |
| Sample Dep               | th (ft bgs) | 2      | 25        |        | 17        | 2      | 21        | 1      | 16        | 2      | 0         |
| Field Sa                 | mple Date   | 1/23   | /2006     | 1/23   | /2006     | 1/23   | /2006     | 1/23   | /2006     | 1/23/  | 2006      |
|                          | QC Code     | FS     |           | FS     |           | ]      | FS        | ]      | FS        | F      | 'S        |
| Paramater                | Criteria    | Result | Qualifier |
| 1,1,2-Trichloroethane    | 1*          | 10     | UJ        | 10     | ) UJ      | 10     | ) UJ      | 10     | UJ        | 10     | ) UJ      |
| 1,1-Dichloroethene       | 5*          | 10     | UJ        | 10     | ) UJ      | 10     | UJ        | 10     | UJ        | 10     | ) UJ      |
| Cis-1,2-Dichloroethene   | 5*          | 8.3    | J         | 10     | ) UJ      | 10     | UJ        | 3      | J         | 3      | J         |
| Ethyl benzene            | 5*          | 10     | UJ        | 10     | ) UJ      | 10     | UJ        | 10     | UJ        | 10     | ) UJ      |
| Isopropylbenzene         | 5*          | 10     | UJ        | 10     | ) UJ      | 10     | UJ        | 10     | UJ        | 10     | ) UJ      |
| Methyl cyclohexane       | NA          | 10     | UJ        | 10     | ) UJ      | 10     | UJ        | 10     | UJ        | 10     | ) UJ      |
| Tetrachloroethene        | 5*          | 87     | J         | 7.1    | J         | 6.3    | J         | 64     | J         | 67     | ' J       |
| Toluene                  | 5*          | 10     | UJ        | 10     | ) UJ      | 10     | ) UJ      | 10     | UJ        | 10     | ) UJ      |
| trans-1,2-Dichloroethene | 5*          | 10     | UJ        | 10     | ) UJ      | 10     | UJ        | 10     | UJ        | 10     | ) UJ      |
| Trichloroethene          | 5*          | 2.7    | J         | 10     | ) UJ      | 10     | UJ        | 1.4    | J         | 1.5    | 5 J       |
| Vinyl chloride           | 2*          | 10     | UJ        | 10     | ) UJ      | 10     | ) UJ      | 10     | UJ        | 10     | ) UJ      |
| Xylene, m/p              | 5*          | 10     | UJ        | 10     | ) UJ      | 10     | UJ        | 10     | UJ        | 10     | ) UJ      |

## Notes:

Only Detected Compounds shown. Samples analyzed for VOCs by USEPA Method OLM04.2.

Results in microgram per liter ( $\mu g/L$ ). Detections are shown in bold.

ft bgs = feet below ground surface

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

U = Not detected at a concentration greater than the reporting limit

J = Estimated value

D = Result reported from a diluted analytical run

Criteria = Values from Technical and Operational Guidance Series (TOGS) 1.1.1, "Ambient Water Quality Standards and Guidance Values and Groundwater Effluent Limitations" (NYSDEC, 1998).

\* = New York State Standard

|                        | Location            | GW-18            | GW-18            | GW-19            | GW-19            | GW-20            | GW-20            |
|------------------------|---------------------|------------------|------------------|------------------|------------------|------------------|------------------|
|                        | Sample Date         | 11/2/2006        | 11/2/2006        | 10/31/2006       | 10/31/2006       | 11/2/2006        | 11/2/2006        |
|                        | Sample ID           | CRGW01802001XX   | CRGW01803001XX   | CRGW01901901XX   | CRGW01902901XX   | CRGW02001901XX   | CRGW02002901XX   |
| Sa                     | mple Depth (ft bgs) | 20               | 30               | 19               | 29               | 19               | 29               |
|                        | QC Code             | FS               | FS               | FS               | FS               | FS               | FS               |
| Parameter              | Criteria            | Result Qualifier |
| 1,1-Dichloroethene     | 5                   | 10 U             | 10 U             | 10 UJ            | 10 U             | 10 U             | 10 U             |
| 2-Butanone             | 50                  | 50 UJ            | 50 UJ            | 50 U             | 50 U             | 50 UJ            | 50 UJ            |
| Acetone                | 50                  | 50 UJ            | 40 J             | 50 U             | 50 U             | 50 UJ            | 14 J             |
| Benzene                | 1                   | 10 U             | 0.82 J           | 10 U             | 10 U             | 10 U             | 10 U             |
| Cis-1,2-Dichloroethene | 5                   | 10 U             |
| Cyclohexane            | NA                  | 10 U             | 10 U             | 10 UJ            | 10 U             | 0.97 J           | 10 U             |
| Ethyl benzene          | 5                   | 10 U             |
| Isopropylbenzene       | 5                   | 10 U             |
| Methyl cyclohexane     | NA                  | 10 U             | 10 U             | 10 U             | 10 U             | 0.88 J           | 10 U             |
| o-Xylene               | 5                   | 10 U             |
| Tetrachloroethene      | 5                   | 10 U             |
| Toluene                | 5                   | 10 U             | 1.1 J            | 10 U             | 10 U             | 0.99 J           | 10 U             |
| Trichloroethene        | 5                   | 10 U             |
| Vinyl chloride         | 2                   | 10 U             |
| Xylene, m/p            | 5                   | 10 U             | 10 U             | 10 U             | 10 U             | 1.4 J            | 10 U             |

## Notes:

Only Detected Compounds shown. Samples analyzed for VOCs by USEPA Method OLM04.2.

Results in microgram per liter (µg/L)

ft bgs = feet below ground surface

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

U = Not detected at a concentration greater than the reporting limit

J = Estimated value

Criteria = Values from Technical and Operational Guidance Series (TOGS) 1.1.1, "Ambient Water Quality Standards and Guidance Values and Groundwater Effluent Limitations" (NYSDEC, 1998).

|                        | Location              | GW-21            | GW-21            | GW-22            | GW-22            | GW-23            | GW-23            |
|------------------------|-----------------------|------------------|------------------|------------------|------------------|------------------|------------------|
|                        | Sample Date           | 11/1/2006        | 11/1/2006        | 11/1/2006        | 11/1/2006        | 11/2/2006        | 11/2/2006        |
|                        | Sample ID             | CRGW02101901XX   | CRGW02102601XX   | CRGW02201801XX   | CRGW02202801XX   | CRGW02301901XX   | CRGW02302701XX   |
| 1                      | Sample Depth (ft bgs) | 19               | 26               | 18               | 28               | 19               | 27               |
|                        | Qc Code               | FS               | FS               | FS               | FS               | FS               | FS               |
| Parameter              | Criteria              | Result Qualifier |
| 1,1-Dichloroethene     | 5                     | 10 U             | 10 U             | 10 U             | 10 U             | 10 UJ            | 10 UJ            |
| 2-Butanone             | 50                    | 50 U             | 2.4 J            | 50 UJ            | 50 U             | 50 U             | 50 UJ            |
| Acetone                | 50                    | 50 U             | 50 U             | 50 UJ            | 50 U             | 50 U             | 50 UJ            |
| Benzene                | 1                     | 10 U             | 1.1 J            | 10 UJ            | 10 U             | 10 U             | 10 UJ            |
| Cis-1,2-Dichloroethene | 5                     | 10 U             | 10 UJ            |
| Cyclohexane            | NA                    | 10 U             | 1.6 J            | 10 U             | 10 U             | 10 UJ            | 10 UJ            |
| Ethyl benzene          | 5                     | 10 U             | 10 UJ            |
| Isopropylbenzene       | 5                     | 10 U             | 10 UJ            |
| Methyl cyclohexane     | NA                    | 10 U             | 1.4 J            | 10 UJ            | 10 U             | 10 U             | 10 UJ            |
| o-Xylene               | 5                     | 10 U             | 10 UJ            |
| Tetrachloroethene      | 5                     | 10 U             | 3.2 J            |
| Toluene                | 5                     | 10 U             | 1.7 J            | 10 U             | 10 U             | 10 U             | 10 UJ            |
| Trichloroethene        | 5                     | 10 U             | 10 U             | 10 UJ            | 10 U             | 10 U             | 10 UJ            |
| Vinyl chloride         | 2                     | 10 U             | 10 U             | 10 U             | 10 U             | 10 UJ            | 10 UJ            |
| Xylene, m/p            | 5                     | 10 U             | 0.95 J           | 10 U             | 10 U             | 10 U             | 10 UJ            |

## Notes:

Only Detected Compounds shown. Samples analyzed for VOCs by USEPA Method OLM04.2.

Results in microgram per liter (µg/L)

ft bgs = feet below ground surface

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

U = Not detected at a concentration greater than the reporting limit

J = Estimated value

Criteria = Values from Technical and Operational Guidance Series (TOGS) 1.1.1, "Ambient Water Quality Standards and Guidance Values and Groundwater Effluent Limitations" (NYSDEC, 1998).

|                        | Location              | GW-24            | GW-24            | GW-25            | GW-25            | GW-25            | GW-01            |
|------------------------|-----------------------|------------------|------------------|------------------|------------------|------------------|------------------|
|                        | Sample Date           | 11/1/2006        | 11/1/2006        | 11/1/2006        | 11/1/2006        | 11/1/2006        | 11/1/2006        |
|                        | Sample ID             | CRGW02401301XX   | CRGW02402301XX   | CRGW02501201XD   | CRGW02501201XX   | CRGW02502201XX   | CRMW00102001XX   |
| :                      | Sample Depth (ft bgs) | 13               | 23               | 12               | 12               | 22               | 20               |
|                        | Qc Code               | FS               | FS               | FD               | FS               | FS               | FS               |
| Parameter              | Criteria              | Result Qualifier |
| 1,1-Dichloroethene     | 5                     | 10 U             |
| 2-Butanone             | 50                    | 50 UJ            | 50 U             |
| Acetone                | 50                    | 50 UJ            | 50 U             |
| Benzene                | 1                     | 10 U             |
| Cis-1,2-Dichloroethene | 5                     | 10 U             |
| Cyclohexane            | NA                    | 10 U             |
| Ethyl benzene          | 5                     | 10 U             |
| Isopropylbenzene       | 5                     | 10 U             |
| Methyl cyclohexane     | NA                    | 10 U             |
| o-Xylene               | 5                     | 10 U             |
| Tetrachloroethene      | 5                     | 10 U             | 6.6 J            |
| Toluene                | 5                     | 10 U             |
| Trichloroethene        | 5                     | 10 U             |
| Vinyl chloride         | 2                     | 10 U             |
| Xylene, m/p            | 5                     | 10 U             |

## Notes:

Only Detected Compounds shown. Samples analyzed for VOCs by USEPA Method OLM04.2.

Results in microgram per liter (µg/L)

ft bgs = feet below ground surface

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

U = Not detected at a concentration greater than the reporting limit

J = Estimated value

Criteria = Values from Technical and Operational Guidance Series (TOGS) 1.1.1, "Ambient Water Quality Standards and Guidance Values and Groundwater Effluent Limitations" (NYSDEC, 1998).

|                        | Location              | GW-05            | GW-06            | GW-10            |
|------------------------|-----------------------|------------------|------------------|------------------|
|                        | Sample Date           | 11/1/2006        | 11/1/2006        | 11/1/2006        |
|                        | Sample ID             | CRMW00501201XX   | CRMW00601301XX   | CRMW01001301XX   |
|                        | Sample Depth (ft bgs) | 12               | 13               | 13               |
|                        | Qc Code               | FS               | FS               | FS               |
| Parameter              | Criteria              | Result Qualifier | Result Qualifier | Result Qualifier |
| 1,1-Dichloroethene     | 5                     | 10 UJ            | 10 UJ            | 4.7 J            |
| 2-Butanone             | 50                    | 50 U             | 50 U             | 50 U             |
| Acetone                | 50                    | 50 U             | 50 U             | 50 U             |
| Benzene                | 1                     | 10 U             | 10 U             | 10 U             |
| Cis-1,2-Dichloroethene | 5                     | 2 J              | 10 U             | 35               |
| Cyclohexane            | NA                    | 10 UJ            | 10 UJ            | 38               |
| Ethyl benzene          | 5                     | 10 U             | 10 U             | 20               |
| Isopropylbenzene       | 5                     | 10 U             | 10 U             | 13               |
| Methyl cyclohexane     | NA                    | 10 U             | 10 U             | 43               |
| o-Xylene               | 5                     | 10 U             | 10 U             | 0.88 J           |
| Tetrachloroethene      | 5                     | 33               | 5 J              | 110              |
| Toluene                | 5                     | 10 U             | 10 U             | 10 U             |
| Trichloroethene        | 5                     | 2.2 J            | 10 U             | 77               |
| Vinyl chloride         | 2                     | 10 U             | 10 U             | 9 J              |
| Xylene, m/p            | 5                     | 10 U             | 10 U             | 7.1 J            |

## Notes:

Only Detected Compounds shown. Samples analyzed for VOCs by USEPA Method OLM04.2.

Results in microgram per liter (µg/L)

ft bgs = feet below ground surface

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

U = Not detected at a concentration greater than the reporting limit

J = Estimated value

Criteria = Values from Technical and Operational Guidance Series (TOGS) 1.1.1, "Ambient Water Quality Standards and Guidance Values and Groundwater Effluent Limitations" (NYSDEC, 1998). Highlighted results in exceed associated criteria

## Table 4.4: Surfacewater VOC Results

| Loca                     | tion Name   | SV        | W-1       | SV              | W-2       |    |   |
|--------------------------|-------------|-----------|-----------|-----------------|-----------|----|---|
| Field                    | Sample Id   | CRSW00    | 100001XX  | CRSW00          | 200001XX  |    |   |
| Sample D                 | epth (feet) | 0         | )-1       | 0               | )-1       |    |   |
| Field Sa                 | mple Date   | 1/23/2006 |           | 1/23            | /2006     |    |   |
|                          | QC Code     | J         | FS        | ]               | FS        |    |   |
| Paramater                | Criteria    | Result    | Qualifier | Result          | Qualifier |    |   |
| Cis-1,2-Dichloroethene   | 5           | 88        | 88        |                 | J         |    |   |
| Tetrachloroethene        | 5           | 17        | 17        |                 | J         |    |   |
| trans-1,2-Dichloroethene | 5           | 0.94 J    |           | 0.94 J          |           | 10 | U |
| Trichloroethene          | 5           | 1.7 J     |           | <b>1.7 J</b> 10 |           |    |   |
| Vinyl chloride           | 2           |           | 3.2 J     |                 | UJ        |    |   |

## Notes:

Only Detected Compounds shown. Samples analyzed for VOCs by USEPA Method OLM04.2. Results in microgram per liter (µg/L) QC Code: FS = Field Sample Qualifiers: U = Not detected at a concentration greater than the reporting limit J = Estimated value

Criteria = Samples collected from basement sumps and potentially representative of groundwater concentrations; therefore results comared to groundwater criteria.

Values from Technical and Operational Guidance Series (TOGS) 1.1.1, "Ambient Water Quality Standards and Guidance Values and Groundwater Effluent Limitations" (NYSDEC, 1998).

All Criteria listed are New York State Groundwater Standards.

Highlighted results exceed associated criteria

Table Created by: ASZ 6/1/06 Table Checked by: CRS 6/9/06

| Table 4.5: Se | oil Vapor | VOC | Results |
|---------------|-----------|-----|---------|
|---------------|-----------|-----|---------|

| Location Name                         | ame GV-1       |           | GV             | -2        | GV      | /-3       | SV             | -1        |
|---------------------------------------|----------------|-----------|----------------|-----------|---------|-----------|----------------|-----------|
| Field Sample ID                       | CRGV00100601XX |           | CRGV00200601XX |           | CRGV003 | 300601XX  | CRSV00100101XX |           |
| Sample Depth (ft bgs)                 | 6-             | 7         | 6-7            |           | 6-7     |           | 1-2            |           |
| Field Sample Date                     | 1/26/2         | 2006      | 1/26/2         | 2006      | 1/26/   | 2006      | 1/26/2006      |           |
| QC Code                               | F              | 5         | F              | 5         | FS      |           | F              | S         |
| Paramater                             | Result         | Qualifier | Result         | Qualifier | Result  | Qualifier | Result         | Qualifier |
| 1,1,2-Trichloro-1,2,2-Trifluoroethane | 0.54           |           | 0.76           | U         | 0.76    | U         | 3.06           | U         |
| 1,2,4-Trimethylbenzene                | 2.85           |           | 2.7            |           | 2.55    |           | 1.96           | U         |
| 1,3,5-Trimethylbenzene                | 0.49           | U         | 0.49           | U         | 1.62    |           | 1.96           | U         |
| 2-Butanone                            | 20.2           |           | 16.5           |           | 14.3    |           | 3.65           |           |
| 2-Propanol                            | 39.4           |           | 25.5           |           | 26.2    |           | 8.83           |           |
| 4-Ethyltoluene                        | 0.59           |           | 0.49           | U         | 0.49    | U         | 1.96           | U         |
| Acetone                               | 113            | D         | 149            | D         | 105     | D         | 114            | В         |
| Benzene                               | 5.33           | J         | 6.38           |           | 3.57    |           | 13.7           |           |
| Butadiene, 1,3-                       | 12             | J         | 21.8           | J         | 8.44    | J         |                | R         |
| Carbon disulfide                      | 1.31           |           | 1.55           |           | 0.9     |           | 3.36           |           |
| Chloromethane                         | 1.66           |           | 1.55           |           | 1.25    |           | 1.23           |           |
| Cyclohexane                           | 0.34           | · U       | 0.34           | U         | 0.34    | U         | 41.3           |           |
| Dichlorodifluoromethane               | 2.38           |           | 2.38           |           | 2.03    |           | 2.38           |           |
| Ethyl acetate                         | 9.83           |           | 4.53           |           | 8.6     |           | 71.3           |           |
| Ethyl benzene                         | 1.47           | J         | 0.95           |           | 1.52    |           | 1.73           | U         |
| Heptane                               | 1.76           |           | 2.17           |           | 1.68    |           | 71             |           |
| Hexane                                | 2.92           |           | 5.38           |           | 2.67    |           | 111            |           |
| Isooctane                             | 0.65           |           | 0.79           |           | 0.61    |           | 1.87           | U         |
| Methylene chloride                    | 0.7            | U         | 1.01           |           | 0.7     | U         | 2.78           | U         |
| o-Xylene                              | 1.17           | J         | 0.82           |           | 0.74    |           | 1.73           | U         |
| Propylene                             | 46.6           | D         | 75.1           | D         | 39.5    | D         | 0.69           | U         |
| Tetrachloroethene                     | 4.21           | J         | 6.04           |           | 4.28    |           | 747            | D         |
| Tetrahydrofuran                       | 0.44           |           | 0.59           | U         | 0.59    | U         | 2.36           | U         |
| Toluene                               | 7.64           | J         | 7.19           |           | 5.08    |           | 18.8           |           |
| Trichloroethene                       | 0.23           | UJ        | 0.23           | U         | 0.23    | U         | 30.2           |           |
| Trichlorofluoromethane                | 1.06           |           | 1.23           |           | 0.84    |           | 2.24           | U         |
| Xylene, m/p                           | 3.08           | J         | 2.08           |           | 1.82    |           | 3.47           | U         |

## Notes:

Only Detected Compounds shown. Samples analyzed for VOCs by USEPA Method TO-15.

Results in microgram per cubic meter ( $\mu g/m^3$ ). Detections are shown in bold.

ft bgs = feet below ground surface

QC Code:

FS = Field Sample

Qualifiers:

U = Not detected at a concentration greater than the reporting limit

J = Estimated value

D = Result is reported from a diluted analytical run

R = Result was rejected

 $\mathbf{B} = \mathbf{A}\mathbf{n}\mathbf{a}\mathbf{l}\mathbf{y}\mathbf{t}\mathbf{e}$  was detected in both the blank and field sample

FIGURES







Document: P:\Projects\nysdec1\projects\nysdec1\projects\Region 8 Dry Cleaning Sites\4.0 Project Deliverables\4.5 Databases\GIS\Map documents\Crystal\_water.mxd PDF: P:\Projects\nysdec1\projects\nysdec1\projects\4.0 Project Deliverables\4.1 Reports\Crystal\Draft Final\E doc deliverables\4.5 Databases\GIS\Map documents\Crystal\_water.mxd PDF: P:\Projects\nysdec1\projects\nysdec1\projects\4.0 Project Deliverables\4.1 Reports\Crystal\Draft Final\E doc deliverables\4.5 Databases\GIS\Map documents\Crystal\_water.mxd PDF: P:\Projects\nysdec1\projects\nysdec1\projects\4.0 Project Deliverables\4.1 Reports\Crystal\Draft Final\E doc deliverables\4.5 Databases\GIS\Map documents\Crystal\_water.mxd PDF: P:\Projects\nysdec1\projects\Region 8 Dry Cleaning Sites\4.0 Project Deliverables\4.1 Reports\Crystal\Draft Final\E doc deliverables\4.1 Beports\Crystal\Draft Final\E doc deliverables\Final\E doc deliverables\Fin



Iment: P\Projects\nysdec1\projects\Region 8 Dry Cleaning Sites\4.0 Project Deliverables\4.5 Databases\GIS\Map documents\Crystal\_water.mxd PDF: P:\Projects\nysdec1\projects\Region 8 Dry Cleaning Sites\4.0 Project Deliverables\4.1 Reports\Crystal\Draft Final\E doc deliverables\Figures\Fig\_\_1\_new.pdf 03/08/2007 4:36 PM crstaples



Document: P1/Projects/nysdec1/projects/Region 8 Dry Cleaning Sites/4.0 Project Deliverables/4.5 Databases/GIS/Map documents/Crystal\_water.mxd PDF: P1/Projects/nysdec1/projects/Region 8 Dry Cleaning Sites/4.0 Project Deliverables/4.1 Reports/Crystal/Draft Final/E doc deliverables/4.5 Databases/GIS/Map documents/Crystal\_water.mxd PDF: P1/Projects/Nysdec1/projects/Ngsdec1/project/Deliverables/4.1 Reports/Crystal/Draft Final/E doc deliverables/Figures/Fig\_4\_2\_new.pdf 03/08/2007 4:40 PM crstaples

# APPENDIX A

# FACILITY INFORMATION REPORT AND PREVIOUS INVESTIGATIONS

| Site:                    | SUGAR CI<br>343 WEST<br>CORNING                        | <u>PBS # : </u><br>8-390267<br>REEK #132<br>PULTENEY STREET<br>5, NY 14830 |                               | NEW Y                        | ORK STA                          | TE DEPA<br>Pe<br>Owner: | ARTMI<br>troleun<br>Facilit<br>WII<br>178<br>WII | ENT OF<br>n Bulk S<br>y Inform<br>LSON F<br>0 WEHI<br>LLIAM | FENVIR<br>Storage F<br>nation R<br>GARMS I<br>RLE DR<br>SVILLE | ONMENTA<br>Program<br>eport<br>NC<br>IVE<br>, NY 14221 | AL CON                    | NSERVATION                  | М                         | ]<br>ail:                   | WIL5<br>1780        | SON F.<br>WEHF              | Printed :<br>ARMS IN<br>RLE DRIV | Pag<br>2/:<br>IC<br>VE<br>NV 1 | e 1 of 1<br>16/2007        | pbsfacr                     | pt_foil.rpt                 |
|--------------------------|--|--|-------------------------------|------------------------------|----------------------------------|-------------------------|--|---|--|--|---------------------------|-----------------------------|---------------------------|-----------------------------|---------------------|-----------------------------|----------------------------------|--------------------------------|----------------------------|-----------------------------|-----------------------------|
| Ope<br>Emerg             | Town: Co<br>erator: WI<br>gency: RIC                   | orning<br>LSON FARMS INC<br>CHARD B HALL                                   | County:                       | Steu                         | ben<br>(607) 9<br>(716) 2        | 962-5949<br>904-4333    | (710<br>Ow<br>Aut                                | 5) 204-4<br>ner Typ<br>h Rep:                               | 333<br>he: Co<br>RI  | orporate or<br>CHARD B                                 | · Comm<br>HALL            | ercial                      |                           | AT                          | ΓN:                 | RICH<br>(716) 2             | ARD B H<br>204-4333              | ALL                            | 7221                       |                             |                             |
|                          | Site Status<br>Site Type:<br>Total Activ<br>Active Cap | : Active<br>Retail Gasoline Sale<br>e Tanks : 3<br>acity : 17,000          | S                             |                              |                                  |                         | Reg<br>Cer                                       | Expire<br>t Printe  | s: ७<br>d: ६   | //28/10<br>8/12/05                                     |                           | Last Inspected              | l: 08/<br>W               | 04/19<br>STEV               | 99<br>'ENSC<br>SPDF | ON<br>ES #                  |                                  | С                              | CBS # :                    |                             |                             |
| (2)<br><u>Tank</u><br>No | (3)<br><u>Tank</u>                                     | (4)<br><u>Status</u>   | (5)<br><u>Date</u><br>Install | (5)<br><u>Date</u><br>Closed | (6)<br><u>Capacity</u><br>(gals) | (7)<br><u>Product</u>   | (8)<br><u>Tank</u><br>Type                       | (9)<br><u>Tank</u><br>IP                                    | (10)<br><u>Tank</u><br>FP                                      | (11)<br><u>Tank</u><br>SC                              | (12)<br><u>Tank</u><br>LD | (13)<br><u>x Tank</u><br>OP | (14)<br><u>Tank</u><br>SP | (15)<br><u>Tank</u><br>Dien | (16)<br><u>Pipe</u> | (17)<br><u>Pipe</u><br>Type | (18)<br><u>Pipe</u><br>FP        | (19)<br><u>Pipe</u><br>SC      | (20)<br><u>Pipe</u><br>L D | (21)<br><u>Date</u><br>Test | (21)<br><u>Next</u><br>Test |
| 006                      | 5  | In Service   | <u>1115tall</u>               | Closed                       | (gais)<br>8 000                  | 0009                    | <u>1ype</u><br>06                                | 03  | 04   | <u>sc</u>  | 02 -                      | 01                          | 01                        | 01                          | 02                  | 08                          | 99                               | <u>sc</u>                      | 07                         | 1051                        | 1081                        |
| 007                      | 5  | In Service   | 3/1/92                        |                              | 8,000                            | 0009                    | 06   | 03  | 04   | 04   | 02                        | 01                          | 01                        | 01                          | 02                  | 08                          | 99                               |                                | 07                         |                             |                             |
| 008                      | 5  | In Service   | 9/1/00                        |                              | 1,000                            | 0012                    | 06   | 03  | 04   | 04   | 02                        | 02                          | 01                        | 02                          | 02                  | 06                          | 00                               |                                | 09                         |                             |                             |
| 001                      | 5  | Closed - Removed   | 6/1/74                        | 3/1/92                       | 4,000                            | 0009                    | 01   | 00  | 00   | 00   | 00                        | 00                          |                           | 02                          | 02                  | 02                          | 00                               |                                |                            | 2/1/92                      |                             |
| 002                      | 5  | Closed - Removed   | 6/1/74                        | 3/1/92                       | 4,000                            | 0009                    | 01   | 00  | 00   | 00   | 00                        | 00                          |                           | 02                          | 02                  | 02                          | 00                               |                                | 1                          | 10/1/86                     |                             |
| 004                      | 5  | Closed - Removed   | 6/1/74                        | 3/1/92                       | 4,000                            | 0009                    | 01   | 00  | 00   | 00   | 00                        | 00                          |                           | 02                          | 02                  | 02                          | 00                               |                                |                            | 10/1/86                     |                             |
| 005                      | 5  | Closed - Removed   | 6/1/74                        | 3/1/92                       | 4,000                            | 0009                    | 01   | 00  | 00   | 00   | 00                        | 00                          |                           | 02                          | 02                  | 02                          | 00                               |                                | 1                          | 2/1/92                      |                             |
| 003                      | 5  | <b>Closed - In Place</b>   | 8/1/84                        | 9/1/00                       | 1,000                            | 0012                    | 01   | 00  | 07   | 00   | 00                        | 00                          |                           | 02                          | 02                  | 02                          | 00                               |                                | 1                          | 4/1/97                      |                             |

| <u>Action (1)</u><br>1. Initial Listing<br>2.Add Tank<br>3. Close/Remove Tank<br>4. Information<br>Correction | Status (4) 1. In-service 2. Temporarily out-of-service 3. Closed-Removed 4. Closed- In Place 5. Tank converted to | Tank Type (8)<br>01. Steel/Carbon Steel/Iron<br>02. Galvanized Steel Alloy<br>03. Stainless Steel Alloy<br>04. Fiberglass Coated Steel<br>05. Steel Tank in Concrete | External Protection (10/18)<br>00. None<br>01. Painted/Asphalt Coating<br>02. Original Sacrificial Anode<br>03. Original Impressed Current<br>04. Fiberglass<br>05. Jacketed | Piping Type (17)     9       00. None     01. Steel/Carbon Steel/Iron       02. Galvanized Steel       03. Stainless Steel Alloy       04. Fiberglass Coated Steel       05. Steel Encased in Concrete | Secondary Containment (11/19)<br>00. None<br>01. Diking (A/G)<br>02. Vault (w/access)<br>03. Vault (w/o access)<br>04. Double-Walled (U/G)<br>05. Synthetic Liner | Piping Location (16)<br>00. No Piping<br>01. Aboveground<br>02. Underground/On-ground<br>03. Aboveground/Underground<br>Combination<br>Pipe Leak Detection (20) |
|---|---|--|--|--|---|---|
| 5. Recondition/Repair/<br>Reline Tank   | Non-Regulated use Product Stored (7)  | 06. Fiberglass Reinforced<br>Plastic (FRP)<br>07. Plastic  | 06. Wrapped (Piping)<br>07. Retrofitted Sacrificial Anode  | 06. Fiberglass Reinforced<br>Plastic (FRP)   | 06. Remote Impounding Area<br>07. Excavation/Trench Liner   | 00. None<br>01. Interstitial Electronic   |
| Tank Location (3)<br>1. Aboveground-contact   | 0000. Empty<br>0001. #2 Fuel Oil<br>0002. #4 Fuel Oil   | 08. Equivalent Technology<br>09. Concrete<br>10. Urethane Clad Steel   | <ul><li>08. Retrofitted Impressed Current</li><li>09. Urethane</li><li>99. Other-please list:*</li></ul>   | 07. Plastic<br>08. Equivalent Technology<br>09. Concrete   | System<br>08. Flexible Internal Liner<br>(Bladder)  | 02. Interstitial Manual Monitoring<br>03. Vapor Well  |
| w/soil<br>2. Aboveground-contact w/<br>impervious barrier   | 0003. #6 Fuel Oil<br>0011. Jet Fuel<br>0008. Diesel   | 99. Other-please list:*<br>Internal Protection (9)   | Tank Leak Detection (12)<br>00 None<br>01.Interstitial Electronic Monitoring   | <ol> <li>Copper</li> <li>Flexible Piping</li> <li>Other-please list:*</li> </ol>   | 09. Modified Double-Walled<br>(A/G)<br>10. Impervious Underlayment  | 04. Groundwater Well<br>07. Pressurized Piping Leak<br>Detector   |
| 3.Aboveground on saddles,<br>legs, stilts, rack, or cradle<br>4. Aboveground with 10%                         | 0009. Gasoline<br>0012. Kerosene<br>0013. Lube Oil  | 00. None<br>01. Epoxy Liner<br>02. Rubber Liner  | 02. Interstitial Manual Monitoring<br>03.Vapor Well<br>04. Groundwater Well  | Overfill Prevention(13)<br>00. None<br>01. Float Vent Valve  | <ol> <li>Double Bottom (A/G)</li> <li>Other-please list:*</li> <li>Spill Prevention (14)</li> </ol>   | 08. Tank Top Sump (Piping)<br>09. Exempt Suction Piping<br>99. Other-please list:*<br>Discenser (15)  |
| or more below ground<br>5. Underground<br>6. Underground vaulted  | 0022. Waste/Used Oil<br>0259. #5 Fuel Oil   | <ul><li>03. Fiberglass Liner (FRP)</li><li>04. Glass Liner</li><li>99. Other-please list:*</li></ul>   | 05. In-Tank System (ATG)<br>06. Impervious Barrier/Concrete Pad (A/G)<br>99. Other-please list *   | 02.High Level Alarm<br>03. Automatic Shut-off<br>04. Product Level Gau   | f 00. None<br>ge(A/G) 01. Catch Basin   | 00. None<br>01. Submersible   |

99. Other-please list:\* 05. Vent Whistle \* If other, please list on a separate sheet including Tank Number 99. Other-please list:\*

99. Other-please list:\*

2642. Used Oil (Fuel)

9999. Other -please list :\*

6. Underground, vaulted,

with access

02. Transfer Station Containment 99. Other - Please list\*

00. None 01. Submersible 02. Suction 03. Gravity

# TEETER Environmental Services, Inc.

RR#1 Box 124B · North Macfee Road · Sayre, PA 18840 Phone: 570.247.7693 · Fax: 570.247.7083

۰

. ..

\_

,

Phase II Environmental Site Assessment Report



# PHASE II ENVIRONMENTAL SITE ASSESSMENT

# Sugar Creek and Crystal Cleaners 343 West Pulteney Street Corning, New York 14830

# **SUBMITTED TO:**

Mr. Donald Stiker 12 Tuscarora Road Addison, New York 14801 OR 5 PARK PLACE ADDISCON, N.Y. 14801

## **PREPARED BY:**

Teeter Environmental Services, Inc.

ĉ

David J. Teeter President

May 25, 2005

# **TABLE OF CONTENTS**

**....** 

| I.    | AUTHORIZATION                        | Page 1 |
|-------|--------------------------------------|--------|
| II.   | OBJECTIVE                            | Page 1 |
| III.  | SCOPE of WORK                        | Page 1 |
| IV.   | SITE DESCRIPTION and HISTORY         | Page 1 |
| V.    | METHODS of INVESTIGATION             | Page 2 |
|       | A. Soil Sampling and Analysis        | Page 2 |
|       | B. Groundwater Sampling and Analysis | Page 3 |
| VI.   | RESULTS                              | Page 3 |
|       | A. General Hydrogeology              | Page 3 |
|       | B. Soil Quality                      | Page 4 |
|       | C. Groundwater Quality               | Page 5 |
| VII.  | SUMMARY and RECOMMENDATIONS          | Page 8 |
| VIII. | LIMITATIONS                          | Page 9 |

# TABLES

| Table 1 | Field Screening and Observations               | Page ! | 5 |
|---------|--|--------|---|
| Table 2 | Laboratory Analytical Summary – Volatiles      | Page 🕻 | 7 |
| Table 3 | Laboratory Analytical Summary - Semi-volatiles | Page 8 | 8 |

# **APPENDICES**

| APPENDIX A        | Figures       |   |
|-------------------|---------------|---|
|                   | Figure 1      | Site Map                                  |
|                   | Figure 2      | Area Map                                  |
|                   | Figure 3      | Aerial View                               |
|                   | Figure 4      | Topographic Setting                       |
|                   | Figure 5      | Contaminant Concentrations in Groundwater |
| <b>APPENDIX B</b> | Subsurface Lo | gs  |
| <b>APPENDIX C</b> | Laboratory Re | port                                      |

# I. AUTHORIZATION

Teeter Environmental Services, Inc. was authorized by Mr. Donald Stiker, 12 Tuscarora Road, Addison, New York to perform a Phase II Environmental Site Assessment (ESA) of the property located at 343 West Pulteney Street, Corning, New York. Mr. Stiker owns the property and leases it to retail and commercial interests. The ESA was performed on April 21, 2005.

# **II. OBJECTIVE**

The objective of the ESA was to determine if subsurface soil and groundwater were impacted with petroleum and solvent-type hydrocarbons originating from historical usage of property as a service station and a dry cleaning operation. Currently, a Sugar Creek convenience store and Crystal Cleaners occupy the site. Sugar Creek also retails gasoline and kerosene.

# **III. SCOPE OF WORK**

The following tasks were performed:

- Completed six (6) soil borings to a maximum depth of 16 feet below ground surface using a Geoprobe<sup>®</sup> direct-push soil sampling rig.
- Obtained soil samples at continuous four (4) foot intervals, observed each for evidence of petroleum impact, characterized lithologically, screened for volatile organic compounds (VOC's) using an organic vapor meter (OVM), and containerized for potential laboratory analysis.
- Submitted groundwater samples from four (4) of the borings for laboratory analysis. Two (2) samples were analyzed for the New York State Department of Environmental Conservation (NYSDEC) short list of regulated compounds for gasoline releases using EPA Method 8260B, STARS list, two (2) samples were analyzed for the NYSDEC short list of regulated compound for fuel oil releases using EPA Method 8270C, STARS list, and two (2) samples were analyzed using EPA Method 8260B full list which includes chlorinated hydrocarbons.
- Prepared the following report of the findings.

# **IV. SITE DESCRIPTION**

The site is located in a mixed commercial and residential area near the western boundary of the city of Corning, New York. The property is generally square with the north and west boundaries approximately 150 feet in length. The east property line is slightly longer.

One single story block building oriented east-west is located at the north end of the site. The building is 128 feet long and 40 feet wide and is separated into three sections to facilitate three tenants. The Sugar Creek convenience store occupies the west side of building. Crystal Cleaners dry cleaning operation is located to the east of Sugar Creek and the third and smaller section of the building at the east end is currently vacant. A laundromat most recently occupied that part of the building. Dry cleaning businesses have apparently been on-site since 1972.

A 6,000-gallon underground storage tank (UST) containing kerosene is located at the southwest corner of the building. The kerosene dispenser is also located at southwest corner off the pavement and up against the wall. The gasoline fuel dispensers are located near the center of the site beneath a free standing canopy. The area beneath the canopy is paved with concrete. Two 8,000-gallon UST's containing the gasoline are situated to the west of the dispensers. Other than beneath the canopy and the concrete protective pads over the UST's, most of the site is paved with asphalt. The area immediately west of the building is grass.

Adjacent properties include residences to the north, northeast, and northwest, Community Bank N.A. to the east across Cutler Avenue, Fazzary Wine & Liquor to southeast across West Pulteney Street, Ontario Carpet and Southern Glass Service to the southwest across West Pulteney Street, and Deck's Auto used car sales to the west across Townsend Avenue.

Refer to Figure 1 in Appendix A for a site map, Figure 2 for map of the area, and Figure 3 for an aerial photographic view of the site and surrounding properties.

# **V. METHODS OF INVESTIGATION**

# A. Soil Sampling and Analysis

Chambers Environmental Group, Inc., Bellefonte, Pennsylvania was contracted to perform the borings under the supervision of David Teeter of Teeter Environmental Services, Inc. The soil borings were completed using a Geoprobe<sup>®</sup> Model 540UD direct-push soil probing rig. Soil samples were obtained by advancing a two (2) inch diameter steel drive point attached to steel drive rods into the subsurface with a diesel-powered percussion hammer. At the desired depth, the point was retracted leaving a two (2) inch diameter open borehole. A two-inch diameter, 48-inch long hollow steel sampling tube with an acetate liner was then attached to the drive rods, set to the bottom of the borehole, and driven the length of the tube. After retrieving the soil core, the drive point was reinserted into the boring and advanced to the next desired sample depth.

Samples were obtained to maximum depth of 16 feet. All soil samples were observed for petroleum impact (sheen, discoloration, odor, etc.) and characterized lithologically. The samples were placed in airtight containers to allow vapors to accumulate in the headspace. The headspace was then screened for VOC's, expressed in parts per million (ppm), using a ThermoEnvironmental Model 580B organic vapor meter (OVM). Since groundwater was encountered and could be sampled, soil samples were not submitted for laboratory analysis.

Refer to Figure 1 in Appendix A for soil boring locations.

# B. Groundwater Sampling and Analysis

Groundwater was obtained from temporary small diameter PVC wells installed in four of the boreholes. The samples were collected by inserting 3/8-inch tubing connected to a Geotech 2 peristaltic pump. Groundwater was pumped out of the wells for a short period of time to reduce turbidity. Samples were then containerized in 40-milliliter zero-headspace vials preserved with hydrochloric acid and packed in an ice-filled cooler. Samples from two borings were also contained one liter amber bottles. The samples were submitted to Eastern Laboratory Services Ltd., South Waverly, PA NYS Laboratory ID #11216) for analysis. Samples contained in the vials from two borings analyzed for the NYSDEC short list of regulated compounds for gasoline releases using EPA Method 8260B, STARS list and samples contained in the vials from two additional boring were analyzed using EPA Method 8260B full list which includes chlorinated hydrocarbons. The samples contained in the amber bottles were analyzed for NYSDEC short list of regulated compound for fuel oil releases using EPA Method 8270C, STARS list.

# VI. RESULTS

# A. General Hydrogeology

The site lies within the Chemung River valley at an approximate elevation of 930 feet above mean sea level. The area specific to the site is topographically flat with steep bedrock hills located about <sup>3</sup>/<sub>4</sub>-mile to the north and <sup>1</sup>/<sub>4</sub>-mile to the south. The Chemung River lies approximately 1,000 feet to the south of the site and flows to the east.

Based on characterization of soil samples from the six borings completed, five of which were advanced to 16 feet and one to 12 feet, surficial geology (unconsolidated material) varies somewhat throughout the site horizontally and vertically. In general, there are two prevailing soil mixtures: brown and reddish brown sand and rounded gravel with lesser amounts of silt and brown and reddish brown silt with lesser amounts of rounded gravel with little sand.

Clayey silt was evident in samples from some boring generally from a depth of 4 to 8 feet. The sand and gravel fractions were fairly loose while the predominantly silty soils were more dense. The site lies in a valley-fill aquifer system where much of the surficial geology consists of outwash sand and gravel deposited by receding glaciers. However, the observed characteristics of the soil samples, namely the silt content and poorly sorted coarser material intermixed suggest an alluvial (stream or river) depositional environment. This is understandable due to the site's proximity to the Chemung River which invariably has meandered throughout the last several thousand years. The clayey silt soils may even represent a lacustrine (lake, lagoon, or swamp) depositional environment where perhaps the river meandered a considerable distance from the site. Some organics (micro roots) were observed in samples from one boring. The roundness of the gravel, however, indicates that running water was responsible for most of the deposition of soils. The variability of grain sizes and matrices is common in glacial and allviual depositional environments.

Depth to groundwater as evidenced by the moisture content of the soil ranged from 10 to 12 feet in most borings. Direction of groundwater flow is almost certainly to the south or southeast toward and with the flow of the Chemung River. It is unusual for groundwater to flow against the direction of major flowing surface water. Horizontal and vertical groundwater flow velocity will likely vary as it will move more readily through the looser and coarser grained soils.

Bedrock was not encountered during the ESA to the maximum boring depth of 16 feet. Bedrock in the area consists of Upper Devonian age Gardeau Formation shales and siltstones of the West Falls Group (Rickard and Fisher, 1970).

Refer to Appendix B for subsurface logs containing lithologic characterization for each sample interval. A topographic map is included as Figure 4 in Appendix A.

# **B.** Soil Quality

Six borings were completed throughout the site adjacent to and/or hydraulically downgradient of potential sources of contaminant releases including the UST's, fuel dispensers, and the dry cleaning shop. Boring locations are indicated on Figure 1 in Appendix A. The intent was to advance the borings until groundwater was encountered or until any contamination had been vertically delineated. In all, 23 soil samples were collected from borings advanced to a depth of 16 feet. Groundwater was generally encountered at a depth of approximately 10 to 12 feet.

No evidence of petroleum or solvent impact such as odor, sheen, discoloration, or separate phase product was observed in any of the samples. All OVM were zero (0) parts per million (ppm) indicating the lack of volatile or semi-volatile vapors in the unsaturated and saturated soils.

Because there was no evidence of contamination, no soil samples were submitted for laboratory analysis.

Refer to Table 1 for a summary of the sampling intervals, OVM readings, and observations.

## Table 1

## **Field Screening and Observations**

| Boring ID | Sampling<br>Interval (feet) | OVM Reading<br>(ppm) | Observations       |
|-----------|-----------------------------|----------------------|--------------------|
| B1        | 0-4                         | 0                    | No observed impact |
|           | 4-8                         | 0                    | No observed impact |
|           | 8-12                        | 0                    | No observed impact |
|           | 12-16                       | 0                    | No observed impact |
| B2        | 0-4                         | 0                    | No observed impact |
|           | 4-8                         | 0                    | No observed impact |
|           | 8-12                        | 0                    | No observed impact |
|           | 12-16                       | 0                    | No observed impact |
| B3        | 0-4                         | 0                    | No observed impact |
|           | 4-8                         | 0                    | No observed impact |
|           | 8-12                        | 0                    | No observed impact |
|           | 12-16                       | 0                    | No observed impact |
| B4        | 0-4                         | 0                    | No observed impact |
|           | 4-8                         | 0                    | No observed impact |
|           | 8-12                        | 0                    | No observed impact |
|           | 12-16                       | 0                    | No observed impact |
| B5        | 0-4                         | 0                    | No observed impact |
|           | 4-8                         | 0                    | No observed impact |
|           | 8-12                        | 0                    | No observed impact |
|           | 12-16                       | 0                    | No observed impact |
| B6        | 0-4                         | 0                    | No observed impact |
|           | 4-8                         | 0                    | No observed impact |
|           | 8-12                        | 0                    | No observed impact |

## April 21, 2005

# C. Groundwater Quality

Groundwater samples were obtained from B1 near the kerosene UST, B3 downgradient of the gasoline UST's and adjacent to the gasoline dispensers, B4 downgradient of the gasoline dispensers, and B5 near and downgradient of Crystal Cleaners. B4 may also be considered downgradient of Crystal Cleaners. The samples from B1 and B3 were analyzed for the NYSDEC short list (STARS list) of volatile petroleum hydrocarbons which target compounds specifically regulated for gasoline releases.

The samples were also analyzed for the NYSDEC short list of semi-volatile petroleum hydrocarbons which target compounds specifically regulated for fuel oil and kerosene releases. Although B1 is hydraulically upgradient of the gasoline UST's, NYSDEC requires that samples tested for semi-volatiles also be analyzed for volatile hydrocarbons as there are some volatile constituents in fuel oils. Samples from B4 and B5 were analyzed for the full EPA Method 8260B compound list which includes chlorinated or solvent-type hydrocarbons. Two in particular, tetrachlorethene (PCE) and trichloroethene (TCE), have historically been used in the dry cleaning industry as primary cleaning agents.

As summarized in Table 2, naphthalene, toluene, and m,p-xylenes were detected in the sample from B1 at concentrations slightly above the applicable regulatory standards. Only naphthalene was detected in the sample from B3 at a concentration below the regulatory standard of 10 micrograms per liter ( $\mu$ g/l). Naphthalene was also evident in the sample from B4 at a level below the regulatory standard. Naphthalene is actually a semi-volatile hydrocarbon that is also a constituent of gasoline. PCE was detected in the sample from B5 at a concentration barely exceeding the regulatory standard of 5 micrograms per liter ( $\mu$ g/l). PCE was also found in the sample from B4 at higher concentration of 43.1  $\mu$ g/l.

Table 2 contains the full EPA STARS list of target compounds. Since the complete EPA Method 8260B list includes over 60 compounds, only the detected chlorinated hydrocarbon is included in the table. Refer to Appendix C for a copy of the full analytical report which includes all target compounds. The detected hydrocarbon concentrations at each sampling point are illustrated on Figure 5 in Appendix A.

There were no semi-volatile hydrocarbons detected in either sample from B1 and B3 above the reporting limit of 10  $\mu$ g/l, including naphthalene which also analyzed for the EPA 8270C method. All detections of naphthalene under the EPA 8260B method were below 10  $\mu$ g/l with the exception of 29.7  $\mu$ g/l in B1.

The analytical results for the semi-volatiles analysis are summarized in Table 3. A copy of the full laboratory report is included in Appendix C.

# Table 2

# Laboratory Analytical Summary Volatile Hydrocarbons and MTBE in Groundwater by EPA Method 8260B STARS and 8260B full list ( $\mu g/l$ )

| April 21, 2005          |           |         |             |         |                    |  |  |
|-------------------------|-----------|---------|-------------|---------|--------------------|--|--|
| Compound                | <b>B1</b> | B3      | <b>B4</b> * | B5*     | NYSDEC<br>Standard |  |  |
| Benzene                 | ND<5.00   | ND<5.00 | ND<5.00     | ND<5.00 | 1                  |  |  |
| n-Butylbenzene          | ND<5.00   | ND<5.00 | ND<5.00     | ND<5.00 | 5                  |  |  |
| sec-Butylbenzene        | ND<5.00   | ND<5.00 | ND<5.00     | ND<5.00 | 5                  |  |  |
| Ethylbenzene            | ND<5.00   | ND<5.00 | ND<5.00     | ND<5.00 | 5                  |  |  |
| Isopropylbenzene        | ND<5.00   | ND<5.00 | ND<5.00     | ND<5.00 | 5                  |  |  |
| p-lsopropyltoluene      | ND<5.00   | ND<5.00 | ND<5.00     | ND<5.00 | 5                  |  |  |
| n-Propylbenzene         | ND<5.00   | ND<5.00 | ND<5.00     | ND<5.00 | 5                  |  |  |
| Naphthalene             | 29.7      | 9.3     | 5.8         | ND<5.00 | 10                 |  |  |
| Toluene                 | 6.2       | ND<5.00 | ND<5.00     | ND<5.00 | 5                  |  |  |
| 1,2,4-Trimethylbenzene  | ND<5.00   | ND<5.00 | ND<5.00     | ND<5.00 | 5                  |  |  |
| 1,3,5-Trimethylbenzene  | ND<5.00   | ND<5.00 | ND<5.00     | ND<5.00 | 5                  |  |  |
| m,p-Xylenes             | 5.2       | ND<5.00 | ND<5.00     | ND<5.00 | 5                  |  |  |
| o-Xylene                | ND<5.00   | ND<5.00 | ND<5.00     | ND<5.00 | 5                  |  |  |
| tert-Butylbenzene       | ND<5.00   | ND<5.00 | ND<5.00     | ND<5.00 | 5                  |  |  |
| MTBE                    | ND<5.00   | ND<5.00 | ND<5.00     | ND<5.00 | 10                 |  |  |
| Total Detected          | 41.1      | 9.3     | 48.9        | 7.0     |                    |  |  |
| Tetrachloroethene (PCE) | ND<5.00   | ND<5.00 | 43.1        | 7.0     | 5                  |  |  |

\*B1 and B3 analyzed using abbreviated compound list. B4 and B5

analyzed using full method compound list to include chlorinated

hydrocarbons. Refer to the laboratory report for the complete list.

µg/l – micrograms per liter

ND - Not detected above the indicated reporting limit

Naphthalene is considered a semi-volatile compound

MTBE is a non-hydrocarbon gasoline additive

## Page 8

# Table 3

| Compound                 | <b>B</b> 1 | B3      | NYSDEC<br>Standard |  |  |
|--------------------------|------------|---------|--------------------|--|--|
| Naphthalene              | ND<10.0    | ND<10.0 | 10                 |  |  |
| Acenaphthene             | ND<10.0    | ND<10.0 | 20                 |  |  |
| Fluorene                 | ND<10.0    | ND<10.0 | 50                 |  |  |
| Phenanthrene             | ND<10.0    | ND<10.0 | 50                 |  |  |
| Anthracene               | ND<10.0    | ND<10.0 | 50                 |  |  |
| Fluoranthene             | ND<10.0    | ND<10.0 | 50                 |  |  |
| Pyrene                   | ND<10.0    | ND<10.0 | 50                 |  |  |
| Benzo(a)anthracene       | ND<10.0    | ND<10.0 | 0.002 or MDL       |  |  |
| Chrysene                 | ND<10.0    | ND<10.0 | 0.002 or MDL       |  |  |
| Benzo (b) fluoranthene   | ND<10.0    | ND<10.0 | 0.002 or MDL       |  |  |
| Benzo (k) fluoranthene   | ND<10.0    | ND<10.0 | 0.002 or MDL       |  |  |
| Benzo (a) pyrene         | ND<10.0    | ND<10.0 | 0.002 or MDL       |  |  |
| Indeno (1,2,3-cd) pyrene | ND<10.0    | ND<10.0 | 0.002 or MDL       |  |  |
| Dibenzo (a,h) anthracene | ND<10.0    | ND<10.0 | 50                 |  |  |
| Benzo (g,h,i) perylene   | ND<10.0    | ND<10.0 | 0.002 or MDL       |  |  |
| Total Detected           | ND         | ND      |                    |  |  |

# Laboratory Analytical Summary Semi-Volatile Aromatic Hydrocarbons in Groundwater by EPA Method 8270C STARS (µg/l)

µg/l – micrograms per liter

ND – Not detected above the indicated reporting limit

MDL -- Method detection limit

# VII. SUMMARY and RECOMMENDATIONS

The Phase II ESA performed by Teeter Environmental Services, Inc. at 343 West Pulteney Street, Corning, New York yielded the following relevant information:

- The property is owned by Mr. Donald Stiker and is leased to parties operating businesses on-site. The single one-story building is divided into three sections with a Sugar Creek convenience store occupying the west part of the building and Crystal Cleaners, a dry cleaning operation, occupies the center section. The east portion of the building is currently vacant and was formerly used as a laundromat.
- Sugar Creek retails gasoline and kerosene. One (1) 6,000-gallon kerosene UST and two (2) 8,000-gallon gasoline UST's are located on-site. One kerosene dispenser is located adjacent to the kerosene tank. Gasoline fuel dispensers are situated near the center of the site under a free standing canopy.
- The site is located in a mixed residential and commercial area. There are no industrial operations adjacent to the site.

- Six (6) soil borings were advanced to a depth of 16 feet (12 feet in one boring) and were located near and hydraulically downgradient of potential sources of petroleum and chlorinated hydrocarbon releases including fuel dispensers, UST's, and the Crystal Cleaners facility. Four-foot soil core samples were obtained at continuous interval to the bottom of each borehole.
- Surficial geology varies horizontally and vertically and generally consists of brown and reddish brown gravelly silt with varying amounts of sand, sandy gravel with little silt, and clayey silt with some sand and gravel. The gravel is rounded and the variations in soil characteristics suggest an alluvial (river) depositional environment. The Chemung River is located approximately 1,000 feet south of the site.
- Depth to groundwater is approximately 10 to 12 feet below ground surface and most likely flows to the south or southeast toward the eastward flowing Chemung River.
- There was no evidence of petroleum or solvent impact such as odor, discoloration, sheen, or separate product in any of the soil samples. All OVM readings were zero (0) parts per million. As such, no soil samples were submitted for laboratory analysis.
- Groundwater samples from four (4) boring were submitted for laboratory analysis. Of the petroleum hydrocarbons, naphthalene was detected in samples from B1, B3, and B4. The concentration of 29.7 micrograms per liter ( $\mu$ g/l) in B1 was slightly above the regulatory standard of 10  $\mu$ g/l. Xylenes and toluene were also detected in B1 at concentrations of 5.2  $\mu$ g/l and 6.2  $\mu$ g/l, respectively, barely exceeding the 5  $\mu$ g/l standard.
- Groundwater samples from B4 and B5 were also analyzed for chlorinated hydrocarbons. Tetrachloroethene (PCE), commonly used in the dry cleaning industry, was detected at concentrations of 43.1  $\mu$ g/l and 7.0  $\mu$ g/l, respectively. The regulatory standard is 5  $\mu$ g/l.

It is clear that mass contamination due to petroleum or solvent releases has not occurred at the site. The concentrations of detected compounds exceeding regulatory standards did not do so by a large margin. Nonetheless, because there were exceedences in the standards, it is required that the results of the ESA be submitted to NYSDEC. NYSDEC may require additional investigation. Otherwise, no additional action is recommended pending review by NYSDEC.

# **VIII. LIMITATIONS**

This report is based on a limited number of soil and groundwater samples and chemical analyses. The conclusions presented in this report are based only on the observations made during this investigation.

The report presents a description of the subsurface conditions observed at each boring location during this investigation. Conclusions and recommendations set forth are applicable only to the facts and conditions at the time of this investigation.

In performing professional services, Teeter Environmental uses the degree of care and skill exercised under similar circumstances by members of the environmental profession practicing in the same or similar locality under similar conditions. The standard of care shall be judged exclusively as of the time these services are rendered and not according to later standards. Teeter Environmental makes no express or implied warranty beyond its conformance to this standard.

Teeter Environmental shall not be responsible for conditions or consequences arising from relevant facts that were concealed, withheld, or not fully disclosed for this report. Teeter Environmental believes that all information contained in this report is factual, however no guarantee is made or implied.
Phase II ESA – Sugar Creek and Crystal Cleaners 343 W. Pulteney Street, Corning, NY May 25, 2005

### **APPENDIX** A

### FIGURES

The ter sense provident of Sciences S.





April 21, 2005

1 in = 990 ft



Phase II ESA Sugar Creek and Crystal Cleaners 343 West Pulteney Street Corning, NY 14830 April 21, 2005

Aerial View April 16, 1995 1 in 280 ft Adapted from USGS Series Topographic Corning Quadrangle 1976



Phase II ESA Sugar Creek and Crystal Cleaners 343 West Pulteney Street Corning, NY 14830 April 21, 2005

Topographic Setting 1 in = 1,460 ft Ν



Phase II ESA – Sugar Creek and Crystal Cleaners 343 W. Pulteney Street, Corning, NY May 25, 2005

### **APPENDIX B**

### SUBSURFACE LOGS

Tector Environmental Services

| 5  |                         | _              |                 |             |               |   |  |  |  |  |  |  |  |
|--|-------------------------|----------------|-----------------|-------------|---------------|---|--|--|--|--|--|--|--|
|  | Teet                    | er Env         | vironn          | nental      | Servic        | ces, Inc. SUBSURFACE LOG  |  |  |  |  |  |  |  |
|  | SITE LO                 | DCATION:       |                 | Cry         | stal Cleane   | ers and Sugar Creek, 343 W. Pulteney Street, Corning, New York, 14830             |  |  |  |  |  |  |  |
|  | CLIENT:                 |                | Mr. Dona        | ald Stiker, | 12 Tuscaro    | ra Road, Addison, New York 14801 WELL/BORING ID: B1                               |  |  |  |  |  |  |  |
|  | START I                 |                | April 2         | 1, 2005     | COMPL         | ETION DATE: April 21, 2005 RECORDED BY: Chris Treese                              |  |  |  |  |  |  |  |
|  | GROUN                   | DWATER         | DEPTH W         | HILE DRIL   | LING:         | ~7 feet GROUNDWATER DEPTH AFTER COMPLETION: NA                                    |  |  |  |  |  |  |  |
|  | WEATH                   | ER CONDI       | ITIONS:         |             | Sunny, 5      | DRILLING CONTRACTOR: Chambers Environmental Group                                 |  |  |  |  |  |  |  |
|  | DRILL R                 | IG: Ge         | -<br>eoprobe® 5 | 40UD I      |               | & TYPE:       2" OD drive point       DRILLER NAME(S):       Keith Skow           |  |  |  |  |  |  |  |
|  | Material Classification |                |                 |             |               |   |  |  |  |  |  |  |  |
| -  | Sample                  | OVM<br>Booding | Sample          | Sampler     | Recovery      |   |  |  |  |  |  |  |  |
|  | No.                     | (ppm)          | (feet)          | Type*       | (inches)      | trace $-1-10\%$ intrie $-11-20\%$ some $-21-35\%$ and $-36-50\%$                  |  |  |  |  |  |  |  |
|  |                         | •              |                 |             |               | f-fine m-medium c-coarse  |  |  |  |  |  |  |  |
| $\frac{1}{2} + \frac{1}{2} + \frac{1}$ |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
| ľ  | 2                       | 0              | 4-8             | MC          | 15            | t brown rounded fmc GRAVEL some SILT. Wet at ~7 feet. No unusual odors            |  |  |  |  |  |  |  |
| u i  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
|  | 3                       | 0              | 8-12            | MC          | 12            | brown m SAND and rounded fmc GRAVEL little rounded c SAND. Wet. No unusual odors. |  |  |  |  |  |  |  |
| -  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
|  | 4                       | 0              | 12-16           | MC          | 15            | brown rounded fmc GRAVEL and mc SAND. Saturated. No unusual odors.                |  |  |  |  |  |  |  |
| -  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
|  |                         |                |                 |             |               | Boring terminated at 16 feet below ground surface.                                |  |  |  |  |  |  |  |
|  |                         |                | _               |             |               |   |  |  |  |  |  |  |  |
|  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
| ļ  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
|  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
|  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
| j  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
|  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
| į  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
| -  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
| ļ  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
| !<br>  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
|  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
|  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
|  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
| ļ  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |
| ╸  | NOTES                   | Groundwat      | er sampled f    | rom borehok | e and submitt | ed for laboratory analysis.   |  |  |  |  |  |  |  |
|  | *MC – 0                 | GEOPROBE       |                 | )RES-       | - SPLIT SPO   | ON DPSS – DIRECT PUSH SPLIT SPOON SH – SHELBY TUBE C – BEDROCK CORE               |  |  |  |  |  |  |  |
| -  |                         |                |                 |             |               |   |  |  |  |  |  |  |  |

| SITE LO | DCATION:       |                       | Cry         | stal Cleane | s and Sugar Creek, 343 W. Pulteney Street, Corning, New York, 14830<br>a Road, Addison, New York 14801 WELL/BORING ID: <b>B2</b>               |  |  |  |  |  |  |
|---------|----------------|-----------------------|-------------|-------------|--|--|--|--|--|--|--|
| CLIENT  | :              | Mr. Dona              | ald Stiker, | 12 Tuscaro  |  |  |  |  |  |  |  |
| START   | DATE:          | April 2               | 1, 2005     | COMPL       | ETION DATE: April 21, 2005 RECORDED BY: Chris Treese   |  |  |  |  |  |  |
| GROUN   | DWATER         | DEPTH W               | HILE DRIL   | Ling:       | ~11 feet GROUNDWATER DEPTH AFTER COMPLETION: NA  |  |  |  |  |  |  |
| WEATH   | ER CONDI       | TIONS:                |             | Sunny, 5    |  |  |  |  |  |  |  |
| DRILL R | lG: Ge         | oprobe <sup>®</sup> 5 | 40UD [      |             | & TYPE: 2" OD drive point DRILLER NAME(S): Keith Skow  |  |  |  |  |  |  |
|         |                |                       |             |             | Material Classification  |  |  |  |  |  |  |
| Sample  | OVM<br>Reading | Sample<br>Interval    | Sampler     | Recovery    | trace - 1-10% little - 11-20% some - 21-35% and - 36-50%   |  |  |  |  |  |  |
| NO.     | (ppm)          | (feet)                | i ype"      | (incries)   | f-fine m-medium c-coarse   |  |  |  |  |  |  |
| 1       | 0              | 0-4                   | MC          | 6           | 4° asphalt $\rightarrow$ 2° rounded m gravel (PEA STONE FILL). Moist. No unusual odors.  |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
| 2       | 0              | 4-8                   | MC          | 1           | trace PEA STONE FILL.  |  |  |  |  |  |  |
| 2       | 0              | 9.10                  |             | 24          | 12" DEA STONE EILL   |  |  |  |  |  |  |
|         | 0              | 0-12                  |             | 24          | TS PEASTONE FILL $\rightarrow$ TT reduisit brown CLAYEY SiLT and rounded the GRAVEL back<br>fmc SAND. Saturated at ~11 feet. No unusual odors. |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
| 4       | 0              | 12-16                 | MC          | 8           | brown SILT and rounded fmc GRAVEL. Saturated. No unusual odors.  |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
|         |                |                       |             |             | Borino terminated at 16 feet below ground surface  |  |  |  |  |  |  |
|         |                |                       |             |             | During terminated at 10 feet below ground surface.   |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
|         |                |                       |             |             |  |  |  |  |  |  |  |
|         |                |                       |             | 1           | · · · · · · · · · · · · · · · · · · ·  |  |  |  |  |  |  |

| SITE LO                            | DCATION: |                       | Cry         | stal Cleane                            | rs and Sugar Creek, 343 W. Pulteney Street, Corning, New York, 14830   |  |  |  |  |  |
|------------------------------------|----------|-----------------------|-------------|--|--|--|--|--|--|--|
| CLIENT                             |          | Mr. Dona              | ald Stiker, | 12 Tuscaro                             | ra Road, Addison, New York 14801 WELL/BORING ID: B3  |  |  |  |  |  |
| START                              | DATE:    | April 2               | 1, 2005     | COMPL                                  | ETION DATE: April 21, 2005 RECORDED BY: Chris Treese   |  |  |  |  |  |
| GROUN                              | DWATER   | DEPTH W               | HILE DRIL   | LING:                                  | ~10 feet GROUNDWATER DEPTH AFTER COMPLETION: NA  |  |  |  |  |  |
| WEATH                              | ER CONDI | TIONS:                |             | Sunny, 5                               | 5° DRILLING CONTRACTOR: Chambers Environmental Grou  |  |  |  |  |  |
|                                    | IG: Ge   | oprobe <sup>®</sup> 5 | 40UD [      | ORILL SIZE                             | & TYPE: 2" OD drive point DRILLER NAME(S): Keith Skow  |  |  |  |  |  |
|                                    |          |                       |             | ······································ | Material Classification  |  |  |  |  |  |
| Sample OVM Sample Sampler Recovery |          |                       |             |  | trace 1-10% little 11-20% some 21-35% and 36-50%   |  |  |  |  |  |
|                                    | (ppm)    | (feet)                |             | (,                                     | f-fine m-medium c-coarse   |  |  |  |  |  |
| 1                                  | 0        | 0-4                   | мс          | 25                                     | 4" asphalt $\rightarrow$ 8" gray silt, sand, f gravel, brick fragments (FILL) $\rightarrow$ 13" reddish brown f SAND |  |  |  |  |  |
|                                    |          |                       | _           |  | and rounded fm GRAVEL little SILT. Moist. No unusual odors.  |  |  |  |  |  |
| 2                                  | 0        | 4 0                   |             | 17                                     | brown (SAND and SILT some rounded for CRAVEL little rounded on SAND. Maint   |  |  |  |  |  |
|                                    | 0        | 4-0                   | IVIC        |  | Drown t SAND and SIL1 some rounded th GRAVEL little rounded mc SAND. Moist.  |  |  |  |  |  |
|                                    |          |                       |             |  |  |  |  |  |  |  |
| 3                                  | 0        | 8-12                  | мс          | 22                                     | reddish brown moderately dense CLAYEY SILT and rounded fm GRAVEL little rounded                                      |  |  |  |  |  |
|                                    |          |                       |             |  | c GRAVEL and c SAND. Saturated at ~10 feet. No unusual odors.  |  |  |  |  |  |
| 4                                  | 0        | 12 16                 | MC          | 10                                     | $2^{\circ}$ it brown soft CLAV $\sim 7^{\circ}$ brown in SAND little rounded is SAND and f CPAV/EL. Wet. No          |  |  |  |  |  |
| 4                                  | 0        | 12-10                 | INIC        | 10                                     | $3 \text{ to bown solid CLAT} \rightarrow 7  brown in SAME have founded CSAME and TGRAVEL. We have unusual odors.$   |  |  |  |  |  |
|                                    |          |                       |             |  |  |  |  |  |  |  |
|                                    |          |                       |             |  |  |  |  |  |  |  |
| _                                  |          |                       |             |  | Boring terminated at 16 feet below ground surface.   |  |  |  |  |  |
|                                    |          |                       |             |  |  |  |  |  |  |  |
|                                    |          |                       | _           |  |  |  |  |  |  |  |
|                                    |          |                       |             |  |  |  |  |  |  |  |
|                                    |          |                       |             |  |  |  |  |  |  |  |
|                                    |          |                       |             |  |  |  |  |  |  |  |
|                                    |          |                       |             |  |  |  |  |  |  |  |
|                                    |          |                       |             |  |  |  |  |  |  |  |
|                                    |          |                       |             |  |  |  |  |  |  |  |
|                                    |          |                       |             | <u> </u>                               |  |  |  |  |  |  |
|                                    |          |                       |             |  |  |  |  |  |  |  |
|                                    |          |                       |             |  | · · · · · · ·_   |  |  |  |  |  |

| SITE LO       | CATION:    |                       | Cry              | stal Cleane          | s and Sugar Creek, 343 W. Pulteney Street, Corning, New York, 14830  |  |  |  |  |  |
|---------------|------------|-----------------------|------------------|----------------------|--|--|--|--|--|--|
| CLIENT:       |            | Mr. Dona              | ald Stiker, '    | 12 Tuscaro           | ra Road, Addison, New York 14801 WELL/BORING ID: B4  |  |  |  |  |  |
| START         | DATE:      | April 2               | 1, 2005          | COMPL                | ETION DATE: April 21, 2005 RECORDED BY: Chris Treese   |  |  |  |  |  |
| GROUN         | <br>DWATER | DEPTH W               | HILE DRIL        | _<br>LING:           | ~12 feet GROUNDWATER DEPTH AFTER COMPLETION: NA  |  |  |  |  |  |
| WEATH         | ER CONDI   | TIONS:                |                  | Sunny, 5             | 5° DRILLING CONTRACTOR: Chambers Environmental Grou  |  |  |  |  |  |
| DRILL R       | IG: Ge     | oprobe <sup>®</sup> 5 | 40UD [           | DRILL SIZE           | & TYPE: 2" OD drive point DRILLER NAME(S): Keith Skow  |  |  |  |  |  |
|               | 0\/M       | Sample                |                  |                      | Material Classification  |  |  |  |  |  |
| Sample<br>No. | Reading    | Interval<br>(feet)    | Sampler<br>Type* | Recovery<br>(inches) | trace – 1-10% little – 11-20% some – 21-35% and – 36-50%   |  |  |  |  |  |
|               | (ppin)     | (ieel)                |                  |                      | f-fine m-medium c-coarse   |  |  |  |  |  |
| 1             | 0          | 0-4                   | мс               | 31                   | 5" asphalt $\rightarrow$ 26" dk brown and black silt, sand f gravel, brick fragments (FILL). Moist. No   |  |  |  |  |  |
|               |            |                       |                  |                      | unusual odors.   |  |  |  |  |  |
|               | 0          | 4.0                   | MC               |                      | E" similar metarial 4" d/ braun SILT trace CLAX and OBCANICS (misse plant roots)   |  |  |  |  |  |
|               | 0          | 4-0                   | - WIC            | 40                   | $3$ similar material $\rightarrow 4$ dk brown SiLT frace CLAY and OKGANICS (micro plant rous) $\rightarrow 39$ it brown moderately dense SILT little gravish and greenish CLAY streaking and motifing. |  |  |  |  |  |
|               |            |                       |                  | ·····                | Moist. No unusual odors.   |  |  |  |  |  |
|               |            |                       |                  |                      |  |  |  |  |  |  |
| 3             | 0          | 8-12                  | MC               | 21                   | 6" similar soils $\rightarrow$ 15" dk brown CLAYEY SILT and rounded fmc GRAVEL little fmc SAND.  |  |  |  |  |  |
|               |            |                       |                  |                      | Moist. No unusual odors.   |  |  |  |  |  |
| 4             | 0          | 12-16                 | MC               | 15                   | 9" similar soits $\rightarrow$ 6" dk brown and reddish silt and f SAND some rounded fm GRAVEL.   |  |  |  |  |  |
|               |            |                       |                  |                      | Saturated. No unusual odors.   |  |  |  |  |  |
|               |            |                       |                  |                      |  |  |  |  |  |  |
|               |            |                       |                  |                      |  |  |  |  |  |  |
|               |            |                       |                  |                      | Boring terminated at 16 feet below ground surface.   |  |  |  |  |  |
|               |            |                       |                  |                      |  |  |  |  |  |  |
|               |            |                       |                  |                      |  |  |  |  |  |  |
|               |            | _                     |                  | _                    |  |  |  |  |  |  |
|               |            |                       |                  |                      |  |  |  |  |  |  |
|               |            |                       |                  |                      |  |  |  |  |  |  |
|               |            |                       |                  |                      |  |  |  |  |  |  |
|               |            |                       |                  |                      |  |  |  |  |  |  |
|               |            |                       |                  |                      |  |  |  |  |  |  |
|               |            |                       |                  |                      |  |  |  |  |  |  |
|               |            |                       |                  |                      | L  |  |  |  |  |  |

| SITE LO       | OCATION:       |                    | Cry                                     | stal Cleane       | rs and Sugar Creek, 343 W. Pulteney Street, Corning, New York, 14830  |  |  |  |  |  |  |
|---------------|----------------|--------------------|---|-------------------|---|--|--|--|--|--|--|
| CLIENT        |                | Mr. Dona           | ald Stiker, <sup>.</sup>                | 12 Tuscaro        | a Road, Addison, New York 14801 WELL/BORING ID: B5  |  |  |  |  |  |  |
| START         | DATE:          | April 2            | 1, 2005                                 | COMPL             | ETION DATE: April 21, 2005 RECORDED BY: Chris Treese  |  |  |  |  |  |  |
| GROUN         |                | DEPTH W            | HILE DRIL                               | LING:             | ~12 feet GROUNDWATER DEPTH AFTER COMPLETION: NA   |  |  |  |  |  |  |
| WEATH         | ER CONDI       | TIONS:             |   | Sunny, 5          | DRILLING CONTRACTOR: Chambers Environmental Grou  |  |  |  |  |  |  |
| DRILL R       | IG: Ge         | oprobe® 5          | 40UD [                                  | ORILL SIZE        | & TYPE: 2" OD drive point DRILLER NAME(S): Keith Skow   |  |  |  |  |  |  |
|               |                |                    |   |                   | Material Classification   |  |  |  |  |  |  |
| Sample<br>No. | OVM<br>Reading | Sample<br>Interval | Sampler<br>Type*                        | Recovery (inches) | trace – 1-10% little – 11-20% some – 21-35% and – 36-50%  |  |  |  |  |  |  |
|               | (ppm)          | (feet)             | .,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, | (                 | f-fine m-medium c-coarse  |  |  |  |  |  |  |
| 1             | 0              | 0-4                | MC                                      | 14                | 4" asphalt $\rightarrow$ 3" gravel ballast $\rightarrow$ 7" silt, sand, gravel (FILL). Moist. No unusual odors. |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
| 2             | 0              | 4-8                | MC                                      | 15                | reddish and dk brown CLAYEY SILT and rounded fm GRAVEL little fmc SAND. Moist. No                               |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
| 3             | 0              | 8-12               | MC                                      | 16                | reddish and dk brown SILT and rounded fmc GRAVEL. Moist. No unusual odors.                                      |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
| 4             | 0              | 12-16              | MC                                      | 6                 | 6" similar soils. Saturated. No unusual odors.  |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
|               |                |                    |   |                   | Boring terminated at 16 feet below ground surface.  |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
|               |                |                    |   | · _ · · ·         |   |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
|               |                |                    |   |                   |   |  |  |  |  |  |  |
| -             |                |                    |   |                   |   |  |  |  |  |  |  |

| E |               |                         |                              |                  |                      |   |
|---|---------------|-------------------------|------------------------------|------------------|----------------------|---|
|   | Teet          | er Env                  | vironn                       | nental           | Servic               | ces, Inc. SUBSURFACE LOG  |
|   | SITE LO       | OCATION:                |                              | Cry              | stal Cleane          | ers and Sugar Creek, 343 W. Pulteney Street, Corning, New York, 14830                                 |
|   | CLIENT        |                         | Mr. Dona                     | ald Stiker,      | 12 Tuscaro           | ra Road, Addison, New York 14801 WELL/BORING ID: B6   |
|   | START         | DATE:                   | April 2                      | 1, 2005          |                      | ETION DATE:April 21, 2005 RECORDED BY:Chris Treese  |
|   | GROUN         | DWATER                  | DEPTH W                      | HILE DRIL        | LING:                | ~11 feet GROUNDWATER DEPTH AFTER COMPLETION: NA   |
|   | WEATH         | ER COND                 |                              |                  | Sunny, 5             | DRILLING CONTRACTOR: Chambers Environmental Group   |
| - | DRILL F       | RIG: Ge                 | eprobe® 5                    | 40UD [           |                      | E & TYPE: <u>2" OD drive point</u> DRILLER NAME(S): Keith Skow  |
| ļ | <u></u>       | · · · · ·               | [                            |                  |                      | Material Classification   |
|   | Sample<br>No. | OVM<br>Reading<br>(nom) | Sample<br>Interval<br>(feet) | Sampler<br>Type* | Recovery<br>(inches) | trace 1-10% little 11-20% some 21-35% and 36-50%  |
|   |               | (                       | (,                           |                  |                      | f-fine m-medium c-coarse  |
|   | 1             | 0                       | 0-4                          | MC               | 18                   | 5" asphalt $\rightarrow$ 13" dk brown and black silt, sand, gravel, brick fragments (FILL). Moist. No |
|   |               |                         |                              |                  |                      | unusual odors.  |
|   | 2             | 0                       | 4-8                          | MC               | 15                   | eddish brown dense CLAYEY SILT and angular to rounded fmc GRAVEL little fmc SAND.                     |
|   |               |                         |                              |                  |                      | Moist. No unusual odors.  |
|   |               |                         |                              |                  |                      |   |
| - | 3             | 0                       | 8-12                         | MC               | 16                   | brown m SAND and rounded fm GRAVEL little SILT. Wet at ~11 feet. No unusual odors.                    |
|   |               |                         |                              |                  |                      | 4   |
| - |               |                         |                              |                  |                      | Boring terminated at 12 feet below around surface   |
| ŀ |               |                         |                              | ·····            |                      |   |
|   |               |                         |                              |                  |                      |   |
|   |               |                         |                              |                  |                      |   |
|   |               |                         |                              |                  |                      |   |
| ┢ |               |                         |                              |                  |                      |   |
|   |               |                         |                              |                  |                      |   |
|   |               |                         |                              |                  |                      | 4   |
|   |               |                         |                              |                  |                      |   |
|   |               |                         |                              |                  |                      |   |
|   |               |                         |                              |                  |                      |   |
|   |               |                         |                              |                  |                      |   |
|   |               |                         |                              |                  |                      |   |
| ┥ |               |                         |                              |                  |                      |   |
| ļ |               |                         |                              |                  |                      |   |
|   | NOTES         |                         | <b>_</b>                     |                  | <u> </u>             | <u></u>   |
|   | *MC -         | Geoprobe                | MACROCO                      | DRE SS-          | - SPLIT SPO          | ON DPSS – DIRECT PUSH SPLIT SPOON SH – SHELBY TUBE C – BEDROCK CORE                                   |
|   |               |                         |                              |                  |                      |   |
|   |               |                         |                              |                  |                      |   |

Phase II ESA – Sugar Creek and Crystal Cleaners 343 W. Pulteney Street, Corning, NY May 25, 2005

### **APPENDIX C**

### LABORATORY REPORT

Techer Favoroamental Services



390 N. Pennsylvania Ave. South Waverly, PA 18840-2826 Phone (570) 888-0169 FAX (570) 888-0717

## **Certificate of Analysis**

| Teeter Environmental<br>RD#1, Box 124B, Macafee 1<br>Sayre PA, 18840 | Road             |                    |   | Project: S<br>Project No: [:<br>Project Manager: E | Sugar Creek<br>none]<br>Dave <u>Teeter</u> |              | <b>Repor</b><br>05/26/05 | <b>ted:</b><br>15:34 |
|--|------------------|--------------------|---|--|--|--------------|--------------------------|----------------------|
| B-1<br>5D22102-01 (Gro   | ound <u>Wate</u> | er)                | Date Sampled: 04/21/05 10:00<br>Date Received: 04/21/05 17:05 |  |  |              |                          |                      |
| Analyte  | Result           | Detection<br>Limit | Units   | Prepared   | Analyzed                                   | Method       | Analyst                  | Notes                |
| SW846/8260B Volatile Organic   | Compour          | lds                |   |  |  |              |                          |                      |
| Benzene  | <5.0             | 5.0                | ug/l  | 04/26/05 00:00                                     | 04/26/05 00:00                             | SW846/8260B  | CY                       |                      |
| n-Butylbenzene   | <5.0             | 5.0                | ug/l  | 04/26/05 00:00                                     | "  | **           | CY                       |                      |
| sec-Butylbenzene   | <5.0             | 5.0                | ug/l  | 04/26/05 00:00                                     |  | u            | CY                       |                      |
| Ethylbenzene   | <5.0             | 5.0                | ug/l  | 04/26/05 00:00                                     |  | н            | CY                       |                      |
| Isopropylbenzene   | <5.0             | 5.0                | ug/l  | 04/26/05 00:00                                     | "  | **           | CY                       |                      |
| p-Isopropyltoluene   | <5.0             | 5.0                | ug/l  | 04/26/05 00:00                                     | "  | u            | CY                       |                      |
| Naphthalene  | 29.7             | 5.0                | ug/l  | 04/26/05 00:00                                     | "  | n            | CY                       | LCCV                 |
| n-Propylbenzene  | <5.0             | 5.0                | ug/l  | 04/26/05 00:00                                     | u.   | **           | CY                       |                      |
| Toluene  | 6.2              | 5.0                | ug/l  | 04/26/05 00:00                                     |  |              | CY                       |                      |
| 1,2,4-Trimethylbenzene   | <5.0             | 5.0                | ug/l  | 04/26/05 00:00                                     |  | н            | CY                       |                      |
| 1,3,5-Trimethylbenzene   | <5.0             | 5.0                | ug/l  | 04/26/05 00:00                                     | "  | ••           | CY                       |                      |
| m,p-Xylene   | 5.2              | 5.0                | ug/l  | 04/26/05 00:00                                     | U  | "            | CY                       |                      |
| o-Xylene   | <5.0             | 5.0                | ug/l  | 04/26/05 00:00                                     | 0  |              | CY                       |                      |
| Methyl tert-butyl ether  | <5.0             | 5.0                | ug/l  | 04/26/05 00:00                                     | п  | "            | CY                       |                      |
| tert-Butylbenzene  | <5.0             | 5.0                | ug/l  | 04/26/05 00:00                                     | **   |              | CY                       |                      |
| Surrogate: 1,2-Dichloroethane-d4                                     |                  | 107 %              | 80-12   | 0  |  |              | CY                       |                      |
| Surrogate: Toluene-d8  |                  | 100 %              | 88-11   | 0  | 11   | **           | CY                       |                      |
| Surrogate: Bromofluorobenzene  |                  | 88.0 %             | 86-11   | 5  | u  | "            | CY                       |                      |
| SW846/8270C Semivolatile Org   | anic Com         | pounds             |   |  |  |              |                          |                      |
| Naphthalene  | <10.0            | 10.0               | ug/l  | 04/28/05 00:00                                     | 04/28/05 00:00                             | SW-846/8270C | RJH                      |                      |
| Acenaphthylene   | <10.0            | 10.0               | ug/l  | 04/28/05 00:00                                     | "  | "            | RJH                      |                      |
| Acenaphthene   | <10.0            | 10.0               | ug/l  | 04/28/05 00:00                                     | "  | "            | RJH                      |                      |
| Fluorene   | <10.0            | 10.0               | ug/l  | 04/28/05 00:00                                     | "  | **           | RJH                      |                      |
| Phenanthrene   | <10.0            | 10.0               | ug/l  | 04/28/05 00:00                                     | n  | "            | RJH                      |                      |
| Anthracene   | <10.0            | 10.0               | ug/l  | 04/28/05 00:00                                     | "  | 11           | RJH                      |                      |
| Fluoranthene   | <10.0            | 10.0               | ug/l  | 04/28/05 00:00                                     | 11   | "            | RJH                      |                      |
| Pyrene   | <10.0            | 10.0               | ug/l  | 04/28/05 00:00                                     | "  | "            | RJH                      |                      |
| Benzo (a) anthracene   | <10.0            | 10.0               | ug/l  | 04/28/05 00:00                                     | "  |              | RJH                      |                      |
| Chrysene   | <10.0            | 10.0               | ug/l  | 04/28/05 00:00                                     |  | "            | RJH                      |                      |
| Benzo (b) fluoranthene   | <10.0            | 10.0               | ug/l  | 04/28/05 00:00                                     | 11   | "            | RJH                      |                      |

Eastern Laboratory Services, Ltd.

Barbara Holman

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.



Reviewed by Barbara Hohman, QA Manager

Page 1 of 10



390 N. Pennsylvania Ave. South Waverly, PA 18840-2826 Phone (570) 888-0169 FAX (570) 888-0717

# **Certificate of Analysis**

| RD#1, Box 124B, Macafe<br>Sayre PA, 18840 | e Road          |                    | I      | Project No: [<br>Project Manager: ] |                                 | <b>Repor</b><br>05/26/05         | ted:<br>15:34 |       |
|---|-----------------|--------------------|--------|-------------------------------------|---------------------------------|----------------------------------|---------------|-------|
| B-<br>5D22102-01 (G                       | 1<br>round Wate | er)                |        |                                     | Date Sampled:<br>Date Received: | 04/21/05 10:00<br>04/21/05 17:05 |               |       |
| Analyte                                   | Result          | Detection<br>Limit | Units  | Prepared                            | Analyzed                        | Method                           | Analyst       | Notes |
| SW846/8270C Semivolatile O                | rganic Com      | pounds             |        |                                     |                                 |                                  |               |       |
| Benzo (k) fluoranthene                    | <10.0           | 10.0               | ug/l   | 04/28/05 00:00                      | н                               |                                  | RJH           |       |
| Benzo (a) pyrene                          | <10.0           | 10.0               | ug/l   | 04/28/05 00:00                      | "                               |                                  | RJH           |       |
| Indeno (1,2,3-cd) pyrene                  | <10.0           | 10.0               | ug/l   | 04/28/05 00:00                      | "                               |                                  | RJH           |       |
| Dibenz (a,h) anthracene                   | <10.0           | 10.0               | ug/l   | 04/28/05 00:00                      | "                               | "                                | RJH           |       |
| Benzo (g,h,i) perylene                    | <10.0           | 10.0               | ug/l   | 04/28/05 00:00                      | **                              | **                               | RJH           |       |
| Surrogate: Nitrobenzene-d5                |                 | 69.4 %             | 42-103 |                                     | 11                              |                                  | RJH           |       |
| Surrogate: 2-Fluorobiphenyl               |                 | 66.8 %             | 44-104 |                                     |                                 | **                               | RJH           |       |
| Surrogate: p-Terphenyl-d14                |                 | 83.0 %             | 55-113 |                                     | н                               | u                                | RJH           |       |

LCCV = Continuing Calibration Verification was below acceptance limits. Results may be biased low.

Eastern Laboratory Services, Ltd.

Barbara Hohman

Reviewed by Barbara Hohman, QA Manager

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

Page 2 of 10





quality 
accuracy 
reliability

390 N. Pennsylvania Ave. South Waverly, PA 18840-2826 Phone (570) 888-0169 FAX (570) 888-0717

# **Certificate of Analysis**

| Teeter Environmental<br>RD#1, Box 124B, Macafee<br>Sayre PA, 18840 | Road      |                    |       | Project: S<br>Project No: [:<br>Project Manager: [ | Sugar Creek<br>none]<br>Dave Teeter |                                  | <b>Repor</b><br>05/26/05 | <b>ted:</b><br>15:34 |
|--|-----------|--------------------|-------|--|-------------------------------------|----------------------------------|--------------------------|----------------------|
| B-3<br>5D22102-02 (Gro   | ound Wate | er)                |       |  | Date Sampled:<br>Date Received:     | 04/21/05 11:10<br>04/21/05 17:05 |                          |                      |
| Analyte  | Result    | Detection<br>Limit | Units | Prepared   | Analyzed                            | Method                           | Analyst                  | Notes                |
| SW846/8260B Volatile Organic                                       | Compour   | ıds                |       |  |                                     |                                  |                          |                      |
| Benzene  | <5.0      | 5.0                | ug/l  | 04/26/05 00:00                                     | 04/26/05 00:00                      | SW846/8260B                      | CY                       |                      |
| n-Butylbenzene   | <5.0      | 5.0                | ug/l  | 04/26/05 00:00                                     | "                                   |                                  | CY                       |                      |
| sec-Butylbenzene   | <5.0      | 5.0                | ug/l  | 04/26/05 00:00                                     | 11                                  | •                                | CY                       |                      |
| Ethylbenzene   | <5.0      | 5.0                | ug/l  | 04/26/05 00:00                                     | U .                                 | n                                | CY                       |                      |
| Isopropylbenzene   | <5.0      | 5.0                | ug/l  | 04/26/05 00:00                                     | 11                                  |                                  | CY                       |                      |
| p-Isopropyltoluene   | <5.0      | 5.0                | ug/l  | 04/26/05 00:00                                     | "                                   | **                               | CY                       |                      |
| Naphthalene  | 9.3       | 5.0                | ug/l  | 04/26/05 00:00                                     | "                                   | "                                | CY                       | LCCV                 |
| n-Propylbenzene  | <5.0      | 5.0                | ug/l  | 04/26/05 00:00                                     | "                                   | "                                | CY                       |                      |
| Toluene  | <5.0      | 5.0                | ug/l  | 04/26/05 00:00                                     | н                                   |                                  | CY                       |                      |
| 1,2,4-Trimethylbenzene   | <5.0      | 5.0                | ug/l  | 04/26/05 00:00                                     |                                     |                                  | CY                       |                      |
| 1,3,5-Trimethylbenzene   | <5.0      | 5.0                | ug/l  | 04/26/05 00:00                                     | н                                   | "                                | CY                       |                      |
| m,p-Xylene   | <5.0      | 5.0                | ug/l  | 04/26/05 00:00                                     | "                                   |                                  | CY                       |                      |
| o-Xylene   | <5.0      | 5.0                | ug/l  | 04/26/05 00:00                                     |                                     | "                                | CY                       |                      |
| Methyl tert-butyl ether  | <5.0      | 5.0                | ug/l  | 04/26/05 00:00                                     | "                                   | "                                | CY                       |                      |
| tert-Butylbenzene  | <5.0      | 5.0                | ug/l  | 04/26/05 00:00                                     | "                                   | "                                | CY                       |                      |
| Surrogate: 1,2-Dichloroethane-d4                                   |           | 109 %              | 80-12 | 0  | н                                   |                                  | CY                       |                      |
| Surrogate: Toluene-d8  |           | 100 %              | 88-11 | 0  | 11                                  | "                                | CY                       |                      |
| Surrogate: Bromofluorobenzene                                      |           | 87.6 %             | 86-11 | 5  | "                                   | "                                | CY                       |                      |
| SW846/8270C Semivolatile Org                                       | anic Com  | pounds             |       |  |                                     |                                  |                          |                      |
| Naphthalene  | <10.0     | 10.0               | ug/l  | 04/28/05 00:00                                     | 04/28/05 00:00                      | SW-846/8270C                     | RJH                      |                      |
| Acenaphthylene   | <10.0     | 10.0               | ug/1  | 04/28/05 00:00                                     | 11                                  | **                               | RJH                      |                      |
| Acenaphthene   | <10.0     | 10.0               | ug/l  | 04/28/05 00:00                                     | "                                   | "                                | RJH                      |                      |
| Fluorene   | <10.0     | 10.0               | ug/l  | 04/28/05 00:00                                     | 11                                  | ••                               | RJH                      |                      |
| Phenanthrene   | <10.0     | 10.0               | ug/l  | 04/28/05 00:00                                     | "                                   | "                                | RJH                      |                      |
| Anthracene   | <10.0     | 10.0               | ug/l  | 04/28/05 00:00                                     | "                                   | "                                | RJH                      |                      |
| Fluoranthene   | <10.0     | 10.0               | ug/l  | 04/28/05 00:00                                     | 11                                  |                                  | RJH                      |                      |
| Pyrene   | <10.0     | 10.0               | ug/l  | 04/28/05 00:00                                     | 19                                  | **                               | RJH                      |                      |
| Benzo (a) anthracene   | <10.0     | 10.0               | ug/l  | 04/28/05 00:00                                     | н                                   |                                  | RJH                      |                      |
| Chrysene   | <10.0     | 10.0               | ug/l  | 04/28/05 00:00                                     | **                                  | "                                | RJH                      |                      |
| Benzo (b) fluoranthene   | <10.0     | 10.0               | ug/l  | 04/28/05 00:00                                     | 11                                  |                                  | RJH                      |                      |

Eastern Laboratory Services, Ltd.

Barbara Holman

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

PA 08380 NY 11216





390 N. Pennsylvania Ave. South Waverly, PA 18840-2826 Phone (570) 888-0169 FAX (570) 888-0717

### **Certificate of Analysis**

| Teeter Environmental<br>RD#1, Box 124B, Macafe | e Road          |                    |        | Reported:                    |                                 |                                  |         |       |  |
|--|-----------------|--------------------|--------|------------------------------|---------------------------------|----------------------------------|---------|-------|--|
| Sayre PA, 18840                                |                 |                    | I      | Project Manager: Dave Teeter |                                 |                                  |         |       |  |
| B-<br>5D22102-02 (G                            | 3<br>round Wate | er)                |        |                              | Date Sampled:<br>Date Received: | 04/21/05 11:10<br>04/21/05 17:05 |         |       |  |
| Analyte  | Result          | Detection<br>Limit | Units  | Prepared                     | Analyzed                        | Method                           | Analyst | Notes |  |
| SW846/8270C Semivolatile O                     | rganic Com      | pounds             |        |                              |                                 |                                  |         |       |  |
| Benzo (k) fluoranthene                         | <10.0           | 10.0               | ug/l   | 04/28/05 00:00               |                                 | 11                               | RJH     |       |  |
| Benzo (a) pyrene                               | <10.0           | 10.0               | ug/l   | 04/28/05 00:00               | "                               | "                                | RJH     |       |  |
| Indeno (1,2,3-cd) pyrene                       | <10.0           | 10.0               | ug/l   | 04/28/05 00:00               | 17                              | "                                | RJH     |       |  |
| Dibenz (a,h) anthracene                        | <10.0           | 10.0               | ug/l   | 04/28/05 00:00               | 17                              | "                                | RJH     |       |  |
| Benzo (g,h,i) perylene                         | <10.0           | 10.0               | ug/l   | 04/28/05 00:00               |                                 |                                  | RJH     |       |  |
| Surrogate: Nitrobenzene-d5                     |                 | <b>6</b> 8.7 %     | 42-103 |                              | 11                              | n                                | RJH     |       |  |
| Surrogate: 2-Fluorobiphenyl                    |                 | 68.4 %             | 44-104 |                              | "                               | ••                               | RJH     |       |  |
| Surrogate: p-Terphenyl-d14                     |                 | 84.6 %             | 55-113 |                              |                                 | **                               | RJH     |       |  |

LCCV = Continuing Calibration Verification was below acceptance limits. Results may be biased low.

Eastern Laboratory Services, Ltd.

Barbara Hohman

Reviewed by Barbara Hohman, QA Manager

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.



Page 4 of 10



390 N. Pennsylvania Ave. South Waverly, PA 18840-2826 Phone (570) 888-0169 FAX (570) 888-0717

## **Certificate of Analysis**

| Teeter Environmental<br>RD#1, Box 124B, Macafee<br>Sayre PA, 18840 | e Road          | _                  |       |                | <b>Report</b><br>05/26/05       | t <b>ed:</b><br>15:34            |         |       |
|--|-----------------|--------------------|-------|----------------|---------------------------------|----------------------------------|---------|-------|
| B-4<br>5D22102-03 (G4  | 4<br>round Wate | er)                |       |                | Date Sampled:<br>Date Received: | 04/21/05 11:50<br>04/21/05 17:05 |         |       |
| Analyte  | Result          | Detection<br>Limit | Units | Prepared       | Analyzed                        | Method                           | Analyst | Notes |
| SW846/8260B Volatile Organi  | c Compoun       | ds                 |       |                |                                 |                                  |         |       |
| Benzene  | < 5.00          | 5.00               | ug/l  | 04/26/05 00:00 | 04/26/05 00:00                  | SW-846/8260B                     | CY      |       |
| Bromobenzene   | < 5.00          | 5.00               | ug/l  | 04/26/05 00:00 | н                               | "                                | CY      |       |
| Bromochloromethane   | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 | **                              |                                  | CY      |       |
| Bromodichloromethane   | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 | 11                              | **                               | CY      |       |
| Bromoform  | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 | 11                              | "                                | CY      |       |
| Bromomethane   | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 |                                 | в                                | CY      |       |
| n-Butylbenzene   | < 5.00          | 5.00               | ug/l  | 04/26/05 00:00 |                                 | D                                | CY      |       |
| sec-Butylbenzene   | < 5.00          | 5.00               | ug/l  | 04/26/05 00:00 | "                               | "                                | CY      |       |
| tert-Butylbenzene  | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 | **                              | "                                | CY      |       |
| Carbon tetrachloride   | < 5.00          | 5.00               | ug/l  | 04/26/05 00:00 | 11                              | "                                | CY      |       |
| Chlorobenzene  | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 | *1                              | "                                | CY      |       |
| Chloroethane   | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 | **                              | "                                | CY      |       |
| Chloroform   | < 5.00          | 5.00               | ug/l  | 04/26/05 00:00 | U                               | "                                | CY      |       |
| Chloromethane  | < 5.00          | 5.00               | ug/l  | 04/26/05 00:00 | 11                              | "                                | CY      |       |
| 2-Chlorotoluene  | < 5.00          | 5.00               | ug/l  | 04/26/05 00:00 | 11                              | н                                | CY      |       |
| 4-Chlorotoluene  | < 5.00          | 5.00               | ug/l  | 04/26/05 00:00 | и                               |                                  | CY      |       |
| Dibromochloromethane   | < 5.00          | 5.00               | ug/1  | 04/26/05 00:00 | 50                              | "                                | CY      |       |
| Dibromomethane   | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 | "                               | "                                | CY      |       |
| 1,2-Dibromoethane (EDB)  | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 | n                               | "                                | CY      |       |
| 1,2-Dibromo-3-chloropropane  | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 | **                              | **                               | CY      |       |
| 1,2-Dichlorobenzene  | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 | н                               | "                                | CY      |       |
| 1,3-Dichlorobenzene  | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 | "                               | **                               | CY      |       |
| 1,4-Dichlorobenzene  | <5.00           | 5.00               | ug/1  | 04/26/05 00:00 |                                 | 11                               | CY      |       |
| Dichlorodifluoromethane  | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 | 11                              | "                                | CY      |       |
| 1,1-Dichloroethane   | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 |                                 | "                                | CY      |       |
| 1,2-Dichloroethane   | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 |                                 | U                                | CY      |       |
| 1,1-Dichloroethene   | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 | 0                               | 11                               | CY      |       |
| cis-1,2-Dichloroethene   | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 |                                 |                                  | CY      |       |
| trans-1,2-Dichloroethene   | <5.00           | 5.00               | ug/l  | 04/26/05 00:00 | "                               | "                                | CY      |       |

Eastern Laboratory Services, Ltd.

Barbara Hohman

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

PA 08380 NY 11216





390 N. Pennsylvania Ave. South Waverly, PA 18840-2826 Phone (570) 888-0169 FAX (570) 888-0717

## **Certificate of Analysis**

| Teeter Environmental<br>RD#1, Box 124B, Macafee<br>Sayre PA, 18840 | Road      |                    | Project: Sugar Creek<br>Project No: [none]<br>Project Manager: Dave Teeter |                |                                 |                                  |         | t <b>ed:</b><br>15:34 |
|--|-----------|--------------------|--|----------------|---------------------------------|----------------------------------|---------|-----------------------|
| B-4<br>5D22102-03 (Gr  | ound Wate | er)                |  |                | Date Sampled:<br>Date Received: | 04/21/05 11:50<br>04/21/05 17:05 |         |                       |
| Analyte  | Result    | Detection<br>Limit | Units  | Prepared       | Analyzed                        | Method                           | Analyst | Notes                 |
| SW846/8260B Volatile Organic                                       | c Compoun | ıds                |  |                |                                 |                                  |         |                       |
| 1,2-Dichloropropane  | <5.00     | 5.00               | ug/l   | 04/26/05 00:00 | 11                              | SW-846/8260B                     | CY      |                       |
| 1,3-Dichloropropane  | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 | **                              |                                  | CY      |                       |
| 2,2-Dichloropropane  | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 | "                               | "                                | CY      |                       |
| 1,1-Dichloropropene  | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 | 17                              | "                                | CY      |                       |
| cis-1,3-Dichloropropene  | <5.00     | 5.00               | ug/l   | 04/26/05 00:00 | **                              | "                                | CY      |                       |
| trans-1,3-Dichloropropene  | <5.00     | 5.00               | ug/l   | 04/26/05 00:00 | "                               | u                                | CY      |                       |
| Ethylbenzene   | <5.00     | 5.00               | ug/l   | 04/26/05 00:00 | "                               | "                                | CY      |                       |
| Hexachlorobutadiene  | <5.00     | 5.00               | ug/l   | 04/26/05 00:00 | *1                              | **                               | CY      |                       |
| Isopropylbenzene   | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 | "                               | 11                               | CY      |                       |
| p-lsopropyltoluene   | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 | **                              | **                               | CY      |                       |
| Methylene chloride   | <5.00     | 5.00               | ug/l   | 04/26/05 00:00 |                                 | 11                               | CY      |                       |
| n-Propylbenzene  | <5.00     | 5.00               | ug/l   | 04/26/05 00:00 | u.                              |                                  | CY      |                       |
| Styrene  | <5.00     | 5.00               | ug/l   | 04/26/05 00:00 |                                 | "                                | CY      |                       |
| 1,1,1,2-Tetrachloroethane  | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 | 11                              | "                                | CY      |                       |
| 1,1,2,2-Tetrachloroethane  | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 | "                               | **                               | CY      |                       |
| Tetrachloroethene  | 43.1      | 5.00               | ug/l   | 04/26/05 00:00 | "                               | "                                | CY      |                       |
| Toluene  | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 |                                 | 11                               | CY      |                       |
| 1,2,3-Trichlorobenzene   | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 | "                               | U                                | CY      |                       |
| 1,2,4-Trichlorobenzene   | <5.00     | 5.00               | ug/l   | 04/26/05 00:00 | u                               |                                  | CY      |                       |
| 1,1,1-Trichloroethane  | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 |                                 | н                                | CY      |                       |
| 1,1,2-Trichloroethane  | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 |                                 |                                  | CY      |                       |
| Trichloroethene  | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 |                                 | "                                | CY      |                       |
| Trichlorofluoromethane   | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 | n                               | **                               | CY      |                       |
| 1,2,3-Trichloropropane   | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 | **                              | n                                | CY      |                       |
| 1,2,4-Trimethylbenzene   | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 |                                 | n                                | CY      |                       |
| 1,3,5-Trimethylbenzene   | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 | 91                              | **                               | CY      |                       |
| Vinyl chloride   | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 | "                               | **                               | CY      |                       |
| o-Xylene   | <5.00     | 5.00               | ug/l   | 04/26/05 00:00 | "                               | "                                | CY      |                       |
| m n-Xvlene   | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00 | "                               |                                  | CY      |                       |

Eastern Laboratory Services, Ltd.

Barbara Hohman

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.



Reviewed by Barbara Hohman, QA Manager



quality 
accuracy 
reliability

390 N. Pennsylvania Ave. South Waverly, PA 18840-2826 Phone (570) 888-0169 FAX (570) 888-0717

## **Certificate of Analysis**

| Teeter Environmental<br>RD#1, Box 124B, Macafee<br>Sayre PA, 18840 | Road     | <u> </u>           | ]      | Project: S<br>Project No: [<br>Project Manager: ] | Sugar Creek<br>[none]<br>Dave Teeter |                                  | <b>Reported:</b><br>05/26/05 15:34 |       |  |
|--|----------|--------------------|--------|---|--------------------------------------|----------------------------------|------------------------------------|-------|--|
| B-4<br>5D22102-03 (Gro   | ound Wat | er)                |        |   | Date Sampled:<br>Date Received:      | 04/21/05 11:50<br>04/21/05 17:05 |                                    |       |  |
| Analyte  | Result   | Detection<br>Limit | Units  | Prepared  | Analyzed                             | Method                           | Analyst                            | Notes |  |
| SW846/8260B Volatile Organic                                       | Compour  | nds                |        |   |                                      |                                  |                                    |       |  |
| Naphthalene  | 5.80     | 5.00               | ug/l   | 04/26/05 00:00                                    | 11                                   | SW-846/8260B                     | СҮ                                 | LCCV  |  |
| Methyl tert-butyl ether  | <5.00    | 5.00               | ug/l   | 04/26/05 00:00                                    | 11                                   |                                  | CY                                 |       |  |
| Surrogate: 1,2-Dichloroethane-d4                                   |          | 110 %              | 80-120 |   | "                                    | "                                | CY                                 |       |  |
| Surrogate: Toluene-d8  |          | 100 %              | 88-110 |   | "                                    | ••                               | CY                                 |       |  |
| Surrogate: Bromofluorobenzene                                      |          | 87.6 %             | 86-115 |   | n.                                   | n                                | CY                                 |       |  |

Eastern Laboratory Services, Ltd.

Barbara Holman

Reviewed by Barbara Hohman, QA Manager

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.







390 N. Pennsylvania Ave. South Waverly, PA 18840-2826 Phone (570) 888-0169 FAX (570) 888-0717

## **Certificate of Analysis**

| Teeter Environmental<br>RD#1, Box 124B, Macafee F<br>Sayre PA, 18840 | Road        |                    |       | Project: S<br>Project No: [<br>Project Manager: [ | ugar Creek<br>none]<br>Dave Teeter |                                  | <b>Repor</b> | <b>ted:</b><br>15:34 |
|--|-------------|--------------------|-------|---|------------------------------------|----------------------------------|--------------|----------------------|
| B-5<br>5D22102-04 (Gro   | und Wate    | er)                |       |   | Date Sampled:<br>Date Received:    | 04/21/05 13:05<br>04/21/05 17:05 |              |                      |
| Analyte  | Result      | Detection<br>Limit | Units | Prepared  | Analyzed                           | Method                           | Analyst      | Notes                |
| SW846/8260B Volatile Organic   | <br>Compoun | ds                 |       |   |                                    |                                  |              | _                    |
| Benzene  | <5.00       | 5.00               | ug/l  | 04/26/05 00:00                                    | 04/26/05 00:00                     | SW-846/8260B                     | CY           |                      |
| Bromobenzene   | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    | "                                  | "                                | CY           |                      |
| Bromochloromethane   | <5.00       | 5.00               | ug/l  | 04/26/05 00:00                                    |                                    | **                               | CY           |                      |
| Bromodichloromethane   | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    |                                    | "                                | CY           |                      |
| Bromoform  | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    |                                    | и                                | CY           |                      |
| Bromomethane   | <5.00       | 5.00               | ug/l  | 04/26/05 00:00                                    |                                    |                                  | CY           |                      |
| n-Butylbenzene   | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    | "                                  | "                                | CY           |                      |
| sec-Butylbenzene   | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    | "                                  | ш                                | CY           |                      |
| tert-Butylbenzene  | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    |                                    | "                                | CY           |                      |
| Carbon tetrachloride   | <5.00       | 5.00               | ug/l  | 04/26/05 00:00                                    |                                    |                                  | CY           |                      |
| Chlorobenzene  | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    | "                                  | **                               | CY           |                      |
| Chloroethane   | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    | "                                  | "                                | CY           |                      |
| Chloroform   | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    | "                                  |                                  | CY           |                      |
| Chloromethane  | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    | "                                  | "                                | CY           |                      |
| 2-Chlorotoluene  | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    | "                                  |                                  | CY           |                      |
| 4-Chlorotoluene  | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    | "                                  | "                                | CY           |                      |
| Dibromochloromethane   | <5.00       | 5.00               | ug/l  | 04/26/05 00:00                                    | 11                                 |                                  | CY           |                      |
| Dibromomethane   | <5.00       | 5.00               | ug/l  | 04/26/05 00:00                                    | "                                  | "                                | CY           |                      |
| 1,2-Dibromoethane (EDB)  | <5.00       | 5.00               | ug/l  | 04/26/05 00:00                                    | "                                  | u.                               | CY           |                      |
| 1,2-Dibromo-3-chloropropane  | <5.00       | 5.00               | ug/l  | 04/26/05 00:00                                    | н                                  | и                                | CY           |                      |
| 1,2-Dichlorobenzene  | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    | "                                  |                                  | CY           |                      |
| 1,3-Dichlorobenzene  | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    | "                                  | "                                | CY           |                      |
| 1,4-Dichlorobenzene  | <5.00       | 5.00               | ug/l  | 04/26/05 00:00                                    |                                    | u                                | CY           |                      |
| Dichlorodifluoromethane  | <5.00       | 5.00               | ug/l  | 04/26/05 00:00                                    | 11                                 | "                                | CY           |                      |
| 1,1-Dichloroethane   | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    | н                                  | n                                | CY           |                      |
| 1,2-Dichloroethane   | < 5.00      | 5.00               | ug/l  | 04/26/05 00:00                                    | н                                  |                                  | CY           |                      |
| 1,1-Dichloroethene   | <5.00       | 5.00               | ug/l  | 04/26/05 00:00                                    | 11                                 |                                  | CY           |                      |
| cis-1,2-Dichloroethene   | <5.00       | 5.00               | ug/l  | 04/26/05 00:00                                    | 19                                 |                                  | CY           |                      |
|  |             | 5.00               |       |   |                                    |                                  |              |                      |

Eastern Laboratory Services, Ltd.

Barbara Hohman

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

Reviewed by Barbara Hohman, QA Manager

Page 8 of 10



ENVIRONMENTAL

390 N. Pennsylvania Ave. South Waverly, PA 18840-2826 Phone (570) 888-0169 FAX (570) 888-0717

# **Certificate of Analysis**

| Teeter Environmental<br>RD#1, Box 124B, Macaf<br>Sayre PA, 18840 | fee Road          |                    |       | <b>Repor</b>   | ted:<br>15:34                   |                                  |         |       |
|--|-------------------|--------------------|-------|----------------|---------------------------------|----------------------------------|---------|-------|
| B<br>5D22102-04 (4   | -5<br>Ground Wate | er)                |       |                | Date Sampled:<br>Date Received: | 04/21/05 13:05<br>04/21/05 17:05 |         |       |
| Analyte  | Result            | Detection<br>Limit | Units | Prepared       | Analyzed                        | Method                           | Analyst | Notes |
| SW846/8260B Volatile Orga  | nic Compoun       | ds                 |       |                |                                 |                                  |         |       |
| 1,2-Dichloropropane  | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | н                               | SW-846/8260B                     | CY      |       |
| 1,3-Dichloropropane  | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | "                               | и                                | CY      |       |
| 2,2-Dichloropropane  | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | 11                              | U U                              | CY      |       |
| 1,1-Dichloropropene  | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 |                                 | "                                | CY      |       |
| cis-1,3-Dichloropropene  | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | "                               | U U                              | CY      |       |
| trans-1,3-Dichloropropene  | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 |                                 | "                                | CY      |       |
| Ethylbenzene   | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | "                               | 11                               | CY      |       |
| Hexachlorobutadiene  | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | **                              | **                               | CY      |       |
| lsopropylbenzene   | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 |                                 | н                                | CY      |       |
| p-Isopropyltoluene   | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | п                               | 11                               | CY      |       |
| Methylene chloride   | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 |                                 | 11                               | CY      |       |
| n-Propylbenzene  | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | "                               |                                  | CY      |       |
| Styrene  | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | "                               | "                                | CY      |       |
| 1,1,1,2-Tetrachloroethane  | < 5.00            | 5.00               | ug/l  | 04/26/05 00:00 | **                              | "                                | CY      |       |
| 1,1,2,2-Tetrachloroethane  | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | "                               |                                  | CY      |       |
| Tetrachloroethene  | 7.00              | 5.00               | ug/l  | 04/26/05 00:00 | n                               | "                                | CY      |       |
| Toluene  | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | "                               |                                  | CY      |       |
| 1,2,3-Trichlorobenzene   | < 5.00            | 5.00               | ug/l  | 04/26/05 00:00 |                                 | "                                | CY      |       |
| 1,2,4-Trichlorobenzene   | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | "                               |                                  | CY      |       |
| 1,1,1-Trichloroethane  | < 5.00            | 5.00               | ug/ì  | 04/26/05 00:00 | H                               | **                               | CY      |       |
| 1,1,2-Trichloroethane  | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | **                              | **                               | CY      |       |
| Trichloroethene  | < 5.00            | 5.00               | ug/l  | 04/26/05 00:00 | "                               | ••                               | CY      |       |
| Trichlorofluoromethane   | < 5.00            | 5.00               | ug/l  | 04/26/05 00:00 |                                 |                                  | CY      |       |
| 1,2,3-Trichloropropane   | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | "                               | u                                | CY      |       |
| 1,2,4-Trimethylbenzene   | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | **                              | **                               | CY      |       |
| 1,3,5-Trimethylbenzene   | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | 18                              | u                                | CY      |       |
| Vinyl chloride   | < 5.00            | 5.00               | ug/l  | 04/26/05 00:00 | "                               | ••                               | CY      |       |
| o-Xylene   | <5.00             | 5.00               | ug/l  | 04/26/05 00:00 | **                              | u                                | CY      |       |
| m,p-Xylene   | < 5.00            | 5.00               | ug/l  | 04/26/05 00:00 | "                               | "                                | CY      |       |

Eastern Laboratory Services, Ltd.

Barbara Hohman

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.





390 N. Pennsylvania Ave. South Waverly, PA 18840-2826 Phone (570) 888-0169 FAX (570) 888-0717

## **Certificate of Analysis**

| Teeter Environmental<br>RD#1, Box 124B, Macafee I<br>Sayre PA, 18840 | Road      |                    | ]      | Project: S<br>Project No: [1<br>Project Manager: [2<br>- | Sugar Creek<br>none]<br>Dave Teeter | 04/21/05 12:05 | <b>Repor</b><br>05/26/05 | <b>ted:</b><br>15:34 |
|--|-----------|--------------------|--------|--|-------------------------------------|----------------|--------------------------|----------------------|
| B-5<br>5D22102-04 (Gro   | ound Wate | er)                |        |  | Date Sampled:<br>Date Received:     | 04/21/05 13:05 |                          |                      |
| Analyte  | Result    | Detection<br>Limit | Units  | Prepared   | Analyzed                            | Method         | Analyst                  | Notes                |
| SW846/8260B Volatile Organic   | Compour   | nds                |        |  |                                     |                |                          |                      |
| Naphthalene  | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00   | "                                   | SW-846/8260B   | CY                       | LCCV                 |
| Methyl tert-butyl ether  | < 5.00    | 5.00               | ug/l   | 04/26/05 00:00   | "                                   | **             | CY                       |                      |
| Surrogate: 1,2-Dichloroethane-d4                                     | _         | 109 %              | 80-120 |  | 11                                  | "              | CY                       |                      |
| Surrogate: Toluene-d8  |           | 99.2 %             | 88-110 |  |                                     | "              | CY                       |                      |
| Surrogate: Bromofluorobenzene  |           | 86.6 %             | 86-115 |  | "                                   | "              | CY                       |                      |

LCCV = Continuing Calibration Verification was below acceptance limits. Results may be biased low.

Eastern Laboratory Services, Ltd.

Barbara Hohman

Reviewed by Barbara Hohman, QA Manager

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.



Page 10 of 10

|  | Ë.  | Lastern<br>390 N. Pennsylva   | Laboratory Services, Ltu.  |  |
|--|---|---|--|--|
| REPORT TO: TES The                       | Eastern Laboratory Sei                                    | Ph  | one: (570) 888-0169  | NEEDED: YES NO   |
|  | REFRIGERATE SAMPLES DW T<br>AFTER COLLECTION.             | DRINKING WATER SL SLUDGE<br>GROUND WATER SO SOIL<br>NIBECE WATER HZ HAZABDOUS | RESULTS ARE BEING USED FOR:<br>NYDOH NYDEC PADEP   | IF YES PLEASE ATTACH<br>IS A QC PACKAGE NEEDED?  |
| CONTACT                                  | / ***   | WASTE WATER OTHER   | PERSONAL OTHER   | IF YES PLEASE ATTACH REQUIREMENTS  |
| PH. #<br>FAX #                           | TRANSPORT TO<br>LABORATORY IN<br>COOLER WITH ICE          | P PLASTIC S SULFUR<br>G CLEAR GLASS SO SODUIN<br>AG AMBER GLASS SO SODUIN     | CHLORIC ACID OH SODIUM HYDROXIDE<br>NC ACID AS ASCORBIC ACID<br>ACID AC ACETIC ACID<br>NSULFITE NH AMMONIUM CHLORIDE | Elpt   |
| BILL TO: TES Inc                         | COMPOSI   |   |  | V RECEIPT  |
| PO#<br>PROJECT DESCRIPTION<br>Sysr Cruck | APLED<br>AMPLING<br>ATRIX<br>YPE - GRAB,<br>7 (INITALS)   | A TYPE<br>EA SIZE<br>MITUE  |  | O<br>O<br>D<br>U<br>D<br>V<br>O<br>V<br>V<br>V<br>V<br>V<br>V<br>V<br>V<br>V<br>V<br>V<br>V<br>V<br>V<br>V<br>V<br>V |
| CONTAINER SAMPLING POINT                 | DATE SAN<br>TIME OF S<br>SAMPLE A<br>SAMPLE A<br>SAMPLE T | ANALYSIS<br>O<br>(PER   | AB<br>TO BE PERFORMED<br>CONTAINER)  | S & applicable areas<br>completely<br>ELS USE ONLY   |
| $\frac{1}{2}$ $B-1$ $H$                  | 2 1000 Gui G DT G   | 4° 1 / 1/ 82  | 60 STARS 8270 STAR   | 5102-01/   |
| 3 <u>3</u> <u>3</u> <u>3</u>             | 115   | 8   | 260 Full List  | - 03/  |
| 4 <u>B-5</u>                             | y res y J J J   |   | + + +  | -04  |
| 6  |   |   |  |  |
| 7  |   |   |  |  |
| 9  |   |   |  |  |
| 10                                       |   |   | Due 5/6/05   |  |
| ELS USE ONLY DELIVERED BY                | 1. SAMPLES WERE: SHIPPE                                   | HAND DELIVERED 4. PRC   | PERLY PRESERVED: Y N   | 7. TRIP BLANKS: Y N  |
|  | 2. CONTAINERS INTACT:                                     | (Y) N 5. WIT  |  |  |
|  | 3. COC COMPLETE:  | (Y) N 6. LAB  | ELS MATCH COC:   |  |
| RELINQUISHED BY:                         | DATE:   | TIME: RECEIVED B  | Y:   | DATE: TIME:  |
|  |   | <b>↓</b>  |  |  |

#### **APPENDIX B**

#### SITE PHOTOGRAPHS

#### **CRYSTAL CLEANERS SITE PHOTOGRAPHS**



South side of the Site building. (9/9/05)



Looking west at north side of Site building. (9/9/05)

 $\label{eq:P:Projects/nysdec1/projects/Region 8 Dry \ Cleaning \ Sites/4.0 \ Project \ Deliverables/4.1 \ Reports/Crystal/For \ Final \ Report/Appendix_B_photos.doc$ 

#### **CRYSTAL CLEANERS SITE PHOTOGRAPHS**



Basement sump surface water sample location (SW-2). (9/9/05)



Basement sump surface water sample location (southeast corner of basement)(SW-1). (9/9/05)

 $\label{eq:P:Projects} P:\ensuremath{\sc ltg} P:\ensuremath{\sc ltg$ 

#### **CRYSTAL CLEANERS SITE PHOTOGRAPHS**



Sub-slab sample in south east corner of Site building (vacant space). (1/26/06)

#### **APPENDIX C**

#### FIELD DATA RECORDS

| Logge<br>Drillin      | g Contract                                     | nd<br>or A  | in SI          | rer            | √ Gr                 |             | Elevation  | Start Date   | 24/06   | Rig Typ                    |                 | n Date         | - 01_<br>24/0                 | 524 or |     |
|-----------------------|--|-------------|----------------|----------------|----------------------|-------------|--|--|---|----------------------------|-----------------|----------------|-------------------------------|--------|-----|
| Drillin<br>Soil D     | g Method                                       | Dy          | Cet<br>Bock D  | Pu             | Sh-                  | P           | rotection Level  | P.I.D. (   | eV)<br>1.0  | Casing                     | Şize            |                | Auge                          | r Size | r _ |
| Depth(Feet)           | Sample No. &<br>Penetration/<br>ecovery (Feet) | Sample Type | SPT Blows/6" D | re Rec./Rqd. % | SPT-N<br>(Blows/Ft.) | Graphic Log | otal Depth<br>26.5<br>S<br>De  | Sample   | ndwater/  | Date<br>SOSN<br>SOSN       | tes on Drilling | ter<br>Scan d) | Vell<br>mitoring<br>Sbace (mo | Borin  | g   |
| 2 3 4 5 6 7 8         | 1.8<br>14:0<br>14:0                            |             |                | CO             |                      |             | 0-1.0 DKBI<br>Sund, MP/SP, c<br>1-2' Ltovange<br>grovel, organ<br>Comp, NP/SP,<br>2'-3' DK olim<br>clay, trace Fir<br>MP, Fryabic,<br>3-4 orage /L<br>gracel, met, H<br>Sligvit organiz<br>4-8' (Compess<br>MS (MSD) - 1<br>Lt orage / 13m<br>of Some Frees<br>Scample Free<br>CP CF + | awn organie Si<br>tamp, PG, Vools,<br>/ Brown Silty S<br>it swell, WG, m<br>c / Brown Silty<br>n gravel, PG,<br>m Stiff, organ<br>.t brown claye<br>osse, WG, SP/v<br>Stified becar<br>luck of soil in<br>comp. The<br>S & - | Aly five<br>Dense<br>Dense<br>Dense<br>Dense<br>,<br>is sent<br>rest<br>ysady,<br>nP,<br>se of<br>Slacue<br>P,<br>of<br>- | SMI-OL<br>SMI-OL<br>SMI-GA | Composition No  |                | PI Me<br>Head                 |        |     |
| 10<br>11<br>12<br>Bob | C 20   | ç.5         | , d.           | Veo            | t pris               | h           |  | <u>006017</u> ,  | -<br>-<br>-   |                            |                 |                |                               |        |     |

9910003(e) L 33







| Clien  | t NYSD<br>ed By<br>Bran                         | NEC<br>den      | Shaw                                  | Sit                  | ie<br>Joun  | Cystal Cleaners Start Date<br>01/25/56   | Sheet N                             | No                | Date   | _ of  |                                       | 9<br>     |
|--|---|-----------------|---------------------------------------|----------------------|-------------|--|-------------------------------------|-------------------|--|---|---------------------------------------|-----------|
| Drillir  | ng Contract                                     | <sup>or</sup> A | TDT                                   |                      |             | Driller's Name<br>Mark Bachner   | Rig Ty                              | pe<br>Ge          | Prob   | c 54  | <br>f00                               |           |
| Drillir  | ng Method                                       | Jve             | etPvs                                 | h                    |             | Protection Level   | Casing                              | Casing Size       |  |   |                                       | pt _      |
| Soil [   | Drilled 8                                       |                 | Rock Drille                           | d                    |             | Total Depth Depth to Groundwate  | /Date                               | <u></u>           | Piez   | Well  | Borin                                 | 9         |
| Depth(Feet)  | Sample No. &<br>Penetration/<br>Recovery (Feet) | Sample Type     | SPT Blows/6"<br>or<br>ore Rec./Rqd. % | SPT-N<br>(Blows/Ft.) | Graphic Log | Sample<br>Description  | USCS<br>Group Symbol                | Votes on Drilling | Aeter<br>d Scan (1) OM   | Aeter<br>(monitorium)<br>(monitorium)<br>(monitorium)<br>(monitorium) | <u> </u>                              | Lab Tests |
| 1-<br>2-<br>3-<br>4-<br>5-<br>6-<br>7-<br>6-<br>7-<br>6-<br>7-<br>6-<br>7-<br>6-<br>7-<br>6-<br>7-<br>6-<br>7-<br>6-<br>7-<br>6-<br>7-<br>6-<br>7-<br>6-<br>7-<br>6-<br>7-<br>6-<br>7-<br>9-<br>9-<br>9-<br>9-<br>9-<br>9-<br>9-<br>9-<br>9-<br>9-<br>9-<br>9-<br>9- | 2.0/40  |                 |                                       |                      |             | D-1 DK Brun orgonie Silly File<br>Sand, damp, roots, MP, PG, Some<br>Fine granel ( C 0.8', Dense<br>1-2.5 Oline's Alty Fire Sand and<br>fine granel, worst, PG, rounded,<br>roots, MP, Dense;<br>2.5 - 3.5; orange/Bran, Sandy clyed<br>granel, morst, WG, SP/MP,<br>3.5 - 4. Lt/bray Bran, cloyed grand<br>WG, wet, SP/MP, base<br>4-24.5 Ofne gravel and cly, WG,<br>MP, moist, M. Dense<br>4-5. Orange rocks/rock film<br>losse<br>5-6 orange / Bran Sund grad<br>and clay, net, MP, WG; M Dense<br>6-8 Brann clear Sand and<br>granel, M Dense, wet, PG, Track<br>collecting sample firm 6-8;<br>CR6800200701XX @ 0830 | Fill-<br>GM<br>- GC<br>- GC<br>- GP |                   | с. I<br>С10<br>(1.0<br>С1.0<br>С1.0<br>С<br>С<br>С<br>С<br>С<br>С<br>С<br>С<br>С<br>С<br>С<br>С<br>С | Ē   |                                       |           |
| 1. 1   |   | ile,            |                                       |                      |             |  |                                     |                   |  |   | · · · · · · · · · · · · · · · · · · · |           |

r i F

|   | Brud   | lan She                                | Grc                  | e<br>ounc   | CNStal Clenners<br>Elevation Start Date<br>01/25/06   | Sheet N              | lo<br>Finisi      |                              |                              | <u> </u><br> |           |
|---|--|--|----------------------|-------------|---|----------------------|-------------------|------------------------------|------------------------------|--------------|-----------|
| Drilling Co   | ntractor<br>7                                  | ADT                                    |                      |             | Driller's Name<br>Marty Bachner   | Rig Ty               | pe Ge             | >> Pr                        | sbe 5.                       | 400          |           |
| Drilling Me   | ethod Div                                      | xet Pus                                | ih                   | F           | Protection Level D P.I.D. (eV)  | Casing               | Size              | 14                           | Auger                        | Size         |           |
| Soil Drilled  | , 8,   | Rock Drille                            | d                    | T           | otal Depth Depth to Groundwater   | /Date                |                   | Piez                         | Well                         | Borin        | g         |
| Depth(Feet)<br>Sample No. &   | Penetration/<br>Recovery (Feet)<br>Sample Type | SPT Blows/6"<br>or<br>Core Rec./Rqd. % | SPT-N<br>(Blows/Ft.) | Graphic Log | Sample<br>Description   | USCS<br>Group Symbol | Notes on Drilling | Pl Meter<br>Field Scan to ov | PI Meter (m<br>Head Space (m |              | Lab Tests |
| 1-19<br>2-19<br>3-11<br>3-12<br>5-16<br>7-72<br>7-72<br>7-72<br>7-72<br>7-72<br>7-72<br>7-72<br>7-7 | .6   |  |                      |             | <ul> <li>0-0.8 Biom / DK gim. Silly fine Sind<br/>and gravel, day to,</li> <li>0.8 - 2.0 Olive silly fine sund and<br/>gravel, wG, SP, days, Slight odor?</li> <li>2-2.5 Sime as above bot black</li> <li>Staining present, Shipht odor, precess<br/>of Direk (glass,</li> <li>2.5-3.5 Drive clay of little gravel,<br/>HP, PG, trace precess fire morter<br/>(wrick-ved), HP, wet,</li> <li>3.5-4. orange Lt Bream Scandy Clay<br/>will some gravel, V. angular to<br/>vounded, wG, wet, HP, Soft,</li> <li>4-5 Grafe Bream, Sand-gravel<br/>Clay, n/glass, wG, mP, leases of<br/>black fire sand, in Dense, morsts,</li> <li>5-5-5 Gray gravel w/ trace sund<br/>PG, DW, NP,</li> <li>58 orange / Bream, Sand gravel<br/>ad clay mikture, wG, Sat/oct,</li> <li>10052,</li> </ul> | Fill<br>GC           |                   | L1.6<br>L1.0                 |                              |              |           |

¢
|     | Client NY<br>Logged By B  | SDE<br>SDE<br>and | <u>z-Re</u><br>-C                      | Site                 | e<br>ounc   | Cnpstal Cleaners<br>Elevation Start Date<br>01/26/06  | No.<br>- 4<br>Sheet                          | No<br>Finis       | Project<br>361<br>N<br>h Date<br>1 (26 | No.<br>2052<br>of | 1      | 2         |
|-----|---|-------------------|--|----------------------|-------------|---|--|-------------------|--|-------------------|--------|-----------|
|     | Drilling Contra   | ctor              | ADT                                    |                      |             | Driller's Name Murty Bacher   | Rig T  | <sup>ype</sup> Ge | io fro                                 | be Sy             | 60     |           |
|     | Drilling Method   | Di                | red Pus                                | ih                   | F           | Protection Level D P.I.D. (eV)  | Casir  | ng Size           | 11/2'                                  | Auger             | Size,  | 12"       |
| · . | Soil Drilled  |                   | Rock Drille                            | d                    | <b>٦</b>    | Total Depth 32  | er/Date                                      |                   | Piez                                   | Well              | Boring | 9         |
|     | Depth(Feet)<br>Sample No. &<br>Penetration/<br>Recovery (Feet)  | Sample Type       | SPT Blows/6"<br>or<br>Core Rec./Rqd. % | SPT-N<br>(Blows/Ft.) | Graphic Log | Sample<br>Description   | USCS<br>Group Symbol                         | Notes on Drilling | Pl Meter<br>Field Scan                 | PI Meter (U       | ]      | Lab Tests |
| 52  | 2 2 2 4.0<br>3 4 5<br>6 1.6<br>7 4.5<br>8 1.6<br>4.5<br>7 4.5<br>8 1.6<br>10 1.0<br>1.0<br>1.0<br>1.0<br>1.0<br>1.0<br>1.0<br>1.0 |                   | , 32'                                  | ojs, d               |             | 0 - 10 Block top, Bedding, Sur<br>grand, dry, boose<br>1.0 - 2.8 oive/Brin grandy chap<br>w intre sund, we, damp, masith<br>2.5 - 3.0 drange brown sundy chap<br>w sure grand, dry i we, SP.<br>3.0 - 4.0 Olive Brown Jorryc Cham<br>Sand-chay - grand with, wet, no, on<br>mstoff.<br>4-5.5 DK Brown Stud - grand<br>With, WG, most, boose, Selmo,<br>5.5-8 Dire Brown, Sand-grand<br>Chap mix, WG in Stuff, wet, MP<br>Collecting Surplet Dys<br>Frans S. Sto 8<br>8-12 Brown Julive Sand-grand<br>Chap w6, wet/Sat, Meshiff dr | Ay Fill<br>GC<br>SC<br>CC<br>SC<br>CC<br>Har | ding La           | 21.9<br>200<br>41.0                    | Associa           | ates-  |           |

e



|    | Project   |  | TEST BORIN   | GLOG  |   |
|----|---|--|--|---|---|
| •  |   | I - fegion   | 8 Dry Acare  | B GW-7  | 3612052036  |
|    | MSDE  | T. Sili  | " (nystal (Ica   | rens Sheet M  | lo of   |
|    | Brilling Contractor   | ndien Shind  |  | ol MoG  | Finish Date   |
|    |   | ADT  | Driller's Name<br>Marty  | Bachner Rig Ty  | pe Geo Pube Syou  |
|    |   | let Dush   | Protection Level   | P.I.D. (eV) Casing  | Size Auger Size   |
|    | Soll Drilled 8  | Rock Drilled   | Total Depth 36 Depth   | to Groundwater/Date   | Piez Well Boring  |
| 51 | A     C     Image: Depth(Feet)       Depth(Feet)     Sample No. &       Penetration/     Penetration/       Recovery (Feet)     Sample Type | SPT Blows/6"<br>or<br>Core Rec./Rqd. %<br>SPT-N<br>(Blows/Ft.) | Do Joji<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Descrip<br>Des | le<br>tion<br>diry, Black,<br>dr<br>nul, odar, weth<br>c. Studfsily ely<br>Stuff, dry,<br>odar<br>c. Stuff, dry,<br>odar<br>c. Stuff, dry,<br>br/sc<br>c.<br>GC | Notes on Drilling       Notes on Drilling       Notes on Drilling       PI Meter       PI Meter       PI Meter       PI Meter       Lab       Lab       Lab       Tests |
|    | 5-1.6/<br>7-1.6/<br>7-1.6/<br>8-<br>1.0/<br>8-<br>1.0/<br>8-<br>1.0/<br>1.0/<br>1.0/<br>1.0/<br>1.0/<br>1.0/<br>1.0/<br>1.0/                |  | 4-45 DK Brun<br>1005e<br>4.5-ab overge,<br>Clay, wet, MP, WC<br>6-8 4 Brun dra<br>Clay, wet/saturate<br>HP/MP, jenses of   | n Sindy Gmet<br>alive growery<br>3:<br>- growery<br>- d zones; WG,<br>  |   |
| 9  | 15t Attmpf-<br>910003(e) L 33   | Refusal @ 7  | jo.b' of oak   | wood blocking Hardi   | ງມຸດ<br>ng Lawson Associates  |

| Logged By  | bec Shew   | Groun                      | d Elevation Start Date<br>01/23/06   | Sheet No of<br>Finish Date<br>01/23 (06   |
|--|--|----------------------------|--|---|
| Drilling Contracto   | ADT  |                            | Driller's Name Murty Bachner   | Rig Type Probe 5400   |
| Drilling Method  | meet push  |                            | Protection Level   | Casing Size   |
| Soil Drilled   | Rock Drilled   | -                          | Total Depth<br>39.5 - Reford Depth to Groundwater.   | /Date Piez Well Boring  |
| Depth(Feet)<br>Sample No. &<br>Penetration/<br>Recovery (Feet)   | Sample Type<br>SPT Blows/6"<br>or<br>Core Rec./Rqd. %<br>SPT-N | (Blows/Ft.)<br>Graphic Log | Sample<br>Description  | USCS<br>Group Symbol<br>Notes on Drilling<br>I Meter<br>eld Scan<br>Meter<br>ead Space<br>Lab Tests   |
| 1-2-2.1<br>2-3-4.0<br>3-4.0<br>3-4.0<br>7-4.0<br>8-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2<br>1-2.2 |  |                            | D-0.2 Black TOP, fill<br>D.2-10 Backfill for tarred area<br>(PPB-390); Black, losse NP,<br>Surdy growel.<br>1.0-1.8 Lt gray five Sard and fore-<br>grover 15, Coarfe Sind, PG, NP,<br>1.0050, dry,<br>1.47-2.0 DK reddish Brown layer,<br>Sitty fore sond of coarse sand, Dy<br>NP/SP, PG, M denge,<br>2.0-2.6 DK gray Sitty fore<br>Sand, oclor, some five grad,<br>darp, M Stiff, fryable.<br>2.6-2.7 White zone, some as above.<br>2.7-4 Brownish once Silty foresad<br>inf tona coarse sond/five grad,<br>darp, MStiff<br>44,2 Black/Brown Sudyclay FIP,<br>PG, MP, Sitt<br>4.2-50 Yellow/olive/Brown, Strathies<br>w orange mell drainced areas, Sandyfore,<br>Sitt, PG, darp, SP, V. Dense.<br>5.0-5.7 Sureas above but moist.<br>5.7-4.0 Sime of shore but dry<br>PG-7 purple 'roots' observed below<br>a D.1' layer of five grae!,<br>More gray zones from 7-8.<br>8-9.5 Olive Brown Five Sady Sitt<br>45-9.5 Sane as 5.7-8.0<br>10.5-12 Lt Brown Sudy Grovel boose. | $\frac{F(1)}{SM} = \frac{C1.0}{SM}$ $\frac{SM}{SM} = \frac{C1.0}{SM}$ $\frac{C1.0}{SM} = \frac{C1.0}{SM}$ $\frac{C1.0}{C1.0}$ $\frac{C1.0}{SM} = \frac{C1.0}{C1.0}$ $\frac{C1.0}{C1.0}$ |

|        |   |                                   |             |                    |                  |                      |             | TEST BORING LOG  | <del>71</del> |                   |                                       |                        |             |              |
|--------|---|-----------------------------------|-------------|--------------------|------------------|----------------------|-------------|--|---------------|-------------------|---------------------------------------|------------------------|-------------|--------------|
|        | Project                                 | A (-                              |             |                    |                  |                      |             | Boring/W   | No.           | F                 | Project                               | No                     |             |              |
| · .    | Client                                  | MSD                               | NEC.        | -                  | lee              | m                    | <u>K</u>    | Dy Cleaners Gw   | - 11          |                   | 3612                                  | -657                   | <u>1031</u> | 6            |
| ····   |   | MYSD                              | EC          | •                  |                  |                      |             | (mystal Cleaners)  | Sheet N       | lo                | <u> </u>                              | _ of                   | 1           | www.content. |
|        | Logged                                  | By Bro                            | in?         | lon S              | sha              | Gro اس               | uno         | d Elevation Start Date<br>ーレーン   |               | Finist<br>c       | n Date<br>>1/24                       | 106                    |             |              |
|        | Drilling C                              | ontracto                          | 'Aı         | PT                 |                  |                      |             | Driller's Name<br>Marrie Badaner   | Rig Ty        | oe<br>G           | eo:pr                                 | ·be                    | 5401        | D            |
|        | Drilling N                              | lethod                            | )iv.        | eet                | Pus              | ·<br>h               | ł           | Protection Level D P.I.D. (eV)   | Casing        | Siz/e             | · ·                                   | Auge                   | Size        | y ec.        |
|        | Soil Drille                             | ed 8'                             | F           | Rock D             | rilled           |                      | -           | Total Depth 32   | er/Date       |                   | Piez                                  | Well                   | Borin       | g            |
| 51     | C C C Depth(Feet)                       | C Penetration/<br>Recovery (Feet) | Sample Type | SPT Blows/6"<br>or | Core Rec./Rqd. % | SPT-N<br>(Blows/Ft.) | Graphic Log | Sample<br>Description<br>Description<br>D. O. I frogen, rootzane, grass<br>DK Brann, snar,<br>O. I - O. G DK Brann silly fine sam<br>organic lasm, dayp. SP, PG, norts,<br>m Desse.<br>O. 6 - 1.0 same as above but of<br>some gracel;<br>I. 0 - 1.3 Lt Brown formye, sandy              | Group Symbol  | Notes on Drilling | O     FI Meter       O     Field Scan | PI Meter<br>Head Space | 3           | Lab Tests    |
| 52     | 4<br>5<br>2<br>7<br>4<br>9              | 1/4:0                             |             |                    |                  |                      |             | Clay w/ give i, demp, MP.<br>M Dense, WCK in Sine.<br>(obble Stuck in Sine.<br>4-5 Lt Brown/oranje Sandy gre<br>WG, NP,<br>5-5.5. Lt Brown/orage graveych<br>VI some sond, het, loose, WG<br>5.5-6.5 Same as above but sa<br>6.5-8 Lt Brown Sandy grael<br>m/ trave fines, ctw, i NP, wG | y GU<br>Y GU  |                   | 4.0                                   |                        |             |              |
|        | 11-11-11-11-11-11-11-11-11-11-11-11-11- |                                   |             |                    |                  |                      |             |  |               |                   |                                       |                        |             |              |
|        | BOB C                                   | 232'.                             | Ti          | zht                | 'a:              | illers               | ?a          | id in Spots.   |               |                   |                                       |                        |             |              |
| E<br>S | 9910003(e)                              | L 33                              |             |                    |                  |                      |             |  | —Hardi        | ng La             | wson A                                | Associ                 | ates—       | l            |



|        |              |   |  |                 |           | TEST BORING                           | GLOG (  | $\gamma$    |                       |               |                |       | ÷     |
|--------|--------------|---|--|-----------------|-----------|---------------------------------------|---|-------------|-----------------------|---------------|----------------|-------|-------|
| Pr     |              | SDE                                     | Z- K                                     | 29.0            | <u>n</u>  | 8 Dudleane                            | Boring/M  | NO.<br>-16  | F                     | Project       | No.            | 520   | >26   |
| CI     | ent N        | SDE                                     | EC                                       | Sit             | ie<br>· ( | Crystal Clea                          | ners  | Sheet N     | 10                    | ŀ             | of             | 1     |       |
| Lo     | gged By<br>B | venda                                   | n Shaw                                   | Gro             | ound      | d Elevation Sta                       | rt Date   |             | Finisl                | n Date        | 106            |       |       |
| Dr     | Illing Contr | actor                                   |  |                 |           | Driller's Name<br>Marky Bor           | la va Part  | Rig Ty      | pe<br>Gee             | , Pn.         | be i           | 340   | 0     |
| Dr     | lling Metho  | od                                      | Push                                     |                 | F         | Protection Level                      | P.I.D. (eV)   | Casing      | Size                  | 1/2 +         | Auger          | Size  | • •   |
| So     | il Drilled   | 8°                                      | Rock Drille                              | d               | 7         | Total Depth / Depth                   | to Groundwater,   | /Date       |                       | Piez          | Well           | Borin | g     |
|        | <u> </u>     | e ()                                    | )"<br>. %                                |                 |           |                                       | karan bin genata karan kanan kanan karan kara |             | Бu                    | Mo            | nitoring       |       |       |
| h(Feel | le No.       | ery (re<br>le Typ                       | slows/6<br>or<br>c./Rqc                  | 'T-N<br>vs/Ft.) | nic Log   | Samp                                  | le  | CS<br>Symbo | Drilli                | (pp           | om)<br>o       | ,     | ests  |
| Dent   | Samp         | Samp                                    | SPT E                                    | SP<br>(Blov     | Grapł     | Descript                              | lion  | US          | otes or               | eter<br>Scan  | eter<br>I Spac | . • . | Lab T |
|        |              |   | Ŭ  |                 |           | 0 25000                               |   |             | z                     | PI M<br>Field | PI M<br>Head   |       |       |
| -      |              |   |  |                 |           | five sand, PG, MP,                    | damp, fryask,   | Sm-<br>Fil? |                       | 4.0           |                |       |       |
| 2.     | 1.5          | <b>,</b>                                |  |                 |           | 2-~3.5 orange                         | Brun, silty   |             |                       |               |                |       |       |
| 3.     | /¼^          |   |  |                 |           | fike sond of grand 1                  | monst, M. Derse   | SW          |                       |               |                |       |       |
| 4.     |              |   |  |                 |           | 3.5 - of Brongish of                  | re Sand/gravel  | Sc          |                       |               |                | -     |       |
| Q.     | -            |   |  |                 |           | M Perse                               |   | G           |                       | 1.0           |                |       |       |
| 6.     | - 19.        |   |  |                 |           | 4- ~6.5 save as                       | 3.5-4   |             |                       |               |                |       |       |
| 7      | - l'y        | 5                                       |  |                 | •         | 6.5-7 Reddish (                       | coarse grad.  | Rock-9      | mel                   |               |                |       |       |
| X      |              |   |  |                 |           | 7-8 Olive Brown                       | , grould  | GC          | n <sub>de</sub> se de |               |                |       |       |
| 6      |              |   |  |                 |           | (lap w/ some some                     | d, 'nct, wo.  |             | •                     |               |                |       |       |
| 7-     |              |   | -  |                 |           | PP' Dark Brun                         | organic   |             |                       |               |                |       |       |
| 10-    |              |   |  |                 |           | sitty sud layer<br>strag organic      | odr.  |             |                       | -             |                |       |       |
| (( -   |              |   |  |                 |           | , , , , , , , , , , , , , , , , , , , | -   |             |                       |               |                |       |       |
| 12 -   |              |   |  |                 |           |                                       | -   |             |                       |               |                |       |       |
|        | 805-         | - 23                                    | 5' P.1                                   |                 | ŀ         | I Fall                                |   |             |                       |               | -              |       |       |
|        |              | • •                                     | - Net                                    | ual j           | ŀ         | and apps .                            |   |             |                       |               |                |       |       |
|        |              |   |  |                 |           |                                       |   |             |                       |               |                |       |       |
| 0010   |              | ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~ | allyn charlen yw a golyddi Blancorraetau |                 |           |                                       |   | —Hardi      | ng La                 | wson A        | ssocia         | ites— |       |

|                                       | Project MS   | DEC - R   | 29TON                               | TEST BO<br>8 Dry Cle   | RING LOG<br>BoringM<br>amery GW-  | No.<br>017               | Project<br>3612                                   | No.<br>20520               | 36        |
|---------------------------------------|--|---|-------------------------------------|--|---|--------------------------|---|----------------------------|-----------|
| i i i i i i i i i i i i i i i i i i i | MSDE   | El  | Site                                | Cnistal (  | leancy  | Sheet N                  | o   | of                         |           |
|                                       | Logged By Bra  | ndon Gran   | Grour                               | d Elevation  | Start Date  | 6                        | Finish Date                                       | 3/06                       |           |
|                                       | Drilling Contractor  | ADT   |                                     | Driller's Name   | lant Bachner.   | Rig Typ                  | Geo Pr  | obe 52/1                   | 10        |
|                                       | Drilling Method  | Seo Pube  | Directp                             | Protection Level   | P.I.D. (eV)   | Casing                   | Sizel/2"  | Auger Size                 | 1'/2"     |
|                                       | Soil Drilled   | Rock Drilled  |                                     | Total Depth<br>23  | Depth to Groundwate   | er/Date                  | Piez  | Well Borii                 | ng        |
| Ç                                     | Depth(Feet)<br>Sample No. &<br>Penetration/<br>Recovery (Feet)   | Sample Type<br>SPT Blows/6"<br>or<br>Core Rec./Rqd. % | SPT-N<br>(Blows/Ft.)<br>Graphic Loo |  | Sample<br>scription   | USCS<br>Group Symbol     | Notes on Drilling<br>PI Meter ()<br>Field Scan () | PI Weter<br>Head Space (mg | Lab Tests |
| 51<br>52<br>1                         | 2 2.1<br>2 2.1<br>4 1.8<br>6 X.0<br>7<br>8<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10 |   |                                     | 0- p. O DK BI<br>MP, PG, 1005<br>1.0-2.0 01,<br>Sand, PG, Sta<br>23:0 Sam<br>2+ Drann.<br>3-4 0 raye,<br>W some fine<br>SP,<br>4- ~S Some a<br>Brum organ<br>~S-6 2+ B<br>M Some fines,<br>to fine graves<br>6-7.524 Brun<br>Graves, Saf, S<br>Arves, Saf, S | on Sardy loam,<br>ie, roots, damp<br>he Brown Sithy fine<br>Ff, Morst, MP,<br>e as (1-2) but<br>(Brown Sardy grine<br>Brown Sardy grine<br>1<br>15 3-4, but DK<br>mz levse tinghit.<br>min Sardy grinel<br>whith, loose, fines<br>WG SP<br>h; lenses of (oarse<br>1 clay, HP, het,<br>un Sudy gravel m<br>ig, loose | Fill<br>Sm<br>Gm<br>- GC |   |                            |           |
|                                       | 27   | 2   | , ,                                 |  |   | Hardir                   | ng Lawson /                                       | Associates-                |           |

•

| Project No : Z( +2/25                   | Log (Slick Up |                 |                        | 10.: MW.   |
|---|---------------|-----------------|------------------------|--|
|   | Project       | Area: Carchal   | Cleaners ery           | Lican  |
| Contractor: ADT                         | Driller: Mark | Radue Method: D | vect Push - G          | seo finto  |
| Logged By: Bunder                       | ~ Shew        | Date Started:   | 1/24106 Comp           | oleted: 01/  |
| Checked By: CR5                         | Date: 4/3/0   | 6 CRGWOOI       | 02601 X X, MS, MI      | )(" 143  |
| Lock Identification:                    | - Flush       | n Mount.        |                        |  |
| Surface Casing Type:                    |               | Elevation       | of top of 938          | 3.05   |
| Steel                                   |               | Elevation       | of top of              | 7.67   |
| Ground Surface Elevation:               |               | Ris             | er Pipe:               | <u>r. 96 r</u>   |
| Samar TOC                               |               |                 | Surface Seal:          | Road!  |
| Surface Casing                          |               |                 |                        |  |
| Diameter: "H'                           | 94            |                 | /                      |  |
| Inside Diameter of                      |               | Borehole [      | Diameter: <u>1/2</u> " |  |
| Surface Casing: 27/8                    |               | Inside Dia      | ameter of 1/2.         |  |
|   |               | Borehole        | e Casing:              |  |
|   | 11 1000       |                 | Class                  | C ila  |
| Top of Well Seal:                       | 11 vias       | Type o          | f Backfill:            | Smarfa   |
| 0" /                                    | → I           | <b>к</b>        | of Riser: Sch 40       | , pvc  |
| Depth/Elevation of<br>Top of Sand:      |               | Riser Inside I  | Diameter:l             |  |
| N/2 /                                   |               |                 | Pro to                 | nile d   |
| Denth/Elevation of                      |               | . Тур           | e of Seal:             | THE CH   |
| Top of Screen:                          |               |                 |                        |  |
| \$ 16.5 B.61 92                         | 2,4 msl       | Type of Sa      | ind Pack: <u>Clean</u> | Sand   |
|   |               |                 |                        | . 1  |
|   |               | Туре с          | of Screen: Slotte      | d PVC  |
|   |               | Slot Size       | x Length: D.016*       | × 10'  |
|   |               | Inside          | Diameter               |  |
|   |               |                 |                        |  |
| Depth/Elevation of                      |               |                 |                        | _  |
| ZE-5 256. 912                           | .4'm) =       | Depth of        | Sediment<br>with Plug: | Summer of the second seco |
| Depth/Elevation of<br>Bottom of Boring: |               |                 |                        |  |
| <u>30</u> <sup>°</sup> / 90             | 8.0 mil       |                 |                        |  |
| Not To Scale                            | -             | · · · · ·       |                        | FIGURE   |
|   |               | MACTEO          |                        |  |

| Project No.: 3612052036                              | Project Name: | MODEC - Repor                            | 8 Dw Cleaners           |
|--|---------------|--|-------------------------|
| <u> </u>   | Project Area: | Chystal alean                            | erf.                    |
| Contractor: ADT Driller:                             | Marty B.      | Method: Direct P                         | ISL - Geo Probe         |
| Logged By: Branden Show                              |               | Date Started: 01 25 10                   | Completed: 01/25106     |
| Checked By: CR5 Dat                                  | te: 4/3/06    |  |                         |
| Lock Identification:                                 | Flish m       | mt                                       |                         |
| Surface Casing Type:                                 |               | Elevation of top of<br>Surface Casing:   | 932.42° msl             |
| Ground Surface Elevation:                            |               | Riser Pipe: _                            | 932.26 mil              |
| Same as TOL  |               | Type of Surface                          | (ment Road Bix Black Te |
| Surface Casing<br>Diameter: 4                        |               | ·  | puth.                   |
| Inside Diameter of<br>Surface Casing: 2 1/4          |               | Borehole Diameter: _                     | 1'12'                   |
|  |               | Inside Diameter of<br>Borehole Casing: _ | ·/2 <sup>°</sup>        |
| Depth/Elevation of                                   |               | Type of Backfill:                        | Cave in.                |
| Top of Well Seal:                                    |               | Type of Riser                            | Sh 40 PVC               |
| Depth/Elevation of<br>Top of Sand:                   |               | Riser Inside Diameter:_                  | / <sup>n</sup>          |
| N/K /  |               | Type of Seal:                            | Bentinite Seal          |
| Depth/Elevation of Top of Screen:                    |               |  |                         |
| 6 1926.4mil  |               | —— Type of Sand Pack: _                  | N/A.                    |
| Nuter Q ~10 _  |               |  | Culler ar               |
|  |               | —— Type of Screen:_                      | Slotted pre             |
|  |               | Slot Size x Length:                      | 0.010 × 10              |
| •  |               | of Screen: _                             |                         |
| Denth/Elevation of                                   |               |  |                         |
| Bottom of Sgreen:                                    |               | Depth of Sediment                        |                         |
| Depth/Elevation of<br>Bottom of Boring:<br>904. 1ms) |               | Sump war Plug                            |                         |
| Not To Scale   |               |  | FIGURE                  |
|  |               | - MACTEC Engineer                        | ing and Consulting Inc  |



| Project No.: 3612052036                 | Project Name:     | MSDEC Region & Dy Man          | ins      |
|---|-------------------|--------------------------------|----------|
|   | Project Area:     | Crystal Cleaners               |          |
| Contractor: ADT Driller                 | r: Marty B.       | Method: Dirat Rish - Geo P     | hobe.    |
| .ogged By: Brindian Shuri               |                   | Date Started: 01/24/06 Complet | ed: 01   |
| Checked By: CRS                         | Date: 4/2/06      | CRGWODOZGOIXX A ITH            | 0        |
|   |                   |                                |          |
| Lock Identification:                    | Flush M           | mt                             |          |
| Surface Casing Type:                    |                   | Elevation of top of 93489      | 1        |
| Ste.                                    |                   | Elevation of top of            | <u> </u> |
| Ground Surface Elevation:               |                   | Riser Pipe:934.70              | ms       |
| Some TOC                                |                   | Type of Surface                | Danak    |
|   |                   | Seal:                          | Non      |
| Surface Casing                          | 499 88            |                                |          |
|   |                   |                                |          |
| Inside Diameter of                      |                   | Borehole Diameter:             |          |
| Surface Casing.                         |                   | Inside Diameter of             |          |
|   |                   | Borehole Casing:               |          |
|   |                   |                                |          |
| Depth/Elevation of                      |                   | ——— Type of Backfill:(a√ℓ ≀∕∕  | _i       |
| Top of Well Seal:                       |                   | - Sch Ho                       | Ove      |
| Depth/Elevation of                      | →                 | Type of Riser: <u>Och 90</u>   | 11       |
| Top of Sand:                            | ¥_                | Riser Inside Diameter:         |          |
| NINI                                    | →   MI   MI       | The Bentonite                  | Scal.    |
| Depth/Elevation of                      |                   |                                | 0        |
| Top of Screen:                          | ater (            |                                |          |
| 8 1 727 msl                             |                   | —— Type of Sand Pack: N [ A    |          |
| 1 Sator Qualo                           |                   |                                |          |
| Vica C 1                                |                   | The former Shatted             | DVC      |
|   |                   | Type of Screen: Goot Troot     | 101      |
|   |                   | Slot Size x Length: 0.0(0 K    | IV       |
|   |                   | of Screen:                     |          |
|   |                   |                                |          |
| Depth/Elevation of<br>Bottom of Screen: |                   |                                |          |
| 18 / 917 ms                             | \                 | Sump with Plug:                |          |
| Depth/Elevation of                      |                   |                                |          |
| 28 / 907 mil                            |                   |                                |          |
| * (aved in.                             | └── <b>▶</b> └─── |                                |          |
| Not To Scale                            |                   | F                              | IGUR     |
|   |                   |                                |          |

|                            |              | Microwel                    | I Completi                     | on Log                        |                         |            | Well No:        |
|----------------------------|--------------|-----------------------------|--------------------------------|-------------------------------|-------------------------|------------|-----------------|
| Project No.: 36            | 12052036     | Project: Regi               | on 8 Group 1 -                 | Crystal Checked               | By: C. Staples          |            | GW-18           |
| Client Name:               | NYSDEO       | C Logged By:                | SHAW                           | Protection Level:             | D Gro                   | ound Ele   | evation: 938.02 |
| Drilling Contract          | or: GEOLOG   | IC                          | Drilling Met                   | hod: DIRECT P                 | USH                     |            | s Name:         |
| Bit Type/Size:             | s            | oil Drilled:                |                                | Rig Type:                     | Start Date:             | JOE MI     | Finish Date:    |
| GEOPROBE F                 | ROD - 1 1/2" | 22'                         |                                | 66DT Track Rig                | 11/2/20                 | 06         | 11/2/2006       |
| Bedrock Interval           |              | N/A                         |                                | P.I.D. (eV): NA               | Casing Size             | N/A        | Auger Size: 2"  |
| Depth (feet)<br>Becovery S | sample ID    | Guaphic Log<br>Mell Diagram | Well<br>Construction<br>Notes  | Notes:                        | all donths in fact from | n ground s | nrfaco)         |
| 2                          |              |                             | Cement: 0 -                    | wen Construction Notes: (     | an depuis in feet froi  | n ground s |                 |
|                            |              |                             | 0.5'                           | Depth to Water Levels:        |                         |            | -               |
| 4                          |              |                             | SandPack/B                     | 11/02/2006: 16.25' (bgs)      |                         |            | -               |
|                            |              |                             | ackfill 0.5 -                  | Top of Riser elevation = 937. | .76                     |            | -               |
| 6                          |              |                             | 6.0                            |                               |                         |            | -               |
| 8                          |              |                             | Schedule 40<br>PVC casing: 0.5 |                               |                         |            | _               |
|                            |              |                             | - 11.9                         |                               |                         |            | _               |
| 10                         |              |                             |                                |                               |                         |            | -               |
| 12                         |              |                             | Bentonite seal:                |                               |                         |            | _               |
|                            |              |                             | 6 -10                          |                               |                         |            | -               |
| 14                         |              |                             | Silica & quartz<br>sand filter |                               |                         |            | -               |
|                            |              |                             | pack, 10 - 23                  |                               |                         |            | -               |
| 16                         |              |                             |                                |                               |                         |            | _               |
| 18                         |              |                             | -screen w/end                  |                               |                         |            | -               |
|                            |              |                             | 21.9                           |                               |                         |            | -               |
| 20                         |              |                             |                                |                               |                         |            | -               |
|                            |              |                             | Sump: 21.9 -                   |                               |                         |            | -               |
| 22                         |              |                             | 22.0                           |                               |                         |            | -               |
| 24                         |              |                             |                                |                               |                         |            |                 |
| - + -                      |              | the second second           |                                | BOW: 22.02'                   |                         |            | -               |
| 26                         |              |                             |                                |                               |                         |            | -               |
| 28                         |              |                             |                                |                               |                         |            | -               |
|                            |              |                             |                                |                               |                         |            | -               |
| 30                         |              |                             |                                |                               |                         |            |                 |
|                            | CTEC         |                             |                                |                               |                         |            | 1 of 1<br>GW-18 |

|                              |             | Microwe         | ll Completi                    | on Log                                |                         |                  | Well No:        |
|------------------------------|-------------|-----------------|--------------------------------|---------------------------------------|-------------------------|------------------|-----------------|
| Project No.: 3612052036      |             | Project: Reg    | ion 8 Group 1 ·                | - Crystal Checked                     | By: C. Staples          |                  | GW-19           |
| Client Name:                 | NYSDEC      | Logged By:      | SHAW                           | Protection Level:                     | D Gr                    | ound Ele         | evation: 939.12 |
| Drilling Contractor:         | GEOLOGIC    |                 | Drilling Met                   | hod: DIRECT                           | PUSH                    | Driller<br>JOE M | S Name:         |
| Bit Type/Size:               | Soil        | Drilled:        | 24.5'                          | Rig Type:                             | Start Date:             | JOE MI           | Finish Date:    |
| GEOPROBE ROD - 1 1           | /2"         |                 |                                | 66DT Track Rig                        | 11/1/20                 | 06               | 11/1/2006       |
| Bedrock Interval             |             | N/A             |                                | P.I.D. (eV): NA                       | Casing Size             | N/A              | Auger Size: 2"  |
| (teet)<br>Recovery Sample ID | Graphic Log | Well Diagram    | Well<br>Construction<br>Notes  | Notes:                                |                         |                  |                 |
|                              |             |                 | Sfc Compltn.                   | Well Construction Notes:              | (all depths in feet fro | m ground s       | urface)         |
| 2                            |             |                 | Cement: 0 -                    | Depth to Water Levels:                |                         |                  | -               |
| 4                            |             |                 | 0.5                            | 11/02/2006: 15.02' (bgs)              |                         |                  | -               |
|                              |             |                 | OON sand                       | Top of Riser Elevation = $9$          | 38.75                   |                  | -               |
| 6                            |             |                 | pack, 0.5 -7                   | · · · · · · · · · · · · · · · · · · · |                         |                  | _               |
|                              |             |                 | Schedule 40                    |                                       |                         |                  | -               |
| 8                            |             |                 | PVC casing                     |                                       |                         |                  | -               |
| 10                           |             |                 |                                |                                       |                         |                  | -               |
|                              |             | <u>,,,,,,,,</u> | Bentonite seal,                |                                       |                         |                  |                 |
| 12                           |             |                 | 5 - 8                          |                                       |                         |                  | -               |
|                              |             |                 |                                |                                       |                         |                  | -               |
| 14                           |             |                 | #10 slot PVC                   |                                       |                         |                  | -               |
| 16                           |             |                 | screen w/end<br>cap, 10.8-20.8 |                                       |                         |                  | -               |
|                              |             |                 |                                |                                       |                         |                  | -               |
| 18                           |             |                 | Silica sand<br>filter pack 8 - |                                       |                         |                  | -               |
| 20                           |             |                 | 21                             |                                       |                         |                  | -               |
|                              |             |                 |                                |                                       |                         |                  |                 |
| 22                           |             |                 | 20.9                           |                                       |                         |                  | -               |
|                              |             |                 |                                |                                       |                         |                  | -               |
|                              |             |                 |                                | BOW: 20.87 (bgs)<br>TD = 24.5' (bgs)  |                         |                  | -               |
| 26                           |             |                 |                                |                                       |                         |                  | -               |
|                              |             |                 |                                |                                       |                         |                  | -               |
| 28                           |             |                 |                                |                                       |                         |                  | -               |
|                              |             |                 |                                |                                       |                         |                  | -               |
|                              | С           |                 |                                | I                                     |                         |                  | 1 of 1<br>GW-19 |

|                              |  | Microwel          | I Completi                    | on Log                      |                          |             | Well No:        |
|------------------------------|--|-------------------|-------------------------------|-----------------------------|--------------------------|-------------|-----------------|
| Project No.: 3612052036      | 1  | Project: Regi     | on 8 Group 1 -                | Crystal Checked             | By: C. Staples           |             | GW-23           |
| Client Name: NY              | SDEC   | Logged By:        | SHAW                          | Protection Level:           | D Gro                    | ound Elev   | vation: 938.35  |
| Drilling Contractor:         | GEOI   | LOGIC             | Drilling Met                  | hod: DIRECT I               | PUSH                     | Driller's   | s Name:         |
| Bit Type/Size:               | Soil   | Drilled:          | 24                            | Rig Type:                   | Start Date:              | JOE ME      | Finish Date:    |
| GEOPROBE ROD - 1 1/2"        |  |                   |                               | 66 DT Track Rig             | 11/2/200                 | )6          | 11/2/2006       |
| Bedrock Interval             | N/A  |                   |                               | P.I.D. (eV): NA             | Casing Size              | N/A         | Auger Size: 2"  |
| (teet)<br>Recovery Sample ID | Graphic Log  | Well Diagram      | Well<br>Construction<br>Notes | Notes:                      |                          |             |                 |
|                              |  |                   | Sfc Compltn.                  | Well Construction Notes:    | (all depths in feet from | n ground su | rface)          |
| 2                            |  |                   | Cement: 0 -                   | Depth to Water Levels:      |                          |             | -               |
| 4                            |  | •                 | 0.5                           | 11/02/2006: 14.79' bgs      |                          |             | -               |
|                              |  |                   | Backfill/                     | Top of Piser Elevation - 93 | 88.11                    |             | -               |
| 6                            |  |                   | 7                             | Top of Riser Elevation = 93 | 0.11                     |             |                 |
|                              |  |                   |                               |                             |                          |             | -               |
| 8                            |  |                   | Bentonite seal,               |                             |                          |             | -               |
|                              |  |                   | 7 - 10                        |                             |                          |             | -               |
| 10                           |  |                   | <b>a a a a</b>                |                             |                          |             | -               |
| 12                           |  | S < 🔂             | Schedule 40<br>PVC casing     |                             |                          |             | -               |
|                              |  |                   |                               |                             |                          |             | -               |
| 14                           |  | <u></u>           | #10 slot PVC                  |                             |                          |             | -               |
|                              |  |                   | cap, 13.5 -                   |                             |                          |             | -               |
| 16                           |  |                   | 23.5                          |                             |                          |             |                 |
| 18                           |  |                   |                               |                             |                          |             |                 |
|                              |  | 20000<br>20000    | Silica sand                   |                             |                          |             |                 |
| 20                           |  |                   | filter pack, 10 -             |                             |                          |             |                 |
|                              |  |                   | 24                            |                             |                          |             |                 |
|                              |  |                   | S                             | POW. 22.61                  |                          |             |                 |
| 24                           | A second se   |                   | 23.5 - 23.6                   | TD = 24.0'                  |                          |             |                 |
|                              | Accession of the second | and the second of |                               |                             |                          |             |                 |
| 26                           |  |                   |                               |                             |                          |             | -               |
|                              |  |                   |                               |                             |                          |             | _               |
| 28                           |  |                   |                               |                             |                          |             | -               |
| 20                           |  |                   |                               |                             |                          |             | -               |
| MACTEC                       |  |                   |                               | 1                           |                          |             | 1 of 1<br>GW-23 |







| Project:       NVSDR-Region & OM Cleanuk       Site:       CNStA Cleanuk         Project:       NUSDR-Region & OM Cleanuk       Date:       Jump 26,200         Project:       Number:       3612052636       Date:       Jump 26,200         Sample Location ID:       O       DH       Date:       OT30       End:       113D         Sample Location ID:       O       DH       Signature of Sampler.       Signature of Sampler.       ProtectiveFt.         Well Depth       Ft.       Measured       Top of Well       Well Riser Stick-up       Ft.       ProtectiveFt.         Depth       Ft.       Measured       Top of Protective       Vell Riser Stick-up       Ft.       Casing/Well Difference         Depth to Water       Ft.       Well Material:       Well Locked?       Well Dia.       2 inch       Protective  |  |
|--|--|
| Project Number:       3612852636       Date:       ymmm 26,2006         Sample Location ID:       0       0       13D         Signature of Sampler:       Signature of Sampler:       Signature of Sampler:         Well Depth       Ft.       Measured       Top of Well         Historical       Top of Protective       (from ground)       Ft.         Casing       Vell No       Vell Dia.       2 inch         Protective       Yes       Yes       Yes         Measured       Pvc       Yes       Yes         Measured       Top of Vell       Well Riser Stick-up       Ft.         Casing       Well Difference       Casing       Protective       Ft.         Depth to Water       Ft.       Well Material:       Well Locked?       Well Dia.       2 inch       Elect. Cond. Probe         S8       Well No       6 inch       Float Activated  |  |
| Time: Start: 0730 End: 1130         Sample Location ID: 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0  |  |
| Sample Location ID:       V - 0 D V       Signature of Sampler:         Well Depth       Ft.       Measured       Top of Well       Well Riser Stick-up       Ft.       Protective       Ft.         Historical       Top of Protective       Casing       Well Riser Stick-up       Ft.       Protective       Ft.         Depth to Water       Ft.       Well Material:       Well Locked?       Well Dia.       2 inch       Protective casing         Bepth to Water       Ft.       Well Material:       Well Locked?       Well Dia.       2 inch       Elect. Cond. Probe         S8       VCLNO       6 inch       Float Activated   |  |
| Well Depth       Ft.       Measured       Top of Well       Well Riser Stick-up       Ft.       Protective       Ft.         Historical       Top of Protective       Casing       Well Riser Stick-up       Ft.       Protective       Ft.         Depth to Water       Ft.       Well Material:       Well Locked Protective       Well Dia.       2 inch       Protective casing         Bepth to Water       Ft.       PVC       Yes       Ges       4 inch       Elect. Cond. Probe         S8       Well No       6 inch       Float Activated       Fto       Float Activated   |  |
| Depth to Water Ft. Well Material: Well Locked P.J.S. Well Dia 2 inch Water Level Equip. Use 4 inch Elect. Cond. Probe 6 inch Float Activated   | €<br>t.  |
| Press. Transducer  | əd:<br>e   |
| Height of Water Column X16 Gal/Ft. (2 in.) =Gal/Vol. Well Integrity: Yes No<br>Ft. Z65 Gal/Ft. (4 in.) =Gal/Vol. Well Integrity: Yes No<br>Prot. Casing Secure<br>Concrete Collar Intact<br>Other  | -  |
| E Purging/Sampling Equipment Used : Decontamination Fluids Used :  |  |
| If Used For)       Purging Sampling       Equipment ID       (✓ All That Apply at Location)         V       V       Peristaltic Pump   | ater   |
| PID: Ambient Air 1.0 ppm Well Mouth 1.0 ppm Purge Data Collected /In-line /Turbid Clear /  | Cloudy   |
| C       Set       @       0128       Gat       @       105       Gat       @       Gat | Gal.   |
| Applytical Parameter / If Sample Preservation Volume Sample Bottle ILot Nos.   | ilinin an                      |
| on Collected Method Required   | <u></u>  |
|  | 130<br>110<br>110<br>110<br>110<br>110<br>110<br>110<br>110<br>110 |
| 3 Deotre (2:30' ad 20' ad 11'  |  |
| FIGURE   | : 4-1<br>ORD<br>LAN  |

9404014D L 22



| F                              | Project: NVSDEC- Legnon & DN () (GAWAYS  | Site: WSTER GLEANAG  |
|--------------------------------|--|--|
| F                              | Project Number: 361205 2036  | Date: January 25, 2006   |
|                                |  | Time: Start: 1414 End: 1530  |
|                                |  | Signature of Sampler:  |
|                                | Well Depth Ft Measured Top of Well<br>Historical Top of Protecti<br>Casing   | Well Riser Stick-up Ft. Protective Ft.<br>ive (from ground) Casing/Well Difference   |
| Level/Well Data                | Depth to Water Ft. Well Material: ) Well Locked?:<br>PVC Yes No  | Well Dia.       2 inch       ProtectiveFt.         Well Dia.       2 inch       Elect. Cond. Probe        6 inch      Float Activated        7ress. Transducer   |
| Water                          | 16 Gal/Ft. (2 in.)<br>Height of Water Column X65 Gal/Ft. (4 in.) =G<br>Ft1.5 Gal/Ft. (6 in.)<br>Gal/Ft. (in.)To  | Sal/Vol. Well Integrity: Yes No<br>Prot. Casing Secure<br>Concrete Collar Intact<br>Other  |
| ation                          | Purging/Sampling Equipment Used :  | Decontamination Fluids Used :  |
| Equipment Documen              | Purging Sampling Equipment ID  Purging Sampling Equipment ID  Submersible Pump Bailer  PVC/Silicon Tubing Feflor/Silicon Tubing Hand Pump In-line Filter Press/Vac Filter  | (✓ All That Apply at Location)<br>Methanol (100%)<br>25% Methanol/75% ASTM Type II water<br>Deionized Water<br>Liquinox Solution<br>Hexane<br>HNO <sub>3</sub> /D.I. Water Solution<br>Potable Water<br>None<br>None |
| Data                           | PID: Ambient Air <u>2.0</u> ppm Well Mouth <u>2.0 ppm</u> Purge Da   | ta CollectedIn-lineCloudy<br>In ContainerColoredOdor   |
| Field Analysis [               | Time Purge Data @ 1458 Get. @ 1519<br>Temperature, Deg. C No Paving 445<br>PH, units<br>Specific Conductivity (µmhos/cm) 2.07<br>Turbidity (NTUS)<br>Oxidation - Reduction, +/- mv<br>Dissolved Oxygen, ppm 6-67<br>DCOT   | Gal. @Gal. @Gal. @Gal.<br>   |
|                                | Analytical Parameter 🖌 If Sample Preservation  | Volume Sample Bottle ILot Nos.   |
| Requirements<br>this Location) | Collected         Method           VOCs         4°C         2           SVOCs         4°C         2           Metals         HN0,4°C         1           Cyanide         NaOH,4°C         1           Nitrate/Sulfate         H,50,4°C         1           Nitrate/Phosphate         H,204,4°C         1   | Required         CRGwoob12201Kx@;500           x1 liter P         GRGw00b012201Kx@;500           x500mLP         GRGw00b01401XA@1520           x1 liter P         x1 liter P   |
| Collectior<br>FRequired at     | $ \begin{array}{c} - \operatorname{Pest/PCB} \\ - \operatorname{TPH} \\ - \operatorname{TOC} \end{array} \begin{array}{c} 4^{\circ}\mathrm{C} \\ H_{s}^{\circ}\mathrm{So}_{*}^{\circ}\mathrm{A}^{\circ}\mathrm{C} \\ H_{s}^{\circ}\mathrm{So}_{*}^{\circ}\mathrm{A}^{\circ}\mathrm{C} \end{array} \begin{array}{c} 2 \\ H_{s}^{\circ}\mathrm{So}_{*}^{\circ}\mathrm{A}^{\circ}\mathrm{C} \end{array} $ | x1 liter AG<br>x1 liter AG<br>x1 liter P   |
| Sample                         | Notes:G<br>G<br>NYSDEC   | FIGURE 4-1<br>ROUNDWATER SAMPLE DATA RECORD<br>QUALITY ASSURANCE PROGRAM PLAN<br>ABB Environmental Services  |

2

ż

4



|  |  | GROUNDWA  | TER SAMPLI   | E FIELD DATA  | RECORD   |
|--|--|---|--|---|--|
| · P  | roject: <u>MSDEC</u>   | - K Im 8  | Dry Cleane   | f Site:   | : Cleaners   |
| P  | roject Number:   | 12052036  | ,  | Date: Ja  | 152 - 1dis   |
|  |  |   |  | Time: Start:  | <u>693</u> End: <u>113</u>   |
| S  | ample Location ID:   | 1-008   |  | Signature of Sa   | mple   |
|  | Well DepthF  | tMeasured<br>Historical   | Top of Well<br>Top of Protect<br>Casing  | Well Riser Stick-<br>ctive (from ground)  | up Ft. Protective Ft.<br>Casing/Well Difference  |
| 'ater Level/Well Data                                  | Depth to Water F   | t. Well Material:<br>PVC<br>SS<br>16 Gal/Ft. (2 in.)  | Well Locked?:<br>Yes Gr  | Gal/Vol. We   | ProtectivePt.     Casing     inch Water Level Equip. Used:     inchElect. Cond. Probe    Float Activated    Press. Transducer  Il Integrity: Yes No  |
| M  | Height of Water Column   | X65 Gal/Ft. (4 in.)<br>1.5 Gal/Ft. (6 in.)<br>Gal/Ft. (in.)   | = L  | Pro<br>Cor<br>Total Gal Purged Oth  | t. Casing Secure<br>herete Collar Intact<br>er   |
| ation  | Purging  | Sampling Equipment Us   | ied :  |   | Decontamination Fluids Used :  |
| Equipment Document                                     | (✔ If Used For)<br>Purging/ Sampling<br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br><br> | Peristaltic Pump<br>Submersible Pump<br>Bailer<br>PVC/Silicon Tubing<br>Teflon/Silicon Tubing<br>Airlift<br>Hand Pump<br>In-line Filter<br>Press/Vac Filter | Equipment ID   | {   | II That Apply at Location)<br>Methanol (100%)<br>25% Methanol/75% ASTM Type II water<br>Deionized Water<br>Liquinox Solution<br>Hexane<br>HNO <sub>3</sub> /D.I. Water Solution<br>Potable Water<br>None<br>DSCeS. DDC |
| Data   | PID: Ambient Air 41.6  | ppm Well Mouth  | )<br>ppm Purge [   | Data CollectedIn-fir  | Sample Observations:<br>— ClearCloudy<br>ontainerColoredOdor   |
| Field Analysis   | Temperature, Deg. C<br>pH, units<br>Specific Conductivity (<br>Turbidity (NTUS)<br>Oxidation - Reduction,<br>Dissolved Oxygen, pp  | @ <u>1'44</u><br><u>13.9</u><br>umhos/cm) <u>1.20</u><br>+/- mv <u>-77</u><br>n <u>(p'1</u>   | _ 5at. @ _ <u>\$</u><br><br>   | <u>Gal</u> @ 1805<br>7.5<br>1.44<br>21000<br>2000<br>20.1   | Gal. @Gal. @Gal.<br>   |
|  | Deptn  | 39.5 - 35.9   |  | 27.5-23   | 4  |
| S  | Analytical Parameter   | ✓ If Sample Pre<br>Collected  | eservation<br>Method   | Volume<br>Required  | Sample Bottle ILot Nos.  |
| Collection Requirement<br>f Required at this Location) | VOCs<br>SVOCs<br>Metals<br>Cyanide<br>Nitrate/Sulfate<br>Nitrate/Phosphate<br>Pest/PCB<br>TPH<br>TOC   | 4°<br>44<br>н<br>н<br>н<br>н<br>н<br>н<br>н<br>н<br>н   | PC <b>/HC</b><br>PC<br>NO,,4°C<br>aOH,4°C<br>SO,4°C<br>SO,4°C<br>C<br>SO,4°C<br>SO,4°C<br>SO,4°C | 2x40 ml<br>2x1 liter AG<br>1x1 liter P<br>1x500mLP<br>1x1 liter P<br>1x1 liter P<br>3x1 liter AG<br>2x1 liter AG<br>1x1 liter P | (RGN00803701×201750<br>(RGN00803701M50730<br>(RGN00803701M50730<br>(RGN008037040)0 1750<br>(RGN00802501×20120  |
| Sample<br>(/1  | 2 Deptus @:  | 25' and 37'   | <br><br>NYSDE(   | GROUNDWAT<br>C QUALITY AS   | FIGURE 4-1<br>ER SAMPLE DATA RECORD<br>SURANCE PROGRAM PLAN  |
| 940401   | 4D L 22  |   |  |   | ADD Environmental Services   |

,ŧ

į

|                          |   |   | UPED SAME   |   |  |   |                         |
|--------------------------|---|---|---|---|--|---|-------------------------|
| P                        | roject: NYSDEC-   | Poerion & Dui   | Aerola  | Siter   | Criktal (1   | I ADRIA   |                         |
| P                        | roject Number: 36   | 2052036   | (10011013   | Date:   | January 20   | 5,2006  | -                       |
|                          |   | -   |   | Time: Sta   | art: 1550  | _ End: _1710  |                         |
| S                        | ample Location ID: 6 W  | -009  |   | Signature   | of Sampler:  |   | — <u> </u>              |
|                          | Well Depth Ft.  | Measured<br>Historical  | Top of Well<br>Top of Protect<br>Casing   | Well Rise<br>ctive (from gro  | er Stick-up Ft.<br>und)  | ProtectiveF<br>Casing/Well Difference   | it.<br>ce               |
| er Level/Well Data       | Depth to Water Ft.  | Well Material:<br>PVC<br>SS   | Well Locked2<br>Yes<br>No   | et Psh<br>Gewell Dia.   | 2 inch<br>2 inch<br>4 inch<br>6 inch   | ProtectiveI<br>Casing<br>Water Level Equip. Us<br>Elect. Cond. Prol<br>Float Activated<br>Press. Transduc | Ft.<br>sed:<br>be<br>er |
| Wate                     | Height of Water Column  | 16 Gal/Ft. (2 in.)<br>65 Gal/Ft. (4 in.)<br>1.5 Gal/Ft. (6 in.)<br>Gal/Ft. (in.   | ) = [   | Gal/Vol.<br>Total Gal Purged  | Well Integrity:<br>Prot. Casing Secure<br>Concrete Collar Inta<br>Other  | Yes No.   | ><br>                   |
| ation                    | Purging/S   | Sampling Equipment U  | sed :   |   | Decontaminat   | lion Fluids Used :  |                         |
| Equipment Documen        | Purging Sampling  | Peristaltic Pump<br>Submersible Pump<br>Bailer<br>PVC/Silicon Tubing<br>Teflon/Silicon Tubing<br>Airlitt<br>Hand Pump<br>In-line Filter<br>Press/Vac Filter | Equipment ID  |   | ( ✓ All That Apply at I<br>Methanol (<br>25% Metha<br>Deionized N<br>Liquinox Sc<br>Hexane<br>HNO <sub>3</sub> /D.1.<br>Potable Wa<br>None | Docation)<br>100%)<br>nol/75% ASTM Type II w<br>Nater<br>Solution<br>Water Solution<br>Nater              | rater                   |
| Data                     | PID: Ambient Air 40   | _ppm_Well Mouth   | 2_ppm Purge [   | Data Collected  | San/ipi<br>Jri-lineTu<br>In ContainerCo  | e Observations:<br>IrbidClearC<br>ploredOdor  | Cloudy                  |
| Field Analysis D         | Tible Purge Data<br>Temperature, Deg. C<br>pH, units<br>Specific Conductivity (μ<br>Turbidity (NTUS)<br>Oxidation - Reduction, +<br>Dissolved Oxygen, ppm | @ <u> 626</u><br><u> 2-3</u><br><u>7-2</u><br>mhos/cm) <u>0.92</u><br>->[@<br>->[@<br>->[0.1]   | _ 61. @ 1650<br>8.9<br>1.46<br>71600<br>71600<br>85                                     | _çal. @   | Gal. @<br>   | Gal. @(   | 3al.                    |
|                          | Depth:  | 20-24'  | 12-11   | 1   |  |   |                         |
| <b>,</b>                 | Analytical Parameter  | ✓ If Sample Pi<br>Collected   | reservation<br>Method   | Volume<br>Required  | Sample Bottle  | ILot Nos.   |                         |
| quired at this Location) | VOCs<br>SVOCs<br>Metals<br>Cyanide<br>Nitrate/Sulfate<br>Nitrate/Phosphate<br>Pest/PCB<br>TPH<br>TOC  |   | 1°C<br>1°C<br>1N0, 4°C<br>1aOH, 4°C<br>1 S0, 4°C<br>1 S0, 4°C<br>1 S0, 4°C<br>1 S0, 4°C | 2x40 ml<br>2x1 liter AG<br>1x1 liter P<br>1x500mLP<br>1x1 liter P<br>1x1 liter P<br>3x1 liter AG<br>2x1 liter AG<br>1x1 liter P | CRGWO  | 0902201 XK (~<br>09014 01 XA @ 1  | <u>763</u> 5            |
|                          | -Repair Cheen on  | Durte wate  | <u>114</u> '.   | GROUND  | WATER SAMPL  | FIGURE 4<br>E DATA RECOR  | 1-1<br>RD               |
| 0401                     | 4D122   |   | NYSDE   | QUALITY   | ASSURANCE  | PROGRAM PL  | AN<br>ces—              |

ļ

į,









| -                    | oject: <u>MSDE</u>  | C-K In  | n 8 Dry clea   | mens Site:  | hrs: Clea   | mers  |
|----------------------|---|---|--|---|---|---|
| Pr                   | oject Number:   | 561205303   | 6  | Date:   | anian 23  | 2006<br>End: 1575   |
| \$                   | mple Location ID:   |   |  |   | t: 1915   |   |
|                      |   | 14-10/14  |  |   | JI Samplet  |   |
| র -                  | Well Depth  | _Ft Measure<br>Historica  | d Top of N<br>N Top of N<br>Casing   | Vell Well Riser<br>Protective (from grou  | Stick-up Ft.<br>nd)<br>Drabe  | Protective Ft.<br>Casing/Well Difference<br>Protective Ft.  |
| IPAN INA MANA        | Depth to Water  | _Ft. Well Material:<br>PVC<br>SS  | Well Locked?<br>Yes  | Grap Well Dia.  | 2 inch<br>4 inch<br>6 inch  | Casing<br>Water Level Equip. Used:<br>Elect. Cond. Probe<br>Float Activated<br>Press. Transducer          |
|                      | Height of Water Column  | 16 Gal/Ft.<br>65 Gal/Ft.<br>1.5 Gal/Ft.<br>Gal/Ft.  | (2 in.)<br>(4 in.) = [<br>(6 in.)<br>(in.)   | Gal/Vol.<br>Total Gal Purged  | Well Integrity:<br>Prot. Casing Secure<br>Concrete Collar Inta<br>Other   | Yes No  |
| non                  | Purgir  | ng/Sampling Equips  | nent Used :  |   | Decontaminat  | lon Fluids Used:  |
| Equipment Docume     | Purging Samph<br>   | ng<br>Peristaltic Pum<br>Submersible P<br>Bailer<br>PVC/Silicon Tu<br>Teflor/Silicon T<br>Airlift<br>Hand Pump<br>In-line Filter<br>Press/Vac Filte | Equipment  |   | ( ✓ All That Apply at L<br>Methanol (1<br>25% Metha<br>Deionized V<br>Liquinox So<br>Hexane<br>HNO <sub>3</sub> /D.I. V<br>Potable Wa<br>None | ocation)<br>00%)<br>nol/75% ASTM Type II water<br>Vater<br>Hution<br>Water Solution<br>ter<br>Post TUbiry |
| ala                  | PID: Ambient Air <u>41</u>  | .O ppm Well Mou   | th <u>LloO ppm</u> Pu  | Irge Data Collected   | Sampli<br>_In-lineTu<br>∠in ContainerCo   | e Observations:<br>rbidClearCloudy<br>ploredOdor  |
|                      | Ting Purge Date<br>Temperature, Deg. (<br>pH, units<br>Specific Conductivity<br>Turbidity (NTUS)<br>Oxidation - Reductio<br>Dissolved Oxygen, p | a @4<br>C7<br>y (μπhos/cm)7<br>op, +/- mv7<br>opm0  | 58 get @ 15<br>7 17<br>17<br>17<br>17<br>17<br>17<br>17<br>17<br>17<br>17<br>17<br>17<br>17<br>1                     | 15 gal. @<br>9<br>7.5<br>41<br>000<br>20<br>1   | Gal. @<br><br>  | Gal. @Gal.  |
|                      | Deptn   | 28'   | +032' 201  | 024'  |   | na an a  |
|                      | Analytical Parameter  | ✓ If Sample<br>Collected  | Preservation<br>Method   | Volume<br>Required  | Sample Bottle   | ILot Nos.   |
| ed at this Location) |   |   | 4°C HC<br>4°C<br>HN0, 4°C<br>NaOH, 4°C<br>H S0, 4°C<br>H S0, 4°C<br>H S0, 4°C<br>H S0, 4°C<br>H S0, 4°C<br>H S0, 4°C | 2x40 ml<br>2x1 liter AG<br>1x1 liter P<br>1x500mLP<br>1x1 liter P<br>1x1 liter P<br>3x1 liter AG<br>2x1 liter AG<br>1x1 liter P | CRGWC   | 21403001XXC-150   |
| adulir               | Notes: 2 Dept   | NSC 30'   | and. 22'   |   | ••••••••••••••••••••••••••••••••••••••  |   |

t, i

į.

|  | national Alexandres De   | 12 - 2224   | Dry Chean  | MM Site:   | CAPSTAL CLE  | in ners  |
|--|--|---|--|--|--|--|
| 1  | roject Number: 36  | 1209 2036   | ·  | Date:  | anvary 13,2  |  |
|  |  |   |  | lime: Sta  | art:   | End: <u>1919</u>   |
| S  | ample Location ID:   | W- 1151   |  | Signature  | of Sampler:  |  |
|  | Well Depth   | Ft Measured<br>Historical   | Top of<br>Top of<br>Casin  | Well Well Rise<br>Protective (from gro   | er Stick-up Ft.<br>und)  | ProtectiveFt.<br>Casing/Well Difference  |
| evel/Well Data   | Depth to Water   | Ft. Well Material:<br>PVC<br>SS   | Weil Locket  | when Grad  | 2 inch<br>4 inch<br>6 inch   | ProtectiveFt.<br>Casing<br>Water Level Equip. Used<br>Elect. Cond. Probe<br>Float Activated<br>Press. Transducer   |
| Water I  | Height of Water Column<br>Ft.  | X16 Gal/Ft. (2<br>X65 Gal/Ft. (4<br>1.5 Gal/Ft. (6<br>Gal/Ft. (   | 2 in.)<br>in.) = [   | Gai/Vol.   | Well Integrity:<br>Prot. Casing Secure<br>Concrete Collar Intac<br>Other   | Yes No   |
| tion   | Purging  | /Sampling Equipmen  | nt Used :  |  | Decontaminati  | on Fluids Used:  |
| Equipment Documer  | Purging Samplin  | Peristaltic Pump<br>Submersible Pum<br>Bailer<br>PVC/Silicon Tubin<br>Teflor/Silicon Tub<br>Airlitt<br>Hand Pump<br>In-line Filter<br>Press/Vac Filter  | Equipmer   | it ID  | ( ✓ All That Apply at La<br>Methanol (11<br>25% Methar<br>Deionized W<br>Liquinox Sol<br>Hexane<br>MNO <sub>3</sub> /D.I. V<br>Potable Wat<br>None | ocation)<br>00%)<br>hol/75% ASTM Type II wate<br>/ater<br>ution<br>Vater Solution<br>er  |
|  |  |   |  |  |  |  |
| ls Data  | PID: Ambient Air <u>L1-6</u><br>Tike Purge Data  | ppm Well Mouth  | <u>C1.0 ppm</u> P<br><u>9 cat. @ 13</u>  | urge Data Collected  | Sample<br>_In-line/fur<br>_In ContainerCol<br>   | o Observations:<br>bidClearClo<br>loredOdor<br>Gal. @ Gal  |
| Field Analysis Data  | PID: Ambient Air <u>L1-6</u><br>Tild Purge Data<br>Temperature, Deg. C<br>pH, units<br>Specific Conductivity<br>Turbidity (NTUS)<br>Oxidation - Reduction<br>Dissolved Oxygen, pp  | ррт Well Mouth<br>@ <u>123</u><br>[]<br>(µmhos/cm) <u>1.44</u><br>7.100<br>т. 42<br>7.100<br>т. 42<br>7.100<br>т. 42<br>7.100<br>т. 42<br>7.100<br>т. 44<br>7.100<br>т. 45<br>т. | <u>     C1.0 ppm</u> P <u>         Ppm</u> <u>         P</u> P | Purge Data Collected<br>25 5 Gat. @<br>1 √<br>26 5 Gat. @<br>27 6 0<br>27 7 0<br>2 | Sample<br>_In-line _I/Tur<br>_In ContainerCol<br>Gal. @  | Observations:<br>bidClearClo<br>loredOdor<br>Gal. @Gal<br>   |
| Fleid Analysis Data  | PID: Ambient Air <u>L1-6</u><br>Till Purge Data<br>Temperature, Deg. C<br>pH, units<br>Specific Conductivity<br>Turbidity (NTUS)<br>Oxidation - Reduction<br>Dissolved Oxygen, pp  | ррт Well Mouth<br>@223<br>[]<br>(µmhos/cm)<br>, +/- mv<br>со.[<br>27-   | 21.0 ppm     P       1     Gat. @ [2]      2     (      2     (      2     (      2     (      2     (      2     (      2     (      2     (      2     (      2     (      2    2      2    2      2    2      2    2      2    2      2    2      2    2  | Purge Data Collected<br>255 Gat. @<br>1.1<br>255 Gat. @<br>1.1<br>255 Gat. @<br>1.1<br>255 Gat. @<br>255 Gat. @  | Sample   | Observations:           rbid        Clear        Clo           lored        Odor          Gal.         @        Gal          Gal.        Gal        Gal  |
| Field Analysis Data  | PID: Ambient Air <u>21-6</u><br>Till Purge Data<br>Temperature, Deg. C<br>pH, units<br>Specific Conductivity<br>Turbidity (NTUS)<br>Oxidation - Reduction<br>Dissolved Oxygen, pp<br><u>De PM</u><br>Analytical Parameter  | ррт Well Mouth<br>@3<br>(µmhos/cm)4<br>(µmhos/cm)4<br>/7<br>/ If Sample<br>Collected  | <u>C1.0 ppm</u> P<br><u>9</u> cat. @ 12<br>2 (<br><u>4</u> -7<br><u>10</u> 21<br><u>0</u> 21<br><u>0</u> 21<br>Preservation<br>Method  | urge Data Collected           v  | Sample<br>_In-line _I/Tur<br>_In ContainerCol<br>Gal. @<br><br><br><br>Sample Bottle I   | Observations:<br>'bidClearClo<br>loredOdorGal. @Gal |
| ents Field Analysis Data   | PID: Ambient Air <u>L1-6</u><br>Tild Purge Data<br>Temperature, Deg. C<br>pH, units<br>Specific Conductivity<br>Turbidity (NTUS)<br>Oxidation - Reduction<br>Dissolved Oxygen, pp<br><u>De PM</u><br>Analytical Parameter  | ррт Well Mouth<br>@_[2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]3<br>[]2]          | Cl.0_ppm         P           9         Cat.         @           20         Data         Cat.           23         Cat.         Qat.           Preservation         Method         Qat.   | Purge Data Collected<br>PS G Gat. @<br>I. {<br>S<br>Volume<br>Required   | Sample   | Observations:           rbid        Clear         _Clo           lored         _Odor          Gal.         @        Gal  |
| enter of the second secon | PID: Ambient Air <u>41-6</u><br>Till Purge Data<br>Temperature, Deg. C<br>pH, units<br>Specific Conductivity<br>Turbidity (NTUS)<br>Oxidation - Reduction<br>Dissolved Oxygen, pp<br><u>De PM</u><br>Analytical Parameter<br><u>VOCS</u><br>SVOCS<br>Metals<br>Cyanide<br>Nitrate/Phosphate<br>Pest/PCB<br>TPH<br>TOC  | ррт Well Mouth<br>@23<br>[2]<br>[2]<br>[2]<br>[2]<br>[2]<br>[2]<br>[2]<br>[2]   | C1.0 ppm     P       9     Cat. @ [2]      2     (1) </td <td>Volume<br/>Required<br/>2x40 ml<br/>2x40 ml<br/>2x1 liter AG<br/>1x1 liter P<br/>1x1 liter AG<br/>2x1 liter AG<br/>1x1 liter P</td> <td>Sample<br/>In ContainerCol<br/>Gal. @<br/><br/>Sample Bottle I<br/><br/><br/><br/><br/></td> <td>bobservations:<br/>ThidClearClou<br/>loredOdor<br/>Gal. @Gal<br/><br/><br/>Lot Nos.<br/>5 07501 KKC1</td>   | Volume<br>Required<br>2x40 ml<br>2x40 ml<br>2x1 liter AG<br>1x1 liter P<br>1x1 liter AG<br>2x1 liter AG<br>1x1 liter P   | Sample<br>In ContainerCol<br>Gal. @<br><br>Sample Bottle I<br><br><br><br><br>   | bobservations:<br>ThidClearClou<br>loredOdor<br>Gal. @Gal<br><br><br>Lot Nos.<br>5 07501 KKC1  |
| 3 Collection Requirements<br>If Required at this Location)   | PID: Ambient Air <u>L1-6</u><br>Tild Purge Data<br>Temperature, Deg. C<br>pH, units<br>Specific Conductivity<br>Turbidity (NTUS)<br>Oxidation - Reduction<br>Dissolved Oxygen, pp<br>De PM<br>Analytical Parameter<br>VOCs<br>SVOCs<br>VOCs<br>SVOCs<br>Metals<br>Cyanide<br>Nitrate/Sulfate<br>Nitrate/Phosphate<br>Pest/PCB<br>TPH<br>TOC<br>Notes: 2 Deot | $(\mu mhos/cm)$ $273$<br>$(\mu mhos/cm)$ $123$<br>$(\mu mhos/cm)$ $1.44$<br>7100<br>7100<br>275<br>15 Sample<br>Collected<br>10<br>10<br>275<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10  | $\begin{array}{c c} C1.0 & ppm & P \\ \hline \\$   | Volume<br>Required<br>2x40 ml<br>2x1 liter P<br>1x1 liter P   | Sample<br>_In-line/Tur<br>In ContainerCol<br>Gal. @<br><br>Sample Bottle I<br><br><br>   | bobservations:<br>rbidClearClou<br>loredOdor<br>Gal. @Gal<br><br>Lot Nos.<br>502501KKC1  |

3

à

| P                        |   | 21.2.0   | VI a Uny Cie   | STIE:   | rysi cici   | 2 2 1 1  |
|--------------------------|---|--|--|---|---|--|
| 1                        | roject Number:  | 20120520   | 36   | Date:   | anvan 2   | 5,100 0<br>End: 1140   |
|                          | male Leasting ID.   |  |  |   | t Samplart  |  |
|                          |   | W-016  |  |   |   |  |
|                          | Well Depth  | _FtMeasu<br>Historic   | red Top of<br>cal Top of<br>Casin  | Well Well Riser<br>Protective (from grou  | Stick-up Ft.<br>nd) Dabe  | Protective Ft.<br>Casing/Well Difference   |
| evel/Well Data           | Depth to Water  | _Ft. Well Material<br>PVC<br>SS  | : Well Locked<br>Yes   | r: Graf Well Dia.   | 2 inch<br>4 inch<br>6 inch  | Water Level Equip. Used<br>Elect. Cond. Probe<br>Float Activated<br>Press. Transducer                        |
| Water L                  | Height of Water Colum   | 16 Gal/F<br>mX65 Gal/F<br>1.5 Gal/F<br>Gal/F   | t. (2 in.)<br>t. (4 in.) = [<br>t. (6 in.)<br>t. (in.)   | Gal/Vol.<br>Total Gal Purged  | Well Integrity:<br>Prot. Casing Secure<br>Concrete Collar Inta<br>Other   | Yes No   |
| lon                      | Purgi   | ng/Sampling Equip  | ment Used :  |   | Decontaminat  | ion Fluids Used :  |
| Equipment Docume         | Purging Sampt   | Ing<br>Peristaltic Pur<br>Submersible I<br>Bailer<br>PVC/Silicon T<br>Teflor/Silicon<br>Airlift<br>Hand Pump<br>In-line Filter<br>Press/Vac Filt | Equipmer   | nt ID   | ( All That Apply at I<br>Methanol ('<br>25% Metha<br>Deionized V<br>Liquinox Sc<br>Hexane<br>HNO <sub>3</sub> /D.I.<br>Potable Wa<br>None | Docation)<br>100%)<br>nol/75% ASTM Type II wate<br>Nater<br>Solution<br>Water Solution<br>Ler<br>Poss. TUBAT |
| Data                     | PID: Ambient Air  | _O_ppm Well Mo   | uth <u>L1.0</u> ppm F  | Purge Data Collected  | In-line<br>In ContainerCo   | e Observations:<br>IrbidClearCk<br>NoredOdor   |
| Field Analysis I         | Time Purge Da<br>Temperature, Deg.<br>pH, units<br>Specific Conductivit<br>Turbidity (NTUS)<br>Oxidation - Reductiv<br>Dissolved Oxygen | ta @<br>C<br>ty (μmhos/cm)<br>on, +/- mv   | 2.0<br>7.5<br>7.5<br>7.0<br>7.5<br>7.0<br>7.5<br>7.5<br>7.5<br>7.5<br>7.5<br>7.5<br>7.5<br>7.5 | <u>-38 gal.</u> @<br><u>-11.4</u><br><u>-5</u><br><u>-21000</u><br><u>-24</u><br><u>-64</u> | Gal. @<br><br>  | Gal. @Ga   |
|                          | Deetn   | ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~  | F. 19.6 19   | <u> </u>  |   |  |
| 67                       | Analytical Parameter  | ✓ If Sample<br>Collected   | Preservation<br>Method   | Volume<br>Required  | Sample Bottle   | ILot Nos.  |
| ient                     | vocs  |  | 4°C/HKV  | 2x40 mi   | CRGWO   | 1602101XX@12   |
| equirem<br>ocation)      | SVOCs<br>Metals<br>Cyanide<br>Nitrate/Sulfate   |  | 4°C<br>HN0,,4°C<br>NaOH,4°C<br>H S0_4°C  | 2x1 liter AG<br>1x1 liter P<br>1x500mLP   | CRGWC   | 16617 01 XA@12   |
| ction Re<br>ed at this l | Nitrate/Phosphate<br>Pest/PCB<br>TPH<br>TCC   |  | H <sup>2</sup> SO <sup>4</sup> ,4°C<br>4°C<br>H <sup>2</sup> SO <sup>4</sup> ,4°C              | 1x1 liter P<br>3x1 liter AG<br>2x1 liter AG   |   |  |
| e Colle<br>If Requir     | Notes: 2 Depth  | s:21 a   | n, 204, 14°C   | 1X1 liter P   |   |  |
| <u> </u>                 |   |  |  |   |   |  |

Į į

ł

į.

E . .

| P  | roject: <u>NSDEC</u>  | - Keym   | 8 Dry Cle  | mens Site:  | Chystal Cle   | aners  |
|--|---|--|--|---|---|--|
| P  |   |  | 2  | Date:<br>Time: St   | January 2   | 3,2006   |
| S  | ample Location ID:  | 1-017  |  | Signatur  | ant: 1005   |  |
|  |   |  |  |   |   |  |
| ata  | Well Depth Ft   | Measured<br>Historical   | Top of<br>Top of<br>Casing   | Well Well Ris<br>Protective (from gr  | er Stick-up Ft.<br>ound) Ft.  | Protective Ft.<br>Casing/Well Difference<br>Protective Ft.   |
| Level/Well D                               | Depth to Water Ft   | . Well Material:<br>PVC<br>SS  | Well Locked  | Grap Well Die   | 2 inch<br>2 inch<br>4 inch<br>6 inch  | Casing<br>Water Level Equip. Used:<br>Elect. Cond. Probe<br>Float Activated<br>Press. Transducer               |
| Water                                      | Height of Water Column<br>Ft.   | 16 Gal/Ft. (2<br>X65 Gal/Ft. (4<br>1.5 Gal/Ft. (6<br>Gal/Ft. (   | in.) = [<br>in.) =<br>in.)   | Gai/Vol.  | Well Integrity:<br>Prot. Casing Secure<br>Concrete Collar Inte<br>d Other   | Yes No<br>   |
| ation                                      | Purging/  | Sampling Equipmen  | t Used :   |   | <u>Decontamina</u>  | tion Fluids Used :   |
| Equipment Docum                            |   | Peristaltic Pump<br>Submersible Pumj<br>Bailer<br>PVC/Silicon Tubin<br>Teflon/Silicon Tub<br>Airlift<br>Hand Pump<br>In-line Filter<br>Press/Vac Filter  | Equipment  |   | All That Apply at<br>Methanol (<br>25% Metha<br>Deionized<br>Liquinox S<br>Hexane<br>HNO <sub>3</sub> /D.I.<br>Potable Wa<br>None<br> | Location)<br>100%)<br>unol/75% ASTM Type II water<br>Water<br>Julution<br>Water Solution<br>ater<br>PeS. TUbry |
| Field Analysis Data                        | PID: Ambient Air ().0<br>Time Purge Data<br>Temperature, Deg. C<br>pH, units<br>Specific Conductivity (µ<br>Turbidity (NTUS)<br>Oxidation - Reduction, -<br>Dissolved Oxygen, ppr | ppm Well Mouth 4<br>@10<br>12.9<br>7.1<br>umhos/cm)5<br>+/- mv40<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>1_5<br>1 | <u>(1.0 ppm</u> Pi<br><u>Gat. @ 11</u><br><u>7</u><br><u>7</u><br><u>7</u><br><u>7</u><br><u>7</u><br><u>7</u><br><u>7</u><br><u>7</u><br><u>7</u><br><u>7</u> | Urge Data Collected<br>2  | Samp  | le Observations:<br>urbidClearCloudy<br>oloredOdor<br>Gal. @Gal.   |
|  | Analytical Parameter  | ✓ If Sample  | Preservation   | Volume  | Sample Bottle   | ILot Nos.  |
| <pre>/ If Required at this Location)</pre> |   | <br>   | 4°C <b>H</b> KU<br>4°C<br>HNO, 4°C<br>NaOH, 4°C<br>H SO, 4°C<br>H SO, 4°C<br>H SO, 4°C<br>H SO, 4°C<br>H SO, 4°C<br>H SO, 4°C                                  | 2x40 ml<br>2x1 liter AG<br>1x1 liter P<br>1x500mLP<br>1x1 liter P<br>1x1 liter P<br>3x1 liter AG<br>2x1 liter AG<br>1x1 liter P | <u>(</u> RGWOI-<br><u>C</u> RGWO  | 1020 DIXK@ 1115  |
| ت  |   |  |  |   | ~   |  |

ļ

į.

| Pr     | roject: $\underline{NTSDEC} - \underline{Region 8 Dry Cleaner}$<br>roject Number: $\underline{3612052036}$  | Date: January 23, 2006<br>Time: Start: 0950 End: 1085  |
|--------|---|--|
| Sa     | ample Location ID:  | Signature of Sampler:  |
| 210    | Well Depth Ft Measured Top of Well<br>Historical Top of Protect<br>Casing   | Well Riser Stick-up Ft. Protective Ft.<br>ive (from ground) Casing/Well Difference<br>Protective Ft.   |
|        | Depth to Water Ft. Well Material: Well Locked2;<br>PVCYes<br>SSNo fm/   | Well Dia.       2 inch       Water Level Equip. Used:        4 inch      Elect. Cond. Probe        6 inch      Float Activated   |
|        | Height of Water Column X16 Gal/Ft. (2 in.)<br>Ft1.5 Gal/Ft. (4 in.) =G<br>Ft1.5 Gal/Ft. (6 in.)<br>Gal/Ft. (To  | tal Gal Purged Other   |
| 5      | Purging/Sampling Equipment Used :   | Decontamination Fluids Used :  |
|        | (✓ If Used For) Purging Sampling Equipment ID Peristaltic Pump Submersible Pump Bailer PVC/Silicon Tubing Teflon/Silicon Tubing Airlift Hand Pump In-line Filter Press/Vac Filter   | ( ✓ All That Apply at Location)<br>Methanol (100%)<br>25% Methanol/75% ASTM Type II water<br>Deionized Water<br>Liquinox Solution<br>Hexane<br>HNO <sub>3</sub> /D.I. Water Solution<br>Potable Water<br>None  |
|        | PID: Ambient Air $4.0$ ppm Well Mouth $4.0$ ppm Purge Data<br>Tiwe Purge Data $@0.958$ cal. $@_000$ c<br>Temperature, Deg. C $(3.4$<br>pH, units<br>Specific Conductivity (µmhos/cm) $1.56$ $56$ $56$ $56$ $56$ $56$ $56$ $56$  | Sample Observations:<br>   |
| _      | Analytical Parameter It Sample Preservation<br>Collected Method I   | Volume Sample Bottle ILot Nos.<br>Required   |
|        | VOCs     4°C /HCl     22       SVOCs     4°C     21       Metals     4°C     11       Cyanide     NaOH,4°C     12       Nitrate/Sulfate     H S0, 4°C     15       Nitrate/Phosphate     H'SO, 4°C     15       Pest/PCB     4°C     33       TPH     H S0, 4°C     15       TOC     H S0, 4°C     15 | $(A0 \text{ ml})$ $(RSwool_{DODOOl_X/C} C)$ $(A1 \text{ liter } AG)$ $(RSwool_{DODOOOl_MS} C)$ $(Ioreg)$ $(A1 \text{ liter } P)$ $(RSwool_{OOS} col_M) C$ $(Ioreg)$ $(A1 \text{ liter } P)$ $(Iiter AG)$ $(Iiter AG)$ $(A1 \text{ liter } P)$ $(Iiter AG)$ $(Iiter P)$ <th< td=""></th<> |
| :<br>• | - Sup pump area in basenent of Gysta<br>- Vust in open water G  | FIGURE 4-1<br>ROUNDWATER SAMPLE DATA RECORD  |



r.

| FIELD DA                            | TA REC                 | ORD - LO                   | W FLOW   | GROUNDWA   | ATER S   | AMPLIN   | G   |  | sh               | eet 1 of 2  |
|-------------------------------------|------------------------|----------------------------|--|--|--|--|---|--|------------------|---|
| PROJECT                             | NYSDE                  | EC Region                  | 18 Clear   | vers   | ana ay ang   |  | e standard de la composition de la comp<br>Composition de la composition de la comp |  | JOB NU           | MBER 36120520761042   |
| LOCATION ID                         | Crustal                | Cleaner                    | S MW-  | 10 FIELD S   | AMPLE ID   | CRM  | WOODC   | 1301X  | R                | EVENT NO.   |
| ACTIVITY                            | START                  | 2:10 0                     | ND 13:3  |  | ETIME  | 1  | 3:25  | ÷  |                  | DATE 11-1-06  |
|                                     | L / PUMP S             | ETTINGS                    | MEASUF   |  |  |  |   |  | CACINIC          |   |
| NITIAL DEPTH<br>TO WATER            |                        | 1.91 fe                    |  | OF WELL RISER  | CASING   | PROTECTIV<br>CASING STI<br>(FROM GRO   |   | feet   | DIFFER           | ENCE feet   |
| FINAL DEPTH<br>TO WATER             | 1                      | 2.14 i                     | WELL D<br>eet (TOR)  |  | feet   | PID<br>AMBIENT AI  | R   | - ppmv   |                  |   |
| SCREEN LENG                         |                        | 10 f                       | PRESSU   |  | psi  | PID WELL<br>MOUTH  |   | ppmv   |                  | RITY: YES NO N/A  |
| TOTAL VOL.<br>PURGED<br>(purge volu | me (milliliters        | gallc<br>per minute) x tim | REFILL<br>ons SETTIN<br>e duration (minu   | G<br>tes) x 0.00026 gal/m  | illiliter)   | DISCHARGE<br>SETTING   |   | nato en Statuta en Statu<br>Statuta en Statuta en St   | LOCK             | $\begin{array}{cccccccccccccccccccccccccccccccccccc$  |
| PURGE DAT                           | <b>A</b>               | ******                     | Panal  | SPECIFIC   |  |  |   | 1  | PUMP             | I .   |
|                                     | DEPTH TO<br>NATER (ft) | PURGE<br>RATE (ml/m)       | TEMP.<br>(+/- deg. C)  | CONDUCTANCE<br>(mS/cm)   | pH<br>(units)  | DISS. O2<br>(mg/L)   | I URBIDITY  | (+/- mV)   | DEPTH (ft)       | COMMENTS  |
| 11:28 1                             | 2.33                   | 200                        | 15.9   | 0.90   | 7.2  | 8.30   | 674   | 20   |                  | ê   |
| 11:34                               | 12.58                  | 175                        | 15.6   | 3.07   | 7.0  | 11.5   | 552   | -50  |                  | air in tubing   |
| 11.39                               | 12.64                  | 175                        | 15.6   | 3.02   | 6.9  | 14.6   | 324   | -60  |                  | air in tubing   |
| 11:44                               | 12.60                  | 150                        | 15.S   | 3.71   | 6.8  | 13.5   | 298   | -58  |                  |   |
| 11:49                               | 12.60                  | 150                        | 15.S   | 2.11   | (0.)   | -7.41  | 102   | - 10   |                  |   |
| 11.54                               | 12.63                  | 150                        | 15.5   | 1.76   | 6.1  | 6.50   | 193   | - 12   |                  |   |
| 11.59                               | 12.61                  | 150                        | 15.5   | 1.61   | 6.6  | 9.51   | 83  | - 10   |                  |   |
| 12:04                               | 12.58                  | 150                        | 15.5   | 1.76   | 6.6  | 0,18   | 00  | - 65   |                  |   |
| 12.04                               | 12.64                  | 150                        | 15.5   | 1.00   | 6.5  | 14.11  | 70  | - 60   |                  | eunoe have fort   |
| 12.00                               | 12.61                  | 150                        | 15,0   | 1.55   | 6.6  | 6.17   | 50  | - 76   |                  | Possible  |
| 13.35                               | 12.67                  | 150                        | 15.3   | 1.61   | 616  | 6.02   | 40  | 1-78   |                  | Lowered 1   |
| 12:46                               | 10.06                  | 150                        | 15.3   | 1.30   | 6.0  | 51   | 142   | -82  |                  |   |
| FOLIPMENT                           | DOCUMEN                | TATION                     | 10.0   | 1,10   | 110.0  | 12.11  | 195   | 1 00   |                  | continued)  |
| TYPE OF                             | PUMP                   |                            | 1997 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 -<br>1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 -<br>1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 -  |  |  | n national<br>An the state   | Ē   | PE OF TUB  | NG               | command y   |
|                                     | CATED MARS             | CHALK BLADD                | ER 🔀   | OTHER perist   | attic-1  | Geopur   | nt k  |  | ISITY POLY       | ETHYLENE  |
|                                     | DEDICATED              | MARSCHALK B                |  |  | alexy and a second second second   |  |   | JOIHER_  |                  | na ana amin'ny tanàna mandritra dia kaominina dia kaominina dia kaominina minina minina minina minina minina mi<br>Ny INSEE dia kaominina mandritra dia kaominina dia kaominina dia kaominina minina minina minina minina minina mi |
| ANALYTICA                           | LPARAMET               | ERS                        | ME   | THOD   | PRE  | SERVATION  | VOLUME  | SAME   | PLE              |   |
| CONTROL                             | L NUMBER               |                            | <br>and extended field and the second as the off<br>and the provided field and the second |  | n<br>Serie reactor<br>Haine al reactor   |  | 3 X 40 MI   |  | <u>יובט</u><br>ר |   |
|                                     | - 25 mi Purge          | (IOW CONC.)                |  | OLMO4.2  | n seine H  | CL/4 DEG. C  | 3 X 40 ML   |  |                  | XX.//   |
| Svoc                                | S                      |                            | Server a server  | OLMO4.2  | 4  | DEG. C   | 2 X 1 L AC  | s _  | ]                | <u> </u>  |
| PEST                                | ICIDES                 |                            |  | OLMO4.2  | 4<br>1.000 - 1.000 - 1.000   | DEG. C   | 2 X 1 L AC  |  | ]                |   |
|                                     | ALS<br>SANESE / IRC    | )N -                       | S S  | ILM04.2<br>W846 6010   | n<br>Maria de H  | NO3 to pH <2   | 1 X 500 M   |  | יייייי<br>ר      | *****   |
|                                     | ATE / CHLOR            | IDE / ALKALINI             | USEP.  | A 375.4 / 325.3 / 310  | ).1 4  | DEG. C   | 1x1LP   | and and a set of the s | jer I            | e Agenerica y provinciale de la companya de la comp   |
| NO2 -                               | NO3                    |                            | USEF   | PA 353.2 / 354.1   | Н  | 2SO4 to pH <   | 2 1 X 500 M   |  | ]                |   |
|                                     |                        |                            | U  | SEPA 415.1   | <b>H</b>   | 2SO4 to pH <   | 2 1 X 250 M   |  |                  |   |
|                                     | IANE / ETHAM           | NE / ETHYLENE              | EP   |  | <b>H</b>   | CL / 4 DEG. (  | , 3 X 40 ML   | · · · · · · · · · ·  |                  |   |
|                                     |                        | ~                          |  | n an   |  | LOCATIO  | NNOTES  |  | معين البي<br>    |   |
| PURGE WAT                           | ER<br>ZED YES          |                            | herosene   | lodor  |  | 1 <sup>N</sup>   |   |  |                  |   |
| NOTES:                              |                        |                            |  | n na na sangar sanga<br>Sangar sangar | The second secon | <ul> <li>Antipation of the second second</li></ul> | 1 4   | 0  | Ca               | sta   |
|                                     |                        |                            |  | lan di<br>Anna dia Manjara Anna  | and a constant<br>Constant of the  | 1947 (MARA)  | MWr_  | . ^  | <u> </u>         | tanana   |
|                                     | × 0                    |                            | n y construction of the second s   |  | a a la construcción de la construcc  | y - Lagagaggan Arthurs (1999)<br>g - Constanting (1997)  | a anna an anna an an an an an an an an a  | wow  |                  |   |
| SIGNATURE                           | Jau                    | ra Sm                      | ith  | and the second   | A Anna Anna Anna Anna Anna Anna Anna An  |  |   | Jr.  |                  |   |
| CHECKED                             | Y                      |                            |  |  |  | n a ser<br>Antonio de la composición<br>A Composición de la c  | 1 + 1   | N. Pul   | teney            |   |
| L                                   |                        |                            |  | The Control of the   | COLOR S  | Spat Aller of  |   | anning and a start of the second second  |                  | )   |

lowflowform.xls

and the second second

| FIELD D                              | ATA R  | ECORD - LO   | OW FLOV                                  | GROUN  | DWATER  | SAMPLI                                      | NG                          |   | Sh                           | eet 20f7   |
|--------------------------------------|--|--|--|--|---|---|-----------------------------|---|------------------------------|--|
|                                      | NUC I  | TC. Pario  | 08 Cloan                                 | PIA  |   |   |                             |   | JOB NU                       | MBER 3612052036104.2                             |
|                                      | 10100  | tol mu   | -10 cult                                 |  |   | 10 CIPI                                     | MINDIDO                     | 120/14                                      |                              |  |
| LOCATION ID                          | , ay   | un nu  | 12                                       | 20   |   |   | 1:25                        | 1001  | -                            | DATE 11-1-06                                     |
| ACTIVITY                             | START  |  | END 70                                   |  |   |   | DICS                        |   |                              |  |
| WATER LEV<br>INITIAL DEPT<br>TO WATE | /EL / PUN<br>R   | IP SETTINGS  | feet MEAS                                | UREMENT POIL<br>OP OF WELL R<br>OP OF PROTEC | NT<br>ISER<br>CTIVE CASING  | PROTECTI<br>CASING ST<br>(FROM GR           |                             | feet  | CASING                       | S / WELL feet                                    |
| FINAL DEPT<br>TO WATE                | R  | 12.14  | feet (TOR                                |  | feet  | PID<br>AMBIENT A                            |                             | ppmv  | DIAMET                       |  |
| SCREEN LEN                           | NGTH   | 10   | feet TO P                                | SURE   | psi   | PID WELL<br>MOUTH                           |                             | ppmv  | INTEGF<br>C<br>CASI          | RITY: YES NO N/A<br>CAP <u>-</u><br>ING <u>-</u> |
| TOTAL VO<br>PURGE<br>(purge vo       | DL.<br>ED<br>Dume (millili   | -4.7 ga  | Illons REFIL<br>MICONS SETT              | L<br>ING<br>nutes) x 0.00020                 | 6 gal/milliliter)   | DISCHARG<br>SETTING                         | ε <u> </u>                  |   | LOCK                         |  |
| PURGE DA                             | TA ( ( ( ))<br>DEPTH T<br>WATER (  | 1 tipuled fri<br>O PURGE<br>ft) RATE (ml/m)  | m pg. ()<br>TEMP.<br>(+/- deg. C)        | SPECIFI<br>CONDUCTA<br>(mS/cm                | C<br>ANCE pH  | DISS. O2<br>s) (mg/L)                       | TURBIDITY<br>(NTU)          | REDOX<br>(+/- mV)                           | PUMP<br>INTAKE<br>DEPTH (ft) | COMMENTS   |
| 12:50                                | 12.60  | 8 150  | 15.3                                     | 1.68   | 6.  | 65.71                                       | 93                          | -80   |                              |  |
| 12:55                                | 12.6   | 5 150  | 15.3                                     | 1.41   | 6.  | 1 5.1                                       | 71                          | -90   |                              | Strpitorecal U-22                                |
| 13.10                                | 12.20  | $\frac{150}{2}$  | 1.15.0                                   | 7.03   | 6.  | $\frac{1}{1}$                               | 300                         | - 20  |                              |  |
| 13.10                                | 10.9   | <u> 100</u>  | 10.4                                     | 6.10   | 0 6   | $\frac{q}{s}$                               | 130                         | -75   |                              |  |
| 13:25                                | 10.0   | Sampl  | 4.2.1                                    | 7.5  | 0 6.  | 5 7.1                                       |                             |   |                              |  |
|                                      |  |  |  |  |   | · · · · · ·                                 |                             |   |                              |  |
|                                      |  | <u> </u>   |  |  |   |   | <u></u>                     |   |                              |  |
|                                      |  | <u> </u>   | 4  |  | - <u>1</u> 272  |   |                             |   |                              |  |
|                                      |  | $\rightarrow \rightarrow \rightarrow \rightarrow$  |  |  |   |   |                             |   |                              |  |
|                                      |  | $\rightarrow$  | K/                                       |  |   | <u></u>                                     |                             |   |                              |  |
|                                      | in a state that the second | `  | Ř  |  | an Arthurst<br>Anna Arthurst  |   |                             |   | -                            |  |
|                                      |  | MENTATION  |  |  | pr. pus   | np. (+                                      |                             |   | NG<br>SITY POLY              | 'ETHYLENE  |
|                                      | N-DEDICA   | TED MARSCHALK  | BLADDER                                  |  | J. A. J. A.   | 1   |                             | OTHER_                                      |                              |  |
| ANALYTIC                             | AL PARA  | METERS   |  | ACTHOD                                       | D   | RESERVATION                                 |                             |   | 1 6                          |  |
| CONTR                                | OL NUMBE   | ER   |  | NUMBER                                       | nasi salaya.  | METHOD                                      | REQUIRE                     |   | TED                          | SAMPLE BOTTLE ID LETTERS                         |
|                                      | Cs - 25 ml P   | Purge (low conc.)  |  | OLCO2.1                                      |   | HCL / 4 DEG.                                | C 3 X 40 ML                 | - L   | <br>P                        | XXI I  |
|                                      | DCs  | nge  |  | OLMO4.2                                      |   | 4 DEG. C                                    | 2 X 1 L A                   | G   | ]                            |  |
| PES                                  | STICIDES   |  |  | OLMO4.2                                      |   | 4 DEG. C                                    | 2 X 1 L A                   | G _   | ]                            |  |
|                                      | TALS   |  | n an | ILM04.2<br>SW846 6010                        |   | HNO3 to pH <                                | 2 1 X 500 N<br>2 1 X 500 N  |   | ]                            |  |
|                                      | FATE / CH  | ILORIDE / ALKALIN  | ITY USI                                  | EPA 375.4 / 325.                             | 3/310.1   | 4 DEG. C                                    | 1 x 1 L P                   |   | ] _                          |  |
|                                      | 2 - NO3  |  | US                                       | EPA 353.2 / 354                              | u cardinario di<br>Manazia  | H2SO4 to pH                                 | <2 1 X 500 N                |   | ]                            |  |
|                                      | C<br>THANE / ET  |  | e de la constante<br>IE                  | USEPA 415.1                                  |   | H2SU4 to pH                                 | <2 1 X 250 M<br>C 3 X 40 MI |   |                              |  |
|                                      | HER  | · · · / · · · · · · · · · · · · · · · ·  |  |  | na fa da lí<br>Talaine se ar  |   |                             |   | ] _                          |  |
| PURGE O                              | BSERVAT  | IONS   |  |  | ala and a second se<br>Second second | LOCATI                                      | ON NOTES                    |   |                              |  |
|                                      |  | YES NO   | - Kevos                                  | ne odo                                       | r Fbs   | And and a second                            |                             |   |                              |  |
| NOTES:                               | • ماهدان الم   |  |  |  |   |   | See p                       | revier.                                     | S PG                         | R  |
|                                      |  |  |  | 11-16  | ar ing Kang   | an an ann an Anna An<br>An Anna Anna Anna A |                             | n - San |                              | P  |
|                                      | _   .  | Cmith  | 6 3                                      | 1211.0                                       |   |   |                             |   |                              |  |
| SIGNATUR                             | E:   | Contraction of the Contraction o |  |  | an a she bulan 1998.<br>Marina a sa Ariya a sa  |   |                             |   |                              |  |
| UNEUKER                              | ът:  |  |  |  |   |   |                             |   |                              |  |
| PROJECT NYSDEC Region & Cleaners JOB NUMBER 36   | NO skhip                         |
|--|----------------------------------|
|  | 12052030144                      |
| LOCATION ID CRYSTAL CLEADERS NW-1 FIELD SAMPLE ID CRMWOOL OZDOLXX. EVENTN  | 10. XZ                           |
| ACTIVITY START 09:48 END 10:45 SAMPLE TIME 10:25 DATE 11-  | -1-06                            |
| WATER LEVEL / PUMP SETTINGS MEASUREMENT POINT  |                                  |
| INITIAL DEPTH<br>TO WATER 15.28 feet TOP OF WELL RISER PROTECTIVE CASING STICKUP DIFFERENCE  | feet                             |
| FINAL DEPTH TO WATER 15.31 feet (TOR) PID AMBIENT AIR ppmv WELL INTERGRITY   | inches                           |
| SCREEN LENGTH 10 feet PRESSURE PID WELL INTEGRITY: YES CAP V   | NO N/A                           |
| TOTAL VOL.<br>PURGED gallons SETTING DISCHARGE LOCKED COLLAR V<br>(purge volume (milliliters per minute) x time duration (minutes) x 0.00026 gal/milliliter)   |                                  |
| PURGE DATA SPECIFIC . PUMP   |                                  |
| DEPTH TO         PURGE         TEMP.         CONDUCTANCE         pH         DISS. 02         TURBIDITY         REDOX         INTAKE           TIME         WATER (ft)         RATE (ml/m)         (+/- deg. C)         (mS/cm)         (units)         (mg/L)         (NTU)         (+/- mV)         DEPTH (ft)         CON  | MENTS                            |
| 9:55 15.61 250 14.2 0.90 8.1 6.40 456 195 20.6 456 NT  | 11 on U-22                       |
| 10:02 15.57 250 14.2 0.90 8.1 5.98 189 185   |                                  |
| 10:08 15,50 175 14.2 0.90 8, 5.84 111 178  |                                  |
| 10:13 15.50 175 4:2 0.40 8: 5.71 205 116   |                                  |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$  |                                  |
| 10.22 $15.50$ $1.50$ $14.2$ $0.10$ $5.10$ $5.10$ $5.10$  | 94<br>1977 - 1977<br>1977 - 1977 |
| 10:25 10:00 100 Sempte   |                                  |
|  |                                  |
|  |                                  |
| 1.1.1.2. Distribution of the sector of the s |                                  |
|  |                                  |
|  |                                  |
| EQUIPMENT DOCUMENTATION TYPE OF TUBING   |                                  |
|  | ·                                |
| NON-DEDICATED MARSCHALK BLADDER  |                                  |
| ANALYTICAL PARAMETERS<br>METHOD PRESERVATION VOLUME SAMPLE   |                                  |
| CONTROL NUMBER NUMBER NUMBER RETHOD REQUIRED COLLECTED SAMPLE BOT  | TLE ID LETTERS                   |
| VOCs - 25 ml Purge (low conc.)         OLCO2.1         HCL / 4 DEG. C         3 X 40 ML        /   | MD                               |
| SVOCs         OLMO4.2         4 DEG. C         2 X 1 L AG        /   |                                  |
| PESTICIDES         OLMO4.2         4 DEG. C         2 X 1 L AG        /  |                                  |
| METALS         ILM04.2         HNO3 to pH <2         1 X 500 ML P  |                                  |
| SW846 6010 ANOS ID PT 2 TX 500 ME P  |                                  |
| USEPA 353.2 / 354.1 H2SO4 to pH <2 1 X 500 ML P  |                                  |
| TOC         USEPA 415.1         H2SO4 to pH <2         1 X 250 ML AG   |                                  |
| METHANE / ETHYLENE EPA Region 1 HCL / 4 DEG. C 3 X 40 ML   |                                  |
|  |                                  |
| PURGE OBSERVATIONS   | . N                              |
| CONTAINERIZED YES NO <u>clear</u>  | T                                |
| NOTES:   | 9                                |
| Filled & MW-1 3  |                                  |
| l litt   |                                  |
| SIGNATURE: ACULACE ADMITCH   |                                  |
| CHECKED BY   |                                  |

| FIELD                       | DATA REC                          | CORD - LC                     | W FLOW  | GROUNDW                    | ATER S                             | AMPLIN  | IG   | kening merulakan melanan pangan pangan pangan kan  |                      | MILING MILING MILING MILING MILING AND |  |
|-----------------------------|-----------------------------------|-------------------------------|---|----------------------------|------------------------------------|---|--|--|----------------------|--|--|
| PROJECT                     | NYSDI                             | EC Region                     | n 8 Clea  | rers                       |                                    |   |  |  | JOB NU               | MBER 3612052031 142  |  |
| LOCATION                    | D Crysta                          | al clear                      | hers MW   | -5 FIELD S                 | AMPLE ID                           | CRM   | WOUS OF  | 201XX  | <u> </u>             | EVENT NO. 2  |  |
| ACTIVITY                    | START O                           | 750                           | END ~915  | SAMPLE                     | ETIME                              | 09  | 1100   |  |                      | DATE 11-1-06   |  |
| WATER LE                    | EVEL / PUMP                       | SETTINGS                      | MEASU   |                            |                                    | PROTECTIN   |  |  | O A O INIC           |  |  |
| INITIAL DEP<br>TO WAT       | TH G                              | 1.15                          | TO  | P OF PROTECTIVE            | CASING                             | NG CASING STICKUP CASING / WELL DIFFERENCE feet   |  |  |                      |  |  |
| FINAL DEP<br>TO WAT         | TH C                              | 7.16                          | WELL D<br>reet (TOR)                                  |                            | feet                               | PID<br>AMBIENT A  |  | ppmv   | DIAMET               | TER inches   |  |
| SCREEN LE                   | INGTH                             | 10                            | PRESS   | JRE                        | psi                                | BI PID WELL PPMV WELL INTEGRITY: YES NO N/A   |  |  |                      |  |  |
| TOTAL V<br>PURG<br>(purge v | OL.<br>ED ~<br>olume (milliliters | 2.5 galle                     | REFILL<br>ons SETTIN<br>ne duration (minu             | G<br>(tes) x 0.00026 gal/m | illiliter)                         | DISCHARGI<br>SETTING  | E  |  | CASI<br>LOCK<br>COLL | ING Z  |  |
| PURGE DA                    | ATA                               |                               | Card Marcan & Handhalton and All All All and an and   | SPECIFIC                   | de anna anna a sa an ta chlainn an | nan an an Arabitan an Angelan an A |  |  | PUMP                 |  |  |
| TIME                        | DEPTH TO<br>WATER (ft)            | PURGE<br>RATE (ml/m)          | TEMP.<br>(+/- deg. C)                                 | CONDUCTANCE<br>(mS/cm)     | pH<br>(units)                      | DISS. O2<br>(mg/L)  | TURBIDITY<br>(NTU)   | REDOX<br>(+/- mV)  | INTAKE<br>DEPTH (ft) | COMMENTS   |  |
| 811                         | 10.41                             | 400                           | 17.1  | 1.29                       | 7.4                                | 5.66  |  | 149  | 12.5                 | very turbid  |  |
| 818                         | 9.69                              | 200                           | 17.1  | 1.42                       | 7.3                                | 5.41  |  | 156  | Ì                    | · (/   |  |
| 823                         | 4.69                              | 200                           | 17.3  | 1.47                       | 7.3                                | 5.11  |  | 164  |                      |  |  |
| 828                         | 9.68                              | 200                           | 17.2  |                            | 7.2                                | 4.86  | 000  | 172  |                      | less turbud  |  |
| 020                         | 9.68                              | 115                           | 17.2  | 1.00                       | 1.2                                | 4.65  | 930  | 176  |                      | 930 NTU on U-22  |  |
| 830                         | 9,55                              | 10                            | 170   | 107                        | $\neg$                             | 4.04  | Vn2  | 102  |                      | Dom  |  |
| 848                         | 9.60                              | 175                           | 17.3  | 179                        | 7.2                                | 457   | 510  | 183  |                      | 5031074 on U-22  |  |
| 853                         | 9.65                              | 175                           | 17.3  | 1.82                       | 7.2                                | 4.47  | 330  | 182  |                      |  |  |
| 858                         | 9.104                             | 175                           | 17.3  | 1.88                       | 7.2                                | 4,44  | 913  | 181  |                      |  |  |
| 900                         |                                   | 100                           | sam   | ple.                       |                                    |   |  |  |                      |  |  |
|                             |                                   | V.                            |   |                            |                                    |   |  |  |                      |  |  |
|                             |                                   | <u> </u>                      | an a canada da sa |                            |                                    |   | and the second state of th | and the second |                      |  |  |
| EQUIPME<br>TYPE C           | NT DOCUMEN<br>OF PUMP             | ITATION                       |   |                            |                                    |   | TY   |  | NG                   |  |  |
|                             | DICATED MARS                      | SCHALK BLADDE<br>MARSCHALK BI |   | OTHER perista              | altic-                             | <u>Geopu</u>  | yr Z   | HIGH DEN   | SITY POLY            | ETHYLENE   |  |
| ANALYTIC                    | AL PARAME                         | TERS                          |   |                            |                                    |   | L  | ]  |                      |  |  |
| CONTR                       | OL NUMBER                         |                               | ME'<br>NUI  | THOD<br><u>MBER</u>        | PRES<br>M                          | ERVATION<br>ETHOD   | VOLUME<br>REQUIRED   | SAMP<br>COLLEC   | LE<br>TED S          | AMPLE BOTTLE ID LETTERS  |  |
|                             | Cs - 25 ml Purge                  | e (low conc.)                 |   | DLCO2.1                    | HC                                 | L/4 DEG. C  | 3 X 40 ML  |  | ]                    | /  |  |
|                             | Cs - 5 ml Purge                   |                               | (   | DLMO4.2                    | HC                                 | L/4 DEG. C  | 3 X 40 ML  | X  | ע ו                  | SX1  |  |
|                             | STICIDES                          |                               | (   | DLMO4.2                    | 4 C<br>4 C                         | EG. C   | 2 X 1 L AG   |  | !<br>                | /  |  |
| ME                          | TALS                              |                               |   | ILM04.2                    | HN                                 | O3 to pH <2   | 1 X 500 ML   | P  | j                    |  |  |
|                             | NGANESE / IRC                     | DN -                          | SI<br>V LISEDA  | V846 6010                  | HN                                 | 03 to pH <2   | 1 X 500 ML   | P  | ]                    |  |  |
|                             | 2 - NO3                           | IDE / ALKALINH                | USEP  | A 353.2 / 354.1            | 1 4 L<br>H2                        | 504 to pH <2  | 1 X 1 L P<br>2 1 X 500 ML  | P  | <br>                 | · · · · · ·  |  |
| Пто                         | C                                 |                               | US  | EPA 415.1                  | H2                                 | SO4 to pH <2  | 2 1 X 250 ML   | AG   | j                    |  |  |
|                             | THANE / ETHAN                     | NE / ETHYLENE                 | EP  | A Region 1                 | HC                                 | L / 4 DEG. C  | 3 X 40 ML  |  | ]                    | /  |  |
|                             | 1ER                               |                               |   |                            | <u> </u>                           |   |  |  | ]                    |  |  |
| PURGE WA                    | TER                               | S                             |   |                            |                                    | LOCATIO   | N NOTES  |  |                      | N  |  |
| CONTAINE                    | RIZED YES                         | the CN                        | urbid, th   | en cleared                 |                                    |   |  | _ Crys   | tall                 | 2 1  |  |
| NOTES:                      |                                   |                               |   |                            |                                    |   |  |  |                      | lert   |  |
|                             | - Shun                            | luit                          |   |                            |                                    |   |  |  |                      | eat  |  |
| SIGNATUR                    | - xuula                           | sonch                         |   |                            |                                    |   |  |  | MW-5                 |  |  |
|                             | 1.                                |                               |   |                            |                                    | 17.7.2.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1  | W.Pult   | ency 3   | -                    |  |  |

| FIELD D                 | ATA REC                | ORD - LO                        | W FLOW G                        | ROUNDWA                                   | TER S           | AMPLIN   | G                            |                       |   | · · ·                  |
|-------------------------|------------------------|---------------------------------|---------------------------------|---|-----------------|--|------------------------------|-----------------------|---|------------------------|
| PROJECT                 | NYSDE                  | C Region :                      | 8 cleane                        |   |                 |  |                              |                       |   | JER 3612052036 47.2    |
|                         | Crysta                 | 1 Cleane                        | rs mw-                          | 6 FIELD S                                 |                 | CRN  | 100060                       | N30IX)                | <u>(</u>                                | EVENT NO. 2            |
| ACTIVITY                | START 1                | 4:50 E                          | ND 17:50                        |   | TIME            |  | 17:38                        |                       |   | TE 11-1-06             |
| WATER LE                | VEL / PUMP S           | SETTINGS                        |                                 |   |                 | PROTECTIV  | F · ·                        |                       | CASING / \                              | WELL                   |
| INITIAL DEPT<br>TO WATE |                        | .41 fe                          | TOP                             | OF PROTECTIVE (                           | CASING          | CASING STI   | CKUP<br>UND) < 9-            | (<br>feet             | DIFFEREN                                |                        |
| FINAL DEP<br>TO WATE    | rh<br>ER               | 11.6 fe                         | HISTORIC<br>WELL DE<br>et (TOR) | PTH                                       | feet            | PID<br>AMBIENT AI  | к € <i>6</i> ,               |                       | DIAMETER                                | ₹ inches               |
| SCREEN LE               | NGTH                   | 10 fe                           | PRESSUR                         | RE  | psi             | PID WELL<br>MOUTH  | < 0                          |                       |   | Y: YES NO N/A          |
| TOTAL VO<br>PURGI       | DL.<br>ED              | 1.4 gallo<br>per minute) x time | REFILL<br>ns SETTING            | es) x 0.00026 gal/m                       | illiliter)      | DISCHARGE  |                              |                       | LOCKEE                                  |                        |
| PURGE DA                |                        |                                 |                                 | SPECIFIC                                  |                 |  |                              |                       | . PUMP                                  |                        |
| TIME                    | DEPTH TO<br>WATER (ft) | PURGE<br>RATE (ml/m)            | TEMP. (+/- deg. C)              | CONDUCTANCE<br>(mS/cm)                    | pH<br>(units)   | DISS. O2<br>(mg/L)   | TURBIDITY<br>(NTU)           | REDOX<br>(+/- mV)     | INTAKE<br>DEPTH (ft)                    | COMMENTS               |
| 17:04                   | 11.40                  | 160                             | 17.8                            | 2.71                                      | 6.9             | 17.1   |                              | 125                   | 13-4-                                   | turbid/silty           |
| 17:09                   | 11.58                  | 60                              | 17.5                            | 2.01                                      | 7.1             | 10.8   |                              | 120                   | 12:11                                   |                        |
| 17:14                   | 11.60                  | 160                             | 12.5                            | 2.10                                      | 7,2             | 85   | factor.                      | 120                   |   |                        |
| 17:24                   | 11.60                  | 160                             | 17.5                            | 3.89                                      | 7.3             | 7.4  | Angeore.                     | 120                   |   | ( <sup>p</sup>         |
| 17:30                   | 11.60                  | 160                             | 17.5                            | 4.61                                      | 7.3             | 7.4  | -bolgsto                     | 120                   |   | 15                     |
| 17:35                   | 11.61                  | 160                             | 17.5                            | 3.64                                      | 7,3             | 6.7  | Providely                    | 125                   |   | 11                     |
| 17:36                   | E                      | mple                            |                                 |   |                 |  |                              |                       |   |                        |
|                         | <u> </u>               |                                 |                                 |   |                 |  |                              |                       |   |                        |
|                         | $\overline{}$          |                                 |                                 |   |                 |  |                              |                       |   |                        |
|                         |                        |                                 |                                 |   |                 |  |                              |                       |   |                        |
| EQUIDME                 | TDOCUMEN               |                                 |                                 |   |                 | ]  |                              | 1                     |   |                        |
|                         | F PUMP                 | TATION                          |                                 | · · · · · · · · · · · · · · · · · · ·     | 11.8            | en de la composition de la composition<br>Reference de la composition de la compos | TY                           |                       | NG                                      |                        |
|                         |                        | SCHALK BLADDE                   |                                 | OTHERPENSE                                | Itic- C         | reopun   |                              | UHIGH DEN             | SITY POLYET                             | HYLENE                 |
|                         |                        | TERS                            | ADDER                           |   |                 |  |                              | J OTHER               | ana |                        |
| CONTR                   |                        |                                 | MET<br>NUM                      | HOD<br>IBER                               | PRES            | ERVATION   |                              | SAMP<br><u>COLLEC</u> | LE<br>CTED SAL                          | MPLE BOTTLE ID LETTERS |
|                         | Cs - 25 ml Purge       | e (low conc.)                   | 0                               | LCO2.1                                    | нс              | CL / 4 DEG. C  | 3 X 40 ML                    |                       | ]                                       |                        |
|                         | Cs - 5 ml Purge<br>DCs |                                 | 0                               | LMO4.2<br>LMO4.2                          | HC<br>4 [       | CL / 4 DEG. C<br>DEG. C  | 3 X 40 ML<br>2 X 1 L AG      |                       |   | <u></u> /              |
|                         | STICIDES               |                                 | O                               | LMO4.2                                    | 4 [             | DEG. C   | 2 X 1 L AG                   |                       | ]                                       |                        |
|                         | TALS<br>NGANESE / IRO  | DN -                            | SW                              | LM04.2<br>/846 6010                       | n<br>H          | 103 to pH <2<br>103 to pH <2   | 1 X 500 ML<br>1 X 500 ML     | .P [                  | ]]                                      |                        |
| su                      | LFATE / CHLOF          | RIDE / ALKALINIT                | Y USEPA                         | 375.4 / 325.3 / 310                       | .1 41           | DEG. C   | 1x1LP                        |                       | ]                                       |                        |
|                         | 2 - NO3<br>C           |                                 | USEPA                           | 353.2 / 354.1<br>EPA 415.1                | H2<br>H2        | 2SO4 to pH <<br>2SO4 to pH <   | 2 1 X 500 ML<br>2 1 X 250 ML | AG                    | ]                                       |                        |
|                         | THANE / ETHA           | NE / ETHYLENE                   | EPA                             | Region 1                                  | н               | CL/4 DEG. C  | 3 X 40 ML                    |                       | ]                                       | //                     |
| ОТ                      | HER                    |                                 |                                 |   | <u>-</u>        |  |                              | L                     | J                                       |                        |
|                         | BSERVATION             | IS                              | <b>P</b> 4                      |   |                 | LOCATIO  | N NOTES                      |                       |   |                        |
| CONTAINE                | RIZED YES              | 5 <u>10 tu</u>                  | rbid                            |   |                 | $\uparrow$   | ſ                            |                       | sal                                     | 14                     |
| NOTES:                  |                        |                                 |                                 |   |                 |  | · · · ·                      | C                     | ryst                                    |                        |
|                         |                        |                                 | · · · · · ·                     |   |                 |  |                              |                       | +                                       |                        |
| SIGNATUS                | E. JALLA               | ATA S.                          | with .                          | in an | -<br>Ben di lar |  |                              | 1 <b>1</b>            | NW-6                                    | 2                      |
| CHECKER                 | BY:                    |                                 |                                 | 8119919-9-930                             |                 | and the second   | W. Putter                    | Vev                   | T                                       |                        |

GROUNDWATTER SAMPLE FIELD DATA RECORD Project: V O 120vp Additional Site: CNStal Cornin Project Number .se 2 Date: ь 140 1220 612 Time: Start: End: 0 3 Ĺ 04.2 Sample Location ID: G W 9 ø 1 Signature of Sampler Well Depth Ft. Measured Well Bieer Stick-up Top of Well Protective Ft Ft Historical Top of Protective (irom ground) Casing/Well Difference Casing đ Protective \_\_\_\_ Ft. Dat Casing Water Level/Well Depth to Water Ft. Well Material: II Locked?; Well Dia. 2 inch Water Level Equip. Used: PVC Yes มณ์ Elect. Cond. Probe 4 inch SS 6 inch Float Activated Press. Transducer .16 Gal/Ft. (2 in.) Well Integrity: Gal/Vol. Yes No Height of Water Celumn X .65 Gal/Ft. (4 in.) Prot. Casing Secure Ft. 1.5 Gal/Ft. (6 in.) Concrete Collar Intact Total Gal Purged \_\_\_ Gal/Ft. (\_\_\_in.) Other Equipment Documentation Purging/Sampling Equipment Used : Decontamination Fluids Used: ( If Used For) Purging Sampling Equipment ID ( All That Apply at Location) Peristaltic Pump Methanol (100%) Submersible Pump 25% Methanol/75% ASTM Type II water Bailer Deionized Water PVC/Silicon Tubing Liquinox Solution Teflon/Silicon Tubing Hexane Airlift HNO<sub>3</sub>/D.I. Water Solution Hand Pump Potable Water In-line Filter None Press/Vac Filter V SP-15 Sample Observations: PID: Ambient Air \_ppm Well Mouth ppm Purge Data Collected ∠ In-line Turbid Fleid Analysis Data \_\_\_ Clear Cloudy In Container Colored V Odor 0-1 Purge Data @ Gal. 6 Gal. @ Gal. a Gal. @ Gal Temperature, Deg. C pH, units Specific Conductivity (µmhos/cm) Turbidity (NTUS) Oxidation - Reduction, +/- my Dissolved Oxygen, ppm Analytical Parameter ✓ If Sample Preservation Volume Sample Bottle ILot Nos. Collected Method Required 2 Sample Collection Requirement V VOCs 4°C 32x40 ml SVOCs 4°C 2x1 liter AG (~ If Required at this Location) Metals 71 HNO. 4°C 1x1-liter P Cyanide NaOH,4°C H S0 ,4°C 1x500mLP Nitrate/Sulfate 1x1 liter P Nitrate/Phosphate H<sup>2</sup>S0, 4°C 1x1 liter P Pest/PCB 4°C 3x1 liter AG H,S0, .4°C TPH 2x1 liter AG TOC H,S0],4°C 1x1 liter, P 18 1022 Notes: FIGURE 4-1 GROUNDWATER SAMPLE DATA RECORD NYSDEC QUALITY ASSURANCE PROGRAM PLAN - ABB Environmental Services 9404014D L22

GROUNDWATER SAMPLE FIELD DATA RECORD ()F Project: o wall Site: 119 Crista lornin Project Number ... Date: Time: Start: 3 End: 11 ĩ 04 Sample Location ID: ΰ Signature of Sampler Well Depth Ft. Measured Top of Well Well Bieer Stick-up Ft Protective Ft Top of Protective Historical (irom ground) Casing/Well Difference Casing g Protective Ft. Dat Casing **Nater Level/Well** Depth to Water Ft. Well Material: Locked? Well Dia. 2 inch Water Level Equip. Used: PVC 4 inch Elect. Cond. Probe SS Float Activated 6 inch Press. Transducer .16 Gal/Ft. (2 in.) Gal/Vol. Well Integrity: Yes No Height of Water Celumn X .65 Gal/Ft. (4 in.) Prot. Casing Secure Ft. 1.5 Gal/Ft. (6 in.) Concrete Collar Intact Total Gal Purged \_\_\_ Gal/Ft. (\_\_\_ in.) Other Equipment Documentation Purging/Sampling Equipment Used : Decontamination Fluids Used: ( If Used For) Purging Sampling Equipment ID ( All That Apply at Location) V Peristaltic Pump Methanol (100%) Submersible Pump 25% Methanol/75% ASTM Type II water Bailer Deionized Water PVC/Silicon Tubing Liquinox Solution Teflor/Silicon Tubing Hexane Airlift HNO<sub>3</sub>/D.I. Water Solution Hand Pump Potable Water In-line Filter None Press/Vac Filter SP-15 Sample Observations: PID: Ambient Air \_ ppm Well Mouth ppm Purge Data Collected /In-line Fleid Analysis Data Turbid Clear Cloudy In Container Colored V Odor @ v . 1 Purge Data Gal. Gal. @ G Gal. @ Gal. @ Gal. Temperature, Deg. C pH, units Specific Conductivity (µmhos/cm) Turbidity (NTUS) Oxidation - Reduction, +/- my Dissolved Oxygen, ppm Analytical Parameter ✓ If Sample Preservation Volume Sample Bottle ILot Nos. Collected Method Required Sample Collection Requirements VOCs 32x40 ml 4°C SVOCs 4°C 2x1 liter AG If Required at this Location) Metals HNO.,4°C 1x1-liter P Cvanide NaOH,4°C H S0,4°C 1x500mLP Nitrate/Sulfate 1x1 liter P Nitrate/Phosphate H<sup>\*</sup>50, 4°C 1x1 liter P Pest/PCB 4°C 3x1 liter AG -----TPH H,S0,,4°C 2x1 liter AG TOC H,S0,4°C 1x1 liter, P 30 Notes: FIGURE 4-1 GROUNDWATER SAMPLE DATA RECORD NYSDEC QUALITY ASSURANCE PROGRAM PLAN - ABB Environmental Services-9404014D L22

es sonno loc Project Boring Well No. SR GW - OIG Project No.5 361206 REGION 8 D.L. - GROUP I Client NYSDEC Site leener Sheet No. Logged By R Finish Date Ground Elevation Start Date MAN Zoob 10 Zodo **Drilling Contracto** Driller's Name Rig Type 2000Tic.N 66 DT mendel Casing Size 11/2 **Drilling Method** P.I.D. (eV) Auger Size **Protection Level** Soil Drilled Total Depth 36 Rock Drilled Depth to Groundwater/Date Piez Well Boring Monitoring or Core Rec./Rqd. % Sample No. & Penetration/ Recovery (Feet) USCS Group Symbol Notes on Drilling Sample Type SPT Blows/6 SPT-N (Blows/Ft.) Depth(Feet) Graphic Log (ppm) Lab Tests Sample PI Meter Head Space Description PI Meter Field Scan 0.0.8 It olive silly fine said. 101 FIL dry/doup isp 0.8 .1.4 Black Sundy grove dry star preces 1.4-B.8 orange Buin sand GC W.O dy squeel, noist MP/SP. 3.8-4 overge brown fine sav days grevel, not mp/rp, sof U 4-4-85. Ity sind of some 6.1 MG fine groved, ding, Dense 22 4.8 Jense of V. Dense reddyn oranje fore silf. 4.8-8 Sine as 4-4.8; COLOR (DAT' LO.1 V-9.6 Simeas 4-4.8 51 On 9.6 - to Brinn clean M Course Surd W (some fines, dry, SP 0 10.12 Brown Sully Sandig graded of some cobbility, SM GM 4.0 [l]Dink 12-16 Bring silt said 5 give of some cobbos (2-15', damp to dry, 5m 6.1 OM NP FIGURE 4-6 TYPICAL TEST BORING LOG NYSDEC QUALITY ASSURANCE PROGRAM PLAN ABB Environmental Services, Inc. '9404014D(z) L33



GROUNDWATER SAMPLE FIELD DATA RECORD 61 VE Project: 1 anna T Site: Additiona Calsta Cornin Project Number: -36 Date: 2612 Time: Start: 164 3 L End: 03 04 Sample Location ID: K Signature of Sampler Well Depth Ft. Measured Well Bieer Stick-up Top of Well Ft. Protective Ft Historical Top of Protective (irom ground) Casing/Well Difference Casing Water Level/Well Data Protective \_\_\_\_ Ft. Casing Depth to Water Ft. Well Material: acked? Well Dia. 2 inch Water Level Equip. Used: PVC Elect. Cond. Probe 4 inch SS 6 inch Float Activated Press. Transducer .16 Gal/Ft. (2 in.) Gal/Vol. Well Integrity: Yes No Height of Water Cetumn X .65 Gal/Ft. (4 in.) Prot. Casing Secure 1.5 Gal/Ft. (6 in.) Ft. Concrete Collar Intact Total Gal Purged \_ Gal/Ft. (\_\_in.) Other Equipment Documentation Purging/Sampling Equipment Used : Decontamination Fluids Used: ( If Used For) Sampling Purging Equipment ID ( All That Apply at Location)  $\underline{V}$ Peristaltic Pump Methanol (100%) Submersible Pump 25% Methanol/75% ASTM Type II water Bailer Deionized Water PVC/Silicon Tubing Liguinox Solution Teflon/Silicon Tubing Hexane Airlift HNO3/D.I. Water Solution Hand Pump Potable Water In-line Filter None Press/Vac Filter SP-15 V Sample Observations: PID: Ambient Air Purge Data Collected \_\_\_\_ppm\_Well Mouth mag ∠ In-line Turbid \_\_ Clear Field Analysis Data Cloudy In Container Colored 1. Odor Purge Data @ 1 Gal. Gal. @ a Gal. @ Gal. @ Gal 5.0 Temperature, Deg. C Curs pH, units Specific Conductivity (µmhos/cm) Turbidity (NTUS) Oxidation - Reduction, +/- mv Dissolved Oxygen, ppm Analytical Parameter ✓ If Sample Preservation Volume Sample Bottle ILot Nos. Collected Method Required Sample Collection Regulments .1 VOCs 4°C 32x40 ml SVOCs 4°C / If Required at this Location) 2x1 liter AG Metais HN0, ,4°C 1x1-liter P Cyanide NaOH,4°C H\_S0\_,4°C 1x500mLP Nitrate/Sulfate 1x1 liter P Nitrate/Phosphate H<sup>2</sup>S0, 4°C 1x1 liter P Pest/PCB 4°C 3x1 liter AG \_\_\_\_TPH H\_S0, ,4°C 2x1 liter AG TOC H,S0,4°C 1x1 liter, P Notes FIGURE 4-1 GROUNDWATER SAMPLE DATA RECORD NYSDEC QUALITY ASSURANCE PROGRAM PLAN - ABB Environmental Services 9404014D L22

CHIZE) HIND MADIN SAMPLE FIELD DATA RECORD Project: VEGION Site: 1.19 CNStal Cornin Project Number: -36 Date: 31 60 Time: Start: 42 End; 17 Sample Location ID: CRG 19 029 5 0 0 Signature of Sampler 1 X X Well Depth Ft. Measured Top of Well Well Biser Stick-up Ft. Protective Ft. Historical Top of Protective (irom ground) Casing/Well Difference Casing Water Level/Well Data Protective Ft. Casing Depth to Water Well Material: Ft. Locked? Well Dia. Water Level Equip. Used: 2 inch PVC 4 inch Elect. Cond. Probe SS 6 inch Float Activated Press. Transducer .16 Gal/Ft. (2 in.) Gal/Vol. Well Integrity: Yes No Height of Water Cetumn X .65 Gal/Ft. (4 in.) Prot. Casing Secure Ft. \_1.5 Gal/Ft. (6 in.) Concrete Collar Intact Gal/Ft. (\_\_in.) Total Gal Purged Other Equipment Documentation Purging/Sampling Equipment Used : Decontamination Fluids Used: ( If Used For) Purging Sampling Equipment ID ( All That Apply at Location) Peristaltic Pump Methanol (100%) Submersible Pump 25% Methanol/75% ASTM Type II water Bailer Deionized Water PVC/Silicon Tubing Liquinox Solution Teflon/Silicon Tubing Hexane Airlift HNO<sub>3</sub>/D.I. Water Solution Hand Pump Potable Water In-line Filter None Press/Vac Filter SP-15 Sample Observations: PID: Ambient Air \_\_ppm Well Mouth nom Purge Data Collected /In-line Turbid Clear Field Analysis Data Cloudy In Container Colored Odor n 2 Purge Data ര Gal. Gal. @ a Gal. @ Gal. @ Gal Temperature, Deg. C Gmo pH, units Specific Conductivity (µmhos/cm) 61 Turbidity (NTUS) 000 Oxidation - Reduction, +/- my Dissolved Oxygen, ppm Analytical Parameter 🖌 If Sample Preservation Volume Sample Bottle ILot Nos. Collected Method Required Sample Collection Requirements **V**vocs 32x40 ml 4°C SVOCs 4°C (~ If Required at this Location) 2x1 liter AG Metais HN0, ,4°C 1x1-liter P Cyanide NaOH,4°C H,S0,4°C 1x500mLP Nitrate/Sulfate 1x1 liter P Nitrate/Phosphate H<sup>\*</sup>S0, 4°C 1x1 liter P Pest/PCB 4°C 3x1 liter AG TPH H,S0, ,4°C 2x1 liter AG TOC H\_S0\_,4°C 1x1 liter, P - 28-130 Notes: FIGURE 4-1 GROUNDWATER SAMPLE DATA RECORD NYSDEC QUALITY ASSURANCE PROGRAM PLAN - ABB Environmental Services 9404014D L22

|                      |  | GROUNDW  | ATER SAM                            | PLE FIELD DA                 | TA RECORI   | <b>)</b>  |   |
|----------------------|--|--|-------------------------------------|------------------------------|---|---|---|
| Pr                   | oject: <u>REGIONY</u>                  | 5 Dry, CLEAM   | E125-61200P                         | I Site: Ao                   | lelifinal (   | Wstal - Cori  | ning                                    |
|                      | bject Numbert                          | 100205   | 10.2                                | Date:                        | 1021  | 12.000  |   |
| 5                    | ample Location ID:                     | 36160360   | 36704.2                             | Time: Star                   | t: <u>330</u>   | End: 14   | 5                                       |
|                      |  | ACIM OFFICE  |                                     | Signature c                  | of Sampler:   |   |   |
|                      | Well DepthF                            | t Measured<br>Historical   | Top of V<br>Top of F<br>Casing      | Vell Well Fier               | Stick-up Ft.  | Protective<br>Casing/Well Diff  | Ft.<br>lerence                          |
| l Date               |  |  |                                     |                              | •   | Protective<br>Casing  | Ft.                                     |
| r Level/Wel          | Depth to Water F                       | t. Well Material:<br>PVC<br>SS   | Well Locked?<br>Yes                 | Well Dia<br>                 | 2 inch<br>4 inch<br>6 inch  | Water Level Equ<br>Elect. Cond<br>Float Activa<br>Press. Tran   | ip. Used:<br>. Probe<br>tted<br>isducer |
| Wate                 | Height of Water Certurn<br>Ft.         | 16 Gal/Ft. (2 i<br>X65 Gal/Ft. (4 i<br>1.5 Gal/Ft. (6 i<br>Gal/Ft. (   | n.) =<br>n.) =<br>in.)              | Gal/Vol.<br>Total Gal Purged | Well Integrity:<br>Prot. Casing Sect<br>Concrete Collar Ir<br>Other | Yes   | No<br>                                  |
| E                    | Puraina                                | Sampling Equipmon  |                                     |                              |   |   |   |
| Equipment Documentat | ( If Used For)<br>Purging Sampling<br> | Peristaltic Pump<br>Submersible Pump<br>Bailer<br>PVC/Silicon Tubin<br>Teflor/Silicon Tubi<br>Airlift<br>Hand Pump<br>In-line Filter<br>Press/Vac Filter | Equipment                           |                              | ( ✓ Ali That Apply i<br>  | at Location)<br>I (100%)<br>thanol/75% ASTM Typ<br>ed Water<br>Solution<br>M.I. Water Solution<br>Water | e II water                              |
| Data                 | PID: Ambient Air                       | ppm Well Mouth   | ppm Pi                              | urge Data Collected          | Sar<br>∠In-line<br>_In Container                                    | mple Observations:<br>_TurbidClear<br>_ColoredOdor  | Cloudy                                  |
| s<br>S               | Purge Data                             | @_^2   | Gal. @                              | Gal. @                       | Gal. @  | Gal @   |   |
| alys                 | Temperature, Deg. C                    |  |                                     |                              |   | Cu. @   | Ga.                                     |
| Ana                  | pH, units<br>Specific Conductivity     | (umbas/am)   |                                     | $= \Sigma$                   | $=$ $\stackrel{\checkmark}{\sim}$                                   |   |   |
| eld                  | Turbidity (NTUS)                       |  |                                     |                              |   | )   |   |
| Ē                    | Dissolved Oxygen, pp                   | , +/- mv   |                                     |                              |   | $\sim$ —  |   |
|                      |  |  | <u> </u>                            |                              |   |   |   |
| on.                  | Analytical Parameter                   | ✓ If Sample<br>Collected   | Preservation<br>Method              | Volume<br>Required           | Sample Bo   | ttle ILot Nos.  |   |
| nen                  |  |  | 4°C                                 | 32x40 ml                     |   |   |   |
| lren<br>tion)        | SVOCs<br>Metals                        |  | 4°C                                 | 2x1 liter AG                 | · · · · · · · · · · · · · · · · · · ·                               |   |   |
| equ                  | Cyanide<br>Nitrate/Sulfate             |  | NaOH,4°C                            | 1x500mLP                     | $\overline{\Delta}$   | 1110  |   |
| this H               | Nitrate/Phosphate                      |  | H <sup>2</sup> S0 <sup>2</sup> ,4°C | 1x1 liter P                  |   | 1710  |   |
| d at                 | Pest/PCB<br>TPH                        |  | 4°C<br>H.S04°C                      | 3x1 liter AG                 |   |   |   |
| llec<br>quire        | TOC                                    |  | H <sub>2</sub> S0,4°C               | 1x1 liter, P                 |   |   |   |
| le Co                | Notes: SLOON'                          | 17 +21   |                                     |                              | •   |   | ······                                  |
| due                  | ALLARY INAL                            | A /  |                                     |                              |   | FIGU  | BE A.1                                  |
| ŝ                    | -hande water                           | · /  |                                     | GROUNDV                      | VATER SAM   | PLE DATA RE   | CORD                                    |
|                      | . –                                    |  | NYS                                 | DEC QUALITY                  | ASSURAN   | CE PROGRAM  | PLAN                                    |
| 940401               | 4D L 22                                |  |                                     |                              | ——— AB  | B Environmental   | Services-                               |

į

À

GROUNDWATER SAMPLE FIELD DATA RECORD Project: T Site: Addi 119 CNStal Cornin Project Number: -3E 02 Date: Time: Start: 330 05 End: 1410 -1 704.2 U Sample Location ID: 9 Signature of Sampler Well Depth Ft. Measured Top of Well Well Biger Stick-up Protective Ft. Historical Top of Protective (irom ground) Casing/Well Difference Casing Water Level/Well Data Protective Ft. Casing Depth to Water FL Well Material: \_ocked?: Well Dia Water Level Equip. Used: 2 inch PVC 4 inch Elect. Cond. Probe SS 6 inch Float Activated Press. Transducer .16 Gal/Ft. (2 in.) Gal/Vol. Well Integrity: Yes No Height of Water Celumn X .65 Gal/Ft. (4 in.) Prot. Casing Secure Ft. 1.5 Gal/Ft (6 in.) Concrete Collar Intact Total Gal Purged \_ Gal/Ft. (\_\_in.) Other Equipment Documentation Purging/Sampling Equipment Used : Decontamination Fluids Used: ( If Used For) Purging Sampling Equipment ID ( All That Apply at Location) Peristaltic Pump Methanol (100%) Submersible Pump 25% Methanol/75% ASTM Type II water Bailer Deionized Water PVC/Silicon Tubing Liquinox Solution Teflon/Silicon Tubing Hexane Airlift HNO3/D.I. Water Solution Hand Pump Potable Water In-line Filter None Press/Vac Filter V SP. 15 Sample Observations: PID: Ambient Air ppm Well Mouth Purge Data Collected nag Turbid /In-line Clear Field Analysis Data \_\_Cloudy 1/ In Container Colored Odor Purge Data @ *,*~~ n Gal. @ Gal. @ Gal. @ Gal. @ Gal Temperature, Deg. C pH, units Specific Conductivity (µmhos/cm) Turbidity (NTUS) Oxidation - Reduction, +/- my Dissolved Oxygen, ppm Analytical Parameter ✓ If Sample Preservation Volume Sample Bottle ILot Nos. Collected Method Required ŋ Sample Collection Requirement **V**vocs 32x40 ml 4°C SVOCs 4°C Required at this Location) 2x1 liter AG Metais HN0, ,4°C 1x1 liter P Cyanide NaOH,4°C H,S0,4°C 1x500mLP tì Nitrate/Sulfate 1x1 liter P Nitrate/Phosphate H,S0, ,4°C 1x1 liter P Pest/PCB 4°C 3x1 liter AG TPH H\_S0\_,4°C 2x1 liter AG -----TOC H,S0, 4°C 1x1 liter, P 30 1 Notes FIGURE 4-1 GROUNDWATER SAMPLE DATA RECORD NYSDEC QUALITY ASSURANCE PROGRAM PLAN - ABB Environmental Services 9404014D L22

Boring/Well No. SR/GW-02 Project REGION 8 D.C. - GROUP I Project No. 203 36170 Client NYSDEC Site 12 Sheet No. Logged By Ground Elevation Start Date Finish Date Loob 0 Zodo 0 **Drilling Contracto** Driller's Name' **Rig Type** de mense 2001011 00 DI Casing Size 11/2 Drilling Method Protection Level Auger Size P.I.D. (eV) Soil Drilled Rock Drilled Total Depth Depth to Groundwater/Date Piez Well Boring, 271-2  $\Box$ Π R Monitoring % Sample No. & Penetration/ Recovery (Feet) SPT Blows/6" or Core Rec./Rqd. % USCS Group Symbol Notes on Drilling Sample Type Depth(Feet) Graphic Log SPT-N (Blows/Ft.) (ppm) Lab Tests Sample PI Meter Head Space PI Meter Field Scan Description 闩 0-0.5 Branisholive Silly Sud Lo.1 rots, R, dup, Sp Fill -1.2 Black suffitie said Whome cly, slap sash, trace -Z it Brown Silly Sant & lottle CC 40 fore gravel, day " miterse, WG 3 2-35 scarfe brown soud crap group noist, we wist ff Li Biern soudy cray, woist CL 4-4.4 Sireas 3.5 +04 621 4.4-8 Brun sty sur squadel NG, some cobbies, some silt GW Moonst said to fine give 40 Bran Michies sand if some Sint 401 coarse suit à truce sille dampive, miderer; 105 tol2 Brun silly Smith 0 Saturation SPIMP, MDanse Nic [l]12-155 Smeas 10.5 to GM 6.1 15.5 to 16 Olive Burn Sendy genel of some oly saturated M Druse, MP GrG FIGURE 4-6 TYPICAL TEST BORING LOG NYSDEC QUALITY ASSURANCE PROGRAM PLAN ABB Environmental Services, Inc.-9404014D(z) L33

GROUNDWATTER SAMIPLE FIELD DATA RECORD Project: 119 CNStal Site: Lornin Project Number - 5 E Date: 67 61Z .05 Time: Start: End; 101 04 c Sample Location ID: 14 Ċ 2 0 b Signature of Sampler Well Depth Ft. Measured Top of Well Well Riger Stick-up Ft Protective Ft. Historical Top of Protective Casing/Well Difference (irom ground) Casing Data Protective Ft. Casino Water Level/Well Depth to Water \_FL Well Material: Locked? Well Dia 2 inch Water Level Equip. Used: PVC Ye Elect. Cond. Probe 4 inch SS 6 inch Float Activated Press. Transducer .16 Gal/Ft. (2 in.) Gal/Vol. Well Integrity: Height of Water Cetumn X Yes No .65 Gal/Ft. (4 in.) Prot. Casing Secure Ft. 1.5 Gal/Ft. (6 in.) Concrete Collar Intact Total Gal Purged \_\_ Gal/Ft. (\_\_in.) Other Equipment Documentation Purging/Sampling Equipment Used : Decontamination Fluids Used: ( If Used For) Purging Sampling Equipment ID ( All That Apply at Location) Peristaltic Pump V Methanol (100%) Submersible Pump 25% Methanol/75% ASTM Type II water Bailer Deionized Water PVC/Silicon Tubing Liquinox Solution Teflor/Silicon Tubing Hexane Airlift HNO<sub>3</sub>/D.I. Water Solution Hand Pump Potable Water In-line Filter None Press/Vac Filter V IF MI/SLOT Sample Observations: PID: Ambient Air \_ppm Well Mouth Purge Data Collected ppm /In-line Turbid Fleid Analysis Data \_\_\_ Clear Cloudy In Container Colored Odor Sundy V . r O. Purge Data @ Gal. Gal. @ ía Gal. a Gal. @ Gal. 3 Temperature, Deg. C pH, units Ö Cins Specific Conductivity (umhos/cm ēss Turbidity (NTUS) >1000 Oxidation - Reduction, +/- my Dissolved Oxygen, ppm X 3 Analytical Parameter ✓ If Sample Preservation Volume Sample Bottle ILot Nos. Collected Method Sample Collection Requirements Required V VOCs 4°C 32x40 ml SVOCs 4°C Required at this Location) 2x1 liter AG Metais HNO.,4°C 1x1 liter P Cyanide NaOH,4°C H\_S0\_,4°C 1x500mLP Nitrate/Sulfate 1x1 liter P Nitrate/Phosphate H<sup>2</sup>S0<sup>1</sup>,4°C 1x1 liter P Pest/PCB 4°C 3x1 liter AG 00 \_\_\_\_TPH H\_S0\_,4°C 2x1 liter AG TOC H[S0],4°C 1x1 liter, P 102 1 Notes: FIGURE 4-1 Sediment. GROUNDWATER SAMPLE DATA RECORD NYSDEC QUALITY ASSURANCE PROGRAM PLAN - ABB Environmental Services-9404014D L22

GROUNDWATER SAMPLE FIELD DATA RECORD Project: ЮN 10/1/1199 CNStal Site: Lornin Project Number ...E Date: 01 103 612 04 Time: Start: End: Sample Location ID: K ъĴ. Ů Signature of Sampler Well Depth Ft. Measured Top of Well Well River Stick-up Ft Protective Ft Historical Top of Protective (irom ground) Casing/Well Difference Casing Data Protective Ft. Casing Water Level/Well Depth to Water \_\_Ft. Well Material: Locked? Well Dia 2 inch Water Level Equip. Used: PVC \_Elect. Cond. Probe 4 inch SS 6 inch Float Activated Press. Transducer .16 Gal/Ft. (2 in.) Gal/Vol. Well Integrity: Yes No Height of Water Celumn X .65 Gal/Ft. (4 in.) Prot. Casing Secure 1.5 Gal/Ft. (6 in.) \_Ft. Concrete Collar Intact Total Gal Purged \_ Gal/Ft. (\_\_in.) Other Equipment Documentation Purging/Sampling Equipment Used : Decontamination Fluids Used: ( If Used For) Purging Sampling Equipment ID ( All That Apply at Location) Peristaltic Pump V Methanol (100%) Submersible Pump 25% Methanol/75% ASTM Type II water Bailer Deionized Water PVC/Silicon Tubing 1/Liquinox Solution Teflon/Silicon Tubing Hexane Airlift HNO<sub>3</sub>/D.I. Water Solution Hand Pump Potable Water In-line Filter None Press/Vac Filter  $\cap Q$ To shill sut Sample Observations: PID: Ambient Air ppm Well Mouth Purge Data Collected ppm Turbid /In-line Data. \_\_ Clear \_Cloudy In Container Colored Odor Fleid Analysis Purge Data ര Gal. @ Gal. @ Gal. @ Gal. @ Gal. Ć Temperature, Deg: C pH, units Const Specific Conductivity (umhos/cm) Turbidity (NTUS) 0200 Oxidation - Reduction, +/- my Dissolved Oxygen, ppm Analytical Parameter ✓ If Sample Preservation Volume Sample Bottle ILot Nos. Collected Method Required Sample Collection Requirements V VOCs 4°C 32x40 ml SVOCs (~ If Required at this Location) 4°C 2x1 liter AG Metais HNO, 4°C 1x1-liter P Cvanide 0.0 NaOH,4°C 1x500mLP Nitrate/Sulfate H, S0 ,4°C 1x1 liter P Nitrate/Phosphate H<sup>2</sup>S0<sup>1</sup>,4°C 1x1 liter P Pest/PCB 4°C 3x1 liter AG TPH H\_S0\_,4°C 2x1 liter AG TOC H\_50,4°C 1x1 liter, P (1 Notes: FIGURE 4-1 GROUNDWATER SAMPLE DATA RECORD NYSDEC QUALITY ASSURANCE PROGRAM PLAN - ABB Environmental Services-9404014D L22

esi Borina Project Boring Well No. SR 6W -022 Project No. 52036 3612062859 REGIONS D.L. - GROUP I Client NYSDEC Site WStal Clenn Sheet No. Logged By Ground Elevation Start Date Finish Date Loob יאאאי 11/01 (20de Driller's Name' Joe Mense **Drilling Contractor Rig Type** zeolofic, 10 Casing Size 11/2 Drilling Method **Protection Level** P.LD. (eV) Auger Size Soil Drilled Rock Drilled Piez Well Boring Total Depth Depth to Groundwater/Date  $\Box$ Monitoring % Sample No. & Penetration/ Recovery (Feet) SPT Blows/6" or Core Rec./Rqd. % Votes on Drilling Group Symbol Sample Type Depth(Feet) Graphic Log (ppm) SPT-N (Blows/Ft.) Lab Tests USCS Sample PI Meter Head Space Description PI Meter Field Scan 0-1 DK Braven Silty loan Frant Fil 10.1 1-15 Cobbies, erushed week PG, 1.5-3 Bran Silty-ely's content Sond, wet, MP, MStiff tec. SCOU 3-4 2+ cray Bren Sand-ely Squiel mix, vet, meltop voots- $\mathcal{F}$ from rear by tree 60 pushed a cobble Lon orcorent. NA -8 to ~11 it Brum/Brun Sund & grovel of some sitt wet/sutwated, in Durse, Lon GM 53 Vale cobsic 10 11-12 clean moonse Said, drup, NP, PG []58 Silve Bran A Board Silvisund's groved of trace 61 SNi Chy, Soft, suturated we GM FIGURE 4-6 TYPICAL TEST BORING LOG NYSDEC QUALITY ASSURANCE PROGRAM PLAN ABB Environmental Services, Inc. 9404014D(z) L33

| N. F.                                 |  | GROUNDWA   | TER SAMPLE  | INICILD DAY                                 | ARECORD  | ar de la Sanan   |   |
|---------------------------------------|--|--|---|---|--|--|---|
| Pr                                    | oject: <u>REGIONS</u>  | Dy, ILEAMER  | 5-6ROVPIT   | Site: Ad                                    | ditinal Cry  | stal - Corn,   | mp  |
| Pr                                    | oject Numbert 361  | 2019 2059  | 0.2   | Date:                                       | 101 12   | ->6  |   |
| SE                                    |  | 1016056030   | 2704.2  | Time: Start:                                | 1140   | End: 1247  |   |
|                                       |  | UNTULIO  |   | Signature of                                | Sampler:   |  |   |
| · · · · · · · · · · · · · · · · · · · | Well Depth Ft.   | Measured<br>Historical   | Top of Well<br>Top of Protect<br>Casing                         | Well River S                                | tick-up Ft.  | Protective<br>Casing/Well Differe  | Ft.<br>Ince                               |
| ll Data                               |  |  |   | -   | en en anter en la construcción<br>Referencia de la construcción de la construcción<br>Referencia de la construcción de la  | Protective   | _Ft.                                      |
| r Level/We                            | Depth to Water Ft.   | Well Material:<br>PVC<br>SS  | Well Lacked?:<br>Yes  |   | 2 inch<br>4 inch<br>6 inch   | Water Level Equip.<br>Elect. Cond. P<br>Float Activated<br>Press. Transd | Uaed:<br>robe<br>l<br>Joer                |
| Wate                                  | Height of Water Cetumn X   | 16 Gal/Ft. (2 in.)<br>65 Gal/Ft. (4 in.)<br>1.5 Gal/Ft. (6 in.)<br>Gal/Ft. (in.) | = [   | Gal/Vol.<br>otal Gal Purged                 | Well Integrity:<br>Prot. Casing Secure<br>Concrete Collar Intact<br>Other  | Yes  | No  |
| E                                     | DursingS   |  | a martaneo de la construction de ante de aconstructura de acons |   | da Marinovan Kanpong dara Jaharan ang penantanan na ka   |  |   |
| mentation                             | <u>Purging/S</u><br>(✔ If Used For)<br>Purging: Sampling                 | Ampling Equipment U:   | <u>sed</u> :<br>Equipment ID                                    |   | Decontaminatio   | en Fluids Used :<br>cation)  |   |
| nt Docu                               |  | Submersible Pump<br>Bailer<br>PVC/Silicon Tubing                                 |   |   | Methanol (10<br>25% Methani<br>Deionized W   | 0%)<br>ol/75% ASTM Type I<br>ater<br>ution                               | l water                                   |
| dulpme                                |  | Airlift<br>Hand Pump<br>In-line Filter   |   |   | Hexane<br>HNO <sub>3</sub> /D.I. W<br>Potable Wate<br>None   | later Solution<br>er   |   |
| ш                                     |  | SR 15  |   |   |  |  |   |
| lata                                  | PID: Ambient Air   | _ppm Well Mouth  | ppm Purge D   | pata Collected                              | Sample<br>In-linefur<br>In ContainerCoi  | Observations:<br>bidClear<br>oredOdor                                    | Cloudy                                    |
| sis D                                 | Purge Data   | @ <u>^}</u>  | _ Gal. @  | Gal. @                                      | Gal. @   | Gal. @   | Gal                                       |
| Analys                                | Temperature, Deg. C<br>pH, units   | 14.2   |   |   |  |  |   |
| leld /                                | Specific Conductivity (µ<br>Turbidity (NTUS)<br>Oxidation - Reduction, + | mhos/cm) <u>0 10 24</u><br>mv  |   |   | $\geq 2$   |  |   |
|                                       | Dissolved Oxygen, ppm  | 2-4  |   |   |  | ×  |   |
|                                       | Analytical Parameter   | ✓ If Sample Pr<br>Collected  | reservation<br>Method   | Volume<br>Required                          | Sample Bottle I  | Lot Nos.   | nine da a constante filon e la para rever |
| emen<br>n)                            |  |  | •c 3  | 2x40 ml<br>2x1 liter AG                     | 94 million and a second s |  |   |
| quin                                  | Metals<br>Cyanide  | H  | HN0, ,4°C<br>NaOH,4°C   | 1x1 liter P<br>1x500mLP                     |  | 124  |   |
| n He                                  | Nitrate/Phosphate  | ł  | H_S0_,4°C<br>H_S0_,4°C  | 1x1 liter P<br>1x1 liter P                  | (  |  |   |
| ectiol<br>ired at                     | Pesupous<br>TPH<br>TOC   | I  | 1°C<br>∃_S0_,4°C<br>∃_S0_,4°C                                   | 3x1 liter AG<br>2x1 liter AG<br>1x1 liter P |  |  |   |
| e Coll                                | Notes: SILCON  | 261 30   | 2 4'  | · · · · · · · · · · · · · · · · · · ·       |  |  |   |
| ampli<br>(                            | AINTI MALTAN   |  | ан таландаан  |   | •  | FIGUR  | E 4-1                                     |
| S                                     | - Presidentiality  |  | NYSDE(  | GROUNDW<br>C QUALITY                        | ATER SAMPL   | E DATA REC   |   |
| 940401                                | 4D L22   |  |   |   | ABB E  | nvironmental Se  | ervices                                   |

į

 $\cos (k_{\rm s}^2 + 1) \cos (k_{\rm s}^2 + 1)^2 = 10^2$ 

ىرى بېرىيىنى بېرىيى بېرى

a company and a second and a second second

a ta anti e contrato,

. A.

|  | <u>Glassica Glassica Gla</u>  | CUUNDWALDR SAWIPL  | <b>ID DIDLID DATA R</b>  |   |                                  |
|--|---|--|--|---|----------------------------------|
| P  | roject: <u>REGIONS</u> DN   | CLEAMERS- GROUP I  | Site: Adelifi  | nal Crystal - Corn  | ing                              |
| P  | roject Numbert  | 2059 10.2  | Date:  | 1 12-26   | <u> </u>                         |
| 6  | ample Leasting ID. COV  | 60520367042  | Time: Start:   | 630 End: 1317   | e                                |
|  | ample Location ID: CIPGIW   | 026018011  | Signature of Samp  | vier:   |                                  |
| t i serie  | Well Depth Ft   | _MeasuredTop of Well<br>_HistoricalTop of Prote<br>Casing  | Well Biser Stick-up<br>ective (irom ground)  | Ft. Protective<br>Casing/Well Diffe   | Ft.<br>rence                     |
| Data   |   |  |  | Protective<br>Casing  | Ft.                              |
| Level/Well   | Depth to Water Ft. Wel  | Material: Well Locked?: #  | Well Dia 2 inc<br>4 inc<br>6 inc   | h Water Level Equip<br>hElect. Cond.<br>hFloat Activate<br>Press. Trans   | b. Used:<br>Probe<br>ad<br>ducer |
| Water  | Height of Water Certumn X   | 16 Gal/Ft. (2 in.)<br>65 Gal/Ft. (4 in.) =<br>1.5 Gal/Ft. (6 in.)<br>Gal/Ft. (in.)   | Gal/Vol. Well In<br>Prot. C<br>Concre<br>_Total Gal Purged Other _   | tegrity: Yes<br>asing Secure<br>te Collar Intact  | No                               |
| entation   | Purging/Samplin<br>(✔ If Used For)<br>Purging Sampling  | ig Equipment Used :  | р  | econtamination Fluids Used :  |                                  |
| Equipment Docum  | Purgung Sampling<br>Period<br>Subr<br>Baile<br>PVC<br>Tefic<br>Airlif<br>Hank<br>In-lir<br>Pres   | Italtic Pump     Equipment ID       nersible Pump  |  | hat Apply at Location)<br>_ Methanol (100%)<br>_ 25% Methanol/75% ASTM Type<br>Deionized Water<br Liquinox Solution<br _ Hexane<br>_ HNO <sub>3</sub> /D.I. Water Solution<br>_ Potable Water<br>_ None | II water                         |
| Jata   | PID: Ambient Airppm   | Well Mouthppm Purge  | Data Collected In-line   | Sample Observations:<br>UrbidClear<br>ainerColoredOdor  | Cloudy                           |
| 1  | Purge Data  | @ 12 Gal. @ \  | Gal. @ Ga  | @ Gal @   |                                  |
| γs   | /   | 14.1   | <u< th=""><th></th><th></th></u<>  |   |                                  |
|  | Temperature, Deg. C   |  |  | L   | Gal.                             |
| Anal   | Temperature, Deg. C<br>pH, units Cons/  |  |  | <u>k                                    </u>  | Gal.                             |
| eld Anal   | Temperature, Deg. C<br>pH, units<br>Specific Conductivity (umhos/c<br>Turbidity (NTUS)  | m) 06487   |  |   | Gal.                             |
| Fleid Anal   | Temperature, Deg. C<br>pH, units<br>Specific Conductivity (umhos/c<br>Turbidity (NTUS)<br>Oxidation - Reduction, +/- mv<br>Dissolved Oxygen, ppm  | m) 0.648<br>51000  |  |   | Gal.                             |
| Fleid Anal   | Temperature, Deg. C<br>pH, units<br>Specific Conductivity (umhos/c<br>Turbidity (NTUS)<br>Oxidation - Reduction, +/- mv<br>Dissolved Oxygen, ppm  | m) 0648<br>SIUUD<br>4.4  |  |   | Gal.                             |
| Field Anal   | Temperature, Deg. C<br>pH, units Cons/<br>Specific Conductivity (µmhtos/c<br>Turbidity (NTUS)<br>Oxidation - Reduction, +/- mv<br>Dissolved Oxygen, ppm<br>Analytical Parameter If Sa<br>Colle  | m) 0.047<br>1000<br>U.U.U.<br>Umple Preservation<br>cted Method  | Volume s<br>Required   | Sample Bottle ILot Nos.   | Gal.                             |
| nents Fleid Anal   | Temperature, Deg. C<br>pH, units Conductivity (µmhos/c<br>Turbidity (NTUS)<br>Oxidation - Reduction, +/- mv<br>Dissolved Oxygen, ppm<br>Analytical Parameter If Sa<br>Colle   | m) 0.047<br>1000<br>Unple Preservation<br>cted Method<br>4°C   | Volume s<br>Required   | Sample Bottle ILot Nos.   | Gal.                             |
| Irements Field Anal  | Temperature, Deg. C         pH, units         Specific Conductivity (µmhos/c         Turbidity (NTUS)         Oxidation - Reduction, +/- mv         Dissolved Oxygen, ppm         Analytical Parameter         VOCs         SVOCs         Metals  | cm) Course<br>Value Courses and  | Volume s<br>Required<br>32x40 ml<br>2x1 liter AG   | Sample Bottle ILot Nos.   | Gal.                             |
| equirements<br>Location) Field Anal  | Temperature, Deg. C         pH, units         Specific Conductivity (µmhos/c         Turbidity (NTUS)         Oxidation - Reduction, +/- mv         Dissolved Oxygen, ppm         Analytical Parameter         VOCs         SVOCs         Metals         Cyanide         Nitrate/Sulfate  | m) Course<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD<br>VICOD | Volume s<br>Required<br>Zx40 ml<br>2x1 liter AG<br>1x1 liter P<br>1x500mLP   | Sample Bottle ILot Nos.   | Gal.                             |
| 1 Regulrements<br>this Location) Field Anal                                    | Temperature, Deg. C         pH, units         Specific Conductivity (µmhts/c         Turbidity (NTUS)         Oxidation - Reduction, +/- mv         Dissolved Oxygen, ppm         Analytical Parameter         VOCs         SVOCs         Metals         Cyanide         Nitrate/Sulfate         Nitrate/Phosphate  | Imple         Preservation           Cted         Method           Imple         Preservation           Imple         Preservation <t< td=""><td>Volume s<br/>Required<br/>Zex40 ml<br/>2x1 liter AG<br/>1x1 liter P<br/>1x500mLP<br/>1x1 liter P<br/>1x1 liter P</td><td>Sample Bottle ILot Nos.</td><td>Gal.</td></t<>  | Volume s<br>Required<br>Zex40 ml<br>2x1 liter AG<br>1x1 liter P<br>1x500mLP<br>1x1 liter P<br>1x1 liter P  | Sample Bottle ILot Nos.   | Gal.                             |
| stion Requirements<br>ad at this Location) Field Anal                          | Temperature, Deg. C         pH, units         Specific Conductivity (µmhos/c         Turbidity (NTUS)         Oxidation - Reduction, +/- mv         Dissolved Oxygen, ppm         Analytical Parameter         VOCs         SVOCs         Metals         Cyanide         Nitrate/Sulfate         Nitrate/Phosphate         Pest/PCB         TPH   | Imple       Preservation         Imple       Preservation         Cted       Method         Imple       Preservation         Imple       Preservati   | Volume s<br>Required<br>Zx40 ml<br>2x1 liter AG<br>1x1 liter P<br>1x500mLP<br>1x1 liter P<br>1x1 liter P<br>1x1 liter AG<br>2x1 liter AG<br>2x1 liter AG   | Sample Bottle ILot Nos.   | Gal.                             |
| ullection Requirements Field Anal<br>quired at this Location)                  | Temperature, Deg. C         pH, units         Specific Conductivity (µmhts/c         Turbidity (NTUS)         Oxidation - Reduction, +/- mv         Dissolved Oxygen, ppm         Analytical Parameter         VOCs         SVOCs         Metals         Cyanide         Nitrate/Sulfate         Nitrate/Phosphate         Pest/PCB         TPH         TOC                                   | Imple       Preservation         Uf.Uf       Imple         Vec       Imple         4°C       Imple         HN0, 4°C       NaOH, 4°C         HN0, 4°C       HaOH, 4°C         HS0, 4°C       HaOH, 4°C         H, S0, 4°C       HaOH, 4°C  | Volume s<br>Required<br>Z2x40 ml<br>2x1 liter AG<br>1x1 liter P<br>1x500mLP<br>1x1 liter P<br>1x1 liter P<br>1x1 liter P<br>3x1 liter AG<br>2x1 liter AG<br>1x1 liter, P                                 | Sample Bottle ILot Nos.   | Gal.                             |
| i Collection Requirements Field Anal   | Temperature, Deg. C         pH, units         Specific Conductivity (µmhos/c         Turbidity (NTUS)         Oxidation - Reduction, +/- mv         Dissolved Oxygen, ppm         Analytical Parameter         VOCs         SVOCs         Metals         Cyanide         Nitrate/Sulfate         Nitrate/Phosphate         Pest/PCB         TPH         TOC                                   | $\begin{array}{c} \hline \hline \\ $   | Volume g<br>Required<br>32x40 ml<br>2x1 liter AG<br>1x1 liter P<br>1x500mLP<br>1x1 liter P<br>1x1 liter P<br>3x1 liter AG<br>2x1 liter AG<br>2x1 liter AG<br>1x1 liter, P                                | Sample Bottle ILot Nos.   | Gal.                             |
| 1ple Collection Requirements<br>(/ If Required at this Location)<br>Field Anal | Temperature, Deg. C         pH, units         Specific Conductivity (µmhos/c         Turbidity (NTUS)         Oxidation - Reduction, +/- mv         Dissolved Oxygen, ppm         Analytical Parameter       If Sa         VOCs         SVOCs         Metals         Cyanide         Nitrate/Phosphate         Pest/PCB         TPH         TOC         Notes:                                | $\begin{array}{c} \hline \hline \\ $   | Volume S<br>Required<br>32x40 ml<br>2x1 liter AG<br>1x1 liter P<br>1x500mLP<br>1x1 liter P<br>1x1 liter P<br>1x1 liter P<br>1x1 liter AG<br>2x1 liter AG<br>1x1 liter, P                                 | Sample Bottle ILot Nos.   | Gal.                             |
| Sample Collection Requirements Field Anal (~If Required at this Location)      | Temperature, Deg. C         pH, units         Specific Conductivity (µmhos/c         Turbidity (NTUS)         Oxidation - Reduction, +/- mv         Dissolved Oxygen, ppm         Analytical Parameter         VOCs         SVOCs         Metals         Cyanide         Nitrate/Phosphate         Pest/PCB         TPH         TOC         Notes:         SUMCL         MML         MML      | $cm) \underbrace{\bigcirc 1000}_{10000}$ $\underbrace{\bigcirc 10000}_{10000}$   | Voiume s<br>Required<br>ZX40 ml<br>2x1 liter AG<br>1x1 liter P<br>1x500mLP<br>1x1 liter P<br>1x1 liter P<br>1x1 liter AG<br>2x1 liter AG<br>1x1 liter AG<br>1x1 liter AG<br>2x1 liter AG<br>2x1 liter AG | Sample Bottle ILot Nos.   | Gal.                             |
| Sample Collection Requirements<br>(/ If Required at this Location) Field Anal  | Temperature, Deg. C         pH, units         Specific Conductivity (µmhos/c         Turbidity (NTUS)         Oxidation - Reduction, +/- mv         Dissolved Oxygen, ppm         Analytical Parameter         VOCs         SVOCs         Metals         Cyanide         Nitrate/Sulfate         Nitrate/Phosphate         Pest/PCB         TPH         TOC         Notes:         SUMML MMM. | $\begin{array}{c} \begin{array}{c} \hline \\ \hline $  | Volume S<br>Required<br>Zx40 ml<br>2x1 liter AG<br>1x1 liter P<br>1x500mLP<br>1x1 liter P<br>1x1 liter P<br>1x1 liter AG<br>2x1 liter AG<br>1x1 liter, P<br>GROUNDWATEI                                  | Sample Bottle ILot Nos.   | Gal.                             |
| Sample Collection Requirements<br>(/ If Required at this Location) Field Anal  | Temperature, Deg. C         pH, units         Specific Conductivity (µmhos/c         Turbidity (NTUS)         Oxidation - Reduction, +/- mv         Dissolved Oxygen, ppm         Analytical Parameter         VOCs         SVOCs         Metals         Cyanide         Nitrate/Phosphate         Pest/PCB         TPH         TOC         Notes:         SUMML WMW                          | Imple       Preservation         Cted       Method         Imple       Preservation         Cted       HN0, 4°C         HN0, 4°C       HSO, 4°C         HSO, 4°C       HSO, 4°C<   | Voiume s<br>Required<br>32x40 ml<br>2x1 liter AG<br>1x1 liter P<br>1x500mLP<br>1x1 liter P<br>1x1 liter P<br>3x1 liter AG<br>2x1 liter AG<br>1x1 liter, P<br>GROUNDWATEI<br>EC QUALITY ASSI              | Sample Bottle ILot Nos.<br>FIGUE<br>R SAMPLE DATA REP<br>URANCE PROGRAM   | Gal.                             |

ì

ġ.

es como 4010 Project Boring/Well No. SR 6W - 923 Project No. 52039 361206215910 REGION 8 D.C. - GROUP I Client NYSDEC Site Crystal grenery Sheet No. Logged By Ground Elevation Start Date Finish Date B Loob .02 07 20do  $\Omega_{\rm b}$ **Drilling Contractor** Driller's Name **Rig Type** Seolofic, N De mans la Casing Size 11/2 Drilling Method **Protection Level** Auger Size P.I.D. (eV) Vec Soil Drilled Rock Drilled Total Depth Depth to Groundwater/Date Piez, Well Boring ハイデ  $\square$  $\Box$  $\Box$ Monitoring % Sample No. & Penetration/ Recovery (Feet) SPT Blows/6" or Core Rec./Rqd. % USCS Group Symbol Notes on Drilling Sample Type Depth(Feet) Graphic Log (ppm) SPT-N (Blows/Ft.) Lab Tests Sample PI Meter Head Space Description PI Meter Field Scan Diois TUpsoil 10.1 amp, ve, spimp ŧ,1] 1.9 1-4. BIGGIE Ashslog mix dry MDense 4.0  $\frac{1}{2}$ 4-42 Sine as 1-4 4.2-8 27 Birm / Birm 40,1 SM silty said & grind of 2.1 IGM Some cobioles, dif, WG, 4.0 Donge J 8-86 Some as 4.2.8 SMG Sib-9 Black Silly said 20.1 SM fines to micrarse, dry 7-12 4 Bunsilly 0 M Micarse sand of inthe fine gived, trag coboks dry Durge, we 3.3 pm GN [] 6.1 1 Sunea 9.7. 12 SM. LD.1 13 61 @15.5 to 16 Olive / redelish Bun sindy crid of some my 14 MIGh 15 FIGURE 4-6 TYPICAL TEST BORING LOG NYSDEC QUALITY ASSURANCE PROGRAM PLAN ABB Environmental Services, Inc. 9404014D(z) L33

|   | GROUNDWATER SAMPLE   | FIELD DATA RECORD  |
|---|--|--|
| P   | OBECT: REGIONS DW, [LEAMERS-GROUP IF   | Site: Additional Crystal - Corning   |
|   | oject Numbert 36100 2059 0.2   | Date: 0 ( /2-56  |
| S   | ample Location ID: $C RGV = 230100000000000000000000000000000000000$   | Signature of Sampler:  |
|   |  | Cignatule Di Salligiel.  |
| r e<br>Transie  | Well Depth Ft Measured Top of Well<br>Historical Top of Protec<br>Casing   | Well Biser Stick-up Ft. Protective Ft.<br>tive (irom ground) Casing/Well Difference  |
| ll Data   |  | - ProtectiveFt.<br>Casing  |
| r Level/We  | Depth to Water Ft. Well Material: Well Locked?: #  | Well Dia.     2 inch     Water Level Equip. Used:      4 inch    Elect. Cond. Probe      6 inch    Float Activated      Press. Transducer  |
| Wate  | Height of Water Celumn       X       .16 Gal/Ft. (2 in.)      (1 in.)         Ft.       .15 Gal/Ft. (6 in.)      (1 in.)      (1 in.)              | Gal/Vol. Well Integrity: Yes No<br>Prot. Casing Secure<br>Concrete Collar Intact<br>Other  |
| 5   | Purging/Sampling Equipment Used :  | Decontamination Eluide Llood   |
| Equipment Documentat  | ( If Used For) Purging Sampling Equipment ID V V Peristaltic Pump Bailer PVC/Silicon Tubing Hand Pump In-line Filter Press/Vac Filter V Vac Filter | (✓ All That Apply at Location)<br>Methanol (100%)<br>25% Methanol/75% ASTM Type II water<br>Deionized Water<br>Liquinox Solution<br>Hexane<br>HNO <sub>3</sub> /D.I. Water Solution<br>Potable Water<br>None   |
| Jata  | PID: Ambient Airppm Well Mouthppm Purge D  | Sample Observations:<br>Data Collected In-line Cloudy<br>In Container Colored Odor   |
| Fleid Analysis [  | Purge Data     @Gal. @       Temperature, Deg. C   | _Gal. @Gal. @Gal.  |
|   | Analytical Parameter V If Sample Preservation  | Voiume Sample Bottle ILot Nos.   |
| e Collection Requirements<br>(If Required at this Location) | Collected     Method   | Required         2x40 ml         2x1 liter AG         1x1 liter P         1x500mLP         1x1 liter P         3x1 liter AG         2x1 liter AG         1x1 liter P         1x1 liter P |
| Sampl   | punge miter: Nysdec  | FIGURE 4-1<br>GROUNDWATER SAMPLE DATA RECORD<br>C QUALITY ASSURANCE PROGRAM PLAN   |
| 140401  | 4U L22   |  |

GROUNDWATER SAMPLE FIELD DATA RECORD Project: Ь 10N Site: Additiona CNStal Cornin Project Number . . C Date: 0120 61Z Time: Start: 0800 os2 End: 04.2 Sample Location ID 0 Signature of Sampler Well Depth Ft. Measured Top of Well Well Riser Stick-up Ft. Protective Ft. Historical Top of Protective (irom ground) Casing/Well Difference Casing Water Level/Well Data Protective Ft Casing Depth to Water Ft. Well Material: Water Level Equip. Used: ocked? Well Dia 2 inch PVC 4 inch \_Elect. Cond. Probe SS 6 inch Float Activated Press. Transducer .16 Gal/Ft. (2 in.) Gal/Vol. Well Integrity: Yes Height of Water Celumn X No .65 Gal/Ft. (4 in.) Prot. Casing Secure Ft. 1.5 Gal/Ft. (6 in.) Concrete Collar Intact Total Gal Purged \_ Gal/Ft. (\_\_in.) Other Equipment Documentation Purging/Sampling Equipment Used : Decontamination Fluids Used: ( If Used For) Purging Sampling Equipment ID ( All That Apply at Location) Peristaltic Pump Methanol (100%) Submersible Pump 25% Methanol/75% ASTM Type II water Bailer Deionized Water PVC/Silicon Tubing Liquinox Solution Teflon/Silicon Tubing Hexane Airlift HNO3/D.I. Water Solution Hand Pump Potable Water In-line Filter None Press/Vac Filter V Sample Observations: PID: Ambient Air \_\_\_\_\_ppm Well Mouth Purge Data Collected \_\_ppm /In-line Turbid Fleid Analysis Data \_ Clear \_\_Cloudy In Container Colored Odor -Purge Data @ Gal. @ Gal. @ Gal. @ Gal. 0 Gal Temperature, Deg. C Cons pH, units Specific Conductivity (unahos/cm Turbidity (NTUS) Oxidation - Reduction, +/- my Dissolved Oxygen, ppm Analytical Parameter 🖌 If Sample Preservation Volume Sample Bottle ILot Nos. Collected Method Required 2 Sample Collection Requirement V VOCs 32x40 ml 4°C SVOCs 4°C (~ If Required at this Location) 2x1 liter AG Metais HNO, 4°C 1x1 liter P Cyanide NaOH,4°C H\_S0\_,4°C 1x500mLP Nitrate/Sulfate 1x1 liter P Nitrate/Phosphate H,S0, 4°C 1x1 liter P Pest/PCB 4°C 3x1 liter AG \_\_\_\_\_\_TPH -----H\_S0\_,4°C 2x1 liter AG H,S0,4°C 1x1 liter, P 26 107 Notes: Brown Silt FIGURE 4-1 GROUNDWATER SAMPLE DATA RECORD NYSDEC QUALITY ASSURANCE PROGRAM PLAN - ABB Environmental Services 9404014D L22

| Project                             |         |        |          |                |               | Test Borin  | ng Lo          | D <b>g</b><br>Boring/Well I   | No. ,       | Pr       | oject N                | о.                   |        |       |
|-------------------------------------|---------|--------|----------|----------------|---------------|---|----------------|-------------------------------|-------------|----------|------------------------|----------------------|--------|-------|
| Ragi                                | 008     | S D.C  |          | Gr             | 22            | PIL_  |                | GW-0                          | 24          | 32       | e1206                  | 2050                 | 1/0    | 2     |
| Client N                            | YSDE    | C      |          | Site           | Crystal Clean |   | ers Sheet No   |                               | )           | <u> </u> | of                     | 1                    |        |       |
| Logged By                           | Smit    | h      |          | Gro            | und           | Elevation Start Date Finish Date<br>11-1-06 Finish Date |                |                               |             |          | 26                     |                      |        |       |
| Drilling Con                        | tractor | 10000  | λIV      | )              | D             | riller's Name   | eal            |                               | Rig Typ     | e bl     | 0 DT                   | -                    |        |       |
| Drilling Met                        | nod     | CH-P   | ustr     |                | P             | rotection Level   | P              | .I.D. (eV)                    | Casing      | Size     | ,"                     | Auger                | Sizer  |       |
| Soil Drilled                        | 16      | Rock D | rilled   |                | Т             | otal Depth  | epth to        | Groundwater $\sim \mathbb{N}$ | /Date       |          | Piez                   | Well                 | Boring |       |
|                                     |         |        | %.       |                |               |   |                |                               | 9           | ing      | Mo                     | nitoring             |        |       |
| (Feet)<br>9 No. <i>2</i><br>ration/ | y (Fee  | ows/6  | c./Rqd   | r-N<br>/s/Ft.) | iic Loç       | Sa  | mple           |                               | Symb        | n Drill  |                        | 8<br>8               |        | Tests |
| Depth<br>Sample<br>Penet            | Recover | SPT B  | Core Rec | SP<br>(Blow    | Graph         | Desc  | criptio        | n                             | US<br>Group | Notes o  | PI Meter<br>Field Scar | ol Meter<br>Head Spa |        | Lab   |
|                                     |         |        |          |                |               | 0-0.3' Asphalt  |                |                               | 112         |          | 0.0                    |                      |        |       |
| 1                                   |         | 35     | 1        |                |               | 0.3-2.5 black<br>asphalt                                | Lsa<br>bas     | ndsgrave                      |             |          | 0,0                    |                      |        |       |
| 2-                                  |         |        |          |                |               |   |                | 1                             |             |          |                        |                      |        |       |
| 3-                                  |         |        |          |                |               | 2.5-4 light br<br>some f-cG                             | Rave           | leyeysm<br>1, subround        | ed CLCL     |          | 0.0                    |                      |        |       |
| 4_                                  |         |        |          |                |               | mcd. plastici   | ty,so          | mast                          | -{          |          |                        |                      |        |       |
| 5-1                                 |         | 15     | 7.       |                |               | 4-6' same   | as a           | bove                          | -           |          | 0.0                    |                      |        |       |
|                                     |         |        |          |                |               |   |                |                               |             |          |                        |                      |        |       |
| Υ                                   |         |        |          |                |               | 6-8' med. brow  | un sa<br>Iitte | andy grav                     | el GM       |          |                        |                      |        |       |
|                                     |         |        |          |                |               | grouded, loosi  | 2,100          | bist.                         |             |          |                        |                      |        |       |
|                                     |         |        |          |                |               | 8-12' same  | as a           | bove                          |             |          | D.C                    |                      |        |       |
| 9-                                  |         | 25     | 7.       |                |               | weter   | 1.0            |                               | GW          |          |                        | 1                    |        |       |
| 10                                  |         |        |          |                |               |   |                |                               | 4           |          |                        |                      |        |       |
| 11-                                 |         |        |          |                |               |   |                |                               | -           |          |                        |                      |        |       |
| -12                                 |         |        |          |                |               | in he mad b   |                |                               |             | •        |                        | +                    |        |       |
| 13-                                 |         | 15     | */ e     |                |               | Gravel, lit<br>graded, lo                               | He s           | 12T, well-<br>saturate        | h En        |          | 0,C<br>]               | <b>`</b>  <br>}      |        |       |
|                                     |         |        |          |                |               |   |                |                               |             |          |                        |                      | - 1.6  |       |
| LL                                  |         |        |          |                |               |   | 0.011          | TY                            |             | TES      | T BOI                  | RING                 | LOG    |       |
|                                     |         |        |          |                |               | NYSDE   |                | ALITY ASS                     | UHAN(       | )E PI    | RUGH                   |                      | LAN    |       |





esi forme lor Project Project No. 52034 361206285910. Boring/Well No. SR 6W - 025 REGION 8 D.L. - GROUP I Client Site NYSDEC Sheet No. liner of Logged By B Ground Elevation Start Date Finish Date SWW Zoob 11 .0 Zodo **Drilling Contractor** Driller's Name Rig Type seolofic, M DP men 00 Casing Size 11/2 Drilling Method Protection Level Auger Size P.I.D. (eV) Total Depth Soil Drilled Rock Drilled Depth to Groundwater/Date Piez Well Boring  $\Box$  $\Delta$ SPT Blows/6" or Core Rec./Rqd. % Monitoring Sample No. & Penetration/ Recovery (Feet) USCS Group Symbol Notes on Drilling Sample Type Depth(Feet) Graphic Log SPT-N (Blows/Ft.) (ppm) Tests Sample PI Meter Head Space PI Meter Field Scan Description Lab . 0.0.8 DK Burr S-14 loan Scool Lo, 刊 0.8-1 Cobble 1-15 DIVE Brown Sand-chy-gined morst, we, in Densekt ff 15-4 Dive silty org whome Sund, StaffAsh (2-3.2, deep HPIMP BACKGA25 4-5.8 JIVE Brun Siltysand LO.1 Squarel and roose, some Consides S.8-8 gray fine scudy Silt -My rittle and; AG, wet/mist, MP Fyable, Some en janic roots/ wood fryning (0.) Graf Cly-Sundisginal GC Wel/Satureted, 1008, WE, HPIMP 0 LUN 12 - 135 Save as 8-12 b.t. V. Soft & Jahrata 135 - 16 Brown Sund- c/4/16 Sul Squel why we saturate FIGURE 4-6 TYPICAL BORING LOG TEST NYSDEC QUALITY ASSURANCE PROGRAM PLAN ABB Environmental Services, Inc. 9404014D(z) L33



GROUNDWATER SAMPLE FIELD DATA RECORD Project: VE OROVP TI Additiona Site: Cn Lornin ta Project Number -3E Date: 612 05 Time: Start: U 1502 End: 500 2 04.2 Sample Location ID: 0 ø 9 K 0 Signature of Sampler Well Depth Ft. Measured Top of Well Well Bieer Stick-up Ft. Protective Ft. Historical Top of Protective (irom ground) Casing/Well Difference Casing Protective Ft. Water Level/Well Dat Casing Depth to Water FŁ Well Material: Locked? Well Dia 2 inch Water Level Equip, Used: PVC Elect. Cond. Probe 4 inch SS 6 inch Float Activated Press. Transducer .16 Gal/Ft. (2 in.) Gal/Vol. Well Integrity: Yes Height of Water Celumn X No \_.65 Gal/Ft. (4 in.) Prot. Casing Secure Ft. 1.5 Gal/FL (6 in.) Concrete Collar Intact Total Gal Purged \_ Gal/Ft. (\_\_in.) Other Equipment Documentation Purging/Sampling Equipment Used : Decontamination Fluids Used: ( If Used For) Purging Sampling Equipment ID ( All That Apply at Location) Peristaltic Pump Methanol (100%) Submersible Pump 25% Methanol/75% ASTM Type II water Bailer Deionized Water **PVC/Silicon Tubing** Liquinox Solution Teflor/Silicon Tubing Hexane Airlift HNO<sub>3</sub>/D.I. Water Solution Hand Pump Potable Water In-line Filter None Press/Vac Filter V SP-15 Sample Observations: PID: Ambient Air ppm Well Mouth ppm Purge Data Collected ∠ In-line Turbid Analysis Data Clear \_Cloudy In Container Colored Odor Purge Data Gal. Gal. @ Gal. @ Gal. @ Gal Temperature, Deg. C pH, units Cust Specific Conductivity (µmhos/cm) Fleid Turbidity (NTUS) Oxidation - Reduction, +/- mv Dissolved Oxygen, ppm Analytical Parameter 🖌 If Sample Preservation Volume Sample Bottle ILot Nos. Collected Method Required ŋ Sample Collection Requirement V VOCs 4°C 32x40 mi SVOCs 4°C (~ If Required at this Location) 2x1 liter AG Metais HN0, 4°C 1x1-liter P Cyanide NaOH,4°C H,S0,4°C 1x500mLP Nitrate/Sulfate 1x1 liter P Nitrate/Phosphate H<sup>\*</sup>S0, 4°C 1x1 liter P Pest/PCB 4°C 3x1 liter AG TPH H\_S0\_,4°C -----2x1 liter AG TOC H,S0,4°C 1x1 liter P +010 14 Notes: FIGURE 4-1 GROUNDWATER SAMPLE DATA RECORD NYSDEC QUALITY ASSURANCE PROGRAM PLAN - ABB Environmental Services 9404014D L22

# APPENDIX D

# SITE SURVEY



| (  |   |   |  |
|--|---|---|--|
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
| 1990 C   |   | 10  |  |
| DATE   | REVIS   | SIONS   | BY   |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  | DRAWING   | ALTERATION  |  |
| WARN<br>Low  | NG: It is a violation of<br>Article 145. Section 720  | the New York State Educ<br>9. Special Provision 2. fo   | ation<br>r any   |
| person   | n unless he is acting u<br>and Professional Enginee   | nder the direction of a   | ,  |
| alter  | an item in any way. If  | an item bearing the seal  |  |
| engine   | er or land surveyor shi   | all affix to the item his s   | eal  |
| date -   | of such alteration, and   | a specific description of   |  |
| the a  | iteration.  |   | _  |
|  |   |   |  |
|  |   |   |  |
|  |   |   | _ I  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  |   |   |  |
|  | BY:   |   | - 1  |
|  |   |   |  |
|  | DATE:   |   | _  |
|  |   |   | -  |
|  |   | NGINEER   | RS I   |
|  | DATE:   | NGINEER<br>d Environmer   | RS<br>Ital   |
|  | DATE:   | NGINEER<br>d Environmer   | CS<br>Ital   |
| JC   | DATE:   | NGINEER<br>d Environmer<br>u engineer<br>rveying, p.(   | C.   |
| JC   | DATE:<br>Civil an<br>SEPH C. L<br>LAND SU<br>2230 F   | NGINEER<br>d Environmer<br>u Engineer<br>rveying, p.(<br>2enfield Roa   | ING A  |
| JC   | DATE:   | NGINEER<br>d Environmer<br>U ENGINEER<br>RVEYING, P.(<br>Penfield Roa<br>New York 1.  | Tral<br>ING A<br>C.<br>d<br>4526   |
| JC   | DATE:   | <b>NGINEER</b><br><b>d Environmer</b><br>U ENGINEER<br>RVEYING, P.(<br><sup>2</sup> enfield Roa<br><sup>2</sup> enfield Roa<br><sup>377-1450</sup><br><sup>377-1450</sup>   | The second secon   |
| Jc   | DATE:   | NGINEER<br>d Environmer<br>U ENGINEER<br>RVEYING, Poa<br>Penfield Roa<br>New York 1<br>377–1450<br>35) 377–12   | ING A<br>c.<br>d<br>4526<br>66   |
| JC   | DATE:<br>Civil an<br>USEPH C. L<br>LAND SUU<br>2230 F<br>Penfield,<br>(585)<br>FAX: (58   | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.(.<br>Penfield Roa<br>New York 1.<br>377–1450<br>35) 377–12   | Tral<br>ING A<br>C.<br>d<br>4526<br>66   |
| JC   | DATE:<br>Civil an<br>DSEPH C. L<br>LAND SUU<br>2230 F<br>Penfield,<br>(585)<br>FAX: (58<br>CT:  | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.(.<br>Penfield Roa<br>New York 1.<br>377–1450<br>35) 377–12   | ING A<br>C.<br>d<br>4526<br>66   |
| JC   | DATE:<br>Civil an<br>USEPH C. L<br>LAND SUU<br>2230 F<br>Penfield,<br>(585)<br>FAX: (58   | NGINEER<br>d Environmer<br>U ENGINEER<br>RVEYING, P.(<br>Penfield Roa<br>New York 1.<br>377–1450<br>35) 377–12  | RS<br>ING A<br>C.<br>d<br>4526<br>66   |
| JC   | DATE:<br>Civil and<br>SEPH C. L<br>LAND SUI<br>2230 F<br>Penfield,<br>(585)<br>FAX: (58<br>CT:  | NGINEER<br>d Environmer<br>U ENGINEER<br>RVEYING, P.(<br>Penfield Roa<br>New York 1<br>377–1450<br>35) 377–12   | ING A  |
| JC   | DATE:   | NGINEER<br>d Environmer<br>U ENGINEER<br>RVEYING, P.(<br>Penfield Roa<br>New York 1<br>377–1450<br>35) 377–12   | <b>C</b><br><b>I</b> NG A<br>C.<br>d<br>4526<br>66   |
|  | DATE:   | NGINEER<br>d Environmer<br>U ENGINEER<br>RVEYING, P.(<br>Penfield Roa<br>New York 1<br>377-1450<br>35) 377-12<br>CLEANERS   | C.<br>Mal<br>ING A<br>4526<br>66   |
| JC<br>JC   | DATE:   | NGINEER<br>d Environmer<br>U ENGINEER<br>RVEYING, P.(<br>Penfield Roa<br>New York 1<br>377–1450<br>35) 377–12<br>CLEANERS<br>NEY STRE   | S<br>Ing A<br>C.<br>d<br>4526<br>666   |
| JC<br>JC<br>PROJEC   | DATE:<br>Civil an<br>DEPH C. L<br>LAND SU<br>2230 F<br>Penfield,<br>(585)<br>FAX: (58<br>CTE<br>CTE<br>CTE<br>CTE<br>CTE<br>CTE<br>CTE<br>CTE                       | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.0<br>Penfield Roa<br>New York 1<br>377–1450<br>35) 377–12<br>Sign Street<br>CLEANERS  | <b>2</b><br>Ital<br>ING A<br>4526<br>66<br>EET   |
| JC<br>JC   | DATE:<br>Civil an<br>USEPH C. L<br>LAND SU<br>2230 F<br>Penfield,<br>(585)<br>FAX: (58<br>ST:<br>CI:<br>CI:<br>CI:<br>CI:<br>CI:<br>CI:<br>CI:<br>CI                | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.0<br>Penfield Roa<br>New York 1:<br>377–1450<br>35) 377–12<br>CLEANERS<br>NEY STRE  | 25<br>ital<br>ING A<br>4526<br>66<br>EET   |
| JC<br>JC<br>PROJEC   | DATE:   | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.(.<br>Penfield Roa<br>New York 1.<br>377–1450<br>35) 377–12<br>CLEANERS<br>NEY STRE   | 25<br>ital<br>ING A<br>d<br>4526<br>66   |
| JC<br>JC<br>PROJEC<br>WE:  | DATE:<br>CVI E<br>CVI an<br>SEPH C. L<br>LAND SUI<br>2230 F<br>Penfield,<br>(585)<br>FAX: (58<br>ST:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>C          | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.d.<br>Penfield Roa<br>New York 1.<br>377–1450<br>35) 377–12<br>CLEANERS<br>NEY STRE   | 25<br>Ing A<br>100 A   |
|  | DATE:<br>Civil and<br>SEPH C. L<br>LAND SUI<br>2230 F<br>Penfield,<br>(585)<br>FAX: (58<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:                     | NGINEER<br>d Environmer<br>U ENGINEER<br>RVEYING, P.(<br>Penfield Roa<br>New York 1.<br>377–1450<br>35) 377–12<br>CLEANERS  | <b>C</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b>   |
| JC<br>JC<br>PROJEC<br>CLIENT<br>MA   | DATE:   | NGINEER<br>d Environmer<br>U ENGINEER<br>RVEYING, P.(<br>Penfield Roa<br>New York 1.<br>377–1450<br>35) 377–12<br>CLEANERS<br>NEY STRE  | C.   |
| JC<br>JC<br>PROJEC<br>WES  | DATE:   | NGINEER<br>d Environmer<br>U ENGINEER<br>RVEYING, P.(<br>Penfield Roa<br>New York 1.<br>377–1450<br>35) 377–12<br>CLEANERS<br>CLEANERS<br>NEY STRE  | <b>C</b> .<br><b>C</b> .<br><b>C</b> .<br><b>C</b> .<br><b>C</b> .<br><b>C</b> .<br><b>C</b> .   |
| JC<br>JC<br>PROJEC<br>WES  | DATE:   | NGINEER<br>d Environmer<br>U ENGINEER<br>RveyING, P.(<br>Penfield Roa<br>New York 1<br>377–1450<br>35) 377–12<br>CLEANERS<br>NEY STRE<br>NEY STRE<br>NEERING, IN<br>ESS STREET<br>D, ME 04101   | <b>C</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b>   |
| JC<br>JC<br>PROJEG<br>WES  | DATE:   | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.0<br>2007 2007 2007<br>New York 1<br>377–1450<br>377–1450<br>35) 377–12<br>CLEANERS<br>CLEANERS<br>NEY STREE<br>NEERING, IN<br>ESS STREET<br>D, ME 04101  | <b>25</b><br>Ital<br>ING A<br>4526<br>66<br>EET  |
| JC<br>JC<br>PROJEG<br>CLIENT<br>MLA  | DATE:<br>Civil an<br>SEPH C. L<br>LAND SUI<br>2230 F<br>Penfield,<br>(585)<br>FAX: (58<br>CTE<br>CTE<br>CTE<br>CTE<br>CTEC ENGI<br>511 CONGR<br>PORTLANI            | NGINEER<br>d Environmer<br>RVEYING, P.0<br>Penfield Roa<br>New York 1<br>377–1450<br>35) 377–12<br>CLEANERS<br>NEY STRE<br>NEY STRE<br>NEERING, IN<br>ESS STREET<br>D, ME 04101   | <b>2S</b><br>ital<br>ING A<br>4526<br>66<br><b>6</b><br><b>6</b><br><b>6</b><br><b>6</b><br><b>6</b><br><b>6</b><br><b>6</b><br><b>6</b><br><b>6</b>   |
|  | DATE:   | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.0.<br>Penfield Roa<br>New York 1:<br>377–1450<br>35) 377–12<br>CLEANERS<br>NEY STRE<br>NEY STRE<br>NEERING, IN<br>ESS STREET<br>D, ME 04101   | 25<br>ital<br>ING A<br>4526<br>66<br>SET   |
| DRAWIN   | DATE:<br>CIVIT AND<br>USEPH C. L<br>LAND SUI<br>2230 F<br>Penfield,<br>(585)<br>FAX: (58<br>TR<br>CTE<br>CTE<br>CTE<br>CTE<br>CTE<br>CONGR<br>PORTLANI<br>16 TITLE: | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.(.<br>Penfield Roa<br>New York 1<br>377–1450<br>35) 377–12<br>CLEANERS<br>NEY STRE<br>NEY STRE<br>NEY STRE<br>NESS STREET<br>D, ME 04101  | 25<br>ital<br>ING A<br>4526<br>66<br>SET   |
| DRAWIN   | CTEC ENGI<br>ST PULTE   | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.(.<br>Penfield Roa<br>New York 1.<br>377–1450<br>35) 377–12<br>CLEANERS<br>CLEANERS<br>NEY STRE<br>NEY STRE<br>NEERING, IN<br>ESS STREET<br>D, ME 04101   | 25<br>ital<br>ING A<br>4526<br>66<br>SET   |
| DROJECT<br>PROJECT<br>CLIENT<br>MLA<br>DRAWIN  | DATE:   | NGINEER<br>d Environmer<br>U ENGINEER<br>RVEYING, P.(.<br>Penfield Roa<br>New York 1.<br>377–1450<br>35) 377–12<br>CLEANERS<br>CLEANERS<br>NEY STRE<br>NEY STRE<br>NEERING, IN<br>ESS STREET<br>D, ME 04101   | <b>C</b> .<br><b>C</b> . |
|  | DATE:   | NGINEER<br>d Environmer<br>U ENGINEER<br>RVEYING, P.(<br>Penfield Ros<br>New York 1.<br>377–1450<br>35) 377–12<br>CLEANERS<br>CLEANERS<br>NEY STRE<br>NEY STRE<br>NEY STRE<br>NEERING, IN<br>ESS STREET<br>D, ME 04101  | C.   |
| DROJEC<br>PROJEC<br>CLIENT<br>MA<br>DRAWIN   | DATE:<br>DATE:<br>DEPH C. L<br>LAND SUI<br>2230 F<br>Penfield,<br>(585)<br>FAX: (58<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:                         | NGINEER<br>d Environmer<br>U ENGINEER<br>RVEYING, P.0<br><sup>30</sup> enfield Roa<br>New York 1<br>377–1450<br>35) 377–12<br>CLEANERS<br>CLEANERS<br>NEY STRE<br>NEY STRE<br>NEERING, IN<br>ESS STREET<br>D, ME 04101  | <b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>C</b><br><b>C</b><br><b>C</b><br><b>C</b><br><b>C</b><br><b>C</b><br><b>C</b><br><b>C</b>   |
| PROJEC<br>CLIENT<br>MA<br>DRAWIN   | DATE:<br>DATE:<br>DEPH C. L<br>LAND SUI<br>2230 F<br>Penfield,<br>(585)<br>FAX: (58<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:                         | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.<br>2007 2014 Roa<br>New York 11<br>377-1450<br>35) 377-12<br>CLEANERS<br>NEY STRE<br>NEY STRE<br>NEERING, IN<br>ESS STREET<br>D, ME 04101  | <b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b>   |
| DROJEG   | DATE:<br>DATE:<br>DEPH C. L<br>LAND SUI<br>2230 F<br>Penfield,<br>(585)<br>FAX: (58<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:                         | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.0<br>2007 2014 Roa<br>New York 1<br>377-1450<br>35) 377-12<br>CLEANERS<br>CLEANERS<br>NEY STRE<br>NEERING, IN<br>ESS STREET<br>D, ME 04101  | <b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b>   |
| PROJECT<br>CLIENT<br>MLA<br>DRAWIN<br>EI<br>DESIGN   | DATE:   | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.<br>2007 - 1450<br>35) 377 - 12<br>CLEANERS<br>NEY STRE<br>NEERING, IN<br>ESS STREET<br>D, ME 04101<br>IMENTA<br>AP   | C.   |
| PROJECT<br>CLIENT<br>ML2<br>DRAWIN<br>EI<br>DESIGN<br>CHECKI   | DATE:   | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.<br>20enfield Roa<br>New York 1-<br>377-1450<br>35) 377-12<br>CLEANERS<br>INEY STRE<br>NEERING, IN<br>ESS STREET<br>D, ME 04101<br>NMENTA<br>AP   | <b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b><br><b>S</b>   |
| DESIGN<br>CHECK  | DATE:   | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, P.<br>20enfield Roa<br>New York 1-<br>377-1450<br>35) 377-12<br>CLEANERS<br>NEY STRE<br>NEY STRE<br>STRE<br>NEY STRE<br>NEY STRE<br>STRE<br>NEY STRE<br>STRE<br>STRE<br>STRE<br>STRE<br>STRE<br>STRE<br>STRE                                    | C.<br>SET<br>C.<br>C.<br>C.<br>C.<br>C.<br>C.<br>C.<br>C.<br>C.<br>C.  |
| DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DRAWIN<br>DR | DATE:<br>DATE:<br>DEPH C. L<br>LAND SU<br>2230 F<br>Penfield,<br>(585)<br>FAX: (58<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:<br>CT:                          | NGINEER<br>d Environmer<br>V ENGINEER<br>RVEYING, Po<br>20nfield Roa<br>New York 1-<br>377-1450<br>35) 377-12<br>CLEANERS<br>NEY STRE<br>NEY STRE<br>STRE<br>NEY STRE<br>NEY STRE<br>STRE<br>NEY STRE<br>STRE<br>STRE<br>STRE<br>NEY STRE<br>STRE<br>STRE<br>STRE<br>STRE<br>STRE<br>STRE<br>STRE | 25<br>ital<br>ING A<br>4526<br>66<br>SET<br>IC.  |

# **APPENDIX E**

## DATA USABILITY SUMMARY REPORT

# DATA USABILITY SUMMARY REPORT 2006 SAMPLING EVENT REGION 8 DRY CLEANERS-CRYSTAL CLEANERS CORNING, NEW YORK

# **Introduction:**

Soil, water, and air samples were collected at the Crystal Cleaners site in January 2006 and submitted for off-site laboratory analyses. Samples were analyzed by Chemtech located in Mountainside, NJ. A listing of samples included in this investigation is presented in Table 1. A summary of analytical results is presented in Appendix D, Tables 1.1-1.5. Samples were analyzed for the following parameters:

- Soil: Contract Laboratory Program (CLP) procedures for volatile organic compounds (VOCs).
- Water: CLP procedures for volatile organic compounds (VOCs)
- Air: EPA Method TO-15 for VOCs

Deliverables for the off-site laboratory analyses included a Category B deliverable as defined in the New York State Department of Environmental Conservation (NYSDEC) Analytical Services Protocols (NYSDEC, 1995; NYSDEC, 2000).

A project chemist review was completed based on NYSDEC Division of Environmental Remediation guidance for Data Usability Summary Reports (NYSDEC, 1997). Laboratory QC limits were used during the data evaluation unless noted otherwise. The project chemist review included evaluations of sample collection, data package completeness, holding times, QC data (blanks, instrument calibrations, duplicates, surrogate recovery, and spike recovery), data transcription, electronic data reporting, calculations, and data qualification. With the exception of the items discussed below, results are interpreted to be usable as reported by the laboratory. The following qualifiers are used in the final data presentation.

U = target analyte is not detected at the reported detection limit

J = concentration is estimated

UJ = target analyte is not detected at the reported detection limit and is estimated

R = target analyte was rejected

Results are interpreted to be usable as reported by the laboratory unless discussed in the following sections.

# **Air - Volatile Organic Compounds**

#### **Blank Contamination**

A detection of 1,1,1-trichloroethane  $(1.64\mu g/m^3)$  was reported in the method blank. An action level was calculated at five times the detection reported in the blank for 1,1,1-trichloroethane. Sample CRSV00100101XX had a detection for 1,1,1-trichloroethane that was less than the action level and was qualified as non-detect (U).

A detection of acetone  $(0.55\mu g/m^3)$  was reported in the method blank. An action level was calculated at ten times the detection found in the blank for acetone. Samples CRGV001000601XX, CRGV00200601XX, and CRGV00300601XX had detections for acetone that were less than the action level and were qualified as non-detect (U).

## Internal Standards

Sample CRGV001000601XX had two internal standards, 1,4-difluorobenzene and chlorobenzene-d5, that had area counts below control limits. Compounds associated with these internal standards in sample CRGV001000601XX were qualified as estimated (J/UJ).

# Initial Calibration

The initial calibration associated with samples CRGV001000601XX, CRGV00200601XX, and CRGV00300601XX had a correlation coefficient that was less than the control limit of 0.995 for hexachloro-1,3-butadiene (0.991). Results for this compound in the above samples were non-detect and were qualified as estimated (UJ). The relative response factor (RRF) for 1,3-butadiene (0.035) was less than the response limit of 0.05. Results for 1,3-butadiene in samples CRGV001000601XX, CRGV00200601XX, and CRGV00300601XX were positive and were qualified as estimated (J). The result for 1,3-butadiene in sample CRVS00100101XX was non-detect and was qualified as rejected (R).

## Laboratory Control Samples

The LCS percent recovery for 1,4-dioxane (8) was below laboratory control limits of 65-135. Since the recovery for 1,4-dioxane was below 10% the non-detect results in sample CRSV00100101XX, CRGV001000601XX, CRGV00200601XX, and CRGV00300601XX were qualified as rejected (R).

# Soil and Water Samples - Volatile Organic Compounds

# **Holding Times and Sample Collection**

The following samples and/or reanalysis (RE)/dilution runs (DL) were outside of the fourteen day holding time. Results for these samples were qualified as estimated (J/UJ).

- Samples CRGW01203401XX (DL), CRGW01502501XX, CRGW01502101XA, CRGW01602101XX, CRGW01601701XA, CRGW01702001XX, and CRGW01701601XA were sampled on 1/23/06 and were analyzed on 2/7/06
- Samples CRGW01202201XA, CRGW01403001XX, and CRGW01402201XA were sampled on 1/23/06 and were analyzed on 2/8/06
- Sample CRGW00501601XD (RE) was sampled on 1/25/06 and analyzed on 2/18/06

#### **Internal Standards**

All three internal standards had area counts that were low and outside of control limits in sample CRGS00500801XX. Results for CRGS00500801XX were all non-detect and reporting limits were qualified as estimated (UJ).

## **Surrogates**

The percent recoveries for all three surrogates were greater than laboratory control limits in samples CRGW00601401XA and CRGW00901401XA indicating a high bias. Positive results in samples CRGW00601401XA and CRGW00901401XA were qualified as estimated (J).

The percent recovery for the surrogate 1,2-dichloroethane-d4 (117) in sample CRGW00902201XXDL was greater than laboratory control limits, indicating a high bias. Positive results in sample CRGW00902201XXDL were qualified as estimated (J).

## **Blank Contamination**

Detections of acetone (4.9  $\mu$ g/L to 20  $\mu$ g/L), methylene chloride (1.0  $\mu$ g/L, 0.82  $\mu$ g/kg), and chloroform (2.4  $\mu$ g/L) were reported in the trip and method blanks. An action level was calculated at ten times the concentration in the blank for acetone and methylene chloride, and five times the concentration for chloroform. Samples CRSW00100001XX, CRSW00200001XX, CRGW00803701XX, CRGW01202201XA, CRGW01403001XX, CRGW01701601XA, CRGS00100601XX, CRGW00702201XA (RE), CRGW01002601XX, CRGW01102201XA, CRGW01302501XX, CRGW007012D, CRGW00402001XB, CRGS00400701XX, CRGW00501601XA, CRGW00601401XA, CRGW00602201XX, CRGW00902201XX, CRGW00501601XDRE, CRGS00200701XX, CRGS00300701XX, and CRGS00500801XX had detections for acetone that were less than the action level and were qualified as non-detect (U). Samples CRGW01602101XX and CRGW01601701XA had detections for chloroform that were less than the action level and was also qualified as non-detect (U).

#### Initial Calibration

The initial calibration associated with samples CRGS00400701XX, CRGS00400701XD, and CRGS00100601XX had a percent relative standard deviation that was greater than the validation limit of 30 for acetone (32). Results for acetone were non-detect in samples and were qualified as estimated (UJ).

The initial calibration analyzed on 1/28/06 had a percent relative standard deviation for trichlorofluoromethane (38.1) that was greater than the control limit of 30. Results for trichlorofluoromethane in samples CRGS00200701XX, CRGS00300701XX, and CRGS00500801XX were non-detect and were qualified as estimated (UJ).

#### **Continuing Calibration**

A continuing calibration percent differences greater than the validation limit of 25 was reported for chloromethane (34.2) and vinyl chloride (28.0). Sample CRSW00100001XX had a positive detection for vinyl chloride and was qualified as estimated (J). Samples CRSW00200001XX, CRGW00803701XX, CRGW00802501XA, and CRGW01203401XX had non-detect results for vinyl chloride and chloromethane and were qualified as estimated (UJ).

A continuing calibration percent differences greater than the validation limit of 25 was reported for chloroethane (31.3), vinyl chloride (27.1), acetone (34.8), methyl acetate (33.1), 2-butanone (33.1), 4-methyl-2-pentanone (31.6), and 2-hexanone (30.0). Results for these compounds in

samples CRGW01502501XX, CRGW01502101XA, CRGW01602101XX, CRGW01601701XA, CRGW01702001XX, and CRGW01701601XA were non-detect and were qualified as estimated (UJ).

A continuing calibration percent differences greater than the validation limit of 25 was reported for chloromethane (37.4), vinyl chloride (32.8), acetone (37.2), methyl-tert-butyl-ether (27.8), methyl acetate (35.4), 1,1-dichloroethane (31.0), 2-butanone (29.3), 4-methyl-2-pentanone (25.5), and 2-hexanone (26.2). Results for these compounds in samples CRGW01202201XA, CRGW01403001XX, CRGW01402201XA, CRGW00401201XA, and CRGW00402001XB were non-detect for these compounds and were qualified as estimated (UJ).

A continuing calibration percent differences greater than the validation limit of 25 was reported for acetone (-43.0), methyl acetate (-29.9), 2-butanone (-36.9), carbon tetrachloride (-34.0), 1,2-dichloroethane (-32.2), and 1,2-dibromo-3-chloropropane (-33.5). Results for these compounds in samples CRGW00703401XX, CRGW01002601XX, CRGW01103001XX, CRGW00703401XD, CRGW00102601XX, CRGW00202601XX, CRGW00201801XA, CRGW00302601XX, CRGW00302601XX, CRGW00302601XX, were non-detect and were qualified as estimated (UJ).

A continuing calibration percent differences greater than the validation limit of 25 was reported for chloromethane (41.9), vinyl chloride (32.7), cyclohexane (32.5), 2-butanone (45.3), 4-methyl-2-pentanone (41.2), and 2-hexanone (44.3). Results for these compounds in samples CRGW00702201XA (RE), CRGW01302501XX, and CRGW01301701XA were non-detect for all compounds except vinyl chloride in two samples. Results were qualified as estimated (UJ). Samples CRGW01302501XX and CRGW01301701XA had positive results for vinyl chloride and were qualified as estimated (J).

A continuing calibration percent differences greater than the validation limit of 25 was reported for acetone (-46.1), 2-butanone (-60.9), 1,1,1-tetrachloroethane (-31.4), 1,2-dichlorethane (-33.4), tetrachloroethene (-34.4), 1,2-dibromo-3-chloropropane (-34.7), and 1,2,4-trichlorobenzene (-33.9). Results for tetrachloroethene in samples CRGW00501601XA, CRGW00601401XA, CRGW00901401XA, CRGW00602201XX, and CRGW00902201XX were positive and were qualified as estimated (J). All other compounds listed above were non-detect and were qualified as estimated (UJ).

A continuing calibration percent differences greater than the validation limit of 25 was reported for 2-hexanone (167.9). The result for 2-hexanone in sample CRGW00501601XDRE was nondetect and was qualified as estimated (UJ).

## Matrix Spike/Matrix Spike Duplicate

The matrix spike duplicate associated with sample CRGW00803701XX had a percent recovery for 1,1-dichloroethene (58) that was below the laboratory established limits. In addition, the relative percent difference between the matrix spike and matrix spike duplicate for 1,1-dichloroethene (29) was also outside of laboratory limits. The result for 1,1-dichloroethene in sample CRGW00803701XX was non-detect and was qualified as estimated (UJ).

## **Tentatively Identified Compounds**

Tentatively identified compounds (TICs) were reported by the laboratory in accordance with CLP method procedures. TICs reported in samples are presented in Table 1.5. Only samples that had TICs reported are included on Table 1.5. If a sample is not listed, no TICs were reported.

| SDG   | Sample Name    | Date Collected | Method   | Parameter        | Туре |
|-------|----------------|----------------|----------|------------------|------|
| X1322 | CRSW00100001XX | 1/23/06        | OLM 04.2 | VOC              | FS   |
| X1322 | CRSW00100001MS | 1/23/06        | OLM 04.2 | VOC              | MS   |
| X1322 | CRSW00100001MD | 1/23/06        | OLM 04.2 | VOC              | MD   |
| X1322 | CRSW00200001XX | 1/23/06        | OLM 04.2 | VOC              | FS   |
| X1322 | CRGW00803701XX | 1/23/06        | OLM 04.2 | VOC              | FS   |
| X1322 | CRGW00803701MS | 1/23/06        | OLM 04.2 | VOC              | MS   |
| X1322 | CRGW00803701MD | 1/23/06        | OLM 04.2 | VOC              | MD   |
| X1322 | CRGW00802501XA | 1/23/06        | OLM 04.2 | VOC              | FS   |
| X1322 | CRGW01203401XX | 1/23/06        | OLM 04.2 | VOC              | FS   |
| X1322 | CRGW01202201XA | 1/23/06        | OLM 04.2 | VOC              | FS   |
| X1322 | CRGW01403001XX | 1/23/06        | OLM 04.2 | VOC              | FS   |
| X1322 | CRGW01402201XA | 1/23/06        | OLM 04.2 | VOC              | FS   |
| X1322 | CRGW01502501XX | 1/23/06        | OLM 04.2 | VOC              | FS   |
| X1322 | CRGW01502101XA | 1/23/06        | OLM 04.2 | VOC              | FS   |
| X1322 | CRGW01602101XX | 1/23/06        | OLM 04.2 | VOC              | FS   |
| X1322 | CRGW01601701XA | 1/23/06        | OLM 04.2 | VOC              | FS   |
| X1322 | CRGW01702001XX | 1/23/06        | OLM 04.2 | VOC              | FS   |
| X1322 | CRGW01701601XA | 1/23/06        | OLM 04.2 | VOC              | FS   |
| X1322 | CRQTOO1XXX01XX | 1/23/06        | OLM 04.2 | VOC              | TB   |
| X1339 | CRGS00100601XX | 1/24/06        | OLM 04.2 | VOC              | FS   |
| X1339 | CRGS00100601XX | 1/24/06        | D2216    | Percent Moisture | FS   |
| X1339 | CRGS00100601MS | 1/24/06        | OLM 04.2 | VOC              | MS   |
| X1339 | CRGS00100601MS | 1/24/06        | D2216    | Percent Moisture | MS   |
| X1339 | CRGS00100601MD | 1/24/06        | OLM 04.2 | VOC              | MD   |
| X1339 | CRGS00100601MD | 1/24/06        | D2216    | Percent Moisture | MD   |
| X1339 | CRGW00102601XX | 1/24/06        | OLM 04.2 | VOC              | FS   |
| X1339 | CRGW00102601MS | 1/24/06        | OLM 04.2 | VOC              | MS   |
| X1339 | CRGW00102601MD | 1/24/06        | OLM 04.2 | VOC              | MD   |
| X1339 | CRGW00703401XX | 1/24/06        | OLM 04.2 | VOC              | FS   |
| X1339 | CRGW00703401XD | 1/24/06        | OLM 04.2 | VOC              | FD   |
| X1339 | CRGW00702201XA | 1/24/06        | OLM 04.2 | VOC              | FS   |
| X1339 | CRGW01002601XX | 1/24/06        | OLM 04.2 | VOC              | FS   |
| X1339 | CRGW01103001XX | 1/24/06        | OLM 04.2 | VOC              | FS   |
| X1339 | CRGW01102201XA | 1/24/06        | OLM 04.2 | VOC              | FS   |
| X1339 | CRGW01302501XX | 1/24/06        | OLM 04.2 | VOC              | FS   |
| X1339 | CGRW01301701XA | 1/24/06        | OLM 04.2 | VOC              | FS   |
| X1339 | CRQTOO2XXX01XX | 1/24/06        | OLM 04.2 | VOC              | TB   |
| X1380 | CRSV00100101XX | 1/26/06        | TO-15    | VOC              | FS   |
| X1377 | CRGS00400701XX | 1/26/06        | OLM 04.2 | VOC              | FS   |
| X1377 | CRGS00400701XX | 1/26/06        | D2216    | Percent Moisture | FS   |
| X1377 | CRGS00400701XD | 1/26/06        | OLM 04.2 | VOC              | FD   |
| X1377 | CRGS00400701XD | 1/26/06        | D2216    | Percent Moisture | FD   |

## TABLE 1

P:\Projects\nysdec1\projects\Region 8 Dry Cleaning Sites\4.0 Project Deliverables\4.1 Reports\Crystal\Draft Final\Appendix E\DUSR\_Crystal\_Jan\_2006.doc

| SDG   | Sample Name    | Date Collected | Method   | Parameter        | Туре |
|-------|----------------|----------------|----------|------------------|------|
| X1377 | CRGW00403001XX | 1/26/06        | OLM 04.2 | SVOC             | FS   |
| X1377 | CRGW00401201XA | 1/26/06        | OLM 04.2 | VOC              | FS   |
| X1377 | CRGV00100601XX | 1/26/06        | TO-15    | VOC              | FS   |
| X1377 | CRGV00200601XX | 1/26/06        | TO-15    | VOC              | FS   |
| X1377 | CRGV00300601XX | 1/26/06        | TO-15    | VOC              | FS   |
| X1377 | CRQTOO4XXX01XX | 1/26/06        | OLM 04.2 | VOC              | TB   |
| X1377 | CRGW00402001XB | 1/26/06        | OLM 04.2 | VOC              | FS   |
| X1362 | CRGS00200701XX | 1/25/06        | OLM 04.2 | VOC              | FS   |
| X1362 | CRGS00200701XX | 1/25/06        | D2216    | Percent Moisture | FS   |
| X1362 | CRGS00300701XX | 1/25/06        | OLM 04.2 | VOC              | FS   |
| X1362 | CRGS00300701XX | 1/25/06        | D2216    | Percent Moisture | FS   |
| X1362 | CRGS00500801XX | 1/25/06        | OLM 04.2 | VOC              | FS   |
| X1362 | CRGS00500801XX | 1/25/06        | D2216    | Percent Moisture | FS   |
| X1362 | CRGW00202601XX | 1/25/06        | OLM 04.2 | VOC              | FS   |
| X1362 | CRGW00201801XA | 1/25/06        | OLM 04.2 | VOC              | FS   |
| X1362 | CRGW00302601XX | 1/25/06        | OLM 04.2 | VOC              | FS   |
| X1362 | CRGW00301801XA | 1/25/06        | OLM 04.2 | VOC              | FS   |
| X1362 | CRGW00502601XX | 1/25/06        | OLM 04.2 | VOC              | FS   |
| X1362 | CRGW00501601XA | 1/25/06        | OLM 04.2 | VOC              | FS   |
| X1362 | CRGW00501601XD | 1/25/06        | OLM 04.2 | VOC              | FD   |
| X1362 | CRGW00602201XX | 1/25/06        | OLM 04.2 | VOC              | FS   |
| X1362 | CRGW00601401XA | 1/25/06        | OLM 04.2 | VOC              | FS   |
| X1362 | CRGW00902201XX | 1/25/06        | OLM 04.2 | VOC              | FS   |
| X1362 | CRGW00901401XA | 1/25/06        | OLM 04.2 | VOC              | FS   |
| X1362 | CRQTOO3XXX01XX | 1/25/06        | OLM 04.2 | VOC              | TB   |

#### **Reference:**

New York State Department of Environmental Conservation (NYSDEC), 1995. "Analytical Services Protocols"; 10/95 Edition; October 1995.

New York State Department of Environmental Conservation (NYSDEC), 1997. "Guidance for the Development of Data Usability Reports"; Division of Environmental Remediation; September 1997.

# DATA USABILITY SUMMARY REPORT 2006 SAMPLING EVENT REGION 8 DRY CLEANERS-CRYSTAL CLEANERS ELMIRA, NEW YORK

## **Introduction:**

Water samples were collected at the Crystal Cleaners site in October-November 2006 and submitted for off-site laboratory analyses. Samples were analyzed by Chemtech located in Mountainside, NJ. A listing of samples included in this investigation is presented in Table 1. A summary of analytical results is presented in Appendix D, Tables 1.6-1.7. Samples were analyzed for the following parameters:

• Water: Contract Laboratory Program (CLP) procedures for volatile organic compounds (VOCs)

Deliverables for the off-site laboratory analyses included a Category B deliverable as defined in the New York State Department of Environmental Conservation (NYSDEC) Analytical Services Protocols (NYSDEC, 2000).

A project chemist review was completed based on NYSDEC Division of Environmental Remediation guidance for Data Usability Summary Reports (NYSDEC, 2002). Laboratory QC limits were used during the data evaluation unless noted otherwise. The project chemist review included evaluations of sample collection, data package completeness, holding times, QC data (blanks, instrument calibrations, duplicates, surrogate recovery, and spike recovery), data transcription, electronic data reporting, calculations, and data qualification. With the exception of the items discussed below, results are interpreted to be usable as reported by the laboratory. The following qualifiers are used in the final data presentation.

U = target analyte is not detected at the reported detection limit

J = concentration is estimated

UJ = target analyte is not detected at the reported detection limit and is estimated

Results are interpreted to be usable as reported by the laboratory unless discussed in the following sections.

# Water Samples - Volatile Organic Compounds

#### **Surrogates**

The percent recovery for the surrogate, d8-toluene (87), was below the laboratory control limits in sample CRGW02302701 indicating a potential low bias. Positive and non-detected results for CRGW02302701 were qualified as estimated (J/UJ).

#### **Blank Contamination**

Detections of acetone (4.8 to 9.1  $\mu$ g/L) were reported in the method blanks and trip blank. An action level was calculated at ten times the maximum acetone detection reported in the blanks. The low level detections of acetone in samples CRGW01901901XX, CRMW00601301XX,
CRGW02102601XX, CRGW02201801XX, CRGW02501201XD, and CRGW02502201XX were less than the action level and were qualified as non-detect (U).

# Initial Calibration

The initial calibration had a percent relative standard deviation that was greater than the control limit of 30 for chloroethane (40). Chloroethane was not detected in the samples, and quantitation limits were qualified as estimated (UJ) in all samples.

# **Continuing Calibration**

Percent differences between the initial calibration average relative response factors (RRFs) and continuing calibration RRFs for dichlorodifluoromethane (27), trichlorofluoromethane (30), chloromethane (24), carbon disulfide (23), 1,1,2-trichlorotrifluoroethane (24), cyclohexane (20), and 1,1-dichloroethene (23) were greater than the control limit of 20. Results for these analytes were non-detect in the associated samples, and quantitation limits for dichlorodifluoromethane, trichlorofluoromethane, carbon disulfide, 1,1,2-trichlorotrifluoroethane, cyclohexane, and 1,1-dichloroethene were qualified as estimated (UJ) in samples CRGW01901901XX, CRMW00501201XX, and CRMW00601301XX.

The percent difference between the initial calibration average RRF and continuing calibration RRF for methyl acetate (70) was above the control limit of 20. Methyl acetate was not detected in the associated samples, and quantitation limits for methyl acetate were qualified as estimated (UJ) in samples CRGW01902901XX, CRMW0102001XX, CRMW01001301XX, CRGW02101901XX, CRGW02102601XX, CRGW02201801XX, and CRGW02202801XX.

Percent differences between the initial calibration average RRFs and continuing calibration RRFs for dichlorodifluoromethane (20), trichlorofluoromethane (22), acetone (-22), and 2-butanone (-25) were greater than the control limit of 20. Results for these analytes were non-detect in the associated samples, and quantitation limits for dichlorodifluoromethane, trichlorofluoromethane, acetone, and 2-butanone were qualified as estimated (UJ) in samples CRGW02201801XX(RE), CRGW02402301XX, CRGW02401301XX, CRGW02501201XX, CRGW02501201XD, CRGW02502201XX, CRGW01802001XX, CRGW01803001XX, CRGW02001901XX, and CRGW02002901XX.

Percent differences between the initial calibration average RRFs and continuing calibration RRFs for dichlorodifluoromethane (27), chloromethane (22), vinyl chloride (24), trichlorofluoromethane (28), 1,1,2-trichlorotrifluoroethane (26), 1,1-dichloroethene (27), carbon disulfide (24), methyl acetate (74), cyclohexane (30), carbon tetrachloride (24), and 1,1,1-trichloroethane (25) were greater than the control limit of 20. These analytes were not detected in the associated samples, and quantitation limits for dichlorodifluoromethane, chloromethane, vinyl chloride, trichlorofluoromethane, 1,1,2-trichlorotrifluoroethane, 1,1-dichloroethene, carbon disulfide, methyl acetate, cyclohexane, carbon tetrachloride, and 1,1,1-trichloroethane in samples CRGW02301901XX and CRGW20302701XX were qualified as estimated (UJ).

# Laboratory Control Samples

Percent recovery for trichloroethene (130) in the LCS associated with a subset of samples was above the laboratory control limits, indicating a potential high bias. The positive detection of trichloroethene in sample CRMW00501201XX was qualified as estimated (J) and may be biased high.

# **Internal Standards**

The instrument response for internal standard 1,4-difluorobenzene in sample CRGW02201801XX(RE) was below control limits. Due to surrogate recovery problems for the initial analysis of CRGW02201801XX, the reanalysis (RE) was selected for final reporting. Results for all target analytes associated with the out of control internal standard were qualified as estimated in sample CRGW02201801XX(RE). The following analytes associated with the non-compliant internal standard were not detected in sample CRGW02201801XX(RE) and quantitation limits were qualified as estimated (UJ): 2-butanone, 1,1,1-trichloroethane, carbon tetrachloride, benzene, 1,2-dichloroethane, trichloroethene, methylcyclohexane, 1,2-dichloropropene, bromodichloromethane, trans-1,3-dichloropropene, cis-1,3-dichloropropene, 1,1,2-trichloroethane, 1,2-dibromoethane, and bromoform.

# **Tentatively Identified Compounds**

Tentatively identified compounds (TICs) were reported by the laboratory in accordance with CLP method procedures. TICs reported in samples are presented in Table 1.7. Only samples that had TICs reported are included on Table 1.7. If a sample is not listed, no TICs were reported.

| SDG   | Sample Name    | Date Collected | Method   | Parameter | Туре |
|-------|----------------|----------------|----------|-----------|------|
| X5203 | CRGW01901901XX | 11/1/06        | OLM 04.2 | VOC       | FS   |
| X5203 | CRGW01902901XX | 11/1/06        | OLM 04.2 | VOC       | FS   |
| X5203 | CRMW00102001XX | 11/1/06        | OLM 04.2 | VOC       | FS   |
| X5203 | CRMW00102001MS | 11/1/06        | OLM 04.2 | VOC       | MS   |
| X5203 | CRMW00102001MD | 11/1/06        | OLM 04.2 | VOC       | MD   |
| X5203 | CRMW00501201XX | 11/1/06        | OLM 04.2 | VOC       | FS   |
| X5203 | CRMW00601301XX | 11/1/06        | OLM 04.2 | VOC       | FS   |
| X5203 | CRMW01001301XX | 11/1/06        | OLM 04.2 | VOC       | FS   |
| X5203 | CRGW02101901XX | 11/1/06        | OLM 04.2 | VOC       | FS   |
| X5203 | CRGW02102601XX | 11/1/06        | OLM 04.2 | VOC       | FS   |
| X5203 | CRGW02201801XX | 11/1/06        | OLM 04.2 | VOC       | FS   |
| X5203 | CRGW02202801XX | 11/1/06        | OLM 04.2 | VOC       | FS   |
| X5203 | CRGW02402301XX | 11/1/06        | OLM 04.2 | VOC       | FS   |
| X5203 | CRGW02401301XX | 11/1/06        | OLM 04.2 | VOC       | FS   |
| X5203 | CRGW02501201XX | 11/1/06        | OLM 04.2 | VOC       | FS   |
| X5203 | CRGW02501201XD | 11/1/06        | OLM 04.2 | VOC       | FD   |
| X5203 | CRGW02502201XX | 11/1/06        | OLM 04.2 | VOC       | FS   |
| X5203 | CRTB001XXX02XX | 11/1/06        | OLM 04.2 | VOC       | TB   |
| X5268 | CRGW01802001XX | 11/2/06        | OLM 04.2 | VOC       | FS   |
| X5268 | CRGW01803001XX | 11/2/06        | OLM 04.2 | VOC       | FS   |
| X5268 | CRGW02001901XX | 11/2/06        | OLM 04.2 | VOC       | FS   |
| X5268 | CRGW02002901XX | 11/2/06        | OLM 04.2 | VOC       | FS   |
| X5268 | CRGW02301901XX | 11/2/06        | OLM 04.2 | VOC       | FS   |
| X5268 | CRGW02302701XX | 11/2/06        | OLM 04.2 | VOC       | FS   |

# TABLE 1

### **Reference:**

New York State Department of Environmental Conservation (NYSDEC), 2000. "Analytical Services Protocols"; June 2000.

New York State Department of Environmental Conservation (NYSDEC), 2002. "Technical Guidance for Site Investigation and Remediation-Appendix 2B"; Draft DER-10; Division of Environmental Remediation; December 2002.

Data Validator: Julie Ricardi January 26, 2007

QA Officer: Chris Ricardi NRCC-EAC

Chris Ricardo

Date: 2/2/07

| Lab Sample Id                         | X1330    | -01       | X1362    | -01         | X1362-02         | X1362-02 X1362-03 |           | ¥1377-01 |           | X1377-02  |           |
|---------------------------------------|----------|-----------|----------|-------------|------------------|-------------------|-----------|----------|-----------|-----------|-----------|
| Lab Sample Delivery Group             | X13      | 39        | X13      | 52          | X1362            | X130              | 52        | X137     | 77        | X137      | 77        |
| Location Name                         | GS-      | 1         | GS-      | 2           | GS-3             | GS-               | 5         | GS-      | 4         | GS-       | 4         |
| Field Sample Id                       | CRGS0010 | 0601XX    | CRGS0020 | -<br>0701XX | CRGS00300701XX   | CRGS0050          | 0801XX    | CRGS0040 | 0701XX    | CRGS0040  | 0701XD    |
| Field Sample Date                     | 1/24/2   | 006       | 1/25/2   | 006         | 1/25/2006        | 1/25/2006         |           | 1/26/2   | 006       | 1/26/2006 |           |
| Qc Code                               | FS       |           | FS       |             | FS               | FS                |           | FS       |           | FD        |           |
| Parameter Name                        | Result   | Qualifier | Result   | Qualifier   | Result Qualifier | Result            | Qualifier | Result   | Qualifier | Result    | Qualifier |
| 1,1,1-Trichloroethane                 | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| 1,1,2,2-Tetrachloroethane             | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| 1,1,2-Trichloro-1,2,2-Trifluoroethane | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| 1,1,2-Trichloroethane                 | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| 1,1-Dichloroethane                    | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| 1,1-Dichloroethene                    | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| 1,2,4-Trichlorobenzene                | 0.58     | J         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| 1,2-Dibromo-3-chloropropane           | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| 1,2-Dibromoethane                     | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| 1,2-Dichlorobenzene                   | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| 1,2-Dichloroethane                    | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| 1,2-Dichloropropane                   | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| 1,3-Dichlorobenzene                   | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| 1,4-Dichlorobenzene                   | 0.59     | J         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| 2-Butanone                            | 57       | U         | 57       | U           | 59 U             | 61                | UJ        | 56       | U         | 58        | U         |
| 2-Hexanone                            | 57       | U         | 57       | U           | 59 U             | 61                | UJ        | 56       | U         | 58        | U         |
| 4-Methyl-2-pentanone                  | 57       | U         | 57       | U           | 59 U             | 61                | UJ        | 56       | U         | 58        | U         |
| Acetic acid, methyl ester             | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Acetone                               | 57       | UJ        | 57       | U           | 59 U             | 61                | UJ        | 56       | UJ        | 58        | UJ        |
| Benzene                               | 0.73     | J         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 0.65      | J         |
| Bromodichloromethane                  | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Bromoform                             | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Bromomethane                          | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Carbon disulfide                      | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Carbon tetrachloride                  | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Chlorobenzene                         | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Chlorodibromomethane                  | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Chloroethane                          | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Chloroform                            | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Chloromethane                         | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Cis-1,2-Dichloroethene                | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| cis-1,3-Dichloropropene               | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Cyclohexane                           | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Dichlorodifluoromethane               | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Ethyl benzene                         | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Isopropylbenzene                      | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Methyl cyclohexane                    | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Methyl Tertbutyl Ether                | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Methylene chloride                    | 10       | U         | 1.5      | J           | 1.6 J            | 12                | UJ        | 11       | U         | 12        | U         |
| o-Xylene                              | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |
| Styrene                               | 11       | U         | 11       | U           | 12 U             | 12                | UJ        | 11       | U         | 12        | U         |

### Appendix E Table 1.1: Soil VOC Results

#### Lab Sample Id X1339-01 X1362-01 X1362-02 X1362-03 X1377-01 X1377-02 Lab Sample Delivery Group X1339 X1362 X1362 X1362 X1377 X1377 Location Name GS-1 GS-2 GS-3 GS-5 GS-4 GS-4 CRGS00100601XX CRGS00200701XX CRGS00300701XX CRGS00500801XX CRGS00400701XX CRGS00400701XD Field Sample Id Field Sample Date 1/24/2006 1/25/2006 1/25/2006 1/25/2006 1/26/2006 1/26/2006 Qc Code FS FS FS FS FS FD Result Qualifier Result Qualifier Result Qualifier Result Qualifier Result Qualifier Result Qualifier Parameter Name Tetrachloroethene 0.64 J 11 U 12 U 12 UJ 11 U 12 U 0.66 J 11 U 12 U 12 UJ 11 U 0.8 J Toluene trans-1,2-Dichloroethene 11 U 11 U 12 U 12 UJ 11 U 12 U trans-1,3-Dichloropropene 11 U 11 U 12 U 12 UJ 11 U 12 U 11 U 12 U 12 UJ 11 U Trichloroethene 11 U 12 U Trichlorofluoromethane 1.6 J 11 UJ 12 UJ 12 UJ 11 U 12 U Vinyl chloride 11 U 11 U 12 U 12 UJ 11 U 12 U 12 U Xylene, m/p 1.3 J 11 U 12 UJ 11 U 12 U

#### Appendix E Table 1.1: Soil VOC Results

Notes:

Results in micrograms per kilogram (µg/kg) Samples analyzed for VOCs by EPA method OLM04.2

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

U = Not detected at a concentration above the reporting limit

J = Estimated Value

| Lab Sample Id  | X1322-05     | X1322-08             | X1322-09            | X1322-10           | X1322-11         | X1322-12         | X1322-13         | X1322-14         | X1322-15         |
|--|--------------|----------------------|---------------------|--------------------|------------------|------------------|------------------|------------------|------------------|
| Lab Sample Delivery Group  | X1322        | X1322                | X1322               | X1322              | X1322            | X1322            | X1322            | X1322            | X1322            |
| Loc Name   | GW-8         | GW-8                 | GW-12               | GW-12              | GW-14            | GW-14            | GW-15            | GW-15            | GW-16            |
| Field Sample Id  | CRGW00803701 | XX CRGW00802501X     | A CRGW01203401XX    | CRGW01202201XA     | CRGW01403001XX   | CRGW01402201XA   | CRGW01502501XX   | CRGW01502101XA   | CRGW01602101XX   |
| Field Sample Date  | 1/23/2006    | 1/23/2006            | 1/23/2006           | 1/23/2006          | 1/23/2006        | 1/23/2006        | 1/23/2006        | 1/23/2006        | 1/23/2006        |
| Oc Code  | FS           | FS                   | FS                  | FS                 | FS               | FS               | FS               | FS               | FS               |
| Param Name   | Result Qua   | ifier Result Qualifi | er Result Qualifier | r Result Qualifier | Result Qualifier |
| 1,1,1-Trichloroethane  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| 1,1,2,2-Tetrachloroethane  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| 1,1,2-Trichloro-1,2,2-Trifluoroethane  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| 1,1,2-Trichloroethane  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| 1,1-Dichloroethane   | 10 UJ        | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| 1,1-Dichloroethene   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| 1,2,4-Trichlorobenzene   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| 1,2-Dibromo-3-chloropropane  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| 1,2-Dibromoethane  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| 1,2-Dichlorobenzene  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| 1,2-Dichloroethane   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| 1,2-Dichloropropane  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| 1,3-Dichlorobenzene  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| 1,4-Dichlorobenzene  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| 2-Butanone   | 50 U         | 50 U                 | 50 U                | 50 UJ              | 50 UJ            | 50 UJ            | 50 UJ            | 50 UJ            | 50 UJ            |
| 2-Hexanone   | 50 U         | 50 U                 | 50 U                | 50 UJ              | 50 UJ            | 50 UJ            | 50 UJ            | 50 UJ            | 50 UJ            |
| 4-Methyl-2-pentanone   | 50 U         | 50 U                 | 50 U                | 50 UJ              | 50 UJ            | 50 UJ            | 50 UJ            | 50 UJ            | 50 UJ            |
| Acetic acid, methyl ester  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Acetone  | 50 U         | 50 U                 | 50 U                | 50 UJ              | 50 UJ            | 50 UJ            | 50 UJ            | 50 UJ            | 50 UJ            |
| Benzene  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Bromodichloromethane   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Bromoform  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Bromomethane   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Carbon disulfide   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Carbon tetrachloride   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Chlorobenzene  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Chlorodibromomethane   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Chloroethane   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Chloroform   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Chloromethane  | 10 UJ        | 10 UJ                | 10 UJ               | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Cis-1,2-Dichloroethene   | 0.85 J       | 3.6 J                | 25                  | 5.5 J              | 5.2 J            | 2.2 J            | 8.3 J            | 7 J              | 10 UJ            |
| cis-1,3-Dichloropropene  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Cyclohexane  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Dichlorodifluoromethane  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Ethyl benzene  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Isopropylbenzene   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Methyl cyclohexane   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Methyl Tertbutyl Ether   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Methylene chloride   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| o-Xylene   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Styrene  | 10 U         | 10 U                 | 10 U<br>210 DI      | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Tetrachioroethene  | 130          | 24                   | 210 DJ              | 42 J               | 13 J             | 3.4 J            | 87 J             | 65 J             | 0.3 J            |
| Toluene  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| trans-1,2-Dichlorogenene   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Trichloroethone  | 10 U         | 10 U                 | 10 0                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            | 10 UJ            |
| Theilding and the second secon | 10 U         | 0.95 J               | 0.9 J               | 3 J                | 1./ J            | 10 UJ            | 2.7 J            | 2.3 J            | 10 UJ            |
| Vinyl ablarida   | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 UI            | 10 UI            | 10 UJ            |
| Videna m/n   | 10 UJ        | 10 UJ                | 10 U                | 10 UI              | 10 UI            | 10 UI            | 10 UI            | 10 UI            | 10 UI            |
| Ayrene, m/p  | 10 U         | 10 U                 | 10 U                | 10 UJ              | 10 UJ            | 10 UJ            | 10 03            | 10 UJ            | 10 UJ            |

Notes: Results in micrograms per liter (µg/L)

Samples analyzed for VOCs by EPA method OLM04.2

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

U = Not detected at a concentration above the reporting limit

J = Estimated value

| Lab Sample Id                         | X1322-16         | X1322-17         | X1322-18         | X1339-04         | X1339-07         | X1339-08         | X1339-09RE       | X1339-10         | X1339-11         |
|---------------------------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|
| Lab Sample Delivery Group             | X1322            | X1322            | X1322            | X1339            | X1339            | X1339            | X1339            | X1339            | X1339            |
| Loc Name                              | GW-16            | GW-17            | GW-17            | GW-1             | GW-7             | GW-7             | GW-7             | GW-10            | GW-11            |
| Field Sample Id                       | CRGW01601701XA   | CRGW01702001XX   | CRGW01701601XA   | CRGW00102601XX   | CRGW00703401XX   | CRGW00703401XD   | CRGW00702201XA   | CRGW01002601XX   | CRGW01103001XX   |
| Field Sample Date                     | 1/23/2006        | 1/23/2006        | 1/23/2006        | 1/24/2006        | 1/24/2006        | 1/24/2006        | 1/24/2006        | 1/24/2006        | 1/24/2006        |
| Qc Code                               | FS               | FS               | FS               | FS               | FS               | FD               | FS               | FS               | FS               |
| Param Name                            | Result Qualifier |
| 1,1,1-Trichloroethane                 | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| 1,1,2,2-Tetrachloroethane             | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| 1,1,2-Trichloro-1,2,2-Trifluoroethane | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| 1,1,2-Trichloroethane                 | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| 1,1-Dichloroethane                    | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| 1,1-Dichloroethene                    | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| 1,2,4-Trichlorobenzene                | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| 1,2-Dibromo-3-chloropropane           | 10 UJ            | 10 U             | 10 UJ            | 10 UJ            |
| 1,2-Dibromoethane                     | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| 1,2-Dichlorobenzene                   | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| 1,2-Dichloroethane                    | 10 UJ            | 10 U             | 10 UJ            | 10 UJ            |
| 1,2-Dichloropropane                   | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| 1,3-Dichlorobenzene                   | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| 1,4-Dichlorobenzene                   | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| 2-Butanone                            | 50 UJ            |
| 2-Hexanone                            | 50 UJ            | 50 UJ            | 50 UJ            | 50 U             | 50 U             | 50 U             | 50 UJ            | 50 U             | 50 U             |
| 4-Methyl-2-pentanone                  | 50 UJ            | 50 UJ            | 50 UJ            | 50 U             | 50 U             | 50 U             | 50 UJ            | 50 U             | 50 U             |
| Acetic acid, methyl ester             | 10 UJ            | 10 U             | 10 UJ            | 10 UJ            |
| Acetone                               | 50 UJ            | 50 U             | 50 UJ            | 50 UJ            |
| Benzene                               | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| Bromodichloromethane                  | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| Bromotorm                             | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| Bromomethane                          | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| Carbon disulfide                      | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| Carbon tetrachioride                  | 10 UJ            | 10 U             | 10 UJ            | 10 UJ            |
| Chlorobenzene                         | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| Chlorodibromolienane                  | 10 UJ            | 10 UJ            | 10 UI            | 10 U             |
| Chlorofenne                           | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| Chloromethana                         | 10 UJ            | 10 UI            | 10 UI            | 10 U             |
| Circl 2 Dishloroothana                | 10 UI            | 2 1              | 2 1              | 10 U             | 0.52 I           | 10 U             | 15 1             | 7.2 1            | 10 U             |
| cis 1.3 Dichloropropene               | 10 UI            | 10 111           | 10 11            | 10 U             | 0.55 J           | 10 U             | 1.5 5            | 7.2 J            | 10 U             |
| Cyclobeyane                           | 10 UI            | 10 UI            | 10 UI            | 10 U             | 10 U             | 10 U             | 10 UI            | 10 U             | 10 U             |
| Dichlorodifluoromethane               | 10 UI            | 10 UI            | 10 UI            | 10 U             |
| Ethyl benzene                         | 10 UI            | 10 UI            | 10 UI            | 10 U             | 10 U             | 10 U             | 10 U             | 281              | 10 U             |
| Isopropylbenzene                      | 10 UI            | 10 UI            | 10 UI            | 10 U             | 10 U             | 10 U             | 10 U             | 1.8 J            | 10 U             |
| Methyl cyclohexane                    | 10 UI            | 10 UI            | 10 UI            | 10 U             | 10 U             | 10 U             | 10 U             | 4.4 I            | 10 U             |
| Methyl Terthutyl Ether                | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| Methylene chloride                    | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| o-Xylene                              | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| Styrene                               | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| Tetrachloroethene                     | 7.1 J            | 67 J             | 64 J             | 11               | 230 D            | 200 D            | 110              | 27               | 0.84 J           |
| Toluene                               | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| trans-1,2-Dichloroethene              | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| trans-1,3-Dichloropropene             | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| Trichloroethene                       | 10 UJ            | 1.5 J            | 1.4 J            | 10 U             | 1.5 J            | 1.3 J            | 1.4 J            | 4.4 J            | 10 U             |
| Trichlorofluoromethane                | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             |
| Vinyl chloride                        | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             | 10 U             | 10 U             | 10 UJ            | 10 U             | 10 U             |
| Xylene, m/p                           | 10 UJ            | 10 UJ            | 10 UJ            | 10 U             | 10 U             | 10 U             | 10 U             | 2.4 J            | 10 U             |

Notes: Results in micrograms per liter (µg/L)

Samples analyzed for VOCs by EPA method OLM04.2

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

U = Not detected at a concentration above the reporting limit

J = Estimated value

| Lab Sample Id                         | X1339-12        | X1339-13           | X1339-14         | X1362-04         | X1362-05         | X1362-06         | X1362-07         | X1362-08         | X1362-09         |
|---------------------------------------|-----------------|--------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|
| Lab Sample Delivery Group             | X1339           | X1339              | X1339            | X1362            | X1362            | X1362            | X1362            | X1362            | X1362            |
| Loc Name                              | GW-11           | GW-13              | GW-13            | GW-2             | GW-2             | GW-3             | GW-3             | GW-5             | GW-5             |
| Field Sample Id                       | CRGW01102201XA  | CRGW01302501XX     | CRGW01301701XA   | CRGW00202601XX   | CRGW00201801XA   | CRGW00302601XX   | CRGW00301801XA   | CRGW00502601XX   | CRGW00501601XA   |
| Field Sample Date                     | 1/24/2006       | 1/24/2006          | 1/24/2006        | 1/25/2006        | 1/25/2006        | 1/25/2006        | 1/25/2006        | 1/25/2006        | 1/25/2006        |
| Qc Code                               | FS              | FS                 | FS               | FS               | FS               | FS               | FS               | FS               | FS               |
| Param Name                            | Result Qualifie | r Result Qualifier | Result Qualifier |
| 1,1,1-Trichloroethane                 | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 UJ            |
| 1,1,2,2-Tetrachloroethane             | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| 1,1,2-Trichloro-1,2,2-Trifluoroethane | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| 1,1,2-Trichloroethane                 | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| 1,1-Dichloroethane                    | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| 1,1-Dichloroethene                    | 10 U            | 1 J                | 1.6 J            | 10 U             |
| 1,2,4-Trichlorobenzene                | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 UJ            |
| 1,2-Dibromo-3-chloropropane           | 10 U            | 10 U               | 10 U             | 10 UJ            |
| 1,2-Dibromoethane                     | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| 1,2-Dichlorobenzene                   | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| 1,2-Dichloroethane                    | 10 U            | 10 U               | 10 U             | 10 UJ            |
| 1,2-Dichloropropane                   | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| 1,5-Dichlorobenzene                   | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| 1,4-Dichlorobenzene                   | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| 2-Butanone                            | 50 UJ           | 50 UJ              | 50 UJ            | 50 UJ            | 50 UJ            | 50 UJ            | 50 UJ            | 50 UJ            | 50 UJ            |
| 2-Hexanone                            | 50 UJ           | 50 UJ              | 50 UJ            | 50 U             |
| 4-Methyl-2-pentanone                  | 50 UJ           | 50 UJ              | 50 UJ            | 50 U             |
| Acetic acid, metnyl ester             | 10 U            | 10 U               | 10 U             | 10 UJ            | 10 U             |
| Acetone                               | 30 U            | 50 U               | 50 U             | 30 UJ            | 30 UJ            | 50 UJ            | 30 UJ            | 30 UJ            | 30 UJ            |
| Bromodiabloromethane                  | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Bromoform                             | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Bromomethane                          | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Carbon disulfide                      | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Carbon disunde                        | 10 U            | 10 U               | 10 U             | 10 UI            | 10 U             |
| Chlorobenzene                         | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Chlorodibromomethane                  | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Chloroethane                          | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Chloroform                            | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Chloromethane                         | 10 UJ           | 10 UJ              | 10 UJ            | 10 U             |
| Cis-1.2-Dichloroethene                | 10 U            | 440 D              | 580 D            | 10 U             | 1.8 J            |
| cis-1,3-Dichloropropene               | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Cyclohexane                           | 10 UJ           | 10 UJ              | 10 UJ            | 10 U             |
| Dichlorodifluoromethane               | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Ethyl benzene                         | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Isopropylbenzene                      | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Methyl cyclohexane                    | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Methyl Tertbutyl Ether                | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Methylene chloride                    | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| o-Xylene                              | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Styrene                               | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Tetrachloroethene                     | 9.7 J           | 390 D              | 820 D            | 0.93 J           | 2.1 J            | 0.74 J           | 0.88 J           | 3 J              | 5.5 J            |
| Toluene                               | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| trans-1,2-Dichloroethene              | 10 U            | 6.5 J              | 6.3 J            | 10 U             |
| trans-1,3-Dichloropropene             | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Trichloroethene                       | 10 U            | 120                | 180              | 10 U             | 0.57 J           |
| Trichlorofluoromethane                | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |
| Vinyl chloride                        | 10 UJ           | 14 J               | 56 J             | 10 U             |
| Xylene, m/p                           | 10 U            | 10 U               | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             | 10 U             |

Notes: Results in micrograms per liter (µg/L)

Samples analyzed for VOCs by EPA method OLM04.2

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

U = Not detected at a concentration above the reporting limit

J = Estimated value

|                                       | r                | 1                | 1                | r                |                  | r                | r                |                  |
|---------------------------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|
| Lab Sample Id                         | X1362-10RE       | X1362-11         | X1362-12         | X1362-13         | X1362-14         | X1377-03         | X1377-04         | X1377-09         |
| Lab Sample Delivery Group             | X1362            | X1362            | X1362            | X1362            | X1362            | X1377            | X1377            | X1377            |
| Loc Name                              | GW-5             | GW-6             | GW-6             | GW-9             | GW-9             | GW-4             | GW-4             | GW-4             |
| Field Sample Id                       | CRGW00501601XD   | CRGW00602201XX   | CRGW00601401XA   | CRGW00902201XX   | CRGW00901401XA   | CRGW00403001XX   | CRGW00401201XA   | CRGW00402001XB   |
| Field Sample Date                     | 1/25/2006        | 1/25/2006        | 1/25/2006        | 1/25/2006        | 1/25/2006        | 1/26/2006        | 1/26/2006        | 1/26/2006        |
| Qc Code                               | FD               | FS               |
| Param Name                            | Result Qualifier |
| 1,1,1-Trichloroethane                 | 10 UJ            | 10 U             | 10 U             | 10 U             |
| 1,1,2,2-Tetrachloroethane             | 10 UJ            | 10 U             |
| 1,1,2-Trichloro-1,2,2-Trifluoroethane | 10 UJ            | 10 U             |
| 1,1,2-Trichloroethane                 | 10 UJ            | 10 U             | 10 U             | 10 U             | 10 U             | 3.9 J            | 10 U             | 10 U             |
| 1,1-Dichloroethane                    | 10 UJ            | 10 U             | 10 UJ            | 10 UJ            |
| 1,1-Dichloroethene                    | 10 UJ            | 10 U             |
| 1,2,4-Trichlorobenzene                | 10 UJ            | 10 U             | 10 U             | 10 U             |
| 1,2-Dibromo-3-chloropropane           | 10 UJ            | 10 U             | 10 U             | 10 U             |
| 1,2-Dibromoethane                     | 10 UJ            | 10 U             |
| 1,2-Dichlorobenzene                   | 10 UJ            | 10 U             |
| 1,2-Dichloroethane                    | 10 UJ            | 10 U             | 10 U             | 10 U             |
| 1,2-Dichloropropane                   | 10 UJ            | 10 U             |
| 1,3-Dichlorobenzene                   | 10 UJ            | 10 U             |
| 1,4-Dichlorobenzene                   | 10 UJ            | 10 U             |
| 2-Butanone                            | 50 UJ            | 50 U             | 50 UJ            | 50 UJ            |
| 2-Hexanone                            | 50 UJ            | 50 U             | 50 UJ            | 50 UJ            |
| 4-Methyl-2-pentanone                  | 50 UJ            | 50 U             | 50 UJ            | 50 UJ            |
| Acetic acid, methyl ester             | 10 UJ            | 10 U             | 10 UJ            | 10 UJ            |
| Acetone                               | 50 UJ            | 50 U             | 50 UJ            | 50 UJ            |
| Benzene                               | 10 UJ            | 10 U             |
| Bromodichloromethane                  | 10 UJ            | 10 U             |
| Bromoform                             | 10 UJ            | 10 U             |
| Bromomethane                          | 10 UJ            | 10 U             |
| Carbon disulfide                      | 10 UJ            | 10 U             |
| Carbon tetrachloride                  | 10 UJ            | 10 U             |
| Chlorobenzene                         | 10 UJ            | 10 U             |
| Chlorodibromomethane                  | 10 UJ            | 10 U             |
| Chloroethane                          | 10 UJ            | 10 U             |
| Chloroform                            | 10 UJ            | 10 U             |
| Chloromethane                         | 10 UJ            | 10 U             | 10 UJ            | 10 UJ            |
| Cis-1,2-Dichloroethene                | 2.1 J            | 10 U             | 10 U             | 2.8 J            | 10 U             | 10 U             | 10 U             | 10 U             |
| cis-1,3-Dichloropropene               | 10 UJ            | 10 U             |
| Cyclohexane                           | 10 UJ            | 10 U             | 10 UJ            | 10 UJ            |
| Dichlorodifluoromethane               | 10 UJ            | 10 U             |
| Ethyl benzene                         | 10 UJ            | 10 U             |
| Isopropylbenzene                      | 10 UJ            | 10 U             |
| Methyl cyclohexane                    | 10 UJ            | 10 U             |
| Methyl Tertbutyl Ether                | 10 UJ            | 10 U             | 10 UJ            | 10 UJ            |
| Methylene chloride                    | 10 UJ            | 10 U             |
| o-Xylene                              | 10 UJ            | 10 U             |
| Styrene                               | 10 UJ            | 10 U             |
| Tetrachloroethene                     | 5.4 J            | 4.1 J            | 7.9 J            | 280 DJ           | 74 J             | 610 D            | 15               | 15               |
| Toluene                               | 10 UJ            | 0.55 J           | 10 U             |
| trans-1,2-Dichloroethene              | 10 UJ            | 10 U             |
| trans-1,3-Dichloropropene             | 10 UJ            | 10 U             |
| Trichloroethene                       | 10 UJ            | 10 U             | 10 U             | 11               | 1.9 J            | 10 U             | 10 U             | 10 U             |
| Trichlorofluoromethane                | 10 UJ            | 10 U             |
| Vinyl chloride                        | 10 UJ            | 10 U             | 10 UJ            | 10 UJ            |
| Xylene, m/p                           | 10 UJ            | 10 U             |
|                                       |                  |                  |                  |                  |                  |                  |                  |                  |

Notes:

Results in micrograms per liter (µg/L)

Samples analyzed for VOCs by EPA method OLM04.2

QC Code:

FS = Field Sample

FD = Field Duplicate

Qualifiers:

U = Not detected at a concentration above the reporting limit

J = Estimated value

| Lab Sample Delivery Group         X1322         X1322           Loc Name         SW-1         SW-2           Field Sample Date         I/23/2006         I/23/2006           Param Name         Result         Qualifier         Result         Qualifier           1,1,1-Trichloroethane         10         0         10         0         0           1,1,2-Trichloroethane         10         0         10         0 <th>Lab Sample Id</th> <th>X1322</th> <th>-01</th> <th colspan="5">X1322-04</th>  | Lab Sample Id                         | X1322    | -01         | X1322-04  |                |  |  |  |
|---|---------------------------------------|----------|-------------|-----------|----------------|--|--|--|
| Loc Name         SW-1         SW-2           Field Sample 1d         CRSW0010001XX         CRSW0020001XX           Field Sample Date         IZ32/2006         IZ32/2006           Qc Code         FS         FS           Param Name         Result         Qualifier         Result         Qualifier           1,1.1-Trichloro-1,2,2-Trifluoroethane         10         10         U         10         U           1,1.2-Trichloroothane         10         U         10         U         10         U           1,1.2-Trichloroothane         10         U         10         U         10         U           1,2-Trichloroothane         10         U         10         U         10         U           1,2-Trichlorobenzene         10         U         10         U         10         U           1,2-Dichlorobenzene         10         U         10  | Lab Sample Delivery Group             | X132     | 22          | X13       | 322            |  |  |  |
| Field Sample 1d         CRSW0010001XX         CRSW0020001XX           Field Sample Date         1/23/2006         1/23/2006           Qe Code         FS         FS           Param Name         Result         Qualifier         Result         Qualifier           1,1,2.7-Trichlorocthane         10         10         10         10           1,1,2.7-Trichlorocthane         10         10         10         11           1,1.2.7-Trichlorocthane         10         10         10         11           1,1.2.7-Trichlorocthane         10         10         10         11           1,1.2.7-Trichlorocthane         10         10         10         11           1,2-Dichrocothane         10         10         10         11           1,2-Dichroropane         10         10   | Loc Name                              | SW-      | -1          | SW        | /-2            |  |  |  |
| Field Sample Date         1/23/2006         1/23/2006           Param Name         Result         Qualifier         Result         Qualifier           1,1,1-Trichloroethane         10         10         10         10           1,1,2-Trichloro-1,2,2-Trifluroethane         10         10         10         10           1,1,2-Trichloro-1,2,2-Trifluroethane         10         10         10         10           1,1,2-Trichloro-1,2,2-Trifluroethane         10         10         10         10           1,1-Dichloroethane         10         10         10         11           1,2-Trichloro-1,2,2-Trifluroethane         10         10         11           1,2-Arrichlorobenzene         10         10         11           1,2-Arrichlorobenzene         10         10         11           1,2-Dichlorobenzene         10         10         11           1,2-Dichlorobenzene         10         10         11           1,3-Dichlorobenzene         10         10         10           1,3-Dichlorobenzene         10         10         10           2-Bexanone         50         10         50         12           2-Hexanone         50         50         10   | Field Sample Id                       | CRSW0010 | -<br>0001XX | CRSW002   | 200001XX       |  |  |  |
| Ope Code         FS         FS           Param Name         Qe Code         FS         Result         Qualifier           1,1.1-Trichloroethane         10         U         10         U           1,1.2,2-Tetrachloroethane         10         U         10         U           1,1.2-Trichloro-1,2.2-Trifluoroethane         10         U         10         U           1,1-Dichloroethane         10         U         10         U           1,1-Dichloroethane         10         U         10         U           1,2-Dirborno-3-chloropropane         10         U         10         U           1,2-Dichoroethane         10         U         10         U           1,2-Dichorobenzene         10         U         10         U           1,2-Dichlorobenzene         10         U         10         U           1,2-Dichlorobenzene         10         U         10         U           1,4-Dichlorobenzene         10         U         10         U           1,4-Dichlorobenzene         50         U         50         U           2-Hexanone         50         U         50         U           4-Methyl-2-pentanoe  | Field Sample Date                     | 1/23/2   | 006         | 1/23/     | 2006           |  |  |  |
| Param Name         Result         Qualifier         Result         Qualifier           11,1-7richloroethane         10         10         10         10           11,2-7richloroethane         10         10         10         10           11,2-7richloroethane         10         10         10         10           1,1-7richloroethane         10         10         10         11           1,1-Dichloroethane         10         10         10         11           1,1-Dichloroethane         10         10         10         11           1,2-Dibromo-3-chloropropane         10         10         10         11           1,2-Dibromo-3-chloropropane         10         10         11         11           1,2-Dibromoethane         10         10         11  | Oc Code                               | FS       | 000         | F         | S              |  |  |  |
| 1.1.1-Trichloroethane         10         10         10         10         10         10         10         11.1.2.Trichloro-1.2.2-Trifluoroethane         10         10         10         10         11.1.2.Trichloro-1.2.2-Trifluoroethane         10         10         10         10         11.1.2.Trichloro-1.2.2-Trifluoroethane         10         10         10         11.1.2.Trichloroethane         10         10         10         11.1.2.Trichloroethane         10         10         11.1.2.Trichloroethane         10         10         11.1.2.Trichloroethane         10         10         11.1.2.Trichloroethane         10         10         11.2.2.Dichlorobenzene         10         10         11.2.2.Dichlorobenzene         10         10         11.2.Dichlorobenzene         10         10         11.2.2.Dichlorobenzene         10         10         11.2.2.D  | Param Name                            | Result   | Oualifier   | Result    | ~<br>Oualifier |  |  |  |
| International internatinteristeme international international international i | 1 1 1-Trichloroethane                 | 10       | U           | 10        | U              |  |  |  |
| International and the second | 1 1 2 2-Tetrachloroethane             | 10       | U           | 10        | U              |  |  |  |
| 1.1.2-Trichloroethane         10         10         10         10           1.1Dichloroethane         10         U         10         U         10         U           1.1Dichloroethane         10         U         10         U         10         U           1.2Dichlorobenzene         10         U         10         U         10         U           1.2-Dichlorobenzene         10         U         10         U         10         U           1.2-Dichlorobenzene         10         U         10         U         10         U           1.2-Dichloropopane         10         U         10         U         10         U           1.4-Dichlorobenzene         10         U         10         U         10         U           2-Butanone         50         U         50         U         20         U         40         U         10         U<   | 1.1.2-Trichloro-1.2.2-Trifluoroethane | 10       | U           | 10        | U              |  |  |  |
| 1-Dickhoroethane         10         10         10         10         10         10         11           1,1-Dichkoroethane         10         U  | 1.1.2-Trichloroethane                 | 10       | Ŭ           | 10        | U              |  |  |  |
| In Dicknown         In U         In U         In U           In Dicknown         In U         In U         In U           1.2-Dibromo-3-chloropropane         In U         In U         In U           1.2-Dibromo-3-chloropropane         In U         In U         In U           1.2-Dibromo-stehne         In U         In U         In U           1.2-Dibromoethane         In U         In U         In U           1.2-Dichlorobenzene         In U         In U         In U           1.3-Dichlorobenzene         In U         In U         In U           1.3-Dichlorobenzene         In U         In U         In U           1.3-Dichlorobenzene         In U         In U         In U           2-Betanone         50         U         50         U           4-Methyl-2-pentanone         50         U         50         U           Acetic acid, methyl ester         In U         In U         In U         In U           Benzene         In U         In U         In U         In U         In U           Bromodichloromethane         In U         In U         In U         In U         In U         In U           Carbon disulfide         In U         <  | 1 1-Dichloroethane                    | 10       | U           | 10        | U              |  |  |  |
| 12.4-Trichlorobenzene         10         0         10         0           1.2.4-Trichlorobenzene         10         U         10         U           1.2-Dibromo-3-chloropropane         10         U         10         U           1.2-Dichlorobenzene         10         U         10         U           1.2-Dichlorobenzene         10         U         10         U           1.2-Dichlorobenzene         10         U         10         U           1.4-Dichlorobenzene         10         U         10         U           2-Butanone         50         U         50         U           2-Butanone         50         U         50         U           2-Butanone         50         U         50         U           4-Methyl-2-pentanone         50         U         50         U           Acetic acid, methyl ester         10         U         10         U           Bromoform         10         U         10         U         10           Bromodichloromethane         10         U         10         U         10           Bromodichloromethane         10         U         10         U         10   | 1 1-Dichloroethene                    | 10       | U           | 10        | U              |  |  |  |
| International Construction         Image of the construction of the constr          | 1.2.4-Trichlorobenzene                | 10       | U           | 10        | U              |  |  |  |
| Distribution         Display  | 1 2-Dibromo-3-chloropropane           | 10       | U           | 10        | U              |  |  |  |
| 12 Distribution         10 U         10 U           1,2-Dichlorobenzene         10 U         10 U           1,3-Dichlorobenzene         10 U         10 U           1,4-Dichlorobenzene         50 U         50 U           2-Butanone         50 U         50 U           2-Hexanone         50 U         50 U           Acetic acid, methyl ester         10 U         10 U           Acetone         50 U         50 U           Bromodichloromethane         10 U         10 U           Carbon tetrachloride         10 U         10 U           Chlorobenzene         10 U         10 U           Chlorodibromomethane         10 U         10 U           Chloroform         10 U         10 U           Chloroform         10 U         10 U           Chloroform         10 U         10 U  | 1.2-Dibromoethane                     | 10       | U           | 10        | U              |  |  |  |
| In Definition         Image of the second secon          | 1.2-Dichlorobenzene                   | 10       | U           | 10        | U              |  |  |  |
| 12 Dichloropropane         10 U         10 U           1,2-Dichloropropane         10 U         10 U           1,3-Dichlorobenzene         10 U         10 U           2-Butanone         50 U         50 U           2-Hexanone         50 U         50 U           2-Hexanone         50 U         50 U           4-Methyl-2-pentanone         50 U         50 U           Acetic acid, methyl ester         10 U         10 U           Bromodichloromethane         10 U         10 U           Carbon tetrachloride         10 U         10 U           Chlorobenzene         10 U         10 U         10 U           Chlorobenzene         10 U         10 U         10 U           Chlorobenzene         10 U         10 U         10 U           Chloroperpane         10 U         10 U         10 U           Chlorofofrm         10 U         10 U   | 1,2-Dichloroethane                    | 10       | U           | 10        | U              |  |  |  |
| 112 Dichlorobenzene         10 U         10 U           1,3-Dichlorobenzene         10 U         10 U           2-Butanone         50 U         50 U           2-Hexanone         50 U         50 U           2-Hexanone         50 U         50 U           Actic acid, methyl ester         10 U         10 U           Acctic acid, methyl ester         10 U         10 U           Benzene         10 U         10 U           Bromoform         10 U         10 U           Bromoform         10 U         10 U           Bromothane         10 U         10 U           Carbon disulfide         10 U         10 U           Chlorodibromomethane         10 U         10 U           Chlorodbromomethane         10 U         10 U           Cis-1,2-D   | 1.2-Dichloropropage                   | 10       | U           | 10        | U              |  |  |  |
| Interference         Interference         Interference           1.4-Dichlorobenzene         10         U         10         U           2-Butanone         50         U         50         U           2-Hexanone         50         U         50         U           4-Methyl-2-pentanone         50         U         50         U           Acetore         50         U         50         U           Acetone         50         U         50         U           Benzene         10         U         10         U           Bromodichloromethane         10         U         10         U           Bromodisulfide         10         U         10         U           Bromomethane         10         U         10         U           Carbon disulfide         10         U         10         U           Carbon disulfide         10         U         10         U           Chlorodibromomethane         10         U         10         U           Chloroform         10         U         10         U         U           Chloroform         10         U         10         U   | 1 3-Dichlorobenzene                   | 10       | Ŭ           | 10        | U              |  |  |  |
| 1.1 Difference         1.0 C         1.0 C         1.0 C           2-Butanone         50 U         50 U         2.0 U           2-Hexanone         50 U         50 U         2.0 U           4-Methyl-2-pentanone         50 U         50 U         50 U           Acctic acid, methyl ester         10 U         10 U         10 U           Bromodichloromethane         10 U         10 U         10 U           Carbon disulfide         10 U         10 U         10 U           Carbon tetrachloride         10 U         10 U         10 U           Chlorodibromomethane         10 U         10 U  | 1.4-Dichlorobenzene                   | 10       | U           | 10        | U              |  |  |  |
| 2 Diamone         30 U         50 U           2-Hexanone         50 U         50 U           4-Methyl-2-pentanone         50 U         50 U           Acetic acid, methyl ester         10 U         10 U           Benzene         10 U         10 U           Bromodichloromethane         10 U         10 U           Bromomethane         10 U         10 U           Carbon disulfide         10 U         10 U           Chlorobenzene         10 U         10 U           Chlorodibromomethane         10 U         10 U           Chlorodifuoromethane         10 U         10 U           Cyclohexane         10 U         10 U           Cyclohexane         10 U         10 U           Dichlorodifluoromethane         10 U         10 U           Sopropylbenzene         10 U         10 U  | 2-Butanone                            | 50       | U           | 50        | U              |  |  |  |
| 2-HCaldoff         30         50         50         U           4-Methyl-2-pentanone         50         U         50         U           Acetic acid, methyl ester         10         U         10         U           Acetic acid, methyl ester         10         U         10         U           Bromodichloromethane         10         U         10         U           Bromodichloromethane         10         U         10         U           Bromomethane         10         U         10         U           Carbon disulfide         10         U         10         U           Chlorobenzene         10         U         10         U           Chlorodibromomethane         10         U         10         U           Chlorodibromomethane         10         U         10         U           Chlorodibromomethane         10         U         10         U           Chloromethane         <  | 2-Hevanone                            | 50       | U           | 50        | U              |  |  |  |
| Actic acid, methyl ester         10         1   | 4-Methyl-2-pentanone                  | 50       | U           | 50        | U              |  |  |  |
| Acetone       10       10       10       10       10         Benzene       10       U       10       U         Bromodichloromethane       10       U       10       U         Carbon disulfide       10       U       10       U         Carbon tetrachloride       10       U       10       U         Chlorobenzene       10       U       10       U         Chlorodibromomethane       10       U       10       U         Chloroform       10       U       10       U         Chloromethane       10       U       10       U         Chloromethane       10       U       10       U         Chloromethane       10       U       10       U         Cyclohexane       10       U       10       U         Cyclohexane       10       U       10       U         Isopropylbenzene       10       U       10   | Acetic acid methyl ester              | 10       | U           | 10        | U              |  |  |  |
| Account       30       0       30       0         Benzene       10       U       10       U         Bromodichloromethane       10       U       10       U         Bromodichloromethane       10       U       10       U         Bromodichloromethane       10       U       10       U         Carbon disulfide       10       U       10       U         Carbon tetrachloride       10       U       10       U         Chlorobenzene       10       U       10       U         Chlorodibromomethane       10       U       10       U         Chloroethane       10       U       10       U         Chloroptorm       10       U       10       U         Chloroptorm       10       U       10       U         Chloroptorethane       88       1.2       J         cis-1,2-Dichloropthene       88       1.2       J         cis-1,3-Dichloropthene       10       U       10       U         Cyclohexane       10       U       10       U       U         Sopropylbenzene       10       U       10       U       U   | Acetone                               | 50       | U           | 50        | U              |  |  |  |
| Bromodichloromethane         10         0         10         0           Bromodichloromethane         10         U         10         U           Bromoform         10         U         10         U           Bromoform         10         U         10         U           Bromoform         10         U         10         U           Carbon disulfide         10         U         10         U           Carbon tetrachloride         10         U         10         U           Chlorobenzene         10         U         10         U           Chlorobenzene         10         U         10         U           Chlorobenzene         10         U         10         U           Chlorothane         10         U         10         U           Cyclohexane         10         U         10         U </td <td>Benzene</td> <td>10</td> <td>U</td> <td>10</td> <td>U</td>   | Benzene                               | 10       | U           | 10        | U              |  |  |  |
| Bromodelinordiality       10       0       10       0         Bromoform       10       0       10       0       0         Bromomethane       10       0       10       0       0         Carbon disulfide       10       0       0       0       0       0         Carbon tetrachloride       10       0       0       0       0       0       0         Chlorobenzene       10       0       0       0       0       0       0       0         Chlorodibromomethane       10       0       0       0       0       0       0         Chlorodibromomethane       10       0 <td< td=""><td>Bromodichloromethane</td><td>10</td><td>U</td><td>10</td><td>U</td></td<>  | Bromodichloromethane                  | 10       | U           | 10        | U              |  |  |  |
| Bromorthane       10       0       10       0         Bromomethane       10       0       10       0       0         Carbon disulfide       10       0       0       0       0       0         Carbon tetrachloride       10       0       0       0       0       0       0       0         Chlorobenzene       10         | Bromoform                             | 10       | U           | 10        | U              |  |  |  |
| Informetinate         10         10         10         0           Carbon disulfide         10         10         10         10         10           Carbon tetrachloride         10         10         10         10         10         10           Carbon tetrachloride         10         10         10         10         10         10           Chlorobenzene         10         10         10         10         10         10           Chlorodibromomethane         10         10         10         10         10         10           Chloroofthane         10         10         10         10         10         10         10           Chloroofthane         10         10         10         10         10         10         10           Chloroofthane         10   | Bromomethane                          | 10       | U           | 10        | U              |  |  |  |
| Carbon tistuffice         10         10         10         0           Carbon tetrachloride         10         10         10         10         10         10           Chlorobenzene         10         U         10         U         10         U           Chlorodibromomethane         10         U         10         U         10         U           Chloroothane         10         U         10         U         10         U           Cis-1,2-Dichloropropene         10         U         10         U         10         U           Cyclohexane         10         U         10         U         10         U           Stopropylbenzene         10         U         10         U         10         U           Methyl cyclohexane         10         U         10         U         10         U         0         U         0 <td>Carbon disulfide</td> <td>10</td> <td>U</td> <td>10</td> <td>U</td>  | Carbon disulfide                      | 10       | U           | 10        | U              |  |  |  |
| Chlorobenzene         10         10         10         10           Chlorobiromomethane         10         U         10         U           Chlorodibromomethane         10         U         10         U           Chlorodibromomethane         10         U         10         U           Chlorodibromomethane         10         U         10         U           Chloroform         10         U         10         U           Chloromethane         10         U         10         U           Cis-1,2-Dichloroethene         88         1.2         J           cis-1,3-Dichloropropene         10         U         10         U           Cyclohexane         10         U         10         U           Dichlorodifluoromethane         10         U         10         U           Ethyl benzene         10         U         10         U           Isopropylbenzene         10         U         10         U           Methyl cyclohexane         10         U         10         U           Methyl renbutyl Ether         10         U         10         U           o-Xylene         10         U   | Carbon tetrachloride                  | 10       | U           | 10        | U              |  |  |  |
| Chlorodilzche         10         10         10         0           Chlorodibromomethane         10         U         10         U           Chlorodibromomethane         10         U         10         U           Chlorooform         10         U         10         U           Chlorooftane         10         U         10         U           Chlorooftane         10         U         10         U           Chlorooftane         88         1.2         J           cis-1,2-Dichloroothene         88         1.2         J           cis-1,3-Dichloropropene         10         U         10         U           Cyclohexane         10         U         10         U         10           Ochorodifluoromethane         10         U         10         U         10         U           Ethyl benzene         10         U         10         U         10         U           Isopropylbenzene         10         U         10         U         10         U           Methyl cyclohexane         10         U         10         U         0         U           o-Xylene         10         U </td <td>Chlorobenzene</td> <td>10</td> <td>U</td> <td>10</td> <td>U</td>  | Chlorobenzene                         | 10       | U           | 10        | U              |  |  |  |
| Chlorodubrohnehme         10         0         10         0           Chloroethane         10         U         10         U           Chloroethane         10         U         10         U           Chloromethane         10         U         10         U           Cis-1,2-Dichloroethene         88         1.2         J           cis-1,3-Dichloropropene         10         U         10         U           Cyclohexane         10         U         10         U           Dichlorodifluoromethane         10         U         10         U           Ethyl benzene         10         U         10         U           Isopropylbenzene         10         U         10         U           Methyl cyclohexane         10         U         10         U           Methyl Tertbutyl Ether         10         U         10         U           Methylene chloride         10         U         10         U           Styrene         10         U         10         U           Toluene         10         U         10         U           trans-1,2-Dichloroethene         0.94         J <td< td=""><td>Chlorodibromomethane</td><td>10</td><td>U</td><td>10</td><td>U</td></td<>  | Chlorodibromomethane                  | 10       | U           | 10        | U              |  |  |  |
| Chlorodinine         10         0         10         0           Chloroform         10         U         10         U         10         U           Chloroform         10         UJ         10         UJ         10         UJ           Cis-1,2-Dichloroethene         88         1.2         J         J         I         I         U         10         U         I         U         I         U         I         U         I         U         I         U         I         U         I         U         I         U         I         U         I         U         I         U         I         U         I         U         I         U         I         U         I  | Chloroethane                          | 10       | U           | 10        | U              |  |  |  |
| Choronethane         10         0         10         0           Chloromethane         10         10         10           | Chloroform                            | 10       | U           | 10        | U              |  |  |  |
| Chronic Hair       10       10       00         Cis-1,2-Dichloropethene       88       1.2       J         cis-1,3-Dichloropropene       10       U       10       U         Cyclohexane       10       U       10       U       10       U         Dichlorodifluoromethane       10       U       10       U       10       U         Ethyl benzene       10       U       10       U       10       U         Isopropylbenzene       10       U       10       U       10       U         Methyl cyclohexane       10       U       10       U       10       U         Methyl Tertbutyl Ether       10       U       10       U       10       U         o-Xylene       10       U       10       U       10       U       10       U         Styrene       10       U       10       U <td< td=""><td>Chloromethane</td><td>10</td><td>UI</td><td>10</td><td></td></td<>   | Chloromethane                         | 10       | UI          | 10        |                |  |  |  |
| Cls1,2-Dichlorobenche         360         1.2         J           cis-1,3-Dichloropropene         10         U         10         U           Cyclohexane         10         U         10         U         10         U           Dichlorodifluoromethane         10         U         10         U         10         U           Ethyl benzene         10         U         10         U         10         U           Isopropylbenzene         10         U         10         U         10         U           Methyl cyclohexane         10         U         10         U         10         U           Methyl rertbutyl Ether         10         U         10         U         10         U           o-Xylene         10         U         10         U         10         U           styrene         10         U         10         U         10         U           tras-1,2-Dichloroethene         17         8.1         J         10         U           tras-1,3-Dichloropropene         10         U         10         U         10         U           Trichlorofluoromethane         10         U   | Cis-1 2-Dichloroethene                | 88       | 03          | 10        | I              |  |  |  |
| Circle Scheme         10         0         10         0           Cyclohexane         10         U         10         U           Dichlorodifluoromethane         10         U         10         U           Ethyl benzene         10         U         10         U           Isopropylbenzene         10         U         10         U           Methyl cyclohexane         10         U         10         U           Methyl rentbutyl Ether         10         U         10         U           o-Xylene         10         U         10         U           o-Xylene         10         U         10         U           Styrene         10         U         10         U           Toluene         10         U         10         U           trans-1,2-Dichloroethene         0.94         J         10         U           Trichloroethene         1.7         J <td< td=""><td>cis-1 3-Dichloropropene</td><td>10</td><td>II</td><td>1.2</td><td>J</td></td<>  | cis-1 3-Dichloropropene               | 10       | II          | 1.2       | J              |  |  |  |
| Cycle between         Cycle between <thcycle bet<="" td=""><td>Cyclobeyane</td><td>10</td><td>U</td><td>10</td><td>U</td></thcycle>   | Cyclobeyane                           | 10       | U           | 10        | U              |  |  |  |
| Definition       10       0       10       0         Ethyl benzene       10       U       10       U         Isopropylbenzene       10       U       10       U         Methyl cyclohexane       10       U       10       U         Methyl Cyclohexane       10       U       10       U         Methyl Tertbutyl Ether       10       U       10       U         o-Xylene       10       U       10       U         o-Xylene       10       U       10       U         Styrene       10       U       10       U         Toluene       10       U       10       U         trans-1,2-Dichloroethene       0.94       J       10       U         trans-1,3-Dichloropropene       10       U       10       U         Trichlorofluoromethane       10       U       10       U         Vinyl chloride       3.2       10       U       U  | Dichlorodifluoromethane               | 10       | U           | 10        | U              |  |  |  |
| Initial Content         Initial Conten         Initial Content         Initial Con  | Ethyl benzene                         | 10       | U           | 10        | U              |  |  |  |
| Isoportion         Isoportion <thisoportion< th="">         Isoportion         Isoporti</thisoportion<>   | Isopropylbenzene                      | 10       | U           | 10        | U              |  |  |  |
| Methyl Yeroloxanc         10         10         10         0           Methyl Terbutyl Ether         10         U         10         U           Methylene chloride         10         U         10         U           o-Xylene         10         U         10         U           Styrene         10         U         10         U           Tetrachloroethene         17         8.1         J           Toluene         10         U         10         U           trans-1,2-Dichloroethene         0.94         J         10         U           trans-1,3-Dichloropropene         10         U         10         U           Trichlorofluoromethane         1.7         J         10         U           Trichlorofluoromethane         10         U         10         U           Vinyl chloride         3.2         J         10         UJ  | Methyl cycloheyane                    | 10       | U           | 10        | U              |  |  |  |
| Methylene chloride         10         10         10         0           Methylene chloride         10         U         10         U         0         U         0         U         0         U         0         U         0         U         0         U         0         U         0         U         0         U         0         U         0         U         0         U         0         U         10         U   | Methyl Tertbutyl Ether                | 10       | U           | 10        | U              |  |  |  |
| Inclusion         Inclusion <thinclusion< th="">         Inclusion         <thinclusion< th="">         Inclusion         <thinclusion< th=""> <thinclusion< th=""> <thinc< td=""><td>Methylene chloride</td><td>10</td><td>U</td><td>10</td><td>U</td></thinc<></thinclusion<></thinclusion<></thinclusion<></thinclusion<>  | Methylene chloride                    | 10       | U           | 10        | U              |  |  |  |
| In U         In U         In U           Styrene         10 U         10 U           Tetrachloroethene         17         8.1 J           Toluene         10 U         10 U           trans-1,2-Dichloroethene         0.94 J         10 U           trans-1,3-Dichloropropene         10 U         10 U           Trichloroethene         1.7 J         10 U           Trichlorofluoromethane         10 U         10 U           Vinyl chloride         3.2 J         10 UJ   | o-Xvlene                              | 10       | Ŭ           | 10        | U              |  |  |  |
| Stylene         10         10         10         10         0           Tetrachloroethene         17         8.1         J         J         Toluene         10         U         10         U         trans-1,2-Dichloroethene         0.94         J         10         U         Trans-1,3-Dichloropropene         10         U         10         U         Trichloroethene         1.7         J         10         U         Trichloroethene         1.7         J         10         U         Trichlorofluoromethane         10         U         10         U  | Sturano                               | 10       | U           | 10        | U              |  |  |  |
| Image: Construction of the image: Construction of th | Tetrachloroethene                     | 10       | 0           | 10<br>& 1 | I              |  |  |  |
| In Iter         In Iter         In Iter         In Iter           trans-1,2-Dichloroethene         0.94 J         10 U           trans-1,3-Dichloropropene         10 U         10 U           Trichloroethene         1.7 J         10 U           Trichloroethene         1.0 U         10 U           Vinyl chloride         3.2 J         10 UJ   | Toluene                               | 1/       | U           | 0.1       | J              |  |  |  |
| Intersection         0.94 J         10 U           trans-1,3-Dichloropropene         10 U         10 U           Trichloroethene         1.7 J         10 U           Trichlorofluoromethane         10 U         10 U           Vinyl chloride         3.2 J         10 UJ   | trans_1.2-Dichloroethana              | 0.04     | I           | 10        | U              |  |  |  |
| Intersection         Intersection<  | trans-1,2-Dichloropropage             | 10       | ,<br>U      | 10        | U              |  |  |  |
| Inclusion         In J         In U           Trichlorofluoromethane         10 U         10 U           Vinyl chloride         3.2 J         10 UJ   | Trichloroethene                       | 10       | I           | 10        | U              |  |  |  |
| Vinyl chloride 3.2 J 10 UJ  | Trichlorofluoromethane                | 1.7      | J           | 10        | U              |  |  |  |
| Vinyi cinoriac         5.2 J         10 UJ           Xvlene m/n         10 U         10 U   | Vinyl chloride                        | 2.2      | I           | 10        | UI             |  |  |  |
|   | Yulene m/n                            | 3.2      | ,<br>U      | 10        | U              |  |  |  |

Appendix E Table 1.3: Surfacewater (Sump) VOC Results

Notes:

Results in micrograms per liter (µg/L) Samples analyzed for VOCs by EPA method OLM04.2 QC Code: FS = Field Sample

Qualifiers:

 $\mathbf{U}=\mathbf{N}\mathbf{o}\mathbf{t}$  detected at a concentration above the reporting limit J = Estimated value

### Appendix E Table 1.4: Soil Vapor VOC Results

| Lab Sample Id                          | X1377-05 |           | X1377-06 |           | X1377-07       |           | X1380-01     |           |
|--|----------|-----------|----------|-----------|----------------|-----------|--------------|-----------|
| Lab Sample Delivery Group              | X13      | 77        | X1377    |           | X1377          |           | X13          | B0        |
| Loc Name                               | GV-      | 01        | GV-      | 02        | GV-            | 03        | SV-0         | 01        |
| Field Sample Id                        | CRGV001  | 00601XX   | CRGV002  | 00601XX   | CRGV00300601XX |           | CRSV0010     | 00101XX   |
| Field Sample Date                      | 1/26/2   | 2006      | 1/26/2   | 2006      | 1/26/2         | 2006      | 1/26/2       | 006       |
| Qc Code                                | FS       | 5         | FS       | 3         | FS             | 3         | FS           | i         |
| Param Name                             | Result   | Qualifier | Result   | Qualifier | Result         | Qualifier | Result       | Qualifier |
| 1,1,1-Trichloroethane                  | 1.47     | UJ        | 1.58     | U         | 1.47           | U         | 7.62         | U         |
| 1,1,2,2-Tetrachloroethane              | 0.69     | U         | 0.69     | U         | 0.69           | U         | 2.75         | U         |
| 1,1,2-Trichloro-1,2,2-Trifluoroethane  | 0.54     |           | 0.76     | U         | 0.76           | U         | 3.06         | U         |
| 1,1,2-Trichloroethane                  | 0.54     | UJ        | 0.54     | U         | 0.54           | U         | 2.18         | U         |
| 1,1-Dichloroethane                     | 0.4      | U         | 0.4      | U         | 0.4            | U         | 1.62         | U         |
| 1,1-Dichloroethene                     | 0.4      | U         | 0.4      | U         | 0.4            | U         | 1.59         | U         |
| 1,2,4-Trichlorobenzene                 | 0.74     | U         | 0.74     | U         | 0.74           | U         | 2.96         | U         |
| 1,2,4-Trimethylbenzene                 | 2.85     |           | 2.7      |           | 2.55           |           | 1.96         | U         |
| 1,2-Dibromoethane                      | 0.77     | U         | 0.77     | U         | 0.77           | U         | 3.08         | U         |
| 1,2-Dichloro-1,1,2,2-tetrafluoroethane | 0.7      | U         | 0.7      | U         | 0.7            | U         | 2.8          | U         |
| 1,2-Dichlorobenzene                    | 0.6      | U         | 0.6      | U         | 0.6            | U         | 2.4          | U         |
| 1,2-Dichloroethane                     | 0.4      | U         | 0.4      | U         | 0.4            | U         | 1.62         | U         |
| 1,2-Dichloropropane                    | 0.46     | UJ        | 0.46     | U         | 0.46           | U         | 1.85         | U         |
| 1,3,5-I rimethylbenzene                | 0.49     | U         | 0.49     | U         | 1.62           |           | 1.96         | U         |
|  | 0.6      | U         | 0.6      | U         | 0.6            | U         | 2.4          | U         |
|  | 0.6      | U         | 0.6      | U         | 0.6            | U         | 2.4          | U         |
| 1,4-Dioxane                            |          | к         |          | к         |                | к         |              | к         |
| 2-Butanone                             | 20.2     |           | 16.5     |           | 14.3           |           | 3.65         |           |
| 2-Hexanone                             | 0.82     | UJ        | 0.82     | U         | 0.82           | U         | 3.27         | U         |
| 2-Propanol                             | 39.4     |           | 25.5     |           | 26.2           |           | 8.83         |           |
| 4-Ethyltoluene                         | 0.59     |           | 0.49     | U         | 0.49           | U         | 1.96         | U         |
| 4-Methyl-2-pentanone                   | 0.82     | UJ        | 0.82     | U         | 0.82           | U         | 3.27         | U         |
| Acetone                                | 113      | D         | 149      | D         | 105            | D         | 114          | В         |
| Allyl chloride                         | 0.31     | U         | 0.31     | U         | 0.31           | U         | 1.26         | U         |
| Benzene                                | 5.33     | J         | 6.38     |           | 3.57           |           | 13.7         |           |
| Benzyl chloride                        | 0.58     | U         | 0.58     | U         | 0.58           | U         | 2.31         | U         |
| Bromodichloromethane                   | 0.67     | UJ        | 0.67     | U         | 0.67           | U         | 2.68         | U         |
| Bromotorm                              | 1.03     | UJ        | 1.03     | U         | 1.03           | U         | 4.14         | U         |
| Bromomethane                           | 0.39     | U         | 0.39     | U         | 0.39           | U         | 1.55         | U         |
| Butadiene, 1,3-                        | 12       | J         | 21.8     | J         | 8.44           | J         |              | R         |
|  | 1.31     |           | 1.55     |           | 0.9            |           | 3.30         |           |
| Carbon tetrachloride                   | 0.63     | UJ        | 0.63     | U         | 0.63           | U         | 2.52         | 0         |
| Chlorobenzene                          | 0.46     | UJ        | 0.46     | U         | 0.46           | U         | 1.85         | U         |
| Chlorodibromomethane                   | 0.85     | UJ        | 0.85     | U         | 0.85           | U         | 3.4          | U         |
| Chloroform                             | 0.27     | 0         | 0.27     | 0         | 0.27           | 0         | 1.06         | 0         |
| Chloromothana                          | 0.49     | U         | 0.49     | U         | 0.49           | U         | 1.95         | 0         |
| Chloromethane                          | 1.00     |           | 1.55     |           | 1.25           |           | 1.23         |           |
|  | 0.4      | 0         | 0.4      | 0         | 0.4            | 0         | 1.59         | 0         |
| Cis-1,3-Dichloropropene                | 0.45     | UJ        | 0.45     | 0         | 0.45           | 0         | 1.02         | 0         |
| Disbloredifluoromethane                | 0.34     | 0         | 0.34     | 0         | 2.02           | 0         | 41.3         |           |
| Ethyl agotata                          | 2.30     |           | 2.30     |           | 2.03           |           | 2.30         |           |
| Ethyl bonzono                          | 9.03     | 1         | 4.00     |           | 1.52           |           | 1 7 2        | 11        |
|  | 1.47     | J         | 0.90     |           | 1.32           |           | 1.73         | 0         |
| Heyeoblerobutediana                    | 1.70     |           | 2.17     |           | 1.00           |           | 4 27         | 11        |
| Hexano                                 | 2.02     | 05        | 5.38     | 00        | 2.67           | 05        | 4.27         | 0         |
|  | 2.52     |           | 0.30     |           | 2.07           |           | 1 07         | 11        |
| Notbyl Torthutyl Ethor                 | 0.03     |           | 0.79     |           | 0.01           |           | 1.07         | 0         |
| Methylana oblarida                     | 0.30     | 0         | 1.01     | 0         | 0.30           | 0         | 2 79         | 0         |
|  | 1.17     | 0         | 0.82     |           | 0.7            | 0         | 2.70         | 0         |
| Propylopo                              | 1.17     | J         | 75.1     | D         | 30.5           | D         | 0.69         | 0         |
| Styrono                                | 40.0     |           | 10.1     |           | 0.80<br>0 / 2  | 0         | 1 7          | 11        |
| Totrachloroothono                      | 0.43     | 1         | 0.43     | 0         | 4.20           | 0         | 747          |           |
| Tetrabydrofuran                        | 4.21     | 5         | 0.04     |           | 4.20           |           | 2.26         |           |
| Toluene                                | 7.64     |           | 7 10     | 5         | 5.09           | 5         | 2.30<br>18 9 | 5         |
| trans-1 2-Dichloroethene               | 0.4      | Ц         | 1.19     | lu –      | 0.00           |           | 1 50         |           |
| trans-1,2-Dichloropropene              | 0.4      | U.I.      | 0.4      | U U       | 0.4            | U         | 1.39         | <u>,</u>  |
| Trichloroethene                        | 0.40     | 111       | 0.40     | 11        | 0.40           | 0         | 20.2         | 5         |
| Trichlorofluoromethano                 | 1.00     | 00        | 1.23     | 5         | 0.23           | 5         | 2.24         |           |
| Vinyl acetate                          | 1.00     |           | 0.25     | lu –      | 0.04           |           | 2.24         | 1         |
| Vinyl bromide                          | 0.33     | U         | 0.33     | U U       | 0.35           | U         | 1.41         | <u>,</u>  |
| Vinyl chloride                         | 0.44     | Ŭ         | 0.44     | Ŭ         | 0.44           | Ŭ         | 1.73         | U<br>U    |
| Xylene m/n                             | 3 09     | Ŭ.        | 2 0.20   | -         | 1.20           | 5         | 3 /17        | U U       |
| , (j.o., in/p                          | 0.00     | ~         | 2.00     | 1         | 1.02           |           | 5.47         | 5         |

Notes:

Results in micrograms per cubic meter (µg/m3) Samples analyzed for VOCs by method TO-15

QC Code:

FS = Field Sample Qualifiers:

U = Not detected at a concentration above the reporting limit

J = Estimated value

D = Results were reported from a duplicate analytical run

R = Rejected result

# Appendix E Table 1.5: Groundwater VOC TICs

| Lab ID                             | X133       | 9-10     |  |  |  |
|------------------------------------|------------|----------|--|--|--|
| Locatin ID                         | GW-10      |          |  |  |  |
| Sample No                          | CRGW010    | 02601XX  |  |  |  |
| Samp Date                          | 1/24/2     | 2006     |  |  |  |
| Units                              | ug/L       |          |  |  |  |
| Parameter                          | Lab Result | Lab Qual |  |  |  |
| Benzene, (1-methyl-1-propenyl)-    | 5          | J        |  |  |  |
| Benzene, (1-methylethyl)-          | 12         | J        |  |  |  |
| Benzene, 1,2,3,4-tetramethyl-      | 9          | J        |  |  |  |
| Benzene, 1,2,3-trimethyl-          | 37         | J        |  |  |  |
| Benzene, 1,3,5-trimethyl-          | 9.7        | J        |  |  |  |
| Benzene, 1-ethenyl-3-ethyl-        | 8.2        | J        |  |  |  |
| Benzene, 1-ethyl-2,4-dimethyl-     | 9.9        | J        |  |  |  |
| Benzene, 1-methyl-4-(1-methylethyl | 10         | J        |  |  |  |
| Benzene, cyclopropyl-              | 11         | J        |  |  |  |
| Indan, 1-methyl-                   | 9.5        | J        |  |  |  |

Notes:

Results reported in microgram per liter ( $\mu$ g/L) Samples anlayzed for VOCs by EPA Method OLM04.2 Qualifiers: J = Estimated value

| Appendix E |                                |  |  |  |  |  |  |  |
|------------|--------------------------------|--|--|--|--|--|--|--|
| Table 1.6: | <b>Groundwater VOC Results</b> |  |  |  |  |  |  |  |

| Lab Sample Id                         | X5203-01         | X5203-02         | X5203-03         | X5203-06         | X5203-07         |
|---------------------------------------|------------------|------------------|------------------|------------------|------------------|
| Lab Sample Delivery Group             | X5203            | X5203            | X5203            | X5203            | X5203            |
| Loc Name                              | GW-19            | GW-19            | MW-01            | MW-05            | MW-06            |
| Field Sample Id                       | CRGW01901901XX   | CRGW01902901XX   | CRMW00102001XX   | CRMW00501201XX   | CRMW00601301XX   |
| Field Sample Date                     | 10/31/2006       | 10/31/2006       | 11/1/2006        | 11/1/2006        | 11/1/2006        |
| Qc Code                               | FS               | FS               | FS               | FS               | FS               |
| Param Name                            | Result Qualifier |
| 1,1,1-Trichloroethane                 | 10 U             |
| 1,1,2,2-Tetrachloroethane             | 10 U             |
| 1,1,2-Trichloro-1,2,2-Trifluoroethane | 10 UJ            | 10 U             | 10 U             | 10 UJ            | 10 UJ            |
| 1,1,2-Trichloroethane                 | 10 U             |
| 1,1-Dichloroethane                    | 10 U             |
| 1,1-Dichloroethene                    | 10 UJ            | 10 U             | 10 U             | 10 UJ            | 10 UJ            |
| 1,2,4-Trichlorobenzene                | 10 U             |
| 1,2-Dibromo-3-chloropropane           | 10 U             |
| 1,2-Dibromoethane                     | 10 U             |
| 1,2-Dichlorobenzene                   | 10 U             |
| 1,2-Dichloroethane                    | 10 U             |
| 1,2-Dichloropropane                   | 10 U             |
| 1,3-Dichlorobenzene                   | 10 U             |
| 1,4-Dichlorobenzene                   | 10 U             |
| 2-Butanone                            | 50 U             |
| 2-Hexanone                            | 50 U             |
| 4-Methyl-2-pentanone                  | 50 U             |
| Acetic acid, methyl ester             | 10 U             | 10 UJ            | 10 UJ            | 10 U             | 10 U             |
| Acetone                               | 50 U             |
| Benzene                               | 10 U             |
| Bromodichloromethane                  | 10 U             |
| Bromoform                             | 10 U             |
| Bromomethane                          | 10 U             |
| Carbon disulfide                      | 10 UJ            | 10 U             | 10 U             | 10 UJ            | 10 UJ            |
| Carbon tetrachloride                  | 10 U             |
| Chlorobenzene                         | 10 U             |
| Chlorodibromomethane                  | 10 U             |
| Chloroethane                          | 10 UJ            |
| Chloroform                            | 10 U             |
| Chloromethane                         | 10 UJ            | 10 U             | 10 U             | 10 UJ            | 10 UJ            |
| Cis-1,2-Dichloroethene                | 10 U             | 10 U             | 10 U             | 2 J              | 10 U             |

| Lab Sample Id             | X520   | X5203-01       |        | 03-02          | X520      | 3-03      | X52    | 03-06     | X520   | 03-07     |
|---------------------------|--------|----------------|--------|----------------|-----------|-----------|--------|-----------|--------|-----------|
| Lab Sample Delivery Group | X5     | 203            | X5     | 203            | X52       | 203       | X5     | 203       | X5     | 203       |
| Loc Name                  | GW     | V-19           | GV     | V-19           | MW        | /-01      | MV     | V-05      | MW     | /-06      |
| Field Sample Id           | CRGW01 | CRGW01901901XX |        | CRGW01902901XX |           | 102001XX  | CRMW00 | 501201XX  | CRMW00 | 601301XX  |
| Field Sample Date         | 10/31  | /2006          | 10/31  | /2006          | 11/1/2006 |           | 11/1/  | /2006     | 11/1/  | 2006      |
| Qc Code                   | F      | ſS             | H      | <b>FS</b>      | FS        |           | FS     |           | F      | S         |
| Param Name                | Result | Qualifier      | Result | Qualifier      | Result    | Qualifier | Result | Qualifier | Result | Qualifier |
| cis-1,3-Dichloropropene   | 10     | U              | 10     | U              | 10        | U         | 10     | U         | 10     | U         |
| Cyclohexane               | 10     | UJ             | 10     | U              | 10        | U         | 10     | UJ        | 10     | UJ        |
| Dichlorodifluoromethane   | 10     | UJ             | 10     | U              | 10        | U         | 10     | UJ        | 10     | UJ        |
| Ethyl benzene             | 10     | U              | 10     | U              | 10        | U         | 10     | U         | 10     | U         |
| Isopropylbenzene          | 10     | U              | 10     | U              | 10        | U         | 10     | U         | 10     | U         |
| Methyl cyclohexane        | 10     | U              | 10     | U              | 10        | U         | 10     | U         | 10     | U         |
| Methyl Tertbutyl Ether    | 10     | U              | 10     | U              | 10        | U         | 10     | U         | 10     | U         |
| Methylene chloride        | 10     | U              | 10     | U              | 10        | U         | 10     | U         | 10     | U         |
| o-Xylene                  | 10     | U              | 10     | U              | 10        | U         | 10     | U         | 10     | U         |
| Styrene                   | 10     | U              | 10     | U              | 10        | U         | 10     | U         | 10     | U         |
| Tetrachloroethene         | 10     | U              | 10     | U              | 6.6       | J         | 33     |           | 5      | J         |
| Toluene                   | 10     | U              | 10     | U              | 10        | U         | 10     | U         | 10     | U         |
| trans-1,2-Dichloroethene  | 10     | U              | 10     | U              | 10        | U         | 10     | U         | 10     | U         |
| trans-1,3-Dichloropropene | 10     | U              | 10     | U              | 10        | U         | 10     | U         | 10     | U         |
| Trichloroethene           | 10     | U              | 10     | U              | 10        | U         | 2.2    | J         | 10     | U         |
| Trichlorofluoromethane    | 10     | UJ             | 10     | U              | 10        | U         | 10     | UJ        | 10     | UJ        |
| Vinyl chloride            | 10     | U              | 10     | U              | 10        | U         | 10     | U         | 10     | U         |
| Xylene, m/p               | 10     | U              | 10     | U              | 10        | U         | 10     | U         | 10     | U         |

Notes: Results in micrograms per liter (µg/L) Samples analyzed for VOCs by EPA Method OLM04.2 QC Code: FS = Field Sample FD = Field Duplicate TB = Trip Blank

Qualifiers:

U = Not detected at a concentration greater than the RL

J = Estimated value

| Lab Sample Id                         | X52    | 03-08     | X5203-09       |           | X5203-10       |           | X5203-11RE     |           | X5203-12       |           |  |
|---------------------------------------|--------|-----------|----------------|-----------|----------------|-----------|----------------|-----------|----------------|-----------|--|
| Lab Sample Delivery Group             | X5     | 203       | X52            | 203       | X5203          |           | X5203          |           | X5203          |           |  |
| Loc Name                              | MV     | V-10      | GW             | -21       | GV             | V-21      | GW-22          |           | GW             | /-22      |  |
| Field Sample Id                       | CRMW01 | 001301XX  | CRGW02101901XX |           | CRGW02102601XX |           | CRGW02201801XX |           | CRGW02202801XX |           |  |
| Field Sample Date                     | 11/1/  | 11/1/2006 |                | 11/1/2006 |                | 11/1/2006 |                | 11/1/2006 |                | 11/1/2006 |  |
| Qc Code                               | F      | rs        | F              | S         | FS             |           | FS             |           | FS             |           |  |
| Param Name                            | Result | Qualifier | Result         | Qualifier | Result         | Qualifier | Result         | Qualifier | Result         | Qualifier |  |
| 1,1,1-Trichloroethane                 | 10     | U         | 10             | U         | 10             | U         | 10             | UJ        | 10             | U         |  |
| 1,1,2,2-Tetrachloroethane             | 10     | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |
| 1,1,2-Trichloro-1,2,2-Trifluoroethane | 10     | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |
| 1,1,2-Trichloroethane                 | 10     | U         | 10             | U         | 10             | U         | 10             | UJ        | 10             | U         |  |
| 1,1-Dichloroethane                    | 10     | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |
| 1,1-Dichloroethene                    | 4.7    | J         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |
| 1,2,4-Trichlorobenzene                | 10     | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |
| 1,2-Dibromo-3-chloropropane           | 10     | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |
| 1,2-Dibromoethane                     | 10     | U         | 10             | U         | 10             | U         | 10             | UJ        | 10             | U         |  |
| 1,2-Dichlorobenzene                   | 10     | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |
| 1,2-Dichloroethane                    | 10     | U         | 10             | U         | 10             | U         | 10             | UJ        | 10             | U         |  |
| 1,2-Dichloropropane                   | 10     | U         | 10             | U         | 10             | U         | 10             | UJ        | 10             | U         |  |
| 1,3-Dichlorobenzene                   | 10     | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |
| 1,4-Dichlorobenzene                   | 10     | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |
| 2-Butanone                            | 50     | U         | 50             | U         | 2.4            | J         | 50             | UJ        | 50             | U         |  |
| 2-Hexanone                            | 50     | U         | 50             | U         | 50             | U         | 50             | U         | 50             | U         |  |
| 4-Methyl-2-pentanone                  | 50     | U         | 50             | U         | 50             | U         | 50             | U         | 50             | U         |  |
| Acetic acid, methyl ester             | 10     | UJ        | 10             | UJ        | 10             | UJ        | 10             | U         | 10             | UJ        |  |
| Acetone                               | 50     | U         | 50             | U         | 50             | U         | 50             | UJ        | 50             | U         |  |
| Benzene                               | 10     | U         | 10             | U         | 1.1            | J         | 10             | UJ        | 10             | U         |  |
| Bromodichloromethane                  | 10     | U         | 10             | U         | 10             | U         | 10             | UJ        | 10             | U         |  |
| Bromoform                             | 10     | U         | 10             | U         | 10             | U         | 10             | UJ        | 10             | U         |  |
| Bromomethane                          | 10     | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |
| Carbon disulfide                      | 10     | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |
| Carbon tetrachloride                  | 10     | U         | 10             | U         | 10             | U         | 10             | UJ        | 10             | U         |  |
| Chlorobenzene                         | 10     | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |
| Chlorodibromomethane                  | 10     | U         | 10             | U         | 10             | U         | 10             | UJ        | 10             | U         |  |
| Chloroethane                          | 10     | UJ        | 10             | UJ        | 10             | UJ        | 10             | UJ        | 10             | UJ        |  |
| Chloroform                            | 10     | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |
| Chloromethane                         | 10     | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |
| Cis-1,2-Dichloroethene                | 35     |           | 10             | U         | 10             | U         | 10             | U         | 10             | U         |  |

| Appendix E |                                |  |  |  |  |  |  |  |
|------------|--------------------------------|--|--|--|--|--|--|--|
| Table 1.6: | <b>Groundwater VOC Results</b> |  |  |  |  |  |  |  |

| Lab Sample Id             | X52    | 03-08     | X520   | 3-09      | X52            | 03-10     | X5203          | 3-11RE    | X52            | 03-12     |  |
|---------------------------|--------|-----------|--------|-----------|----------------|-----------|----------------|-----------|----------------|-----------|--|
| Lab Sample Delivery Group | X5     | 5203      | X52    | X5203     |                | X5203     |                | X5203     |                | 203       |  |
| Loc Name                  | MV     | V-10      | GW-21  |           | GW-21          |           | GW-22          |           | GW-22          |           |  |
| Field Sample Id           | CRMW01 | 1001301XX | CRGW02 | 101901XX  | CRGW02102601XX |           | CRGW02201801XX |           | CRGW02202801XX |           |  |
| Field Sample Date         | 11/1   | /2006     | 11/1/2 | 11/1/2006 |                | 11/1/2006 |                | 11/1/2006 |                | 11/1/2006 |  |
| Qc Code                   | I      | FS        | F      | FS        |                | FS        |                | FS        |                | FS        |  |
| Param Name                | Result | Qualifier | Result | Qualifier | Result         | Qualifier | Result         | Qualifier | Result         | Qualifier |  |
| cis-1,3-Dichloropropene   | 10     | U         | 10     | U         | 10             | U         | 10             | UJ        | 10             | U         |  |
| Cyclohexane               | 38     |           | 10     | U         | 1.6            | J         | 10             | U         | 10             | U         |  |
| Dichlorodifluoromethane   | 10     | U         | 10     | U         | 10             | U         | 10             | UJ        | 10             | U         |  |
| Ethyl benzene             | 20     |           | 10     | U         | 10             | U         | 10             | U         | 10             | U         |  |
| Isopropylbenzene          | 13     |           | 10     | U         | 10             | U         | 10             | U         | 10             | U         |  |
| Methyl cyclohexane        | 43     |           | 10     | U         | 1.4            | J         | 10             | UJ        | 10             | U         |  |
| Methyl Tertbutyl Ether    | 10     | U         | 10     | U         | 10             | U         | 10             | U         | 10             | U         |  |
| Methylene chloride        | 10     | U         | 10     | U         | 10             | U         | 10             | U         | 10             | U         |  |
| o-Xylene                  | 0.88   | J         | 10     | U         | 10             | U         | 10             | U         | 10             | U         |  |
| Styrene                   | 10     | U         | 10     | U         | 10             | U         | 10             | U         | 10             | U         |  |
| Tetrachloroethene         | 110    |           | 10     | U         | 10             | U         | 10             | U         | 10             | U         |  |
| Toluene                   | 10     | U         | 10     | U         | 1.7            | J         | 10             | U         | 10             | U         |  |
| trans-1,2-Dichloroethene  | 10     | U         | 10     | U         | 10             | U         | 10             | U         | 10             | U         |  |
| trans-1,3-Dichloropropene | 10     | U         | 10     | U         | 10             | U         | 10             | UJ        | 10             | U         |  |
| Trichloroethene           | 77     |           | 10     | U         | 10             | U         | 10             | UJ        | 10             | U         |  |
| Trichlorofluoromethane    | 10     | U         | 10     | U         | 10             | U         | 10             | UJ        | 10             | U         |  |
| Vinyl chloride            | 9      | J         | 10     | U         | 10             | U         | 10             | U         | 10             | U         |  |
| Xylene, m/p               | 7.1    | J         | 10     | U         | 0.95           | J         | 10             | U         | 10             | U         |  |

Notes: Results in micrograms per liter (µg/L) Samples analyzed for VOCs by EPA Method OLM04.2 QC Code: FS = Field Sample FD = Field Duplicate TB = Trip Blank

Qualifiers:

U = Not detected at a concentration greater than the RL

J = Estimated value

| Lab Sample Id                         | X5203-13         | X5203-14         | X5203-15         | X5203-16         | X5203-17         |  |
|---------------------------------------|------------------|------------------|------------------|------------------|------------------|--|
| Lab Sample Delivery Group             | X5203            | X5203            | X5203            | X5203            | X5203            |  |
| Loc Name                              | GW-24            | GW-24            | GW-25            | GW-25            | GW-25            |  |
| Field Sample Id                       | CRGW02402301XX   | CRGW02401301XX   | CRGW02501201XX   | CRGW02501201XD   | CRGW02502201XX   |  |
| Field Sample Date                     | 11/1/2006        | 11/1/2006        | 11/1/2006        | 11/1/2006        | 11/1/2006        |  |
| Qc Code                               | FS               | FS               | FS               | FD               | FS               |  |
| Param Name                            | Result Qualifier |  |
| 1,1,1-Trichloroethane                 | 10 U             |  |
| 1,1,2,2-Tetrachloroethane             | 10 U             |  |
| 1,1,2-Trichloro-1,2,2-Trifluoroethane | 10 U             |  |
| 1,1,2-Trichloroethane                 | 10 U             |  |
| 1,1-Dichloroethane                    | 10 U             |  |
| 1,1-Dichloroethene                    | 10 U             |  |
| 1,2,4-Trichlorobenzene                | 10 U             |  |
| 1,2-Dibromo-3-chloropropane           | 10 U             |  |
| 1,2-Dibromoethane                     | 10 U             |  |
| 1,2-Dichlorobenzene                   | 10 U             |  |
| 1,2-Dichloroethane                    | 10 U             |  |
| 1,2-Dichloropropane                   | 10 U             |  |
| 1,3-Dichlorobenzene                   | 10 U             |  |
| 1,4-Dichlorobenzene                   | 10 U             |  |
| 2-Butanone                            | 50 UJ            |  |
| 2-Hexanone                            | 50 U             |  |
| 4-Methyl-2-pentanone                  | 50 U             |  |
| Acetic acid, methyl ester             | 10 U             |  |
| Acetone                               | 50 UJ            |  |
| Benzene                               | 10 U             |  |
| Bromodichloromethane                  | 10 U             |  |
| Bromoform                             | 10 U             |  |
| Bromomethane                          | 10 U             |  |
| Carbon disulfide                      | 10 U             |  |
| Carbon tetrachloride                  | 10 U             |  |
| Chlorobenzene                         | 10 U             |  |
| Chlorodibromomethane                  | 10 U             |  |
| Chloroethane                          | 10 UJ            |  |
| Chloroform                            | 10 U             |  |
| Chloromethane                         | 10 U             |  |
| Cis-1,2-Dichloroethene                | 10 U             |  |

| Appendix E |                                |  |  |  |  |  |  |  |
|------------|--------------------------------|--|--|--|--|--|--|--|
| Table 1.6: | <b>Groundwater VOC Results</b> |  |  |  |  |  |  |  |

| Lab Sample Id             | X5203-13         | X5203-14         | X5203-15         | X5203-16         | X5203-17         |  |
|---------------------------|------------------|------------------|------------------|------------------|------------------|--|
| Lab Sample Delivery Group | X5203            | X5203            | X5203            | X5203            | X5203            |  |
| Loc Name                  | GW-24            | GW-24            | GW-25            | GW-25            | GW-25            |  |
| Field Sample Id           | CRGW02402301XX   | CRGW02401301XX   | CRGW02501201XX   | CRGW02501201XD   | CRGW02502201XX   |  |
| Field Sample Date         | 11/1/2006        | 11/1/2006        | 11/1/2006        | 11/1/2006        | 11/1/2006        |  |
| Qc Code                   | FS               | FS               | FS               | FD               | FS               |  |
| Param Name                | Result Qualifier |  |
| cis-1,3-Dichloropropene   | 10 U             |  |
| Cyclohexane               | 10 U             |  |
| Dichlorodifluoromethane   | 10 UJ            |  |
| Ethyl benzene             | 10 U             |  |
| Isopropylbenzene          | 10 U             |  |
| Methyl cyclohexane        | 10 U             |  |
| Methyl Tertbutyl Ether    | 10 U             |  |
| Methylene chloride        | 10 U             |  |
| o-Xylene                  | 10 U             |  |
| Styrene                   | 10 U             |  |
| Tetrachloroethene         | 10 U             |  |
| Toluene                   | 10 U             |  |
| trans-1,2-Dichloroethene  | 10 U             |  |
| trans-1,3-Dichloropropene | 10 U             |  |
| Trichloroethene           | 10 U             |  |
| Trichlorofluoromethane    | 10 UJ            |  |
| Vinyl chloride            | 10 U             |  |
| Xylene, m/p               | 10 U             |  |

Notes: Results in micrograms per liter (µg/L) Samples analyzed for VOCs by EPA Method OLM04.2 QC Code: FS = Field Sample FD = Field Duplicate TB = Trip Blank Qualifiers: U = Not detected at a concentration greater than the RL

U = Not detected at a concentration gi

| Lab Sample Id                         | X52     | 03-18     | X526           | 68-01     | X5268-02       |           | X5268-03       |           | X5268-04       |           |
|---------------------------------------|---------|-----------|----------------|-----------|----------------|-----------|----------------|-----------|----------------|-----------|
| Lab Sample Delivery Group             | X5      | 5203      | X5             | 268       | X5268          |           | X5268          |           | X5268          |           |
| Loc Name                              | (       | QC        | GW             | /-18      | GW             | /-18      | GW-20          |           | GV             | V-20      |
| Field Sample Id                       | CRTB001 | XXX02XX   | CRGW01802001XX |           | CRGW01803001XX |           | CRGW02001901XX |           | CRGW02002901XX |           |
| Field Sample Date                     | 11/1    | /2006     | 11/2/2006      |           | 11/2/2006      |           | 11/2/2006      |           | 11/2/2006      |           |
| Qc Code                               | ]       | ГВ        | F              | 'S        | F              | S         | ŀ              | TS .      | FS             |           |
| Param Name                            | Result  | Qualifier | Result         | Qualifier | Result         | Qualifier | Result         | Qualifier | Result         | Qualifier |
| 1,1,1-Trichloroethane                 | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| 1,1,2,2-Tetrachloroethane             | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| 1,1,2-Trichloro-1,2,2-Trifluoroethane | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| 1,1,2-Trichloroethane                 | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| 1,1-Dichloroethane                    | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| 1,1-Dichloroethene                    | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| 1,2,4-Trichlorobenzene                | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| 1,2-Dibromo-3-chloropropane           | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| 1,2-Dibromoethane                     | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| 1,2-Dichlorobenzene                   | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| 1,2-Dichloroethane                    | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| 1,2-Dichloropropane                   | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| 1,3-Dichlorobenzene                   | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| 1,4-Dichlorobenzene                   | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| 2-Butanone                            | 50      | U         | 50             | UJ        | 50             | UJ        | 50             | UJ        | 50             | UJ        |
| 2-Hexanone                            | 50      | U         | 50             | U         | 50             | U         | 50             | U         | 50             | U         |
| 4-Methyl-2-pentanone                  | 50      | U         | 50             | U         | 50             | U         | 50             | U         | 50             | U         |
| Acetic acid, methyl ester             | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| Acetone                               | 8.1     | J         | 50             | UJ        | 40             | J         | 50             | UJ        | 14             | J         |
| Benzene                               | 10      | U         | 10             | U         | 0.82           | J         | 10             | U         | 10             | U         |
| Bromodichloromethane                  | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| Bromoform                             | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| Bromomethane                          | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| Carbon disulfide                      | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| Carbon tetrachloride                  | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| Chlorobenzene                         | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| Chlorodibromomethane                  | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| Chloroethane                          | 10      | U         | 10             | UJ        | 10             | UJ        | 10             | UJ        | 10             | UJ        |
| Chloroform                            | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| Chloromethane                         | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |
| Cis-1,2-Dichloroethene                | 10      | U         | 10             | U         | 10             | U         | 10             | U         | 10             | U         |

| Lab Sample Id             | X52     | X5203-18  |        | X5268-01  |        | X5268-02  |        | X5268-03  |        | X5268-04  |  |
|---------------------------|---------|-----------|--------|-----------|--------|-----------|--------|-----------|--------|-----------|--|
| Lab Sample Delivery Group | X5      | X5203     |        | X5268     |        | X5268     |        | X5268     |        | 268       |  |
| Loc Name                  | (       | QC        |        | GW-18     |        | GW-18     |        | GW-20     |        | /-20      |  |
| Field Sample Id           | CRTB001 | XXX02XX   | CRGW01 | 802001XX  | CRGW01 | 803001XX  | CRGW02 | 001901XX  | CRGW02 | 002901XX  |  |
| Field Sample Date         | 11/1    | /2006     | 11/2/  | 2006      | 11/2   | /2006     | 11/2/  | /2006     | 11/2/  | 2006      |  |
| Qc Code                   | 1       | ГВ        | F      | S         | ŀ      | <b>S</b>  | F      | TS .      | F      | 'S        |  |
| Param Name                | Result  | Qualifier | Result | Qualifier | Result | Qualifier | Result | Qualifier | Result | Qualifier |  |
| cis-1,3-Dichloropropene   | 10      | U         | 10     | U         | 10     | U         | 10     | U         | 10     | U         |  |
| Cyclohexane               | 10      | U         | 10     | U         | 10     | U         | 0.97   | J         | 10     | U         |  |
| Dichlorodifluoromethane   | 10      | U         | 10     | UJ        | 10     | UJ        | 10     | UJ        | 10     | UJ        |  |
| Ethyl benzene             | 10      | U         | 10     | U         | 10     | U         | 10     | U         | 10     | U         |  |
| Isopropylbenzene          | 10      | U         | 10     | U         | 10     | U         | 10     | U         | 10     | U         |  |
| Methyl cyclohexane        | 10      | U         | 10     | U         | 10     | U         | 0.88   | J         | 10     | U         |  |
| Methyl Tertbutyl Ether    | 10      | U         | 10     | U         | 10     | U         | 10     | U         | 10     | U         |  |
| Methylene chloride        | 10      | U         | 10     | U         | 10     | U         | 10     | U         | 10     | U         |  |
| o-Xylene                  | 10      | U         | 10     | U         | 10     | U         | 10     | U         | 10     | U         |  |
| Styrene                   | 10      | U         | 10     | U         | 10     | U         | 10     | U         | 10     | U         |  |
| Tetrachloroethene         | 10      | U         | 10     | U         | 10     | U         | 10     | U         | 10     | U         |  |
| Toluene                   | 10      | U         | 10     | U         | 1.1    | J         | 0.99   | J         | 10     | U         |  |
| trans-1,2-Dichloroethene  | 10      | U         | 10     | U         | 10     | U         | 10     | U         | 10     | U         |  |
| trans-1,3-Dichloropropene | 10      | U         | 10     | U         | 10     | U         | 10     | U         | 10     | U         |  |
| Trichloroethene           | 10      | U         | 10     | U         | 10     | U         | 10     | U         | 10     | U         |  |
| Trichlorofluoromethane    | 10      | U         | 10     | UJ        | 10     | UJ        | 10     | UJ        | 10     | UJ        |  |
| Vinyl chloride            | 10      | U         | 10     | U         | 10     | U         | 10     | U         | 10     | U         |  |
| Xylene, m/p               | 10      | U         | 10     | U         | 10     | U         | 1.4    | J         | 10     | U         |  |

Notes: Results in micrograms per liter (µg/L) Samples analyzed for VOCs by EPA Method OLM04.2 QC Code: FS = Field Sample FD = Field Duplicate TB = Trip Blank

Qualifiers:

U = Not detected at a concentration greater than the RL

J = Estimated value

| Lab Sample Id                         | X52    | 68-05     | X5268-06       |           |  |
|---------------------------------------|--------|-----------|----------------|-----------|--|
| Lab Sample Delivery Group             | X5     | 268       | X5268          |           |  |
| Loc Name                              | GV     | V-23      | GV             | W-23      |  |
| Field Sample Id                       | CRGW02 | 301901XX  | CRGW02302701XX |           |  |
| Field Sample Date                     | 11/2   | /2006     | 11/2/2006      |           |  |
| Qc Code                               | F      | TS .      | FS             |           |  |
| Param Name                            | Result | Qualifier | Result         | Qualifier |  |
| 1,1,1-Trichloroethane                 | 10     | UJ        | 10             | UJ        |  |
| 1,1,2,2-Tetrachloroethane             | 10     | U         | 10             | UJ        |  |
| 1,1,2-Trichloro-1,2,2-Trifluoroethane | 10     | UJ        | 10             | UJ        |  |
| 1,1,2-Trichloroethane                 | 10     | U         | 10             | UJ        |  |
| 1,1-Dichloroethane                    | 10     | U         | 10             | UJ        |  |
| 1,1-Dichloroethene                    | 10     | UJ        | 10             | UJ        |  |
| 1,2,4-Trichlorobenzene                | 10     | U         | 10             | UJ        |  |
| 1,2-Dibromo-3-chloropropane           | 10     | U         | 10             | UJ        |  |
| 1,2-Dibromoethane                     | 10     | U         | 10             | UJ        |  |
| 1,2-Dichlorobenzene                   | 10     | U         | 10             | UJ        |  |
| 1,2-Dichloroethane                    | 10     | U         | 10             | UJ        |  |
| 1,2-Dichloropropane                   | 10     | U         | 10             | UJ        |  |
| 1,3-Dichlorobenzene                   | 10     | U         | 10             | UJ        |  |
| 1,4-Dichlorobenzene                   | 10     | U         | 10             | UJ        |  |
| 2-Butanone                            | 50     | U         | 50             | UJ        |  |
| 2-Hexanone                            | 50     | U         | 50             | UJ        |  |
| 4-Methyl-2-pentanone                  | 50     | U         | 50             | UJ        |  |
| Acetic acid, methyl ester             | 10     | UJ        | 10             | UJ        |  |
| Acetone                               | 50     | U         | 50             | UJ        |  |
| Benzene                               | 10     | U         | 10             | UJ        |  |
| Bromodichloromethane                  | 10     | U         | 10             | UJ        |  |
| Bromoform                             | 10     | U         | 10             | UJ        |  |
| Bromomethane                          | 10     | U         | 10             | UJ        |  |
| Carbon disulfide                      | 10     | UJ        | 10             | UJ        |  |
| Carbon tetrachloride                  | 10     | UJ        | 10             | UJ        |  |
| Chlorobenzene                         | 10     | U         | 10             | UJ        |  |
| Chlorodibromomethane                  | 10     | U         | 10             | UJ        |  |
| Chloroethane                          | 10     | UJ        | 10             | UJ        |  |
| Chloroform                            | 10     | U         | 10             | UJ        |  |
| Chloromethane                         | 10     | UJ        | 10             | UJ        |  |
| Cis-1,2-Dichloroethene                | 10     | U         | 10             | UJ        |  |

| Lab Sample Id             | X52    | 68-05     | X5268-06                |           |  |
|---------------------------|--------|-----------|-------------------------|-----------|--|
| Lab Sample Delivery Group | X5     | 268       | X                       | 5268      |  |
| Loc Name                  | GV     | V-23      | GW-23<br>CRGW02302701XX |           |  |
| Field Sample Id           | CRGW02 | 301901XX  |                         |           |  |
| Field Sample Date         | 11/2   | /2006     | 11/2                    | 2/2006    |  |
| Qc Code                   | H      | FS        | ]                       | FS        |  |
| Param Name                | Result | Qualifier | Result                  | Qualifier |  |
| cis-1,3-Dichloropropene   | 10     | U         | 10                      | UJ        |  |
| Cyclohexane               | 10     | UJ        | 10                      | UJ        |  |
| Dichlorodifluoromethane   | 10     | UJ        | 10                      | UJ        |  |
| Ethyl benzene             | 10     | U         | 10                      | UJ        |  |
| Isopropylbenzene          | 10     | U         | 10                      | UJ        |  |
| Methyl cyclohexane        | 10     | U         | 10                      | UJ        |  |
| Methyl Tertbutyl Ether    | 10     | U         | 10                      | UJ        |  |
| Methylene chloride        | 10     | U         | 10                      | UJ        |  |
| o-Xylene                  | 10     | U         | 10                      | UJ        |  |
| Styrene                   | 10     | U         | 10                      | UJ        |  |
| Tetrachloroethene         | 10     | U         | 3.2                     | J         |  |
| Toluene                   | 10     | U         | 10                      | UJ        |  |
| trans-1,2-Dichloroethene  | 10     | U         | 10                      | UJ        |  |
| trans-1,3-Dichloropropene | 10     | U         | 10                      | UJ        |  |
| Trichloroethene           | 10     | U         | 10                      | UJ        |  |
| Trichlorofluoromethane    | 10     | UJ        | 10                      | UJ        |  |
| Vinyl chloride            | 10     | UJ        | 10                      | UJ        |  |
| Xylene, m/p               | 10     | U         | 10                      | UJ        |  |

Notes:

Results in micrograms per liter ( $\mu$ g/L) Samples analyzed for VOCs by EPA Method OLM04.2 QC Code: FS = Field Sample FD = Field Duplicate TB = Trip Blank Qualifiers:

U = Not detected at a concentration greater than the RL

J = Estimated value

### Appendix E Table 1.7: Groundwater VOC TICs

| Lab Sample Id                      | X5203-08 |           | X5203-10 |           | X5268-02 |                | X5268-03  |                | X5268-04  |                |  |
|------------------------------------|----------|-----------|----------|-----------|----------|----------------|-----------|----------------|-----------|----------------|--|
| Lab Sample Delivery Group          | X5       | 203       | X5203    |           | X5268    |                | X5268     |                | X5268     |                |  |
| Loc Name                           | MV       | MW-10     |          | GW-21     |          | GW-18          |           | GW-20          |           | GW-20          |  |
| Field Sample Id                    | CRMW01   | 001301XX  | CRGW02   | 102601XX  | CRGW01   | CRGW01803001XX |           | CRGW02001901XX |           | CRGW02002901XX |  |
| Field Sample Date                  | 11/1/    | /2006     | 11/1     | /2006     | 11/2     | /2006          | 11/2/2006 |                | 11/2/2006 |                |  |
| Qc Code                            | F        | rs        | ŀ        | TS        | FS       |                | FS        |                | FS        |                |  |
| Param Name                         | Result   | Qualifier | Result   | Qualifier | Result   | Qualifier      | Result    | Qualifier      | Result    | Qualifier      |  |
| 1,2,3-Trimethylbenzene             | 49       | NJ        |          |           |          |                |           |                |           |                |  |
| 1,2-Diethylbenzene                 | 28       | JN        |          |           |          |                |           |                |           |                |  |
| 1,3-Dimethyl-4-ethylbenzene        | 29       | JN        |          |           |          |                |           |                |           |                |  |
| 1,4-Dimethylcyclohexane            | 57       | JN        |          |           |          |                |           |                |           |                |  |
| 2,3,4-Trimethylpentane             | 50       | JN        |          |           |          |                |           |                |           |                |  |
| 3-Methylhexane                     | 35       | JN        |          |           |          |                |           |                |           |                |  |
| 4-Methyl-2-hexene (c,t)            | 31       | JN        |          |           |          |                |           |                |           |                |  |
| Benzene, 1-butynyl-                | 38       | NJ        |          |           |          |                |           |                |           |                |  |
| Cyclohexane, 1-methyl-4-methylene- | 39       | JN        |          |           |          |                |           |                |           |                |  |
| Propylene                          |          |           | 19       | JN        | 12       | JN             | 8.4       | JN             | 7         | JN             |  |
| 1-PROPENE, 2-METHYL-               |          |           |          |           | 9.5      | JN             |           |                |           |                |  |
| 2-Butene-Trans                     |          |           |          |           |          |                | 6.9       | JN             | 6.1       | JN             |  |

Notes: Results in micrograms per liter ( $\mu$ g/L) Samples analyzed for VOCs by EPA Method OLM04.2 QC Code: FS = Field Sample Qualifiers: JN = Analyte was tenatively identified and the value is estimated

Created By: WBC 1/29/06 Checked By: ASZ 2/5/07